78-575

1

WALKER, William Wilmot, Jr., 1949-SOME ANALYTICAL METHODS APPLIED TO LAKE WATER QUALITY PROBLEMS.

Harvard University, Ph.D., 1977 Environmental Sciences

University Microfilms International, Ann Arbor, Michigan 48106



1977

WILLIAM WILMOT WALKER, Jr.

ALL RIGHTS RESERVED

INFORMATION TO USERS

This material was produced from a microfilm copy of the original document. While the most advanced technological means to photograph and reproduce this document have been used, the quality is heavily dependent upon the quality of the original submitted.

The following explanation of techniques is provided to help you understand markings or patterns which may appear on this reproduction.

- The sign or "target" for pages apparently lacking from the document photographed is "Missing Page(s)". If it was possible to obtain the missing page(s) or section, they are spliced into the film along with adjacent pages. This may have necessitated cutting thru an image and duplicating adjacent pages to insure you complete continuity.
- 2. When an image on the film is obliterated with a large round black mark, it is an indication that the photographer suspected that the copy may have moved during exposure and thus cause a blurred image. You will find a good image of the page in the adjacent frame.
- 3. When a map, drawing or chart, etc., was part of the material being photographed the photographer followed e definite method in "sectioning" the material. It is customary to begin photoing at the upper left hand corner of a large sheet and to continue photoing from left to right in equal sections with a small overlap. If necessary, sectioning is continued again — beginning below the first row and continuing on until complete.
- 4. The majority of users indicate that the textual content is of greatest value, howaver, a somewhat higher quality reproduction could be made from "photographs" if essential to the understanding of the dissertation. Silver prints of "photographs" may be ordered at additional charge by writing the Order Department, giving the catalog number, title, author and specific pages you wish reproduced.
- 5. PLEASE NOTE: Some pages may have indistinct print. Filmed as received.

Xerox University Microfilms 300 North Zeeb Road

Ann Arbor, Michigan 48106

SOME ANALYTICAL METHODS APPLIED TO LAKE WATER QUALITY PROBLEMS

T

ť

()

A thesis presented

by

William Wilmot Walker, Jr.

to

The Division of Engineering and Applied Physics

in partial fulfillment of the requirements

for the degree of

Doctor of Philosophy

in the subject of

Engineering

Harvard University

Cambridge, Massachusetts

February, 1977

Copyright Reserved by the Author

THESIS ABSTRACT

"Some Analytical Methods Applied to Lake Water Quality Problems"

William W. Walker, Jr.

A variety of analytical methods are demonstrated and evaluated in the context of assessing lake water quality problems. The techniques are drawn from the general areas of exploratory data analysis, parameter estimation, sensitivity analysis, and error analysis. Both empirical and theoretical approaches are taken in examining and modelling the behavior of a cross-section of lakes, as well as the behavior of a single lake in temporal and spatial domains. Proper use of these techniques is suggested as a means of improving efficiencies and assessing inaccuracies in converting scientific principles and observed data into environmental impact projections and management strategy designs.

Data characterizing the phosphorus balances, morphology, and hydrology of 105 northern temperate lakes provide a basis for demonstrating empirical approaches. Using nonlinear regression techniques, a two-parameter model for the phosphorus retention coefficient is shown to have stable parameter values, but increasing standard errors of estimate across the oligotrophic, mesotrophic, and eutrophic states. An error analysis estimates the independent variable, parameter, and model error components of retention coefficient and lake phosphorus concentration predictions. Stepwise discriminant and principal component analyses are employed to develop a model for estimating lake trophic state, expressed in probabilistic terms, from uncertain estimates of phosphorus loading, mean depth, and hydraulic residence time. The implications for the design of monitoring programs to provide data for use of these models are discussed. The limitations of the models are assessed in relation to various characteristics of the data base upon which they are developed.

Extensive water quality data from Onondaga Lake, New York, and its tributaries provide a basis for demonstrating the roles of and methods for exploratory data analysis in preliminary assessments. Theoretical approaches are taken in developing a model for vertical stratification in the lake. The model is estimated using nonlinear programming algorithms for locating maxinum likelihood estimates of parameters in dynamic systems. The stability of optimal parameter estimates under different hydrologic and meteorologic conditions is demonstrated using data from different years of the survey. Error and sensitivity analyses are performed in applying the model to assess the effects of an outfall design on vertical mixing in the lake.

Also discussed are the potential roles of the methods demonstrated in addressing some of the problems and deficiencies which have been characteristic of efforts at modelling natural systems, specifically including: parameter estimation problems (particularly in dynamic models); imbalances with regard to model complexity and data availability; misuse of model projections, as related to the scarcity of attempts to estimate confidence limits and to quantify error sources; misuse of models, as related to misinterpretations of "verification" tests and results. Control of these types of problems is characterized as being more difficult in systems where relatively complex modele must be employed and where the quality and quantity of independent variable data are relatively low. The stability of optimal parameter estimates across lakes and along temporal dimensions within lakes is suggested as an effective indicator of model generality and a useful criterion for model verification. PREFACE

Our civilization is such that environmental quality objectives often apparently conflict with other short- and long-term interests and requirements. We cannot express the costs and benefits of environmental quality management on the same scales and rely upon our political process to weigh and compare them in selecting, in theory, the best options for society as a whole. In contributing to this process, the scientific and engineering community has assumed responsibilities to assess the cultural impacts upon the environment and to suggest specific, feasible alternatives for achieving quality standards. These efforts require functional representations of natural systems, or "models" which transcend the descriptive analyses historically characteristic of environmental studies.

4

245.1

If the environmental modelling community were assembled along a gradient of empiricism, the extreme left might be found performing stepwise linear regressions, without much regard for physics, chemistry, or biology, and the extreme right, integrating hundreds of simultaneous differential equations, without much regard for observed data. At an inquisition, the left could be accused of curve-fitting and proposing models with limited realism and generality, and the right, of being narrow-minded and ignoring the truth in the data. Most of the community would attempt to use both data and theory as bases for developing models. For analysis of most environmental systems, efficiency would be maximized somewhere in the middle, as determined by quantity and quality of data and by extent and validity of scientific theory. However, the middle is also an area with many pitfalls, not the least hazardous of which is to accept correlation as sufficient proof of causation.

At its conception during undergraduate years, my approach to environmental modelling was, I suppose, somewhere to the right of the middle, a position attributed chiefly to a background in physical science. Experience in the Environmental Systems Program at Harvard has developed my ability to move to the left, according to the needs of a particular problem. Hope-



 (\cdot)

fully, it has also increased my awareness of the many pitfalls in this area. Interactions with faculty and students working on problems that were, for the most part, quite different from my own provided exposure to analytical techniques and approaches which were generally new to me, but which seemed promising as means of fortifying my own approaches. Identifying those techniques which were both valid and useful for application to the types of problems of my concern was a difficult and frustrating educational exercise, aided considerably by conversations with faculty members and fellow students.

The guidance of Professor Joseph J. Harrington has been invaluable. Based upon his experience with a wide variety of techniques and upon his general grasp of the environmental field, he has provided many stimulating suggestions and thoughtful responses. His proposal that my thesis focus on the concept of technique rather than on a particular theory or model eventually eliminated much doubt and confusion as to possible directions for my graduate studies. His demonstrations of faith and patience have earned him my life-long gratitude and respect.

Professor Harrington's guidance has been more-than-adequately supplemented by the comments and suggestions of Professors Ralph Mitchell, Frederick E. Smith, and Harold A. Thomas, Jr. The time, interest, and insights of these gentiemen are gratefully acknowledged.

The opportunity to exchange ideas with fellow students in the program has been a valuable aspect of my graduate experience. Associations with Kenneth H. Reckhow, who shares an interest and involvement in lake water quality problems, have been particularly meaningful.

Extensive data from Onondaga Lake, New York, have provided an excellent basis for many of the analytic efforts in the thesis. The Onondaga County program is one of the most intensive monitoring efforts ever undertaken in and around a lake of this size. I salute the County for its endeavors and hope that proper use of the data they have gathered will someday contribute to improvements in the lake's unfortunate condition. The following people have been especially helpful in providing information and insight relative to Onondaga Lake and its environs : Dr. Cornelius B. Murphy, Jr. and Mr. Greg Welter (O'Brien and Gere, Engineers); Mr. Randy R. Ott and Mr. Donald Lawler (Onondaga County Department of Drainage and Sanitation); Mr. William Embree, Mr. Kenneth Darmer, and Mr. Al Randal (U.S. Geological Survey); Dr. Jay Bloomfield (New York State Department of Environmental Conservation); Mr. James Rooney (U.S. E.P.A., Region II); Mr. Richard Clough (Allied Chemical Corporation). Their interest and cooperation are gratefully acknowledged.

τv

I am grateful to the National Science Foundation for its financial support of these efforts.

The moral and clerical support of Ms. Martha J. Ploetz have been an essential factor in the draft and final preparation of this volume.

Finally, I would like to dedicate this work to my parents.

William W. Walker, Jr. Harvard University February 1977

1

(i

TABLE OF CONTENTS

(

();

()

| | | | Page |
|------|------------------|--|-------|
| LÍST | OF FI | GURES | viíi |
| list | of tai | BLES | xiii |
| SYNO | PSIS | | xvi |
| 1.0 | INTRO | DUCTION | 1-1 |
| | 1.1 | Objectives | 1-1 |
| | 1.2 | Causation and Correlation | 1-3 |
| | 1.3 | Types of Models | 1-5 |
| | 1.4 | Model Specification. Estimation. and | |
| | | Verification | 1-7 |
| | 1.5 | Model Applications | 1-20 |
| | 1.6 | Introduction to Illustrations in Subsequent | |
| | | Chapters | 1-27 |
| | Refer | ences | 1-29 |
| 2.0 | Methoi Probli | DS FOR ASSESSMENT OF LAKE WATER QUALITY EMS | 2-1 |
| | 2.1 | Introduction | 2-1 |
| | 2.2 | Lake Classification | 2-2 |
| | 2.3 1 | Nutrient Balance Rationale | 2-14 |
| | | 2.3.1 Factors Influencing Nutrient Sources | 2-16 |
| | | 2.3.2 Estimation Methods | 2-24 |
| | | 2.3.3 Monitoring Program Design | 2-35 |
| | | 2.3.4 Interpretation Problems | 2-46 |
| | 2.4 | A Review of Lake Models | 2-53 |
| | | 2.4.1 Empirical Models | 2-54 |
| | | 2.4.2 Theoretical Models | 2-62 |
| | 2.5 | Analysis of Empirical Approaches | 2-78 |
| | | 2.5.1 Lake Data Base Description | 2-79 |
| | | 2.5.2 Phosphorus Retention Models | 2-91 |
| | | 2.5.3 Error Analysis | 2-107 |
| | | 2.5.4 Trophic State Prediction | 2-134 |
| | | 2.5.5 Discussion of Misclassified Lakes | 2-157 |
| | | 2.5.6 Application Strategies | 2-162 |
| | : | 2.5.7 Implications for Monitoring Program | |
| | | Desig. | 2-179 |
| | | 2.5.8 Summary | 2-188 |
| | : | 2.5.9 Suggestions for Future Work | 2-193 |
| | 2.6 | General Comments and Conclusions | 2-196 |
| | Refere | ences / | 2-202 |

Standing and a standard and a s

2

TABLE OF CONTENTS (continued)

ĩ

(

(

| | | | Page |
|-----|------------|--|-------|
| 3.0 | EXPI | ORATORY ANALYSIS OF ONONDAGA LAKE WATER | |
| | QUAI | JITY DATA | 3-1 |
| | • • | Takus Australian | 21 |
| | 3.1 | Introduction | 3-1 |
| | | 3.1.1 General Description | 3-1 |
| | | 3.1.2 Water Quality issues | 3-4 |
| | | 3.1.3 Plans for Pollution Adatement | 3-0 |
| | - - | 3.1.4 Scope of Chapter | 3-10 |
| | 3.2 | Data Base 2.2.1. taka and Mathukawa Nakan Gualitu | 3-11 |
| | | 2.2.2 Makes Source Wrostwerk Dist Crossing | 3-11 |
| | | 5.2.2 Metro Sewage Treatment Plant Operating | 3-17 |
| | | 2 2 2 Hudrology | 3-19 |
| | | 3.2.5 Hydrology | 3-24 |
| | 3 3 | Nathada | 3-26 |
| | 2.3 | 3.3.1 t-Tests for Worigontal Mixing | 3-26 |
| | | 3.3.2 GRID Displays | 3-27 |
| | | 3.3.3 Line Plots | 3-28 |
| | | 3.3.4 Mass Balances | 3-33 |
| | 3.4 | Results and Discussion | 3-37 |
| | | 3.4.1 Horizontal Mixing | 3-37 |
| | | 3.4.2 Vertical Mixing | 3-40 |
| | | 3.4.3 Phytoplankton and Nutrients | 3-53 |
| | | 3.4.4 Dissolved Oxygen | 3-60 |
| | | 3.4.5 Mass Balances | 3-65 |
| | 3.5 | Eutrophication Assessment | 3⊷78 |
| | | 3.5.1 Phosphorus as a Controlling Factor | 3-78 |
| | | 3.5.2 Light as a Controlling Factor | 3-84 |
| | 3.6 | Conclusions | 3-96 |
| | Refe | rences | 3-100 |
| | | | |
| 4.0 | A MC | DEL FOR VERTICAL STRATIFICATION IN ONONDAGA | |
| | LAKE | | 4-1 |
| | 4.1 | Objectives - The Outfall Design Issue | 4-1 |
| | 4.2 | Review of Lake Vertical Stratification Models | 4-7 |
| | | 4.2.1 Mass Transport Models | 4-8 |
| | | 4.2.2 Mechanical Energy Balance Models | 4-11 |
| | | 4.2.3 Statistical Studies | 4-20 |
| | 4.3 | Model Development | 4-28 |
| | | 4.3.1 System Definition | 4-32 |
| | | 4.3.2 Forcing Functions | 4-36 |
| | | 4.3.3 System Functions | 4-41 |
| | | 4.3.4 Parameter Values | 4-46 |
| | 4.4 | Implementar lon | 4-48 |

v1

TABLE OF CONTENTS (continued)

ľ

(

| | | Page |
|----------|---|-------|
| 4.5 | Parameter Estimation | 4-53 |
| | 4.5.1 Methods | 4-53 |
| | 4.5.2 Implementation | 4-60 |
| | 4.5.3 Lake Data | 4-63 |
| | 4.5.4 Results | 4-65 |
| | 4.5.5 Interpretation of Parameter Estimates | 4-76 |
| | 4.5.6 Analysis of Residuals | 4-81 |
| 4.6 | Model Application | 4-99 |
| | 4.6.1 Sensitivity Analysis | 4-99 |
| | 4.6.2 Case Simulations | 4-110 |
| | 4.6.3 Error Analysis | 4-127 |
| 4.7 | Conclusions and Recommendations | 4-133 |
| | 4.7.1 Model Adequacy | 4-133 |
| | 4.7.2 Data Adequacy | 4-136 |
| | 4.7.3 Methods Adequacy | 4-137 |
| | 4.7.4 Implications for Outfall Design | 4-141 |
| Refe | erencés | 4-144 |
| 5.0 OVER | RVIEW . | 5-1 |
| APPENDIX | A - Results of t-Tests for Horizontal Mixing in Onondaga Lake | A-1 |
| APPENDIX | B - GRID Displays of Onondaga Lake Water | |
| | Quality Data | A-4 |
| APPENDIX | C - Line Plots of Volume-Averaged Onondaga | • |
| | Lake Water Quality Data | A-12 |
| APPENDIX | D - Mass Balances on Onondaga Lake | A-19 |
| APPENDIX | E - Plots of Monthly-Average Meteorologic and Hydrologic Data Used as Boundary Conditions in Simulating Vertical Mixing | |
| | in Onondaga Lake | A-30 |

<u>Page</u>

LIST OF FIGURES

ľ

 $\mathbf{C}^{\mathbf{c}}$

| 1.4-1 | Stages in Model Development | 1-8 |
|-------|---|-------------------|
| 2.2-1 | Dynamic Lake Classification According to Uttormark and Wall | 2 - 12 |
| 2.3-1 | Relationship Between Effective and Actual Sample Sizes in Calculating a Mean Generated by a Lag One Markov Process | 2-43 |
| 2.4-1 | Vollenweider's First Model for Predicting Lake Trophic State | 2-56 |
| 2.4-2 | Vollenweider's Second Model for Predicting Lake Trophic State | 2-56 |
| 2.4-3 | Dillon's Model for Predicting Lake Trophic State | 2-56 |
| 2.4-4 | Biological and Chemical Systems Diagram for Lake Ontario Model | 2-66 |
| 2,4-5 | Control Pathways in the Lower Charles River Model | 2-66 |
| 2,4-6 | Effect of Mixed Depth upon Biomass Potential According to Lorenzen and Mitchell | 2-72 |
| 2.5-1 | Normal Probability Plot of Retention Coefficient Residuals | 2-95 |
| 2.5→2 | Observed and Estimated Phosphorus Retention Coefficients | 2-96 |
| 2.5-3 | Observed and Estimated Phosphorus Outlet Concentrations | 2-96 |
| 2,5-4 | Dependence of First-Order Phosphorus Sedimentation Coef- ficient on Hydraulic Residence Time | 2-100 |
| 2.5-5 | Outlet Conccentration Residual Variance for Various Residence Time Intervals | 2-111 |
| 2.5-6 | Estimated Model and Measurement Error Components of Outlet Concentration Residuals | 2-113 |
| 2.5-7 | Estimated Model and Measurement Error Components of Retention Coefficient Residuals | 2-113 |
| 2.5-8 | Variance Components of Retention Coefficient Predictions | 2-118 |
| 2.5-9 | Detectability of Model Errors in Estimating the Retention Coefficient as a Function of Residence Time and Model to Measurement Error Variance Ratio | 2-122 |

Ć

 $\mathbf{C}^{\mathbf{I}}$

| <u> </u> | | Page |
|----------|--|--------------------|
| 2.5~10 | Variance Components of Outlet Phosphorus Concentration Predictions | 2-125 |
| 2.5-11 | Relationship Between Mean Summer Chlorophyll a and Total Phosphorus Concentration at Spring Overturn According to Dillon and Rigler | 2-128 |
| 2.5-12 | Variance Components of Mean Summer Chlorophyll a Predictions | 2-132 |
| 2.5-13 | Lake Classification According to Vollenweider's First Model | 2~150 |
| 2.5-14 | Lake Classification According to Vollenweider's Second Model | 2-150 |
| 2.5-15 | Lake Classification According to Dillon's Model with Observed Retention Coefficients | 2-151 |
| 2.5-16 | Lake Classification According to Dillon's Model with Estimated Retention Coefficients | 2-151 |
| 2.5-17 | Lake Classification According to Model Derived from Stepwise Discriminant Analysis on COEST vs. L Axes | 2 - 155 |
| 2.5-18 | Lake Classification According to Model Derived from Stepwise Discriminant Analysis on L vs. QS/(1-RE) Axes | 2-155 |
| 2.5-19 | Trophic State Probabilities Versus Canonical Variable X | 2-164 |
| 2.5-20 | Lines of Constant Eutrophic Probability on L vs. QS/(1-RE) Axes | 2-165 |
| 2.5-21 | Lines of Constant Oligotrophic Probability on L vs. QS/(1-RE) Axes | 2-165 |
| 2.5-22 | Contour Plots of Trophic State Probabilities vs. X and Coefficient of Variation of X for Normal and Lognormal Error Distributions | 2-177 |
| 2.5-23 | Coefficient of Variation of Mean Phosphorus Flux Estimate Against Sample Sizs for Various Stations in the Cross- Florida Barge Canal Study | 2-182 |
| 3.1-1 | Onondaga Lake | 3-2 |
| 3.1-2 | Onondaga Lake/Watershed | 3-3 |
| 3.2-1 | Monthly Water Balance | 3-23 |

TV

, : , :

ć

(:

| 3.3 - 1 | Surface Area and Volume vs. Depth | 3-30 |
|--------------------|--|------|
| 3.4-1 | Density | 3-43 |
| 3.4-2 | Density Gradient | 3-43 |
| 3.4-3 | Line Plots of Temperature and Chloride Gradients | 3-47 |
| 3.4-4 | Chemical and Thermal Components of Total Density Gradient | 3-48 |
| 3.4-5 | Buoyant Potential Energy Deficit | 3-51 |
| 3.4-6 | Line Plot - Chlorophyll | 3-54 |
| 3,4-7 | Line Plot - Total Inorganic Phosphorus | 3-55 |
| 3.4-8 | Line Plot - Ortho-Phosphorus | 3-56 |
| 3.4-9 | Phytoplankton and Zooplankton Populations | 3-58 |
| 3.4-10 | Dissolved Oxygen, Percent of Saturation | 3-62 |
| 3,4-11 | Line Plots - DO, DO/DO Bat, DO Sat-DO | 3-64 |
| 3.4-12 | Dissolved Oxygen, Percent of Saturation, and Density Gradient | 3-66 |
| 3.4-13 | Relationships Among Phytoplankton Standing Crop, Gross Photosynthesis, and Photosynthetic Efficiency for Onondaga Lake Compared with Results of Brylinsky and Mann | 3-74 |
| 3.5-1 | Onondaga Lake Transparency Measurements | 3-88 |
| 3.5-2 | Light- and Nutrient-Limited Biomass under Various Conditions | 3-90 |
| 4.2-1 | Entrainment Ratio Versus Richardson Number - Experimental Data of Kato and Phillips | 4-15 |
| 4.2-2 | Mechanical Energetics of Entrainment | 4-17 |
| 4.2-3 | Mechanical and Thermal Energy Transformations in a Lake According to Stefan and Ford | 4-19 |
| 4.2-4 | Plot of Blanton's Data on Entrainment Rate vs. Stability for Temperate Lakes | 4-23 |
| 4.2-5 | Blanton's Data on Entrainment Rate Versus Mean Depth | 4-25 |

Page

۸

(

Ĉ

(

| 4.2-6 | Snodgrass's Data on Vertical Transport Coefficient vs. Mean Depth | 4-25 |
|--------|--|--------------|
| 4.3-1 | System Diagram for the Onondaga Lake Vertical Stratifi- cation Model | 4-29 |
| 4.3-2 | Control Pathways in the Onondaga Lake Vertical Strati- fication Model | 4-30 |
| 4.4.1 | Schematic of Model Subroutines | 4-50 |
| 4.5-1 | Schematic of Model and Parameter Estimation Subroutines | 4-62 |
| 4,5-2 | Approximate 95% Confidence Regions for Estimates of Parameters a ₁₄ and a ₂₂ Obtained for Various Years | 4-75 |
| 4.5-3 | Observed(*) and Estimated(-) Temperatures Versus Time | 4-83 |
| 4.5-4 | Observed(*) and Estimated(-) Chlorides Versus Time | 4-84 |
| 4.5-5 | Observed(*) and Estimated(-) Densities Versus Time | 4-85 |
| 4.5-6 | Temperature Residuals Versus Time | 4-86 |
| 4.5-7 | Chloride Residuals Versus Time | 4-87 |
| 4.5-8 | Density Residuals Versus Time | 4-88 |
| 4.5-9 | Observations Versus Predictions | 4-89 |
| 4.5-10 | Histograms of Residuals | 4-9 0 |
| 4.5-11 | Normal Probability Plots of Residuals | 4-91 |
| 4.5-12 | Mean Residuals Versus Time of Year | 4-94 |
| 4.5-13 | Mean Residuals Versus Year | 4-95 |
| 4.6-1 | Temperature Sensitivity Coefficients Versus Time | 4-101 |
| 4.6-2 | Chloride Sensitivity Coefficients Versus Time | 4-102 |
| 4.6-3 | Density Sensitivity Coefficients Versus Time | 4-103 |
| 4.6-4 | Simulated Total and Non-Advective Hypolimmic Dilution Rates | 4-107 |
| 4.6-5 | GRID Display of Dissolved Oxygen | 4-107 |

Page

ľ

(

| | | Page |
|--------------------|---|-------|
| 4.6-6 | Simulated Mean Density Gradients for Various Cases | 4-117 |
| 4.6-7 | Simulated Maximum Annual Density Gradients for Various Cases | 4-117 |
| 4. 6 -8 | Simulated Mean Hypolimnic Dilution Rates for Various Cases | 4-118 |
| 4.6-9 | Simulated Minimum Annual Hypolimnic Dilution Rates for Various Cases | 4-118 |
| 4.6-10 | Simulations of Actual 1968-74 Conditions | 4-119 |
| 4.6-11 | Simulations of Case 6 Conditions | 4-120 |
| 4.6-12 | Simulations of Case 7 Conditions | 4-121 |
| 4.6-13 | Simulations of Case 13 Conditions | 4-122 |

~~~

# LIST OF TABLES

C

Ċ

**(**\*

|        |                                                                                                       | <u>rage</u> |
|--------|-------------------------------------------------------------------------------------------------------|-------------|
| 2.2-1  | Principal Characteristics of Oligotrophic, Eutrophic,<br>and Dystrophic Lakes According to Welch      | 2-4         |
| 2.2-2  | Point System for Uttormark and Wall's Lake Condition<br>Index                                         | 2-10        |
| 2.3-1  | Typical Values of Nutrient Runoff Coefficients for<br>Various Land Uses According to Uttormark et al. | 2-19        |
| 2.3-2  | Methods of Estimating Nutrient Fluxes Investigated<br>in the Cross-Florida Barge Canal Study          | 2-29        |
| 2.3-3  | Optimal Allocation of Samples Among Tributaries for<br>Nutrient Balance Estimation                    | 2-38        |
| 2.5-1  | Tabulation of Lake Data                                                                               | 2-80        |
| 2.5-2  | Identification of Variables and Analyses of Variance<br>Across Trophic Groups                         | 2-83        |
| 2.5-3  | Correlations Among Variables in Analysis                                                              | 2-89        |
| 2.5-4  | Parameter Estimates for Phosphorus Retention Model                                                    | 2~94        |
| 2.5-5  | Comparisons of Phosphorus Retention Models                                                            | 2-98        |
| 2.5-6  | Histograms of Observed and "Corrected" Pirst-Order<br>Sedimentation Coefficients for Total Phosphorus | 2-99        |
| 2.5-7  | Summary of Distributions and Initial Discriminating<br>Powers of Variables in Discriminant Analysis   | 2-136       |
| 2.5-8  | Discriminant Analysis - Vollenweider's First Model                                                    | 2-140       |
| 2.5-9  | Discriminant Analysis - Vollenweider's Second Model                                                   | 2-140       |
| 2.5-10 | Discriminant Analysis - Dillon's Model                                                                | 2-142       |
| 2.5-11 | Discriminant Analysis - Stepwise                                                                      | 2-142       |
| 2.5-12 | Disoriminant Analysis - Canonical Variable Derived<br>from Stepwise Analysis                          | 2-146       |
| 2.5-13 | Summary of Statistics Characterizing Discriminant<br>Analyses                                         | 2-147       |
| 2.5-14 | Comparisons of Model Performance in Classifying 105<br>Lakes                                          | 2-149       |

Page

. . .

# LIST OF TABLES (continued)

T.

(

(\*\*

|        |                                                                                            | Page              |
|--------|--------------------------------------------------------------------------------------------|-------------------|
| 2.5-15 | Misclassified Lakes                                                                        | 2-158             |
| 2.5-16 | First-Order Error Analysis for Estimates of Canonical<br>Variable X                        | 2-168             |
| 2.5-17 | Eutrophic Probabilities for Normal Error Distribution<br>of X                              | 2-173             |
| 2.5-18 | Eutrophic Probabilities for Lognormal Error Distribution<br>of X                           | 2-174             |
| 2,5-19 | Oligotrophic Probabilities for Normal Error Distribution<br>of X                           | 2–175             |
| 2.5-20 | Oligotrophic Probabilities for Lognormal Error Distri-<br>bution of X                      | 2–176             |
| 2.5-21 | Effect of Uncertainty in X on Rational Design Values<br>for Lognormally Distributed Errors | 2-17 <del>9</del> |
| 2.5-22 | Effect of Uncertainty in X on Rational Design Values<br>for Normally Distributed Errors    | 2-179             |
| 3.2-1  | Onondaga Lake Survey - Numbers of Samples by Station<br>and Year                           | 3-13              |
| 3.2-2  | Onondaga Lake Survey - Components Monitored                                                | 3-16              |
| 3.2-3  | Availability of Flow Data for Onondaga Lake Hydrologic<br>Balance                          | 3-20              |
| 3.2-4  | Regression Models Used to Estimate Missing Flow Obser-<br>vations                          | 3-21              |
| 3.2-5  | Onondaga Lake Water Balance                                                                | 3-22              |
| 3.2-6  | Summary of Meteorologic Data : 1968-74                                                     | 3-25              |
| 3.4-1  | Summary of Results of t-Tests for Horizontal Mixing                                        | 3-38              |
| 3.4-2  | Summary of Mid-summer Vertical Stratification Patterns<br>in Onondaga Lake, by Component   | 3-41              |
| 3,4-3  | Mass Balances - Onondaga Lake                                                              | 3-67              |
| 3.4-4  | Onondaga Lake - Ultimate Oxygen Demand Balance                                             | 3-76              |
| 3.4-5  | Onondaga Lake - Present and Projected Ultimate Oxygen<br>Demand Loadings                   | 3 <b>-7</b> 9     |

- 4

.

# LIST OF TABLES (continued)

C

(

(

| 3.5-1 | Allowable Phosphorus Loadings for Onondaga Lake<br>According to Various Models                                                                        | 3-81  |
|-------|-------------------------------------------------------------------------------------------------------------------------------------------------------|-------|
| 4.2-1 | Essential Aspects of Deep Reservoir Temperature<br>Models Evaluated by Parker et al.                                                                  | 4-8   |
| 4.3-1 | Water, Heat, and Chloride Fluxes                                                                                                                      | 4-31  |
| 4.3~2 | Forcing Function Definitions                                                                                                                          | 4-37  |
| 4.3-3 | System Function Definitions                                                                                                                           | 4-42  |
| 4.3-4 | Parameter Definitions and Values                                                                                                                      | 4-47  |
| 4.4-1 | Model Subroutines                                                                                                                                     | 4-49  |
| 4.5-1 | Parameter Estimation Subroutines                                                                                                                      | 4-61  |
| 4.5-2 | Temperature and Chloride Data Used for Parameter<br>Estimation                                                                                        | 4-64  |
| 4.5-3 | Parameter Estimates and Confidence Regions Based<br>upon All Data                                                                                     | 4-68  |
| 4.5-4 | Characteristics of Temperature and Chloride Residuals                                                                                                 | 4-69  |
| 4.5-5 | Best Parameter Estimates for Individual Years                                                                                                         | 4-74  |
| 4.5-6 | Results of Residuals Analysis                                                                                                                         | 4-82  |
| 4.6-1 | Definitions of Cases Studied                                                                                                                          | 4–111 |
| 4.6-2 | Results of Case Simulations                                                                                                                           | 4-116 |
| 4.6-3 | Parameteric and Residual Variance Components for<br>Each Case and Predicted Variable                                                                  | 4-131 |
| 4.7-1 | Comparisons of Standard Errors of Estimate for Tem-<br>perature Predictions in Onondaga Lake with Those<br>Typical of the M.I.T. Deep Reservoir Model | 4-135 |

Page

л۷

( :

C

Our ability to assess the impacts of cultural interventions upon natural systems provides a partial basis for design of resource management programs and for policy-making in the midst of other societal interests. While, historically, much of environmental management has been dictated by trial and error, we now require more rational methods which facilitate both the organization of experience at the system level (empirical approaches) and construction of mechanistic models with projective capabilities (theoretical approaches). These approaches entail functional understanding which transcends the purely descriptive analyses historically characteristic of environmental studies. Rational assessments rely upon two general information sources: data and scientific principles. This thesis is intended as a demonstration and evaluation of some techniques for synthesizing these information sources in formulating environmental impact assessments. A variety of quantitative techniques are demonstrated and evaluated in the context of analyzing lake water quality problems. These techniques are drawn from the general areas of exploratory data analysie, parameter estimation, sensitivity analysis, and error analysis. They are appropriate for use at various stages of the analytic process and are employed in examining the behavior of a cross-section of lakes, as well as that of a single lake in temporal and spatial domains.

Assessments of cultural eutrophication problems in lakes are dependent upon two types of models: source models and lake models. The factors driving these systems and thus determining lake response can be classified as naturally or culturally mediated. The former are uncontrollable, while

### SYNOPSIS (continued)

**(** )

the latter often represent decision variables or control points for the design and implementation of management strategies. A review of the literature suggests that, while much data have been compiled, (e.g., nutrient export versus land use), the development of generalized source models is considerably behind that of lake models. The latter can be classified as "theoretical" or "empirical", according to whether or not they attempt to eimulate internal mechanisms. These types of models are reviewed and compared with regard to complexity, applicability, accuracy, and generality.

Quantitative assessments of possible errors involved in use of source or lake models are generally lacking, suggesting that, as yet, there is an insufficient basis for rational lake water quality management program designs. In predicting lake response to a given management strategy, errors are introduced through uncertainty in the independent variables and parameter estimates, as well as through inadequacies in the models themselves. Given estimates of these individual error sources, some fairly simple methodologies can be employed to track these errors through the analysis and to compare their contributions to total prediction error. This can identify particular data or model deficiencies and thus provide a basis for efficient allocation of monitoring, experimental, and modelling effort. The effects of projection error can also be considered in final designs or policy recommendations. The difficulty and data requirements of such an exercise increase with model complexity. XV11

#### SYNOPSIS (continued)

( );

Based upon data for 105 northern temperate lakes, empirical lake models are developed and compared with existing models for steady-state predictions of phosphorus retention coefficients and lake trophic states as functions of three factors: total phosphorus loading, L (g/m<sup>2</sup>-year), lake mean depth, Z (m), and mean hydraulic residence time, T (years). The approach is based upon five assumptions: (1) mass balance; (2) completely-mixed conditions; (3) phosphorus limitation of lake ecosystems; (4) first-order kinetics for phosphorus sedimentation; (5) the possible influence of lake morphometric and hydrologic factors upon nutrient dynamics and trophic state response. The accuracy and value of these models are shown to be limited by various features of their data base, including measurement or estimation errors in the independent or dependent variables, multicollinearity, nonsteady-state conditions existing in the lakes during sampling, and the subjectivity involved in classifying lakes. In this context, it is difficult to identify model deficiencies due to effects of other controlling factors or aspects of system behavior which are ignored. Because of the data-dependency of these empirical models, their general superiority over existing models is not claimed. The primary emphases are upon the approaches taken and techniques employed in their development and evaluation and upon the strategies proposed for their application. A general focus is upon the development and use of error estimates.

Modifying Vollenwieder's<sup>9</sup> model based upon first-order kinetics, the following expression is proposed for use in predicting the phosphorus retention coefficient:

 $C_{1}$ 

$$1 - RE = \frac{1}{1 + KT}$$
$$K = a(\frac{1}{m})^{b}z^{c}$$

Using nonlinear regression techniques, the model is shown to have stable parameter values, but increasing standard errors of estimate across the oligotrophic, mesotrophic, and eutrophic states. The parameter expressing depth dependence, c, is not found to be significantly different from zero in any of the data groups tested. The best estimates of parameters a and b suggest the following equation:

$$1 - RE = \frac{1}{1 + 0.824T^{-454}}$$

Through an error analysis, the accuracy of a lake phosphorus prediction is shown to be data-limited (or source-model-limited) in lakes with low hydraulic residence times and retention-model-limited in lakes with high residence times. It is demonstrated that model errors would be statistically detectable relative to measurement errors in only about one third of the lakes employed in the analysis. By coupling this model with Dillon and Rigler's<sup>3</sup> equation relating chlorophyll concentration to lake phosphorus concentration, the accuracy of a chlorophyll prediction is shown to be limited by errors in predicting chlorophyll from lake phosphorus, not by errors in predicting lake phosphorus from estimates of L, Z, and T. This suggests that the retention model is adequate for use with state-of-

#### SYNPOSIS (continued)

the-art empirical chlorophyll models and that future work should be focused on improving the latter, possibly by incorporating the effects of algal growth limitation by light or other nutrients.

Discriminant analysis is employed to evaluate and compare existing models for trophic state prediction and to derive an optimal linear classification model for this data set. The analysis suggests the general inferiority of Vollenweider's<sup>8</sup> first model, in which lakes are classified in the L versus Z axes, as compared with Vollenweider's<sup>10</sup> second model (L versus QS), with Dillon's<sup>2</sup> model (CO), or with an optimal model derived from stepwise discriminant analysis (L versus COEST). It is more difficult to distinguish among the last three models, however. The stepwise model misclassifies an average of 14.5% of the lakes, with most of the errors centered in the mesotrophic group.

The primary advantage of posing the classification problem as a formal discriminant analysis is that it permits predictions of trophic states on probabilistic terms. Using the results of the stepwise analysis, equations are developed which permit estimation of trophic state probabilities as functions of L, T, and Z. The "rational" phosphorus loading allocation for a given lake is suggested as that loading which satisfies a certain maximum probability (risk) of eutrophic classification, typically 54:

$$L_{.05} = 0.0310 \left[\frac{2}{T} (1 + 0.824T^{.454})\right]^{.815}$$

XX

 $C^{\perp}$ 

Means of incorporating the effects of uncertainty in the independent variables (Z, T) and in the source models (used to estimate L as a function of management strategy) into the estimation of trophic state probabilities and into the rational loading allocation scheme are also developed. Efficiencies of loading allocations derived from the approach are shown to be limited in part by independent variable errors and dependent variable (classification) errors in the original data set.

Extensive data from Onondaga Lake, New York,<sup>4,5</sup> and its tributaries are used to illustrate potential roles and methods for exploratory data analysis and data reduction in preliminary assessments. Onondaga is a saline, eutrophic lake used primarily for municipal and industrial waste disposal purposes by Metropolitan Syracuse. The lake is introduced by describing major water quality issues, plans for pollution abatement, and water quality and quantity data availability. The variations of major water quality components are summarized and displayed along temporal and spatial dimensions in order to elucidate trends and seasonal variations and to determine the extents of horizontal and vertical mixing in the lake. The use of cubic splines<sup>7</sup> as a technique for examining seasonal and long-term variations in time series data is demonstrated along with a variety of computerized mapping and display technologies. Original methods for estimating and displaying mass balances on a continuous basis are also employed.

Statistically significant horizontal variations in some water quality

ĩ

components are explained by the location of a sewage outfall in the lake. Horizontal differences are small, however, compared with vertical and seasonal variations, which are interpreted relative to the effects of dilution, density, and differences in the chemical and biological reactions occurring in the epilimnion and hypolimnion. Some important trends observed during the 1968-75 monitoring period include: reductions in phosphorus, chromium, and silica levels, reductions in vertical temperature and chloride gradients, increases in hypolimnic dissolved oxygen levels, and the near disappearance of a formerly dominant blue-green algal population. Mass balances on several chemical constituents are formulated at monthly intervals over a five-year period and interpreted (1) in light of the chemical, physical, and biological processes considered to be of importance in the lake ecosystem; (2) in relation to specific pollution abatement programs which were implemented during that period; (3) with regard to the implications for the potential improvement of lake water quality through future implementation of specific point and non-point source control measures.

A modelling problem is designed to estimate the impact of the design of an outfall for future discharge of a saline industrial/municipal waste upon density stratification and vertical mixing rates in Onondaga Lake. The dynamic model developed to address this issue serves as a context in which to demonstrate the use of nonlinear programming algorithms for estimation of parameters in dynamic models, sensitivity analysis, and error analysis. In this effort, considerable modifications to Bard's<sup>1</sup> program for parameter estimation in nonlinear dynamic systems are made to improve Ϊ.

performance, flexibility, and the interpretative value of its output.

The modelling exercise demonstrates that essential aspects of vertical mixing in the lake can be simulated using a spatially and temporally aggregated model, consisting of two mixed compartments driven by monthlyaverage meteorologic and hydrologic boundary conditions. The model simulates temperature and chloride variations, based upon balances of mechanical energy, thermal energy, and mass. Parameter estimates indicate that density-dependent vertical exchange in the lake can be simulated by assuming that a constant percent (47%) of the turbulent kinetic energy introduced at the surface by wind shear stress is used to increase the buoyant potential energy of the system by mixing the hypolimnic and epilimnic waters. Empirical parameters are shown to be stable when estimated separately based upon data from different years of the survey. An error analysis indicates that prediction variance associated with uncertainty in the parameter estimates is generally small compared with residual variance, suggesting the adequacy of the data for parameter estimation purposes. The standard errors of temperature predictions are shown to be comparable to those typical of much more complex and disaggregated thermal stratification models<sup>6</sup>.

A sensitivity analysis compares the effects of wind mixing, salt discharge, and industrial cooling water consumption on lake density gradients. Simulated vertical mixing rates correlate with observed increases in hypolimnic dissolved oxygen levels over the 1968-74 monitoring xxiii

( )

period. Simulations indicate that discharge of the combined municipal/ industrial effluent into the hypolimnion would induce permanent density stratification under the meteorologic conditions of 1968-74, whereas discharge into the epilimnion at an initial dilution ratio of 16 or greater would result in an average of nearly two periods of vertical circulation annually, thus minimizing the impact of salt disposal on lake mixing. Variations in simulated density gradients and mixing rates attributed to parameter uncertainty are small relative to those attributed to yearto-year fluctuations in hydrologic/meteorologic conditions and to those induced by alternative management strategies. The variety of economic and environmental aspects of the outfall design issue are also discussed.

A final chapter summarizes the roles and limitation of the various methods and approaches demonstrated in the thesis. Exploratory data analyses are useful in providing important descriptive information, but independent evidence generally forms the basis for the functional understanding required to predict system behavior. The potential of nonlinear programming algorithms for use in estimating parameters in general models is high, subject to economic constraints on model complexity and requirements for reasonably unbiased estimates of independent variables. The importance of considering the quantity and quality of available independent variable data in specifying models is stressed. For a variety of reasons, error analyses yield very approximate results, particularly in cases of complex models. Comparisons of error sources can still yield useful information for assessing model and data adequacies, because the terms of the

### SYNOPSIS (continued)

ď.

total error equation often differ by orders of magnitude. Estimates of prediction errors should not be relied upon too heavily as bases for probabilistic projections and rational designs, particularly if a complex model is employed.

Generality is a key model attribute. Empirical lake models have demonstrated reasonable generality in associating the average conditions of lakes in a given geographic region with nutrient loading and morphometric factors. Theoretical lake models have successfully simulated behavior of individual lakes in time. The generality of empirical models along temporal dimensions and that of theoretical models among different lakes have not been demonstrated, due to inadequacies in the data and/or in the models themselves. These weaknesses suggest that future efforts should strive to demonstrate generality by estimating and verifying models based upon time series data from more than one lake system. Using the estimation techniques demonstrated here, the stability of optimal parameter estimates along temporal and system dimensions could be examined and used as a partial basis for assessment of generality. Applied to key parameters, this strategy would partially eliminate the necessity of using parameters derived from laboratory experiments, the results of which are often of limited validity under field conditions. Such an approach would depend upon the availability of adequate data and upon the feasibility of expressing essential functional relationships in concise terms, based upon current understanding of lake ecosystems.

### LIST OF SYMBOLS - SYNOPSIS

ζ.

 $(\cdot)$ 

lake mean depth Z (m) = mean hydraulic residence time (years) т z/T surface overflow rate (m/year) QS = = total phosphorus loading per unit area (g/m<sup>2</sup>year) L = average inlet total phosphorus concentration  $(g/m^3)$ CI average outlet total phosphorus concentration  $(q/m^3)$ co observed phosphorus retention R 1 - (CO/CI)≡ coefficient (dimensionless) RE estimated phosphorus retention coefficient (dimensionless) (1 - RE) CI estimated average outlet total COEST = phosphorus concentration  $(g/m^3)$ effective first order sedimentation coefficient for К total phosphorus (year<sup>-1</sup>) a,b,c =regression parameters

XXV1

( :

- Bard, Y., "Nonlinear Parameter Estimation and Programming", Program No. 360D-13.6.003, Share Program Library Agency, Triangle Universities Computation Center, Research Triangle Park, North Carolina, December 1967.
- Dillon, P.J., "The Phosphorus Budget of Cameron Lake, Ontario: The Importance of Flushing Rate to the Degree of Eutrophy of Lakes", <u>Limnology and Oceanography</u>, Vol. 20, No. 1, pp. 28-45, January 1975.
- Dillon, P.J. and F.H. Rigler, "The Phosphorus-Chlorophyll Relationship in Lakes", <u>Limnology and Oceanography</u>, Vol. 19, No. 1, pp. 767-773, September 1974.
- O'Brien and Gere, Engineers, "Onondaga Lake Monitoring Program", Annual Report: to Onondaga County, Department of Public Works, 1970-74.
- 5. Onordaga County, New York, "Onordaga Lake Study", US EPA, Water Quality Office, Publication 11000 FAE 4/71, April 1971.
- Parker, F.L., B.A. Benedict, and C. Tsai, "Evaluation of Mathematical Models for Temperature Prediction in Deep Reservoirs", National Environmental Research Center, Office of Research and Development, US EPA, Document No. EPA-660/3-75-038, June 1975.
- Reinsch, C.H., "Smoothing by Spline Functions", <u>Numerical Mathematics</u>, Vol. 10, No. 117, 1967.
- 8. Vollenweider, R.A., "Scientific Fundamentals of the Eutrophication of Lakes and Flowing Waters, with Particular Reference to Nitrogen and Phosphorus as Factors in Eutrophication", Organization of Economic Cooperation and Development, Paris, September 1970.
- Vollenweider, R.A. "Possibilities and Limits of Elementary Models Concerning the Budget of Substances in Lakes", <u>Arch. Hydrobiol.</u>, Vol. 66, No. 1, pp. 1-36, April 1969.
- Vollenweider, R.A., "Input-Output Models", unpublished manuscript, 1973.

#### 1.0 INTRODUCTION

#### 1.1 Objectives

Increasing demands imposed upon our natural resources as a result of population growth, urbanization, and industrialization have made it necessary to take specific steps for protection of these resources. Our ability to manage them depends in part upon our understanding of the important structural and functional aspects of the natural systems which we are trying to protect. This understanding provides a basis for impact assessment and input to a decision-making process involving economic and social, as well as environmental interests. Methods for interpretation and synthesis of data and theory in formulating impact assessments are of primary concern in this work.

Environmental impact assessment generally entails the use of models and data. Models, defined as "abstract representations of form and function"<sup>6</sup>, may range from "rules of thumb", which qualitatively capture essential aspects of system dynamics, to complex, mathematical simulation models. In a management context, the role of a model is to relate "response" variables to "stimulus" variables. Some of the latter typically include potentially controlled boundary conditions or manageable aspects of the system itself. In this way, a model provides a basis for control strategy designs and impact assessments. Data can be grouped into two general categories according to whether they describe boundary conditions or system states and, as such, represent either stimulus or response variables. Both types of data are required for the detection and assessment of problems and for the estimation and application of models.

### 1.1 Objectives (continued)

Z

It is evident that, in analyzing a given system, a planner, engineer, or manager must make numerous decisions in the process of gathering, developing, and synthesizing data and theory in order to formulate and apply a model. In evaluating a given policy, the goal of the analyst should be to utilize available monitoring and modelling resources most effectively to creat his "best" estimate of the state of the system under the conditions imposed by that policy. A variety of factors could be considered in the definition of "best", including that the estimate should have minimum bias, minimum variance, and be robust to errors in the data or theory or to errors in judgment or implementation committed by the analyst himself. The hope is that policy recommendations derived from this process are independent of the particular analytic decisions made, provided that the choices are among apparently equally valid approaches and are internally consistent.

Generally, the availability of a wide selection of techniques to the sxperienced analyst can increase his freedom of choice, perspective, and ability to make most effective use of the resources available to him. This thesis is intended as a demonstration and evaluation of some techniques appropriate for ecosystem analysis. It is suggested that facility with such techniques should serve as a supplement to, rather than a substitute for familiarity with the physical, chemical, and biological aspects of the systems being monitored and modelled. Various methods are demonstrated in the context of the general problem of lake water quality management, but could be applied as well in analyses of other types of environmental systems.

### 1.2 Causation and Correlation

C.

( )

In some respects, the general problems confronting modellers of environmental systems are perhaps more closely related to those confronting social scientists than to those confronting physical scientists. Because of the size, complexity, and relatively uncontrollable nature of the systems being studied, research tends to be more non-experimental than experimental in form. While various aspects of the systems can be studied individually in controlled laboratory or semi-controlled microcosm experiments, rarely is there an opportunity to experiment with the system as a whole. Under such conditions, causation can be difficult to establish. Regardless of the level of empiricism of a model, a statistical fit of uncontrolled system behavior establishes association, but not causation<sup>4,6</sup>.

In systems whose internal forms and functions are well-understood theoretically, causal assumptions are perhaps less risky than in systems whose components are less well-understood and which are represented by more empirical models, particularly if the independent variables of such systems are correlated with each other<sup>4</sup>. Regardless of the type of model, causality cannot be absolutely established in a purely logical sense unless there is an opportunity for controlled experimentation. We can observe changes in environmental systems following accidents or implementation of management strategies, but these may occur in the contexts of other, natural variations and are not controlled situations.

Ĩ.

ſ

C

On the other hand, environmental systems are real and must conform to the laws of physical and biological science. These types of absolutes do not have analogues in many other non-experimental fields. They include, for example, the law of continuity (mass balance) and the laws of thermodynamics. Models at all levels of empiricism must conform to such laws. These constraints on the systems and models can greatly facilitate analyses and inferences of causation at a practical level. Because of these aspects, a purely statistical approach to the analysis of natural systems would generally not be acceptable. A better approach would be founded upon physical, chemical, and biological principles and flavored with statistical notions of "systems", "models", and "data".

X

Models exist along a gradient of empiricism. All are ultimately based upon observations, but they differ with regard to the level at which the observations are made. Less empirical, or more theoretical models attempt to simulate mechanisms and are based upon observations of the behavior of isolated components of the system under controlled laboratory conditions and/or upon accepted principles. These models tend to be complex and often require the estimation of relatively large numbers of parameters whose values cannot always be derived directly from laboratory studies. Design of controlled experiments to test mechanistic theories and/or develop parameter estimates for use in large-scale simulation models can be difficult, because of the potential for over-simplification. For this reason, such experimental results often cannot be directly applied to simulations of field conditions. More empirical, or "black box" types of models are based upon observed associations of system behavior with properties of the system itself or of its boundary conditions. Such models cannot be properly applied outside of the range of conditions in which they have been calibrated. Despite this lack of generality, the "black-box" models are usually relatively simple and do not suffer as much from the parameter estimation difficulties characteristic of the more complex, theoretically-based systems models.

Another type of gradient is complexity. Above, it was implied that the complexity and empiricism axes are not independent of each other, in that empirical models tend to be less complex than theoretical

### 1.3 Types of Models (continued)

T

ones. However, this is not necessarily the case. Regardless of the level of empiricism, each model is applied at a level of temporal and spatial aggregation, which should be selected based upon characteristics of the system, types and amounts of data available, and types of management strategies being evaluated. Another type of complexity pertains to model characteristics within each spatial element and time domain. In a theoretical model, this is determined by the number of processes or mechanisms simulated. In an empirical one, it is a function of the number of independent and dependent variables employed.

There are numerous trade-offs involved in the specification of model complexity. If important mechanisms or independent variables are not included, biased parameter estimates and a distorted view of system dynamics can result  $^{4,6}$ . On the other hand, overly-complex models are cumbersome, not "parsimonious"<sup>5</sup>, and have demanding data requirements, both for specification of boundary conditions and for estimation of parameters. Errors in such data can lead to distortion and bias, just as can omission of important variables or processes. The expense of implementing a complex model can limit the analyst'e opportunities to fully explore and exploit the model in various ways. Optimally, a model should probably be as simple as possible, given the important processes in the system and the intended uses for the model.
# 1.4 Model Specification, Estimation, and Verification

There are essentially three steps involved in the development of a model: specification, estimation, and verification. These steps, outlined in Figure 1.4-1, rely upon four types of information: (1) established theory; (2) extraneous data obtained from experiments or other natural systems; (3) historical data characterizing the boundary conditions of the system(s) being modelled; (4) historical data characterizing the states of the system(s) being modelled.

Model specification involves definition of (1) the system control boundaries; (2) the variables of importance; (3) equations relating the variables. This step relies upon established theory, as well as upon analysis and interpretation of data characterizing the system and its boundary conditions. The roles of data at this stage can be significant, particularly in highly empirical models. An extreme example would be the use of stepwise linear regression, in which the selection of independent variables is dictated by the data. In another situation, data might be analyzed to determine the levels of spatial and temporal aggregation which are appropriate. The selection of independent or forcing variables at this stage can be critical. If important ones are omitted, biased parameter estimates and a distorted representation of system dynamics could result. This type of problem has led some investigators to believe that, in nonexperimental research, model specification is the most important of the three steps described in this section. The hypothesized model

1-7

:...





Figure 1.4-1

C

(

C:

is of the general form\*:

$$y_i = f(x_i, p), \quad i = 1, n, \quad (1.4 - 1)$$

where,

1

- $\chi_i = \text{vector of state variables for case}$ or observation *i*, dimension *n*
- $x_i =$  vector of independent variables of boundary conditions, dimension n.
- $\underline{p}$  = vector of parameter values, dimension  $n_{p}$
- n<sub>i</sub> = total number of cases or historical measurements

If the model is being developed to represent a number of different systems,  $n_i$  corresponds to the number of cases for which observations are available. If the model is dynamic, and being used to simulate one system,  $n_i$  corresponds to the number of times at which system measurements are available. In the latter case, the equation must

\* Notation: A, <u>A</u>, and <u>A</u> represent a scalar, a vector, and a matrix, respectively.  $\underline{A}^{T} = \text{transpose of vector } \underline{A}$  $\underline{A}^{-1} = \text{inverse of matrix } \underline{A}$  $\text{det } (\underline{A}) = \text{determinant of matrix } \underline{A}$  à.

be integrated through time, so that the elements of  $\underline{y}_{\underline{i}}$  are also functions of the historical values of elements of the  $\underline{y}_{\underline{i}}$  and  $\underline{x}_{\underline{i}}$  vectors.

The process of parameter estimation transforms a hypothesized model into a calibrated one. This involves the use of extraneous data and/or a matched set of observations made on the boundary conditions and on the system. Generally, complex, theoretically-based models have relied more heavily upon the former, while the simpler, empirical models have relied primarily upon the latter as bases for parameter estimates. There is no reason, however, why theoretical models could not rely upon data of both types, provided that model verification is carried out carefully. The relative difficulty of estimating parameters of theoretical models based upon system data has probably been the primary reason for the scarcity of such attempts. The complexity, and often time-variable nature of these models renders the estimation problem somewhat more involved, but, in many cases, equally as feasible as linear least squares, given the availability of computing resources.

Parameter estimation involves specification of the elements of the vector  $\underline{p}$ . For those elements which are estimated from observations of the system, the objective is to select the parameter values which maximize the agreement between observations and model predictions. In the case of a one-variable model,  $(n_y = 1)$ , a commonly employed objective is to minimize the sums of squares of the residuals:

$$e_{i} = y_{i} - f(\underline{x}_{i}, \underline{p})$$
 (1.4 - 2)  
 $\phi = \sum_{i=1}^{n_{i}} e_{i}^{2}$  (1.4 - 3)

Bard<sup>2</sup> has shown that, for the general case, maximum-likelihood estimates of the parameter values can be obtained by maximizing a function of the following form:

. .

$$G_{ML} = -\frac{n_y n_i}{2} \ln(2\pi) - \frac{n_i}{2} \ln[\det(\underline{y})] - \frac{1}{2} \sum_{i=1}^{n_i} \frac{e_i^T [\underline{y}^{-1}]}{i} = (1.4 - 4)$$

where,

÷

ſ.

Ć

C .

$$G_{ML} \approx$$
 generalized maximum likelihood function  
 $\underline{Y}_{Y} \approx$  covariance matrix of residuals  
 $\underline{e}_{i} \approx$  vector of residuals for observation i

G<sub>ML</sub> represents the logarithm of the posterior probability of the observations, given the predictions derived for a set of parameter estimates. While other general criteria have been employed, maximum likelihood estimators have generally been the most popular for use

in parameter estimation problems<sup>2</sup>, because of the desirable properties discussed below.

Parameter estimates derived from maximization of the above function will be unbiased, possess minimum variance, and be normally distributed when the following conditions hold<sup>2</sup>:

(1) The residuals have zero means.\*

$$E(e_{ij}) = 0.$$
  $i = 1, n_i$  (1.4-5)  
 $j = 1, n_y$ 

(2) The residuals from different observations (cases) are uncorrelated:

$$E(e_{ij}e_{lk}) = 0. \quad i \neq l$$
 (1.4-6)  
j,k = 1, n<sub>y</sub>

(3) The residualy from each observation have identical normal distributions with covariance matrix  $\chi_{y}$ :

$$E(e_{ij}e_{ik}) = \{\underline{v}_{y}\}_{jk}$$
  $j,k = 1, n_{y}$  (1.4-7)

\* E(x) = expected value of x

C:

イーエイ

In order to apply equation (4) some assumptions must be made about the appropriate form of the covariance matrix of residuals,  $\underline{V}_{y}$ . In one respect,  $\underline{V}_{y}$  can be known beforehand, known to within a multiplicative constant, or unknown. In another,  $\underline{V}_{y}$  can be assumed to be a general, symmetric matrix or a diagonal one. In the diagonal case, the residuals for different observed variables are assumed to be uncorrelated. The appropriate form for the objective function derived from the above " general equation depends upon the assumptions made about  $\underline{V}_{y}$ . In the general case,  $\underline{V}_{y}$  is assumed to be unknown and with potentially non-zero off-diagonal elements, and equation (4) reduces to:

$$G_{ML} = -\frac{n_{i}n_{y}}{2} \left[1 + \ln(2\pi)\right] - \frac{n_{i}}{2} \ln\left[\det\left(\frac{1}{n_{i}}\right)\right] \qquad (1.4-8)$$

$$\underline{\underline{A}} = \sum_{i=1}^{n_i} \underline{\underline{e}}_i^{\mathrm{T}} \underline{\underline{e}}_i \qquad (1.4-9)$$

where,

r

#### $\underline{A}$ = moment matrix of residuals

Likelihood functions appropriate for other assumed forms for the residual covariance matrix,  $\chi_y$ , are given in Bard <sup>2</sup>. In situations

C

where prior estimates of the parameters are available, based, for example, upon extraneous experimental data or data from other natural systems, Bayee Theorem<sup>2</sup> can be applied and another term added to  $G_{\rm ML}$ to account for the prior information, which is assumed to be given in the form of a vector of mean  $\underline{p}^{\circ}$  and covariance matrix  $\underline{V}_{,\circ}$ :

$$G_{\rm B} = G_{\rm ML} - \frac{1}{2} (\underline{p} - \underline{p}^{\rm O})^{\rm T} \underline{v}_{\rm p}^{-1} (\underline{p} - \underline{p}^{\rm O})$$
 (1.4-10)

In this way, information obtained both from extraneous sources and from the system itself can be incorporated into the parameter estimates. These extraneous sources could also include the opinions of experts familiar with aspects of the systems represented by various parameters, provided that such "opinions" can be translated into mean parameter values and corresponding standard deviations.

For some models, the parameter vector maximizing  $G_{ML}$  can be solved for explicitly, e.g. multiple linear regression. In many cases, however, the models are nonlinear and/or dynamic and finding the maximum of the objective function takes on the form of a nonlinear programming problem. Various search<sup>2</sup> or gradient<sup>1,2</sup> methods can be employed to locate the solution. Since the cost of evaluating the objective function for a given set of parameter values can be appreciable, particularly in the case of a dynamic model, methods which require a minimal number of

ų.

function evaluations are generally preferred. Bard<sup>2</sup> recommends the use of gradient methods, which require estimates of the first and, in some cases, second derivatives of the objective function with respect to the parameter values in order to quide convergence. These derivatives can be computed from analytic derivatives of the model equations or by using finite-difference techniques. Starting with an assumed set of parameters, gradient methods are essentially iterative, hill-climbing algorithms with select step length and direction in parameter space to guide convergence to the solution. Typically, the process is stopped when fractional changes in each element of the parameter vector with each iteration become less than some specified amount ( $\sim$  0.0001)<sup>2</sup>. The "solution" obtained in such a way cannot be guaranteed to be global, because the response surface may be irregular, with several local optima. Because of this aspect, the algorithm can usually be started from a number of different locations in parameter space to test for the significance of alternative solutions.

The use of nonlinear parameter estimation routines in dynamic modelling of environmental systems has been demonstrated by a number of investigators. Radha Krishnan et al.<sup>14</sup> have done extensive work in evaluating various algorithms for parameter estimation in dynamic models of streams using eimulated data. Hornberger et al.<sup>9</sup> have attempted to apply Bard's algorithms<sup>1,2</sup>to experimental data for the purpose of estimating kinetic parameters in algal growth rate formulations. Yih and Davidson<sup>16</sup> have evaluated three algorithms for use in estimating effective longitudinal disper-

sion coefficients in a dynamic salinity intrusion model of the Delaware River Estuary. Johnston and Pilgrim<sup>10</sup> have demonstrated the use of formal optimization methods in estimating the parameters of a watershed model. A recursive estimation algorithm (the Extended Kalman Pilter) has been used by Beck and Young<sup>3</sup> in systematically identifying the structure of a model for dissolved oxygen in England's River Cam.

One important advantage in posing the parameter estimation problem in a formal way is that approximate estimates of the covariance matrix of the parameters can be derived<sup>2</sup>:

$$[\underline{H}]_{jk} = (\frac{\partial^2 G}{\partial p_j \partial p_k}) \qquad (1.4-11)$$

$$\underline{\underline{V}}_{\mathbf{p}} \stackrel{\simeq}{=} \left(\underline{\underline{H}}^{-1}\right)_{\underline{p}} \stackrel{\simeq}{=} \underline{\underline{p}}^{*} \tag{1.4-12}$$

where,

×.

(

 $H = "Hessian" \simeq matrix of second partial$ derivatives of the objective functionwith respect to the parameters $<math display="block">\frac{V}{P} = covariance matrix of parameter estimates$ G = G<sub>ML</sub> or G<sub>B</sub> = log-likelihood functionp\* = parameter vector of solution

The required derivatives are estimated in the process of locating the solution, if a gradient method is employed. The covariance matrix

is useful (a) in interpreting the relative significance of various parameters and the processes or mechanisms they represent; (b) in assessing the adequacy of the amount of data used for parameter estimation purposes; (c) in estimating the portion of model prediction error attributed to uncertainty in the parameter estimates; (d) in estimating parameter confidence regions. It should be noted that estimates of the covariance matrix obtained according to equation (12) are only approximate in the case of a nonlinear model.

If unknown, the covariance matrix of residuals can also be  $\frac{2}{2}$  estimated from the following formula :

$$\underline{\underline{\mathbf{y}}}_{\mathbf{r}} = \underline{\underline{\mathbf{y}}}_{\mathbf{y}} = \frac{1}{n_{\mathbf{i}}} \underbrace{\underline{\underline{\mathbf{h}}}}{n_{\mathbf{j}}} \underbrace{\underline{\underline{\mathbf{h}}}}{n_{\mathbf{j}}}$$
(1.4-13)

The covariance matrix is useful in assessing the predictive abilities of the model. The covariances of the residuals can k attributed to measurement errors in the observations and to model error.

Once the parameters have been estimated, regardless of whether or not the formal procedure outlined above is employed, the model is considered calibrated. The next step, verification, (Figure 1.3-1) tests the adequacy of the model for its intended uses. If the model fails the test, the entire specification, estimation, and verification procedure

may be repeated with a revised model. If it passes, the model is considered suitable for application.

The term "verification" has been used and misused with a variety of different meanings in the environmental modelling field. As noted above in Section 1.2, we rarely establish the truth of a model, but rather we test whether or not that truth can be rejected. Because we cannot conduct controlled experiments, "verification" is usually just a test for association and not, in itself, sufficient. We usually rely upon independent evidence or established principles as bases for assumptions of causation, a necessary condition for true "verification". It is important to consider that verification also depends upon the intended uses for the models. Highly empirical models cannot be verified for use under conditions outside of the range in which they have been calibrated, i.e. the range in which the systems themselves have been observed. Whereas theoretical models of relatively well-understood systems may be verified for such use, provided that all of the relevant processes are incorporated and independent means of estimating parameters are available.

The key to testing whether or not a model can be rejected is in analyzing the residuals. A variety of tests can be applied to the residuals in order to determine whether the principle assumptions of the parameter estimation exercise (equations (5) to (7)) have been violated  $^{7}$ . The residuals can be plotted against the independent variables in the model to reveal possible inadequacies in assumed

( )

functional forms or against variables not included in the model to determine whether potentially important factors have been ignored. For dynamic models, serial correlation in the residuals may reflect the effects of errors in specification of boundary conditions or the effects of factors not considered in the model. The accuracy of the model, as gauged by the standard deviations of the residuals, can be compared with expected measurement errors (which would represent lower limits for residual standard errors), with the desired accuracy for model applications or with the accuracies of alternative models. The last could serve as a partial basis for model selection.

A supplemental criterion for verification is parameter stability. To test for this, data can be divided into two or more groups and optimal parameters estimated separately for each group. The parameters could then be tested for statistically-significant differences among groups. If differences are not detected, a single set of parameter values could be considered to be equally appropriate over the range of conditions represented by the various groups. This would tend to support the generality of the model and the consistent quality of the independent variable data. For maximum efficiency, the parameters should probably be re-estimated based upon all the data, once the model is considered "verified". This strategy is similar to that of Beck and Young<sup>3</sup>, who used the temporal stability of recursive parameter estimates as evidence for correct identification of a model for dissolved oxygen in a river.

1

There are a variety of approaches and techniques which can facilitate full and proper use of a model. Three aspects of concern in this work are sensitivity analysis, simulation, and error analysis. These are introduced briefly below.

Potentially useful information about some of the important controlling factors and processes in a system can be derived from sensitivity analysis. Patten has suggested that such an analysis be used as a means of identifying control strategies and has outlined a general approach for application to lakes. Miller has shown that sensitivity analysis can yield useful information, even under conditions of uncertainty in the model or parameter estimates. Sensitivities of model simulations to parameter values can be used as a basis to assess important controlling processes within the system, while sensitivities to independent variables can indicate the relative controlling influences of external factors, some of which may be manageable, possibly acting as "decision veriables" in a design problem. This type of analysis can also be used as a basis for simplification of the model. A useful way of expressing sensitivities are as normalized first partial derivatives of the model dependent variables with respect to the factor being studied:

$$\frac{S_{j}}{S_{j}} = \frac{\partial Y}{\partial w_{j}}$$
(1.5-1)

where,

- $\underline{S}_{j}$  = vector of sensitivities of state variables with respect to factor j

Multiplication by w<sub>j</sub> essentially corrects for differences in factor scales and permits direct comparisons among different factors. The coefficients can be determined analytically or by using finite-difference techniques. In a dynamic model, the sensitivity coefficients are timevariable.

Control strategies can be viewed as modifications to the vector of independent variables. Most often, many of the independent variables are not influenced by management strategies. Meteorologic regimes are usually not controllable, but can have significant effects upon the systems being studied. For this reason, control strategies should be evaluated in the context of the variability induced by fluctuations in such factors. In some situations, "critical" conditions can be easily defined and employed in simulating management strategies. A classic example of this is evaluating the effects of biochemical oxygen demand discharges on dissolved oxygen in a river under low-flow conditions. In others, "critical" conditions may not be as evident, and simulations of historical records under the conditions imposed by each management

strategy can be used as a basis for evaluating and comparing alternative schemes. If sufficient historical records are not available, a synthetic record might be generated with time series models identified and estimated from data obtained from the geographical region being studied or from one similar to it. The range of system variability induced by fluctuations in the uncontrollable factors can be compared with the range induced by alternative management strategies as a means of estimating the potential detectability of changes induced by such strategies.

Error analysis can be a useful and revealing aspect of model application. Generally, covariance in model predictions can be attributed to three error components:<sup>2</sup>

> Total Parameter + Variable + Residual Error Error Error Error

$$\underline{\Psi}_{\hat{Y}} \stackrel{\simeq}{=} \left(\frac{\partial \hat{Y}}{\partial p}\right) \underbrace{\Psi}_{p} \left(\frac{\partial \hat{Y}}{\partial p}\right)^{T} + \left(\frac{\partial \hat{Y}}{\partial x}\right) \underbrace{\Psi}_{x} \left(\frac{\partial \hat{Y}}{\partial x}\right)^{T} + \underbrace{\Psi}_{r}$$
(1.5-2)

In order to evaluate the total error, we need to estimate three covariance matrices:  $\underline{V}_p$ ,  $\underline{V}_x$ ,  $\underline{V}_x$ . The last can be estimated by comparing model predictions with system observations and is usually

ľ

obtained in the process of estimating the model parameters. The parts of  $\frac{V}{P}$  corresponding to those parameters which have been estimated directly from system observations can be estimated according to equation 1.3-12. The covariances of the remaining parameters and  $\frac{V}{2x}$  must be estimated independently, often with only very approximate results. The required sensitivity vectors can be obtained from differentiation of the model equations, again using analytic derivatives or finitedifference methods. The error should be evaluated for each case or management strategy studied. Although the individual covariance matrices may be invariant, the sensitivities and total prediction error may change from one case to the next. For a dynamic model, the total error and the first two terms in equation (2) are also time-variable.

The residual error component includes both measurement and model errors:

Residual \_ Measurement + Model Error Error Error

 $\underline{\underline{v}}_{r} = \underline{\underline{v}}_{m} + \underline{\underline{v}}_{e}$ (1.5-3)

These two components usually cannot be distinguished without some independent evidence. For instance, measurement error could be

(

estimated from replicate observations.

If the model and measurement error terms can be separated, a comparison of the parameter error term with the model error term can provide a basis for deciding whether future efforts should be directed at improving the model or at obtaining additional data. The covariances of the parameters can usually be reduced by gathering additional data for model calibration, whereas changes in the model would generally be required in order to reduce the model error component. In the interest of minimizing total prediction variance, future efforts would be directed toward data acquisition or model improvement according to whether the parameter error term is greater or less than the model error term, respectively. The relative sensitivities of these terms to investments in reducing them might also be considered, as discussed below.

In cases where parameter or independent variable errors are significant, Thomas has suggested that an error analysis could also be used as a basis for guiding data acquisition and research efforts. For the one-variable case, and combining the parameter and independent variable vectors into a general vector  $\underline{w}$ , the variance attributed to uncertainty in the elements of  $\underline{w}$  is given by:

$$\sigma_{\hat{y}}^{2} \simeq \left(\frac{\partial \hat{y}}{\partial w}\right) \underbrace{Y}_{w} \left(\frac{\partial \hat{y}}{\partial w}\right)^{T}$$
(1.5-4)

**.** 

If the off-diagonal elements of  $\underline{V}_{,,}$  are assumed to be zero:

$$\sigma_{\hat{\mathbf{y}}}^{2} \approx \sum_{i=1}^{n_{\mathbf{w}}} \left( \frac{\partial \hat{\mathbf{y}}}{\partial \mathbf{w}_{i}} \right)^{2} \sigma_{\mathbf{w}_{i}}^{2}$$
(1.5-5)

Given the objective of minimizing the total prediction variance, one strategy would be to invest additional resources into reducing the variance of the component contributing most to the total. According to this scheme, the various terms of equation (5) would be compared and the factor corresponding to the largest term would serve as a focus for additional monitoring and/or research efforts. Another strategy would consider cost-effectiveness by taking into account the fact that the variances of the parameters or independent variables may have different sensitivities to investment, i.e., some eetimates might be more difficult to improve than others. According to this scheme, Thomas suggested that the variables could be ranked according to:

$$\begin{pmatrix} \frac{\partial \hat{y}}{\partial w_{i}}^{2} \begin{pmatrix} \frac{\sigma_{w_{i}}}{\sigma_{\hat{y}}} \end{pmatrix} \begin{pmatrix} \frac{\partial \sigma_{w_{i}}}{\partial c_{i}} \end{pmatrix}$$
(1.5-6)

where,

( )

(

Harrington<sup>8</sup> has reviewed applications of these and similar strategies to the design of ambient monitoring programs. In a similar vein, Moore<sup>12</sup> has proposed the use of Kalman Filtering algorithms to permit estimation of system states based both upon model simulations and upon field measurements. Using a minimum variance criterion, he has shown how this scheme can lead to rational designs for aquatic ecosystem monitoring programs. Strategies such as these can be used to guide research and/or monitoring programs and represent potentially fruitful areas of model application.

Another benefit derived from an error analysis is the capability of placing confidence limits on model predictions. Estimates of the first and second moments of predictions can be used, for example, to calculate the probability that implementation of a given management strategy will result in satisfaction of a system standard or quality criterion. Alternatively, management strategies can be designed to satisfy a given probability of achieving system standards. A probabilistic projection, as compared with a deterministic one, provides the policy-maker with a better decision basis and the analyst with an escape route.

#### 1.6 Introduction to Illustrations in Subsequent Chapters

Chapter 2 introduces the general problem of cultural eutrophication in lakes and reviews some of the roles of monitoring data and models in this application. Using data from a cross-section of lakes, some empirical approaches are demonstrated in the development, estimation, and evaluation of models for prediction of phosphorus concentration and lake trophic state as functions of phosphorus input rates, and hydrologic and morphometric characteristics. Techniques employed include nonlinear parameter estimation and discriminant analysis. Error analyses are employed to (1) propose a scheme for rational allocation of sampling effort in a monitoring program designed to gather data for lake nutrient budget estimation; (2) assess the relative sizes of independent variable, parameter, model, and measurement errors involved in prediction of lake phosphorus and chlorophyll concentrations; (3) develop a means of incorporating the effects of uncertainty in decision variables on rational design values for lake nutrient loading allocation. It concludes with a discussion of model selection criteria. The study is cross-sectional in nature and provides a basis for a more detailed, longitudinal case study of Onondaga Lake, New York, in Chapters 3 and 4.

Chapter 3 introduces Onondaga Lake: its environment, historical problems, and proposed solutions. The bulk of the chapter describes and explores the existing data base on the lake and its tributaries, one of the most extensive data bases of its type known to this author. Techniques for reduction and display of the data are employed to reveal

1.6 Introduction to Illustrations in Subsequent Chapters (continued)

(

í

1

variations and associations in time and space. A method for examining time series behavior is explored. Techniques for estimating and displaying mass balances on a continuous basis are developed and demonstrated. While attempts are made to mechanistically interpret some of the observed relationships, the formal use of models is not stressed. A summation of proposed management strategies for the lake is used in combination with the results of data analyses to define the modelling problem addressed in Chapter 4.

In Chapter 4, a model for vertical stratification in Onondaga Lake is developed, estimated, and applied. The objective is to evaluate the potential impact of the design and location of a waste outfall upon general features of lake mixing. In the context of this problem, the use of nonlinear programming algorithms for estimation of parameters in dynamic models is demonstrated and evaluated. Other techniques employed include sensitivity analysis, simulation, and error analysis. The chapter concludes with some general comments on the adequacies of the model, the data, and the techniques employed to address the defined problem, as well as some recommendations for outfall design.

#### **REFERENCES - CHAPTER 1**

đ.

ſ

1

- Bard, Y., "Comparison of Gradient Methods for the Solution of Nonlinear Parameter Estimation Problems", <u>Society of Industrial</u> <u>and Applied Mathematics</u>, <u>Journal of Numerical Analysis</u>, Vol. 7, No. 1, March 1970.
- 2. Bard, Y., <u>Nonlinear Parameter Estimation</u>, Academic Press, New York, 1974.
- Beck, B. and P. Young, "Systematic Identification of DO-BOD Model Structure", Journal of the Environmental Engineering Division, <u>American Society of Civil Engineers</u>, Vol. 102, No. EE5, pp. 909-927, October 1976.
- 4. Blalock, H.M., "Causal Inferences in Non-Experimental Research", University of North Carolina Press, Chapel Hill, North Carolina, 1961.
- 5. Box, G.E.P. and G.M. Jenkins, <u>Time Series Analysis</u>, Forecasting, and <u>Control</u>, Holden-Day, San Francisco, 1970.
- 6. DeNeufville, R. and J.H. Stafford, <u>Systems Analysis for Engineers</u> and Managers, McGraw-Hill, New York, 1971.
- 7. Draper, N.R. and H. Smith, <u>Applied Regression Analysis</u>, Wiley, New York, 1966.
- Harrington, J.J., "Monitoring Program Design A Systems Approach", presented at "A Seminar and Workshop on Water Monitoring Technology", University of Delaware, May 1976.
- Hornberger, G.M., M.G. Kelly, and T.C. Lederman, "Evaluating a Mathematical Model for Predicting Lake Eutrophication", Virginia Water Resources Research Center, Virginia Polytechnic Institute and State University, Bulletin 82, November 1975.
- Johnston, P.R. and D.H. Pilgrim, "Parameter Optimization for Watershed Models", <u>Water Resources Research</u>, Vol. 12, No. 3, pp. 477-486, June 1976.
- 11. Miller, D.R., "An Experiment in Sensitivity Analysis on an Uncertain Model", Simulation, Vol. 23, No. 3, September 1974.
- Moore, S.F., "Estimation Theory Applications to Design of Water Quality Monitoring Programs", <u>Journal of the Hydraulics Division</u>, <u>American Society of Civil Engineers</u>, Vol. 99, No. H45, pp. 815-831, 1973.

- Patten, B.C., "The Need for an Ecosystem Perspective in Eutrophication Modelling", in Middlebrooks, E.J., D.H. Falkenborg, and T.E. Maloney, Eds., <u>Modeling the Eutrophication Process</u>, Ann Arbor Science, Ann Arbor, Michigan, 1974.
- 14. Radha Krishnan, D.P., J.J. Lizcano, L.E. Erickson, and L.T. Fan, "Evaluation of Methods for Estimating Stream Water Quality Parameters in a Transient Model from Stochastic Data", <u>Water Resources Bulletin</u>, <u>American Water Works Association</u>, Vol. 10, No. 5, pp. 899-913, October 1974.
- Thomas, H.A., Jr., "Operations Research in Disposal of Liquid Radioactive Wastes in Streams", Harvard Water Resources Group, December 1965.
- Yih, S. and B. Davidson, "Identification in Nonlinear, Distributed Parameter Water Quality Models", <u>Water Resources Research</u>, Vol. 11, No. 5, pp. 693-704, October 1975.

#### 2.1 Introduction

(

A lake can be viewed as a system which responds in various ways to oulturally or naturally induced changes in its boundary conditions. Assessment of lake water quality problems is a good context in which to illustrate some of the analytic techniques and strategies discussed in Chapter 1. In the following chapter, the roles of data and models in managing lake water quality are discussed with an emphasis on the general problem of cultural eutrophication. Introductory sections review aspects of lake classification, nutrient balance estimation, and lake modelling. In Section 2.5, data from over 100 northern-temperate lakes are used to evaluate and compare some of the simpler lake models which have been proposed. Nonlinear regression and discriminant analyses are employed to develop empirical models for phosphorus retention and trophic state projection. Strategies for application of these models are discussed with a particular emphasis on parameter estimation, error analysis, and data requirements. The chapter concludes with a brief overview and discussion of model selection criteria in Section 2.6. Besides providing a context in which to illustrate some of the various techniques discussed in Chapter 1, this chapter serves as a partial basis for a more detailed case study of Onondaga Lake in Chapters 3 and 4.

ſ

Lakes have been classified according to various physical, chemical, and biological characteristics. Since lakes exist on a continuum, classification into discrete states is rather artificial, but serves as a basis for organization and comparison. While classifications are defined on arbitrary scales, some may be more theoretically-based than others. Schemes for use in water quality management should relate to established criteria, which, in turn, should reflect potential for beneficial use and, possibly, some fundamental measures of ecosystem health.

An important aspect to consider is that lakes are generally not static entities. The term "eutrophication" is generally used to describe the natural evolution of lakes from deep, nutrientpoor systems with low biological productivity to shallow, nutrientrich systems with high productivity, and, eventually, into swamps and meadows. The phrase "cultural eutrophication" more accurately reflects water quality management concerns because it describes the acceleration of this natural process as a result of cultural interventions. This acceleration is generally considered to be a result of enhanced flow of nutrients into the lake ecosystem.

Traditionally, biological productivity has served as a primary basis for classification of lakes into three basic types: "oligotrophic," "eutrophic," and "dystrophic." The history and basis of this classification scheme have been reviewed by Welch<sup>98</sup> T

and are summarized in Table 2.2-1. Oligotrophic lakes are relatively low in productivity, organic matter and nutrient content, and relatively high in hypolimnetic dissolved oxygen and transparency, while eutrophic lakes have opposite characteristics. The term "mesotrophic" refers to an intermediate state between the above. Dystrophic lakes are distinguished by their high humic acid content, which renders a characteristic yellow or brown color. Organic matter in these lakes is generally allochthonous (originating in the watershed), whereas, organic matter in oligotrophic and eutrophic lakes is autochthonous (produced within the lake). The effect of light limitation limits the submerged, biological productivity of dystrophic lakes to relatively low levels, although oxygen levels are similar to those found in eutrophic lakes, due to decay of organic matter entering the lake from external sources. The natural succession of oligotrophic lakes is toward the eutrophic state, that of eutrophic lakes, toward swamps, and that of dystrophic lakes, toward peat bogs. While there are some quantitative aspects to this traditional classification scheme, it is primarily descriptive.

A number of investigators  $^{73,75,88,92}$  have suggested some more quantitative classification schemes in attempts to remove some of the subjectivity from the traditional method while retaining essentially the same interpretation. These schemes, reviewed by Uttormark and Wall<sup>92</sup>, have been based upon such indices as minimum hypolimnio dissolved oxygen, transparency, chlorophyll-a, available

# 2-4

# Table 2.2-1

C

ť

(

# Principal Characteristics of Oligotrophic, Eutrophic,

and Dystrophic Lakes According to Welch 98

| the second se |                                                                                                                                                                                                                                                     |                                                                                                                                                                                                                                                                                       |                                                                                                                                                                                                                                 |
|-----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Characteristics                                                                                                 | Olizatrophie type                                                                                                                                                                                                                                   | Eutrophie type                                                                                                                                                                                                                                                                        | Dystroyble type                                                                                                                                                                                                                 |
| General form relations                                                                                          | Deept shoals narrow or absects volume<br>of hypolymniun greater than epilimpion                                                                                                                                                                     | Shallew; shoals broad; volume of bype-<br>limnion emailer than epilimnion                                                                                                                                                                                                             | Drep to shallow; fe bog surroundings of<br>in ohl resourcine                                                                                                                                                                    |
| Color of water                                                                                                  | . Blue to green; characteristic color siway,<br>prodominant                                                                                                                                                                                         | Green to yellow and brownish green;<br>characteristic color aften concessed by                                                                                                                                                                                                        | Vellaw to brawn; characteristic order<br>always predominant                                                                                                                                                                     |
| Тгасарагеогу                                                                                                    | Ilich                                                                                                                                                                                                                                               | Lower: ereatually very low                                                                                                                                                                                                                                                            | Low                                                                                                                                                                                                                             |
| Chemical content of water                                                                                       | Electrolytes veriable; relatively peor in<br>N and P; humie materials absent; Ca<br>variable                                                                                                                                                        | Electrolytes variable; tich is N and F;<br>humle materials elight; Ca ebundana,<br>seldam erany                                                                                                                                                                                       | Electrolyteslow; poor in N and P; humle<br>materials abundant; Ca aranty                                                                                                                                                        |
| Burpended detritue                                                                                              | Malmai quastites<br>Poor la argenie materials; noi putraccat                                                                                                                                                                                        | Large quantities<br>Rich in autochthonous,? puirescent or-<br>genie materials                                                                                                                                                                                                         | Humis materials abundant<br>Poor la suisekthusous, humis materiala<br>rich la slitchthusous materiale                                                                                                                           |
| Dimelyed suyges essioni<br>a. In summer                                                                         | Oxygen content telatively uniform from<br>authors to bottom; so increase in ther-<br>mediase; abundant in Sypohimator;<br>deep watar with 60-70 % acturation;<br>for or no putteristive processes in<br>bottom depends                              | In deep fakes of this type, exygen ton-<br>test in thermocilos suddenly and<br>atrongly increased; little or us exygen<br>in hypolizator; exygen decline of deep<br>waters D-60, 5 asturation, eddons abovy<br>40 S; vigorous purrelation in bottem                                   | As in eutrophic type                                                                                                                                                                                                            |
| 5. In winter under Jos                                                                                          | At is remain                                                                                                                                                                                                                                        | deposite<br>In deep lakes as in oligetrophic type) in<br>shallow fakes, stygen may fall to 6 %<br>enturation at bottom                                                                                                                                                                | As in suizophie type                                                                                                                                                                                                            |
| Cause of oxygen decline                                                                                         | ••••••                                                                                                                                                                                                                                              | In summer, due to plankton and bottom<br>muda; in winter, due to bottom muda                                                                                                                                                                                                          | Due to ellockthosous superaded and<br>deposited hunte deposits; combination<br>with Fo                                                                                                                                          |
| Litteral plante                                                                                                 | Sciaty                                                                                                                                                                                                                                              | Abundant .                                                                                                                                                                                                                                                                            | Benaty                                                                                                                                                                                                                          |
| Plankton                                                                                                        | Quantitatively poor; preset up to large<br>depths; diarank vertical migration wide<br>in reage.<br>Water bloom rare; Chlorophycese and to<br>some extent the distona dominate over<br>the fichisophycese; true hypolizinion<br>phytoplankton absets | Quantulatively sich; diarnal vertical<br>migration restricted to small distance<br>Water bloom sbundant; Bebisophycess<br>and distance together with Chrysomaa-<br>addae and the Peridialscene predomi-<br>nating over the Chicophycess; true<br>hypolization phytopicaliton abustant | Phytoplaaktes quantitativsly seasty<br>Water bloom absest or very rave; Schlac-<br>physess and distens, Chlorophycess,<br>Chrysomassdales, Peridialacess, and<br>damids present; true hypothales<br>phytoplaaktes abuadasi (±). |
| ·                                                                                                               |                                                                                                                                                                                                                                                     | (±)                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                 |
| Distinction between litteral and pro-<br>fundal scars.                                                          | Indistinct; indicated only by termina-                                                                                                                                                                                                              | Bherpiy defined; due chiefly to charge of<br>dissolved exygen relation                                                                                                                                                                                                                | As in europhie type                                                                                                                                                                                                             |
| A. Qualitative                                                                                                  | Bish in aprolos) requiring high arygan<br>outpout: Tanylarous fauna; Corolico<br>actor prusest                                                                                                                                                      | Poor in species; with wide toleration of<br>axygen-content changes; Chironomas<br>fauna; Corribe almost always pressay                                                                                                                                                                | Still poorer in specias; wide toleration at<br>oxygen-contest changes; Chironomus<br>fauna; bottom deposite eventually<br>devoid of founa; Coreles prasat (ecca-                                                                |
| A Quantitative                                                                                                  | Relatively richt is gessent, inn than<br>3,000 bus above 200 animals per sq. m.<br>or 3-4.g. day weight                                                                                                                                             | Rich: 2.000-10.000 salanis perso, m. =<br>20-100 g. dry wight                                                                                                                                                                                                                         | aionally esty Corries)<br>Poer; at most, 10-22 animals per eq. 18-3<br>bottom layer often sone                                                                                                                                  |
| a. Distribution with depth                                                                                      | Vallorith                                                                                                                                                                                                                                           | Not vallers                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                 |
| Qualitative correlation between plankton and the profundel found.                                               | Pressal                                                                                                                                                                                                                                             | Absent                                                                                                                                                                                                                                                                                | Bettern fauns always pears phyto-<br>plankton ports sorplankton often tich                                                                                                                                                      |
| Deep-dwelling coregosies febre                                                                                  | Abradaas                                                                                                                                                                                                                                            | Present only in anceptional inclusion                                                                                                                                                                                                                                                 | Always skame is advaared dystrephic<br>lakes                                                                                                                                                                                    |
| Buereslan                                                                                                       | fale outrophie type                                                                                                                                                                                                                                 | Into punt, anomp, mendow man                                                                                                                                                                                                                                                          | Into post bag                                                                                                                                                                                                                   |

2 (Lunsolarische Terminalogie Abderhulden's Hardback der hielogisches Arbeitemetherten, by permission of Urban und Sekwarsesberg, Berlin, P Autschthanous: produced within the some basie; allochthoneus; erignaling eisewhere.

ſ

and total nitrogen and phosphorus concentrations, and alkalinity. For collections of lakes, these indices are often highly correlated with one another, reflecting their common association with the eutrophication process. Because of the particular effects of light limitation, dystrophic lakes are often excluded or treated separately.<sup>73</sup> Likewise, lakes in which primary production is dominated by aquatic macrophytes generally do not fit into classification schemes developed on lakes dominated by phytoplanktonic production.<sup>73,88</sup> Some schemes have been based upon rankings and have been used primarily for comparisons within a given group of lakes.<sup>88</sup> Other investigators have attempted to derive objective scales for more general application.<sup>73,75,88,92</sup> Both types of schemes, however, depend directly upon the particular lake data set employed as a basis and are accordingly restrictive in scope of application.

Shannon and Brezonik<sup>73,74</sup> have led the work on subtropical lakes with a unique, multivariate ~ statistical approach. Their data base consisted of 55 North and Central Florida lakes. Cluster analysis of water quality data was used to classify the lakes into distinct groups, which roughly corresponded to the classical trophic states. The first principal component of the correlation matrix of traditionally-employed trophic state indicator variables was used to derive a formula for a so-called "trophic state index" (TSI):

TSI = 5.19 + 0.919 
$$\frac{1}{5D}$$
 + 0.800 COND + 0.896 TON + 0.738 TP (2.2-1)  
+ 0.942 PP + 0.862 CHA + 0.634  $\frac{1}{CR}$ 

#### where

C

(

|                |   |                                                               | Mean  | Standard<br>Deviation |
|----------------|---|---------------------------------------------------------------|-------|-----------------------|
| $\frac{1}{SD}$ | = | Inverse Transparency<br>(m) <sup>-1</sup>                     | 0.84  | 0.77                  |
| COND           | = | Specific Conductance ·<br>(micromhos/cm)                      | 93.1  | 101.3                 |
| ton            | = | Total Organic Nitrogen<br>(mg/l)                              | 1.02  | 0,82                  |
| TP             | = | Total Phosphorus<br>(mg/l)                                    | 0.125 | 0.177                 |
| ₽₽             | = | Planktonic Primary<br>Production<br>(mg-C/m <sup>3</sup> -hr) | 44.8  | 82.3                  |
| СНА            | = | Chlorophyll-a<br>(mg/m <sup>3</sup> )                         | 16.9  | 19.8                  |
| $\frac{1}{CR}$ | - | Inverse Cation Ratio                                          | 1.47  | 1,63                  |
| 2.11           | = | $[(Na + K)/(Ca + Mg)]^{-1}$                                   |       |                       |

The coefficients in equation (1) are appropriate for use with standardized values of the respective variables. This expression explained about 70% of the variance in the correlation matrix of the indicator variables. Oligotrophic lakes generally had TSI

#### 2.2 Lake Classification (continued)

values for less than 3, mesotrophic lakes, between 3 and 7, and eutrophic lakes, greater than 7.

The TSI offers some distinct advantages over traditional classification schemes. It provides a more objectives means of ranking lakes according to water quality using a continuous scale which is both more sensitive and more realistic than the traditional, discrete trophic states. The coefficients of variation of five of the seven indicator variables in equation (1) were greater than one, however. This reflects considerable skewness in the distributions of these variables and suggests that a transformation of the variables may have been appropriate prior to the principal component analysis. By promoting normality in the distributions of the variables, transformations would have rendered the analysis more in agreement with the principal conditions for efficiency of the multivariate techniques employed.

As a second phase of their work, Shannon and Brezonik demonstrated correlations between TSI values and land use patterns, population densities, and nitrogen and phosphorus loadings. Multiple regressions of TSI on land use and population density factors explained 81.5% of the TSI variance, while regressions of TSI on estimated phosphorus and nitrogen loadings explained up to 67%. These kinds of relationships were suggested as means of evaluating watershed management strategies for the control of eutrophiciation.

[.

In view of the results of Vollenweider<sup>95</sup> and Dillon<sup>16</sup>, the results might have been revealing if the effects of lake morphometric and hydrologic properties had been taken into account. The significance of the regressions of TSI on nutrient loadings is questionable, since nutrient concentrations contributed directly to the calculated TSI values. In the interest of distinguishing between causes and effects, one might argue against the inclusion of nutrient concentrations in the list of trophic indicators. It would seem more logical to allow the TSI to depend only upon variables which reflect observed ecosystem response or which relate to the potential for beneficial use. Nutrients are generally viewed more as causes than effects and should therefore be excluded from the TSI, according to this rationale.

Despite these criticisms, the general approach of Shannon and Brezonik provides some potentially valuable management tools for application in regional planning. TSI values provide an objective means of ranking lakes with regard to observed water quality. The statistical relationships between TSI values and watershed characteristics provide means of evaluating management strategies and of suggesting feasible subjects for water quality restoration and protection programs. However, the approach is highly empirical and at most as good as the data used to develop the ranking schemes and relationships. Potential effects of data include the effects of original lake selection, measurement errors, and assumptione (

and techniques employed in reducing the data to useful form. More confidence could be placed in results of this type if essentially the same results could be derived from an independent, randomly-selected set of lakes and watersheds in the same geographical region. In addition, the generality of their model could perhaps be improved by the incorporation of more theoretical factors in the analysis, such as lake flushing rate and depth. Perhaps the major contribution of Shannon and Brezonik was the demonstration that multivariate techniques can be applied successfully to problems of this type in order to reduce dimensionality and thereby provide a concise and useful summary of the data.

Uttormark and Wall<sup>92</sup> developed a system for classifying lakes based upon subjective information. Noting that data were generally lacking to permit the application of a classification scheme based directly upon water quality measurements, they devised the "Lake Condition Index" (LCI), which was based upon the point system outlined in Table 2.2-2. In classifying 1100 Wisconsin lakes, Uttormark and Wall constructed a questionnaire which facilitated estimation of LCI values by regional agency personnel who were generally familiar with the lakes in question. In comparing the LCI values with the traditional trophic state rankings for those lakes for which data were available, oligotrophic lakes were generally found to have LCI values of 4 or less, mesotrophic lakes, (

between 5 and 9, and eutrophic lakes, between 10 and 23. Comparing multiple responses for the same lakes, Uttormark and Wall found that estimated LCI values were reproducible to within  $\pm$  2 units for 89% of the lakes tested. The source of this variation was found to lie primarily in the use impairment component, which, in turn, correlated with the particular professional concerns or recreational interests of the individuals or agencies being polled. Based upon a national survey, they concluded that the LCI technique could be used to correctly classify 70-80% of the lakes in the U.S. with surface areas greater than 40 hectares.

#### Table 2.2-2

Point System for Uttormark and Wall's Lake Condition Index

| Component          |       | <u>Range</u> a |
|--------------------|-------|----------------|
| Dissolved Oxyge    | 0~6   |                |
| Transparency       | 0-4   |                |
| <b>Fis</b> h Kills | 0,4   |                |
| Use Impairment     |       | <u>0-9</u>     |
|                    | TOTAL | 0-23           |

a Lower scores are more desirable.

×.

As a second phase of their study, Uttormark and Wall suggested how LCI values might be related to nutrient loading and watershed development factors to demonstrate that the classification scheme could serve as a workable data base for lake renewal and management programs. They suggested that a normalized nutrient loading parameter could be estimated for each lake based upon measured or estimated loadings and upon lake morphometric properties. The normalized loading would be an indicator of eutrophication potential. For instance, according to Vollenweider's first model\* (Pigure 2.4-1), the normalized loading would be given by  $L/Z^{*6}$ , or, according to his second model\* (Figure 2.4-2), it would be given by  $L/QS^{*5}$ . In the absence of measured loading data, land use factors, such as urban drainage area/lake surface area could be employed as surrogates. A plot of the normalized loading against LCI would, in effect, depict potential problems versus observed problems.

A major contribution of Uttormark and Wall was that such a plot (Figure 2.2-1) could be used as a device for "dynamic" classification of lakes. Their basic assumption was that deviations from predicted conditions were due to non-steady-state conditions. Accordingly, four regions could be identified in Figure 2.2-1. Lakes in Region I, lower left, would have low observed and low potential eutrophication problems and would be considered to be in no immediate danger. Conversely, lakes in Region III, upper right,

\* These models will be discussed in detail in Section 2.4.



đ.

(

Ç




Ľ

( :

would be problem lakes, in which extensive nutrient abatement may be necessary in order to effect lasting improvements. Lakes in Region II, upper left, would be considered high priorities for protective actions, since their location on the plot suggests that currently "acceptable" water quality may be deteriorating. Conversely, lake in Region IV, lower right, would be considered likely candidates for restoration programs, since long-term benefits may be possible without extensive nutrient source abatement.

This scheme for dynamic classification of lakes has some obvious applications in regional management of water quality. The basic assumption is that deviations from the perfect fit of potential versus observed eutrophication problems are due to non-steady-state conditions. Of course, data errors would contribute substantially to such deviations. Model error would also be another source of deviations, since all of the factors contributing to variations in the capacities of lakes to handle nutrient influx without adverse effects would not be accounted for in a simple model of the Vollenweider type. With these qualifications in mind, Uttormark and Wall's dynamic classification scheme appears to be promising as a screening method in designing regional water quality management programs. Its potential applications are not restricted to cases in which the LCI is used as a measure of observed lake water quality.

The nutrient balance is an essential element in any analysis of lake water quality problems from a eutrophication viewpoint. Regardless of the complexity of the particular lake model being employed, long-term predictions of water quality can be heavily dependent upon accurate estimation of the boundary conditions imposed by the lake's nutrient inputs, particularly if projections are being made in the nutrient-limited state of the system. Under such conditions, it would be impossible to calibrate or verify a given lake model unless the nutrient inputs were estimated accurately. By quantifying the relative contributions of various nutrient sources in a lake's drainage basin, the balance also serves to identify problem origins and potential control points.

In formulating a nutrient balance, an estimate must be obtained for each potential source in the watershed, averaged over the time scale of interest. The latter is determined by the requirements of the particular lake model being employed. Nutrient sources can be grouped initially into general categories according to the types of mechanisms involved, including hydrologic, cultural, meteorologic, geologic, and biologic processes<sup>52</sup>. Hydrologic fluxes enter or leave the lake in surface- or ground-water flows. Cultural fluxes represent point discharges of municipal, industrial, or agricultural wastes. Meteorologic fluxes are mediated by atmospheric phenomena. Inputs include dustfall and precipitation; outputs, entrainment of nutrients in spray from the water surface. Geologic fluxes represent transfers to or from the lake's sediment, as 2~14

( :

Ć

governed by water and sediment chemistry. Biologic fluxes include contributions from migrant waterfowl, insect emergence, plant or animal harvesting, nitrogen fixation, denitrification, and sediment exchanges.

Because the geologic and many of the biologic fluxes involving the sediment are difficult to estimate or measure directly, the lake's sedimant is usually included within the control boundaries of the system. The accumulation term represents the net sum of all fluxes not estimated directly. Hydrologic, cultural, and, to some extent, meteorogic fluxes are relatively easy to measure and generally represent the primary terms in the balance.

#### 2.3.1 Factors Influencing Nutrient Sources

(

Point sources of nutrients are influenced by a number of factors, including tributary population, types and amounts of industrial or commercial activity, and types and degrees of waste treatment. The kinds of materials being cycled through the urban and industrial systems are also of importance (e.g., detergents). Point sources generally exhibit marked diurnal, weekly, and seasonal periodicities which reflect variations in cultural activities<sup>58,82</sup>. In contrast to nonpoint sources, they are relatively easy to measure and trace.

The flux of nutrients from a watershed can be viewed as leakage from the resident terrestrial ecosystems. Likens and Borman<sup>52</sup> have discussed the various processes which are responsible for the input, cycling, and loss of nutrienta in terrestrial systems. Generally, cultural interventions tend to open the relatively closed nutrient cycles of natural terrestrial environments. This results in an increased flux of nutrients from the watershed and in an increased potential for water quality problems. The concept that management of water quality entails management of the terrestrial environments has become the fundamental theme of programs for controlling nonpoint sources of water pollution<sup>2,52</sup>.

Some nutrients are more tightly conserved than others in terrestrial systems . Biological uptake acts to conserve

4

nutrients by converting soluble forms to forms which are less labile in the hydrologic cycle. Soil chemistry is an important regulating factor. Erosion is a principal export mechanism for nutrients which can be strongly adsorbed to soil particles and are therefore filtered from percolating waters (e.g., phosphorus or ammonia). The degree of adsorption depends in part upon soil pH and clay composition<sup>48</sup>. In contrast, nitrate nitrogen generally does not adsorb and is readily leached from the soil in surface runoff and groundwaters <sup>8,51</sup>. The reduction in the potential for contact between surface waters and soils is considered to significantly increase nutrient mobility and export from urban areas <sup>31</sup>.

The work of Likens et al.<sup>8,34,51</sup> at Hubbard Brook in New Hampshire has demonstrated the relatively closed nature of the nutrient cycles in an undisturbed forest. The measured export rate of phosphorus from one portion of the watershed was 21 g/ha-yr, compared with an input of 108 g/ha-yr in precipitation and an annual internal cycling of 1900 g-P/ha-yr in leaf fall alone.<sup>8</sup> Thus, only a relatively small fraction of the phosphorus cycled within the watershed escaped in the hydrologic outflows and the system appeared to be accumulating phosphorus at the rate of 87 g/ha-yr. A similar area which had been clearcut had an average export rate of 203 g-P/ha-yr, an increase of about 10-fold over the undisturbed system. Most of this increase

(\* · ·

was attributed to the particulate fraction (> 1 mm); the corresponding increase in the dissolved and fine-particulate phosphorus export after clear-cutting was about two-fold. The export of nitrate nitrogen increased about 50-fold and remained at that level for at least two years after clear-cutting.

Increases in nutrient export following agricultural or urban development have also been well-documented 2,19,52,53,89,91 Table 2.3-1 presents the results of an extensive literature review conducted by Uttormark et al. 91 to summarize the available data on the relationships between land use and nutrient export. The wide distribution of export rates within each land use classification can be attributed to variations in such factors as land use intensity  $5^2$ , geology 19,89, soil type 19,89,93 and drainage basin morphology 43,89. In addition, the types of fertilizers, the timing and techniques of fertilizer application, soil structure, geography, and animal population characteristics are considered to be critical factors in regulating nutrient losses from agricultural watersheds . Population densities, street sanitation practices, extents of paved surfaces, and sewerage system characteristics regulate losses from urban watersheds . Errors in the mean flux estimates could also contribute substantially to the variability of export rates within each land use classification. Such errors would be characteristic of estimates derived from sampling programs of

| Table | 2.3-1 |  |
|-------|-------|--|
|-------|-------|--|

Ľ

(

(

-----

Typical Values of Nutrient Runoff Coefficients for Various Land Uses According to Uttormark et al.

| · · ·        | NO <sub>3</sub> -N+NH <sub>4</sub> -N<br>kg/ha/yr |     |     | Total-N<br>kg/ha/yr |     |     |
|--------------|---------------------------------------------------|-----|-----|---------------------|-----|-----|
| .Land use    | High                                              | Low | Ave | High                | Low | Ave |
| Urban        | 5.0                                               | 1.0 | 2.0 | 10.0                | 2.5 | 5.0 |
| Forests      | 3.0                                               | 0.5 | 1.6 | 5.0                 | 1.0 | 2.5 |
| Agricultural | 10.0                                              | 1.O | 5.0 | 10.0                | 2.0 | 5.0 |

|              | Diss.inorg-P<br>kg/ha/yr |      |      | Total-P<br>kg/ha/yr |      |     |
|--------------|--------------------------|------|------|---------------------|------|-----|
|              | High                     | Low  | Ave  | High                | Low  | Ave |
| Urban        | 2.0                      | 0.5  | 1.0  | 5.0                 | 1.0  | 1.5 |
| Forests      | 0.1                      | 0.01 | 0.05 | 0.6                 | 0.05 | 0.2 |
| Agricultural | 0.5                      | 0.05 | 0.1  | 1.0                 | 0.1  | 0.3 |

### 2.3.1 Factors Influencing Nutrient Sources (continued)

insufficient frequency or length to provide an adequate basis for calculation of a long-term-average flux. Hydrologic variations may also be important, since, as discussed below, concentration tends to vary less than flux in streams not dominated by point sources.

The factors discussed above are responsible for variations in long-term-average nutrient export rates from watershed to watershed. Temporal variations for a given watershed, as induced by climatologic changes on various time scales, are also of concern. An understanding of such variations is critical to the interpretation of measurements of export rates obtained over a particular time period. Because different types of terrestrial ecosystems respond differently to climatologic variations, one would expect that both the mean and the variability of the nutrient export rate would depend upon land use distributions and upon various geologic and morphologic characteristics of the watershed.

The effect of variation in streamflow on nutrient concentrations is of primery concern. Flow may influence concentration via several mechanisms  $^{9,39,41,62}$ , including: (a) energy effects (streambed or land surface scouring during periods of high flows, heavy precipitation, and/or surface runoff); (b) residence time effects (higher flows permitting less time for entrapment and

Ċ

utilization of nutrients in upstream terrestrial or aquatic ecosystems); (c) dilution effects (dilution of upstream point-source discharges or groundwater base flows during high flow periods). The interaction of all these mechanisms determines the net response of nutrient concentrations in a given stream to changes in flow. The responses of dissolved, suspended, and particulate fractions are often quite different, reflecting the relative importance of various export mechanisms.

Generally, rivers which are not dominated by upstream point-source discharges tend to show less variation in total nutrient concentration than in flux (concentration × flow). This is true both in response to short- and long-term variations in hydrologic regime. For example, in developing a 6-year nutrient budget for Shagawa Lake, Mauleg et al.<sup>58</sup> monitored nutrient concentrations at weekly intervals on the Burntside River, a major tributary with a heavily forested watershed. The ranges of the reported yearly-average flows, phosphorus fluxes, and flowweighted mean phosphorus concentrations were  $34.6 - 83.5 \times 10^6 \text{ m}^3/\text{yr}$ , 429-973 Kg/yr, and 11.7-14.9  $mg/m^3$ , respectively. The corresponding coefficients of variation were 0.30, 0.26, and 0.09, indicating that year-to-year variations in concentration were much less significant than variations in flow or mass flux. No consistent relationship between yearly average concentration and flow was evident. Similarly, results from Hubbard Brook 39,51

## 2.3.1 Factors Influencing Nutrient Sources (continued)

1

and from the Burntside River<sup>58</sup> indicate a relatively small response of total phosphorus or total nitrogen concentration to seasonal flow variations in forested watersheds.

Less extensive data exist to determine whether other types of watersheds respond similarly to changes in hydrologic regime. Kilkus et al. found that ortho-phosphorus and ammonia nitrogen concentrations correlated slightly positively with flow in some primarily agricultural watershed in Central Iowa. They suggested that this could be a result of an increased proportion of relatively high-concentration runoff during high flow periods. Cahill et al. found a negative correlation between total and ortho-phosphorus concentrations and streamflow during steady-stage conditions on the Brandywine River, which drains a highly diverse watershed in southeastern Pennsylvania and Delaware. They suggested that dilution of upstream point sources during higher flow periods was primarily responsible for the observed behavior. Likewise, Wang and Evans<sup>96</sup> found an inverse relationship between ortho-phosphorus concentration and flow on the Illinois River. In a study of flow and concentration data from various water quality sampling stations in the U.S., Enviro Control observed dilution effects in 8 stations out of 26 for ortho-phosphorus, but in only one station out of 42 for total phosphorus. The National Eutrophication Survey analyzed concentration and flow data from over 200 watereheds and concluded

## 2.3.1 Factors Influencing Nutrient Sources (continued)

Ċ

that, on the average, total phosphorus and total nitrogen concentrations varied with the -0.11 and -0.6 power of flow, respectively.

The response of stream concentrations to storm events is also of interest. Observed increases in suspended nutrient concentrations during rising river stages have been attributed to the scouring of land surfaces and river bottoms with increasing flows  $^{9,27,41}$ . In the Enviro Control study mentioned above  $^{27}$ , total phosphorus was found to exhibit this behavior in 21 out of 41 stations, whereas ortho-phosphorus exhibited it in 3 out of 26. If the scouring of streambeds is an adequate explanation, the response of a stream to a given storm would depend not only upon the spatial and temporal rainfall intensity pattern, but also upon the antecedent dry-weather period, as the latter would determine the amount of material stored in the stream bed at the beginning of the storm. The consequences of this phenomenon on the design of tributary sampling programs will be discussed in a subsequent section.

Strategies for estimating the terms of a nutrient balance can be classified as direct or indirect. Direct evaluation entails the integration of flow and concentration measurements obtained at or near the point of discharge of each source into the lake. Indirect evaluation is based upon established relationships between fluxes or concentrations and the causallyrelated factors discussed in the previous section.

The current state-of-the-art is such that direct estimation methods are generally preferable. Indirect methods are most useful in extrapolating estimates derived from direct measurements. For example, the nutrient fluxes from a given watershed might be estimated from measurements on an adjacent watershed with similar land use, geologic, and morphometric characteristics, assuming a drainage area proportionality or equivalent concentrations 62,86. Nutrient fluxes from sewage treatment plants can be estimated from population and type of treatment. Fer capita estimation of nutrient sources from shoreline residences with septic systems has been commonly employed . The latter estimates are suspect due to the complexities of the factors determining the performance of such disposal systems. Unfortunately, direct measurement of contributions from shoreline septic systems is an also difficult task , and this can contribute substantial uncertainty to the total mutrient budgets of some lakes

## 2.3.2 Estimation Methods (continued)

(

Comprehensive models for indirect estimation of nonpoint sources do not exist as yet. The relatively wide variations of nutrient export rates for various land use categories (Table 2.3-1) suggest that other factors will have to be included before sufficient accuracy and reliability can be achieved<sup>53</sup>. The development of these models is essential to the understanding and eventual control of nonpoint sources and thus to lake water quality management in general.

Direct evaluation of nutrient fluxes presents some particular sampling and estimation problems. In order to obtain an average flux estimate over a time scale  $\Delta t$ , the following integral must be evaluated:

$$\overline{W}_{\Delta t} = \frac{1}{\Delta t} \int_{0}^{\Delta t} q_{t} c_{t} dt \qquad (2.3-1)$$

where

| ٩ <sub>t</sub> | Ð | instantaneous flow (vol/time | e)          |
|----------------|---|------------------------------|-------------|
| сt             | 5 | instantaneous concentration  | (mass/vol)  |
| W              | = | average flux over time At    | (mass/time) |

T.

Typically, continuous measurements of flow are available. If continuous, flow-weighted composite sampling is employed to determine concentration, the integral can be evaluated directly and the only errors in the estimate are the usual ones associated with the measurement and analytical process. If only grab sample concentration data are available, however, assumptions must be made about the behavior of concentration between sampling times. This requires some assumptions which introduce estimation errors, as well as measurement errors.

The various methods of computing average fluxes from continuous flow and grab-sample concentration data differ in their assumptions regarding the behavior of concentration between sampling times. The characteristics of the stream, in particular, the relationship between concentration and flow, determine which method is appropriate. Generally, calculation methods which assume a constant flux are appropriate for point-source dominated systems, while methods which assume constant concentration are more appropriate for use in systems dominated by nonpoint sources. These concepts are discussed in greater detail below.

The appropriate calculation technique also depends upon the time scale of interest. For instance, if a continuous flux estimate is desired, the data can be treated as a time series and interpolation techniques can be employed to estimate fluxes between sampling

3

<u>ا</u>

times<sup>69</sup>. Interpolation can be done on the flux versus time axes if flux is found to be less variable than concentration, or on the concentration versus volume (cumulative flow) axes if the opposite is true. If the time scale of interest is long enough compared with the sampling frequency, the data can be aggregated and treated as samples from a stationary population. With appropriate adjustment in the equivalent sample size to account for any serial correlation in the observations<sup>4</sup>, this permits estimation of both a mean and its variance. The latter is useful in establishing confidence limits and in identifying relative needs for additional stream sampling.

In a study for the Army Corps of Engineers, Meta Systems<sup>62</sup> developed nutrient budgets on various reaches of two major river systems in north-central Florida: the Oklawaha and the Withlacoochee. For each reach, total nitrogen, total phosphorus, and BOD<sub>5</sub> balances were developed on monthly, yearly, and eight-year-average time scales. Point sources of nutrients were generally unimportant in these rivers. In this effort, this author investigated a number of methods for estimating average fluxes from continuous flow and grab-sample concentration data. For each station, nutrient, and time scale, both the mean and variance of the flux estimate were derived. The latter permitted evaluation of the statistical significance of the nutrient accumulation terms in various reaches, relative to the errors inherent in the input and output estimates.

#### 2.3.2 Estimation Methods (continued)

T

Monthly balances were based upon regression models which related concentration to flow, season, and time (trend). In these models, it was necessary to incorporate interactions between season and flow, since the slopes of the concentration versus flow regression lines were found to vary significantly with season at many of the stations. These models explained between 23% and 70% of the variance in the log-transformed concentration data at various stations. They were used in combination with the continuous flow records to generate flux estimates and mass balances at monthly intervals.

To estimate yearly- and long-term-average fluxes, five methods were investigated and compared, based upon the computed variances of the respective estimates. The formulas employed are given in Table 2.3-2. Method 1 involved the direct averaging of the products of sampled concentrations and corresponding flows. According to Method 2, the concentration and flow data were averaged independently. Method 3 was based upon the flow-weighted average concentration and the average flow. Methods 4 and 5 were based upon regression models which related sampled concentration to a sampled flow. The model formulations were concentration versus inverse flow and log (concentration) versus log (flow), respectively.

On both time scales, the variances of the estimates derived

Table 2.3-2

Methods of Estimating Nutrient Fluxes Investigated in the Cross-Plorida Barge Canal Study<sup>62</sup>

| Techniques of Estimating Mutrient Loadings from Grab Sample<br>Concentration Data and Continuous Flow Measurements                                         |
|------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Definitions:                                                                                                                                               |
| C = grab sample concentration on sampling day i                                                                                                            |
| q = flow corresponding to sample i                                                                                                                         |
| H = number of samples                                                                                                                                      |
| 9 • average flow in period of interest                                                                                                                     |
| Ly = average estimated loading (q x c) using Mathod j                                                                                                      |
| V{Lj} = variance of Lj                                                                                                                                     |
| O • everage concentration                                                                                                                                  |
| C <sup>e</sup> = flow-weighted average concentration.                                                                                                      |
| a,b = regression parameters                                                                                                                                |
| •1 = error term                                                                                                                                            |
| 5 • standard error of estimate = standard deviation of e.                                                                                                  |
| Method I: Average Loading                                                                                                                                  |
| $\mathbf{L}_{1} = \frac{1}{\mathbf{N}_{s}} \sum_{i=1}^{H} \mathbf{C}_{s_{i}} \mathbf{q}_{s_{i}}$                                                           |
| $v_{\mathrm{IL}_{1}} = \frac{\sum_{j=1}^{\mathrm{H}_{n}} \left(q_{\sigma_{j}} c_{\sigma_{j}} - L_{1}\right)^{2}}{N_{\sigma}(\mathrm{H}_{\sigma} - 1)^{2}}$ |

.



)

ſ

from Methods 1 and 2 were generally higher than those derived using Methods 3, 4, or 5. Besides giving estimates with relatively high variance, it can be shown that Method 2, in which flow and concentration are averaged independently, gives biased flux estimates if concentration is not independent of flow, whereas the other methods all give asymptotically unbiased estimates. In deriving long-term (eight-year) flux estimates, no general distinction could be made among the last three methods. Methods 3, 4, and 5 gave somewhat sharper estimates of phosphorus, BOD<sub>5</sub>, and nitrogen fluxes, respectively. In estimating yearly-average fluxes, Method 3 was found to develop an advantage over Methods 4 and 5, in that it required estimation of only one, as opposed to two parameters.

Based upon considerations of minimum bias, variance, and computation effort, the results of the Cross-Florida Barge Canal Study  $^{62}$  indicated that Method 3 was preferable overall. For large sample sizes, the regression methods offered an advantage only if the relationships between concentration and flow were significant. Method 3 involved multiplying the flow-weighted average sample concentration times the average flow over the entire period of concern. This amounts to a "ratio estimate" of the mean and assumes that, on the average, flux is proportional to flow. Snedecor and Cochran<sup>77</sup> note that this type of estimate works best in systems where the variance of the dependent variable (flux)

#### 2.3.2 Estimation Methods (continued)

increases with the magnitude of the independent variable (flow). They also note that the ratio estimate (Method 3) is superior to direct averaging (Method 1) only if:

$$\rho_{XY} > \frac{C_X}{2C_Y}$$

(2.3-2)

where

| ρω | = | correlation coefficient betwee | J |
|----|---|--------------------------------|---|
| ~1 |   | independent variable X and     |   |
|    |   | dependent variable Y           |   |

C<sub>X</sub>, C<sub>Y</sub> = coefficients of variation of X and Y, respectively

In this case, the above criterion will be approximately satisfied if the slope of a log (concentration) versus a log (flow) regression is greated than -0.5. As dilution effects become important, the slope would approach -1 and direct averaging (Method 1) would become preferable. For systems not dominated by point sources, however, concentration would be expected to be a weak function of flow\*, and the ratio estimate would be generally preferable.

\*See Section 2.3.1.

### 2.3.2 Estimation Methods (continued)

T

In computing the variances of the flux estimates in the Florida study, independence of the samples was assumed. Serial correlation in the observations would tend to decrease the effective sample sizes for calculation of the mean and variance <sup>4</sup>. As will be discussed in the next section, such effects would be expected to be small at the sampling frequences of less than one per month, typical of the data used in the Florida study. As sampling frequency increases, however, serial correlation may distort the mean and variance estimates derived directly from the equations in Table 2.3-2. Interpolation techniques which treat the data as time series may become superior to the methods discussed above. An approach to evaluating calculation techniques under these conditions is discussed below.

A Monte-Carlo study would be useful as a means of evaluating sampling strategies and calculation methods for estimating the mean and variance of nutrient flux in streams of various characteristics. Stochastic models for streamflow and concentration could be identified and estimated from any high-frequency sampling data available. The models would incorporate trends, seasonal effects, and stochastic variations<sup>37,50,59</sup> Relationships between flow and concentration could be built into the deterministic and/or stochastic elements of these models. They could be used to generate synthetic time series of flow and concentration which could be sampled at various frequencies. Various calculation

Ľ

methods could be applied to estimate the average fluxes from the sampled data. This would permit comparison and evaluation of the methods with regard to bias and variance. The accuracy of the variance estimates derived from various methods could also be assessed. Alternative model formulations and/or parameter values could be used to test the effects of different stream characteristics on the relative performances of the estimation methods and upon the sampling frequencies required to provide estimates with given error bounds.

In designing monitoring programs to provide basic data for nutrient balance estimation, the particular characteristics and periodicities of the sources must be considered. In the case of cultural streams, concentration or flux data may exhibit cyclical variations at daily or weekly frequencies. Such periodicity has been observed in quantity and quality data from .58,82 municipal sewage effluents . Industrial effluents may be characterized by a high degree of variability. Because of these aspects, continuous, flow-weighted composite sampling has been most-often employed in streams of these types. However, continuous sampling of nutrient concentrations has been done only rarely in tributaries . Accordingly, flux estimates in these cases are most-often derived from grab-sample concentration measurements. If this is the case, grab-sampling should be scheduled at roughly equal volume, rather than equal time increments. This would increase temporal frequency during high runoff periods (e.g. spring).

One problem in this regard is the potential impact of storm events on loading estimates derived from infrequent grab sampling. As discussed previously (Section 2.3.1), total phosphorus concentrations have been observed to increase during rising flows as a result of the scouring of land surfaces and/or bottom deposits  $^{9,27,34,41}$ . Brief periods of high flows and high concentrations can introduce considerable skewness into

the distribution of total mass flux. As a result, average fluxes estimated from dry-weather grab-sampling could be considerably in error. One factor which would tend to reduce the significance of this phenomenon is that it is generally associated with the particulate nutrient fractions  $^{34,41}$ . 41 Keup has suggested that these materials settle out fairly rapidly in downstream river reaches as storm flows subside or eventually in lakes. The likelihood that they enter into lake nutrient cycles is relatively small. However, periodic grab-sampling should probably be supplemented with some sampling during storm events in order to observe the variations of nutrient concentrations in the dissolved, suspended, and gross particulate fractions. If large increases in the dissolved and suspended fractions are evident over the course of a storm, then periodic grab sampling alone may not provide a sufficient basis for mean flux estimation. The particular storm events sampled should have fairly long antecedent dry-weather periods, so as to allow sufficient time for materials to accumulate in upstream river beds or on land surfaces.

The standard design for tributary sampling programs in lake management schemes has typically been to sample all tributaries at a given frequency, usually monthly or biweekly. Given the objective of obtaining a total nutrient loading estimate which is unbiased and has minimum viariance, it would

Ĺ

be of interest to consider whether appreciable benefits would be derived from optimizing the allocation of sampling effort among the various streams, subject to a fixed total monitoring cost. This problem is initially posed as a problem in mathematical programming and solved in Table 2.3-3. The objective is to obtain a total loading estimate with minimum variance, subject to a fixed cost constraint which is initially assumed to be expressed as a fixed total number of samples, N. The allocation is assumed to be made among M tributaries, each characterized by a variability  $k_i$  and loading  $l_i$ . The solution is given by:

$$\frac{\mathbf{n}_{\mathbf{i}}}{\mathbf{N}} = \frac{\mathbf{k}_{\mathbf{i}} \mathbf{c}_{\mathbf{i}} \mathbf{q}_{\mathbf{i}}}{\mathbf{M}} = \frac{\mathbf{s}_{\mathbf{i}}}{\mathbf{M}}$$
(2.3-3)  
$$\sum_{\substack{\mathbf{k} \in \mathbf{i}} \mathbf{c}_{\mathbf{i}} \mathbf{q}_{\mathbf{i}}} \sum_{\substack{\mathbf{k} \in \mathbf{i}} \mathbf{s}_{\mathbf{i}}} \mathbf{s}_{\mathbf{i}} \mathbf{s}_{\mathbf{i}}$$

Thus, three factors determine the fraction of the total sampling effort optimally allocated to stream  $i : k_i$ , variability;  $c_i$ , average concentration; and  $q_i$ , average flow. In designing a sampling program according to this scheme, the parameters could be estimated <u>a priori</u> based upon a preliminary survey, land use, and hydrologic data.

The potential benefits of optimizing the sampling program can be estimated by comparing the variance of the

# Table 2.3-3

(

C

Optimal Allocation of Samples Among Tributaries for Nutrient Balance Estimation

# Problem Definition:

| Select:      | n <sub>i</sub> , i=l, M                                          |
|--------------|------------------------------------------------------------------|
| To Minimize: | $\sigma_{T}^{2} = \sum_{i=1}^{M} k_{i}^{2} z_{i}^{2} / n_{i}$    |
| Subject to:  | M<br>Σ n <sub>1</sub> = N<br>1=1                                 |
| Where:       | n <sub>i</sub> = number of samples allocated<br>to stream i      |
|              | M = total number of streams                                      |
|              | $\sigma_{\mathbf{T}}^2 = \text{variance of total flux estimate}$ |
|              | N = total number of samples                                      |
|              | k <sub>i</sub> = variability of stream i                         |
|              | ℓ <sub>1</sub> = average flux of stream i                        |

Solution: LaGrange Method of Undetermined Multipliers

LaGrangian = Y = 
$$\sum k_{1}^{2} \ell_{1}^{2}/n_{1} + \lambda(\sum n_{1} - N)$$
  
 $\frac{\partial Y}{\partial n_{1}} = -\frac{k_{1}^{2} \ell_{1}^{2}}{n_{1}^{2}} + \lambda = 0$ ,  $i=I,M$   
 $\therefore \qquad n_{1} = k_{1}\ell_{1}/\sqrt{\lambda}$   
 $\frac{\partial Y}{\partial \lambda} = \sum n_{1} - N = 0$   
 $+ \sum n_{1} = N = \sum k_{1}\ell_{1}/\sqrt{\lambda}$   
 $\neq \sqrt{\lambda} = \sum k_{1}\ell_{1}/N$   
 $\therefore \qquad \frac{n_{1}}{N} = \frac{k_{1}\ell_{1}}{\sum k_{1}\ell_{1}} = \frac{k_{1}q_{1}c_{1}}{\sum k_{1}q_{1}c_{1}}$ 

# 2.3.3 Monitoring Program Design (continued)

estimate obtained from the allocation according to equation (3) with the variance of the estimate for the case in which all streams are sampled equally:

$$\sigma_{\rm TE}^2 = \frac{M}{N} \sum_{i=1}^{N} s_i^2 \qquad (2.3-4)$$

$$\sigma_{T^{*}}^{2} = \frac{1}{N} \left( \sum_{i=1}^{N} s_{i} \right)^{2}$$
(2.3-5)

$$\sigma_{T^{*}}^{2} = \frac{(\Sigma s_{i})^{2}}{M \Sigma s_{i}^{2}} = \frac{1}{1 + \overline{cv}_{s}^{2}}$$
(2.3-6)

where

č

C

(

$$\sigma_{TE}^{2} = \text{variance of estimate with equal} \\ \text{sampling allocations} \\ \sigma_{T^{\pm}}^{2} = \text{variance of estimate with optimal} \\ \text{sampling allocations} \\ \overline{CV}_{s} = \text{coefficient of variation of } s_{i}, \\ \text{i=1.M} \\ \end{array}$$

If all  $s_i$  are equal,  $\overline{CV}_s = 0$ , and the optimal and equal allocations coincide. As the differences in the streams become more pronounced, reflecting increasing levels of  $\overline{\text{CV}}_{\mathrm{g}}$  ,

8

the potential benefits of optimization increase.

The nutrient budgets of five lakes in the National Eutrophication Survey 86 have been examined to determine the possible effects of optimal sample allocations on the variances of the mean loading estimates for some typical tributary groupings. In estimating the optimal allocations, it was assumed that the weighting factors, s, , were proportional to the total phosphorus loadings, i.e., the variabilities were assumed to be equal. Variance ratios according to equation (6) ranged from 0.15 to 0.88, corresponding to a range in the ratios of confidence limits of 0.39 to 0.94. The most pronounced effect was observed in a lake which had seven influent streams, one of which accounted for 96.7% of the total loading. The least pronounced effect was observed in a lake with two inlet streams at a loading ratio of about two to one. Thus, it appears that in some cases, appreciable benefits could be achieved from optimizing the sampling allocations.

One means of improving the above analysis would be to employ a more realistic cost constraint, which would not be strictly proportional to the total number of samples. One such constraint would be:

$$C = a n_{\underline{i}}^{\max} + b N \qquad (2.3-7)$$

where,

C

$$C = \text{total cost}$$

$$\max_{i} = \max \operatorname{maximum value of } n_{i}$$

$$a,b = \text{cost parameters}$$

The first term represents the fixed costs of deploying a sampling crew a total of  $n_1^{\text{MAX}}$  days during the year of monitoring. The second term reflects the incremental sampling and analytical costs per sample.

A second improvement would be to account for possible effects of serial correlation in the observations on the variance of a mean flux estimate for a given stream. Bayley and Hammersly<sup>4</sup> have shown that serial dependence can be accounted for by using an equivalent sample size,  $n_i^e$ , in calculating the variance of the mean estimate;

$$\frac{j=n_{i}-1}{\prod_{i=1}^{e} = \frac{1}{n_{i}} + \sum_{j=1}^{e} (n_{j}-j) \rho_{jt}}$$
(2.3-8)

where,

Ć

t = sample interval = 365/n<sub>i</sub> days ρ = lag jt serial correlation coefficient jt

A number of investigators  $^{37,50,59,60}$  have shown that highfrequency components of stream water quality data can be modelled reasonably well as a first order, autoregressive (Markov) process with a serial correlation coefficient in the range of 0.7 to 0.9 at one day lag. For such models, Matalas and Langbein<sup>57</sup> have shown that equation (8) reduces to:

$$\frac{1}{n_{i}^{e}} = \frac{1}{n_{i}} + \frac{2}{n_{i}^{2}} \left( \frac{\rho^{t}}{(\rho^{t} - n\rho^{t} + n - 1)}}{(\rho^{t} - 1)^{2}} \right)$$
(2.3-9)

where,

ρ = lag l day autoregression coefficient

This equation is plotted on log-log scales in Figure 2.3-1 for various values of  $\rho$ . At a monthly sampling frequency, the effect is relatively insignificant and  $n_i^e \approx n_i$ . As

# Figure 2.3-1

**(**)

(

(

Relationship Between Effective and Actual Sample Sizes in Calculating a Mean Generated by a Lag One Markov Process





# 2.3.3 Monitoring Program Design (continued)

**(** -

sampling frequency increases, the effect becomes more pronounced. For values of between 0.8 and 0.9, there is little difference in effective sample size between weekly and daily sampling frequencies. Thus, if the objective is to estimate a mean value, a weekly sampling program would provide about as much useful information as a daily one. Littenmaier <sup>50</sup> has discussed the implications of this on monitoring program design and has employed the effective sample size concept in studying the powers of nonparametric tests to detect trends in water quality data. Incorporation of these effects into the mathematical programming model would tend to make the optimal allocations more equal than indicated by equation (3).

In real applications, equation (3) might be used as a rough indication of desirable sampling program design from a nutrient budget standpoint. Incorporation of a more realistic cost function and consideration of the effects of serial correlation on the variance of the mean would tend to suggest a more even allocation of sampling effort than that indicated by equation (3). The benefit in terms of increased precision in the loading estimate would tend to increase as the similarities of the sampled streams decrease. In many cases, equal allocation of sampling effort would not be far from the optimal design. Since monitoring programs are usually multi-objective,

# 2.3.3 Monitoring Program Design (continued)

Ĩ

other factors may have to be considered before arriving at the final program design.

Rationally, the size of the sampling effort should depend upon the required accuracy of the nutrient budget estimates. This, in turn, should depend upon the requirements of the particular lake model employed and upon the potential impact of errors in the flux estimates upon the decisions made or management policies selected. Historically, however, little effort has been made to place confidence limits on nutrient budgets. The design of sampling programs has been dictated by "rules of thumb" and budgetary contraints. As the stateof-the-art advances, the increased interest in making probabilistic water quality projections should eventually provoke an awareness of the errors involved in nutrient budget estimates. Systematic approaches to lake monitoring, modelling, and managing efforts will, in turn, involve more rational monitoring program designs.

### 2.3.4 Interpretation Problems

( :

A lake's nutrient budget which has been estimated based upon monitoring data obtained during a given period may not reflect average conditions. If a lake model is to be used to predict long-term water quality changes resulting from various management strategies, long-term-average nutrient loadings from various sources in the watershed must be estimated. Likewise, if there are lake-specific parameters in a steady-state model, the parameter values would have to be estimated from observations made on the lake or its outflow under steady-state conditions. Fluctuations in cultural, hydrologic, and meteorologic factors can cause year-to-year variations in the nutrient dynamics of watersheds and lakes. These variations introduce uncertainty into the extrapolation of monitored input and output fluxes to long-term-average conditions.

There are no data known to the author which would establish whether significant long-term variations in the export of nutrients from watersheds under culturally-stable conditions exist. The errors inherent in the measurement of these fluxes based upon grab sampling may have obscured our ability to detect such variations, however. The time scale of the natural evolution of terrestrial ecosystems and any resultant changes in nutrient export is probably beyond the planning horizon. Since concentration is generally a weak function of flow in tributaries not dominated by point sources, the flux for an average hydrologic year can be

approximated as the product of the measured flux and the ratio of average flow to the flow during the sampled year. This assumes that the watershed is under fairly stable conditions and that no significant interventions which would influence nutrient trapping or release within the watershed occur during the period of sampling. The National Eutrophication Survey<sup>86</sup> has used essentially this procedure, with the exception that the ratio of flows was raised to the .89 power, assuming that average phosphorus concentration varies as the -.11 power of flow. This slope was derived from an analysis of flow and concentration data from over two hundred watersheds with fairly heterogeneous characteristics. In the case of point sources, which are not hydrologically-mediated, fluxes are usually assumed to be independent of flow. If the distinction between point and nonpoint sources cannot be easily made, the relationship between concentration and flow can be examined graphically and, if a significant slope is apparent, a regression model can be employed to estimate loading during an average hydrologic year  $^{9,33}$ . Assuming that watershed conditions are relatively stable during sampling, the adjustment of loading to average conditions can thus be made relatively easily.

Estimating the output and accumulation terms of a lake's nutrient budget under average conditions is a much more difficult task. In order to do this, assumptions must be made about the

dynamics of the nutrient trapping mechanisms within the lake. Estimates of the retention coefficient, defined as the fraction of the influent phosphorus which does not leave in the outflows, are required for the applications of Dillon's model , which will be discussed in the next section. In converting observed phosphorus and nitrogen outflows to an average hydrologic year, the National Eutrophication Survey (NES)<sup>86</sup> has employed the same techniques used in converting the nutrient inflows. The following analysis suggests that this may have introduced some significant errors in the accumulation rates, retention coefficients, and average outlet concentrations reported by the NES. This is particularly true for the lakes which were sampled in 1972, which was a relatively wet year, characterized by a yearly flow to average flow ratio of about two, for many of the midwestern and northeastern lakes sampled. Tropical storm Agnes occurred in late June of 1972 and could have had a profound effect on the nutrient and hydrologic budgets of many of the lakes sampled that year.

Assume that the lake's nutrient budget can be summarized in the following four terms, which are measured during a given year:

| <sup>L</sup> a | = | diffuse source loading | (g-P/m <sup>2</sup> -yr) |
|----------------|---|------------------------|--------------------------|
| L<br>P         | = | point source loading   | (g-P/m <sup>2</sup> -yr) |
| Ŀ              | = | outflow                | (g-P/m <sup>2</sup> -yr) |
| La             | • | accumulation rate      | (g-₽/m <sup>2</sup> -yr) |
From a steady-state mass balance:

.

$$L_a = L_d + L_p - L_o$$
 (2.3-10)

In converting these terms to long-term averages, the NES has essentially employed the following conversion scheme:

$$L_{d}' = L_{d} q^{.89}$$
  
 $L_{p}' = L_{p}$  (2.3-11)  
 $L_{o}' = L_{o} q^{.89}$ 

where

£

ĺ

(

q = average-year flow/sampled-year flow.

The corresponding expressions for the sampled and average retention coefficients are given by:

$$R = \frac{L_{a}}{L_{p} + L_{d}} = \frac{L_{p} + L_{d} - L_{o}}{L_{p} + L_{d}}$$
(2.3-12)

. .

$$R' = \frac{L_{p} + L_{d} q^{\cdot 89} - L_{o} q^{\cdot 89}}{L_{p} + L_{d} q^{\cdot 89}}$$
(2.3-13)

€

$$R' = \frac{L_d - L_o}{L_d} = R, \text{ independent of } q \quad (2.3-14)$$

For L >> L ,

$$R' = \frac{L_{p} - L_{q}^{,89}}{L_{p}}$$
(2.3-15)

The analysis shows that, for lakes dominated by nonpoint sources, the corrected retention coefficient, R', equals the sampled retention coefficient, and is, therefore, independent of flow. This result is contrary to the empirical equation of Dillon and Kirchner<sup>42</sup>, which expresses R as a function of QS, the surface overflow rate (= outflow/ surface area). This equation predicts that R should decrease from one to zero with increasing flows. The above result is ( )

also contrary to the intuitive argument that higher flows should permit less time for the physical, chemical, and biological reactions responsible for phosphorus trapping to occur within the lakes and should therefore result in lower retention coefficients. For the lakes dominated by point sources, the predicted response of the retention coefficient to changes in flow is correct in sign, but likely to be in error, since no consideration has explicitly been given to the factors controlling nutrient trapping in the lake.

Another difficulty in interpreting lake outflux data is that of assuming steady-state. One would expect the nutrient dynamics to respond faster to changing conditions in low residence time lakes. However, equilibration of lakewater/sediment exchanges may occur on a much longer time scale than indicated by the hydraulic residence time<sup>48,56</sup>. There is a general problem of deciding whether the nutrient outflux is in equilibrium with long-term or short-term hydrologic, and meteorelogic, or loading conditions. The net result of these problems is that measurements of nutrient outflows, accumulation rates, and retention coefficients are relatively difficult to interpret in terms of average conditions, unless the measurements have been taken over sn extended period of time. Such a period should probably encompass at least three hydraulic residence times, or one year, whichever is greater, and under hydrologic, meteorologic, and nutrient ЗĨ.

loading conditions which are relatively stable and not far from average. One primary consequence is that lake models which rely upon measured lake concentrations or retention coefficient values, such as Dillon's model<sup>17</sup>, have more demanding data requirements than those which rely upon estimates of inlet fluxes alone. The data requirements are more demanding both with regard to the number of stations sampled and to the required length of the monitoring period. T

Models which have been proposed for use in managing lake water quality can be characterized along various dimensions, including empirical versus theoretical, deterministic versus stochastic, stratified versus mixed, and dynamic versus steadystate. There is a correspondingly complex array of assumptions, applicabilities, and data requirements. To facilitate discussion of existing models, they have been separated into two categories: "empirical" and "theoretical". Models which focus on nutrient concentrations, relying upon observed correlations between nutrient concentrations and sutrophic symptoms, have been included in the "empirical" group. The relatively complex systems models which attempt to directly predict lake response by simulating the various physical, chemical, and biological processes have been included in the "theoretical" group. Since the empirical-theoretical dimension is continuous, the distinct classification of the models is somewhat artificial and has been made only as a means of organizing the discussion. Generally, models of the former group have relatively low data requirements and are applicable to long-term management criteria, while models of the latter group require more data and are applicable to short-term, critical statss, as well as long-term criteria.

#### 2.4.1 A Review of Empirical Models

**(**):

The models of Vollenweider<sup>93,95</sup> and Dillon<sup>16-18</sup> are based upon empirical relationships among trophic states, phosphorus input rates, and various morphometric or hydrologic variables. Derived on data from northern temperate lakes, these models are applicable to steady-state, yearly-average conditions and focus on phosphorus as the limiting nutrient. They rely heavily upon correlations between total phosphorus concentrations (mean annual or at spring overturn) and lake trophic state, or indicators thereof, such as mean midsummer chlorophyll-a concentrations.<sup>14</sup> These models are characterized by their simplicity and relative ease of application. They have been posed in graphical form and require relatively little data to implement.

The areal loading of total phosphorus (g/m<sup>2</sup> lake surfaceyear) is a critical variable in each of these models. Accordingly, the kinds of data and methods employed in the formulation of a lake's nutrient balance, as discussed previously (Section 2.3), are particularly important to successful implementation. The models will be discussed in the chronological order of the development, which also corresponds roughly to decreasing empiricism and increasing data requirements.

Vollenweider's first model<sup>93</sup> predicts trophic state as a function of total phosphorus loading, L  $(g/m^2-yr)$  and mean depth, Z (m) (Figure 2.4-1). In developing this model,

C

Vollenweider plotted data from 18 lakes on log L versus log Z axes and drew two lines which roughly separated the three trophic states. Eutrophic lakes were generally located above a "dangerous" loading line, and oligotrophic lakes, below a "permissible" loading line. The slopes of these parallel lines are about 0.6, indicating that a doubling in mean depth would permit a 60% increase in loading without a change in trophic state. This is the first, simplest, and perhaps most widely cited of any general eutrophication model.

Vollenweider's second model<sup>95</sup> was developed on an expanded data base of 31 lakes. It predicts trophic state as a function of phosphorus loading and surface overflow rate, the latter given by:

$$QS = \frac{Q}{A} = \frac{Z}{T}$$
 (2.4-1)

where



I.

۲.

(

(

# 2.4.1 <u>A Review of Empirical Models</u> (continued)

(

(

This model, depicted in Figure 2.4-2, incorporates the notion that lake flushing rate, 1/T, in addition to mean depth, determine the response of a lake's trophic state to a given nutrient loading. The model is somewhat more theoretically-based than Vollenweider's earlier attempt, although the locations of the lines in Figure 2.4-2 were still empirically determined. The slopes of these lines are about 0.5, indicating that a doubling in eurface overflow rate would permit a 50% increase in phosphorus loading without a change in trophic state. This model has been referred to extensively in the working papers of the EPA's National Eutrophication Survey<sup>.86</sup>.

Dillon<sup>16-18</sup> has extended Vollenweider's second model by coupling it more closely with the phosphorus mass balance. He defined the retention coefficient, R, as the fraction of the influent phosphorus which is trapped in lake sediments, or, equivalently, the fraction which does not leave in the lake's outflow. According to mass balance equation, the retention coefficient is given by:

$$R = 1 - \frac{QS \cdot CO}{L} = 1 - \frac{CO}{CI} \qquad (2.4-2)$$

where

( )

| R  | Ŧ | retention coefficient        |                     |
|----|---|------------------------------|---------------------|
| 0  | = | average outlet concentration | (g/m <sup>3</sup>   |
| 21 | = | average inlet concentration  | (g/m <sup>3</sup> ) |

In Dillon's model (Figure 2.4-3), the lines delimiting trophic state correspond to solutions of the steady-state phosphorus balance equation for mean outlet concentrations of 0.01 and  $0.02 \text{ g/m}^3$ , respectively. In Figure 2.4-3, Dillon's original axes<sup>18</sup>, L(1-R)T versus Z, have been transformed to the equivalent form, L versus QS/(1-R). This transformation isolates the design variable, L, on the y-axis and permits a more direct comparison with Vollenweider's models.

In applying Dillon's model, the retention coefficient must be determined, in addition to L, Z, and T. For a given lake, R can be estimated from input/output measurements, or, if such are not available, from an empirical equation derived by Kirchner and Dillon<sup>42</sup>, which allows estimation of R as a function of the surface overflow rate:

$$RD = 0.426 e^{-.271 QS} + 0.574 e^{-.00949 QS}$$
(2.4-3)

### 2.4.1 A Review of Empirical Models (continued)

This equation had a correlation coefficient of 0.94 with data from fifteen oligotrophic and mesotrophic lakes, primarily in southern Ontario.

In using the model to estimate "dangerous" and "permissible" loading levels, the assumption is made that R (or RD ) is independent of average inlet or average outlet concentrations. This is equivalent to assuming that phosphorus decays from the lake system in a first-order fashion. Such an assumption seems contrary to the saturation kinetics typically employed in modelling algal productivity as a function of phosphorus concentration 47. These kinetics predict a zero-order dependence at high phosphorus concentrations, with a half-saturation parameter in the range of 0.01  $g/m^3$ . Other, chemical or physical mechanisms for phosphorus removal (e.g., precipitation, flocculation, and particulate settling) may exhibit first-order behavior, however. Time series. data on the response of the retention coefficient to a reduction or increase in phosphorus loading for a number of lakes would be helpful in clarifying this issue. Alternatively, if an empirical model for the retention coefficient (such as equation (3)) could be shown to have parameter values which are independent of concentration (or loading), then the first-order assumption could be applied with increased confidence in predicting a lake's ultimate (steady-state) response to a reduction or increase in loading.

# 2.4.1 A Review of Empirical Models (continued)

Dillon<sup>15</sup> has reviewed early attempts at simple, theoretically based input/output phosphorus models. The basic model of this type was initially proposed by Vollenweider<sup>94</sup>. This scheme represents a lake as a completely mixed reactor from which phosphorus is removed via a simple, first-order reaction. The steady-state solution for the average outlet (= average lake) phosphorus concentration is given by:

$$CO = \frac{L}{QS + \sigma Z}$$
 (2.4-4)

where

 $( \ )$ 

ĺ

Vollenweider found that  $\sigma$ , the so-called "sedimentation coefficient", was generally not constant across the limited number of lakes for which data were available. He subsequently made a series of modifications to equation (4) to attempt to account for various inadequacies, in particular the completelymixed assumption <sup>15,95</sup>.

A host of more theoretically-based models of phosphorus

# 2.4.1 A Review of Empirical Models (continued)

£

Ć

dynamics in lakes have grown out of the early attempts of vollenweider. Complexities of various forms have been incorporated into equation (4), resulting in multi-box versions<sup>39,56,63,78</sup> (epilimnion, hypolimnion, and/or sediment), multi-component versions<sup>63,78</sup> (dissolved P and particulate P), and non-steady-state versions <sup>56,63,78</sup>, the latter intended to simulate seasonal and/or long-term responses. The common factor in all of these models is the explicit consideration of phosphorus or forms thereof. This distinguishes these models from the more complex dynamic ecosystem models, which explicitly consider other chemical and biological components and are often built upon elaborate hydrodynamic models<sup>10,20,21-24,72</sup> In a management context, however, the more empirical models of Vollenweider and Dillon have received the most attention and application.

#### 2.4.2 A Review of Theoretical Models

(

Relatively complex systems models have also been proposed for use in managing lake water quality. Their theoretical bases, flexibility, and ability to handle spatial and temporal heterogeneities offer potential advantages over the relatively simple models discussed above. In application, they can be used to make predictions which relate directly to short- or long-term water quality management criteria. This permits straightforward comparisons of management strategies<sup>67</sup>. Using simulation and sensitivity analysis techniques, these models can also be employed to investigate the relative importance of various mechanisms in controlling water quality. Such results can be used, in turn, to suggest effective control strategies<sup>65</sup>.

The complexities of these models render them relatively difficult to apply. Their data requirements are extensive and lack of sufficient data often severely limits effective and proper use. The elaborate spatial and temporal resolutions of these models are often not warranted in view of the natures of the data bases used for verification, parameter estimation, and for specification of boundary conditions. Typically, large numbers of parameters must be estimated, based upon observations of experimental systems or of the lakes themselves. Because some of the important processes are not well-understood theoretically, numerous structural assumptions are also required. The multi-dimensionality and inter-dependence of the structural § .

and parametric assumptions make it difficult to assess potential errors in model projections. Finally, in contrast to some of the simple models discussed in the previous section, specialized professionals and computers are required for successful implementation.

Applications of these types of models can be grouped into two generaly categories: ecologically-oriented studies and water-quality-oriented studies. The former are characterized as relatively fundamental, scientific investigations, whereas the latter pertain more directly to specific lakes and management problems and are more in the realm of engineering. The theoretical components shared by these models have gradually evolved, based upon trial-and-error simulationa and upon independent experimental evidence gathered in laboratory microcosms or in situ. With regard to autotrophic communities, recent theoretical contributions have been made by Bannister 2 (light effects) and Grenney et al. $^{30}$  (nutrient storage effects). Lassiter47 has conveniently summarized the functional forms frequently used in representing various chemical and biological processes in aquatic ecosystems. This work is a useful reference in model building, although an analogous compilation of parameter estimates is lacking.

Ecological investigations have focused on the structures

1

and interactions of the aquatic food chain<sup>29,49,64,76</sup>. The general goal of this type of work has been to determine to what extent the interactions between and within the various trophic levels can be simulated based upon current understanding of the individual processes and relationships comprising the system. In this work, qualitative agreement between observed and simulated behavior is generally sought, but direct, quantitative comparisons of observations and predictions are rare. Typically, the consequences of structural assumptions on the behavior of the systems of equations have been examined and used to generate hypothesis concerning important controlling factors 76 An increased emphasis on multispecificity in these models has evolved. Simulating the seasonal succession of algal species as controlled by physical, predatory, and nutritional factors has been the subject of works by Lehman <u>et al</u>. and Grenney<sup>29</sup>. "CLEAN"<sup>64</sup> (Comprehensive Lake Ecosystem Analyzer) is a recent attempt at an elaborate ecosystem model, having grown out of the efforts of the International Biological Program.

The coupling of ecosystem models with water quality models and their application in a management context have grown principally out of the early work of Chen and Orlob<sup>10</sup> and Thomann <u>et al<sup>20</sup></u>. The latter group has published modelling studies of the Sacramento River<sup>21</sup>, Potomac Estuary<sup>23</sup>, and the Great Lakes<sup>22,24,25,83</sup>, among others. The underlying structural 2~64

(:

assumptions of these models have not changed much. Increasing emphasis has been placed upon an effective display of results and sensitivity analysis  $^{24}$ . The work of Thomann <u>et al</u>. on Lake Ontario<sup>24</sup> is a good example of the effectiveness of sensitivity analysis in studying the dynamics of these systems. A diagram of the control pathways typical of models of this sort is given in Figure 2.4-4.

The evolution from single-specific to multi-specific representations of the algal community is an important advance in promoting realism and permitting more direct comparisons of model predictions with observed seasonal succession in algal populations 6,46,66,67,70. In a management view, the capability of differentiating among the various desirable and undesirable algal groups would been obvious benefit resulting from the development of multi-specific models. Scavia et al. 70 applied a onedimensional, single-specific model initially developed and calibrated for Lake Ontario to each of the Great Lakes. They found that it was necessary to "tune" two important parameters in ordar to achieve "acceptable" agreement between observed and predicted chlorophyll and nutrient levels in each of the lakes. One of the parameters was the half-saturation constant for algal growth on phosphorus; the other determined the effect of food concentration on zooplankton growth rate. Since the Great Lakes roughly span the path of a single lake through the trophic

2~65



Figure 2.4-4 Biological and Chemical Systems Diagram for

 $(\cdot)$ 

ſ

٠.



Transparence

(

1

states in time, they concluded that a single-specific model with constant parameters would be inadequate to simulate the eutrophication or recovery of a given lake:

"For an ecological model to be able to predict ecological changes occurring during eutrophication, it must include at least saveral compartments in each level of the food chain to allow natural selection to be simulated. In this way, "recalibration" will take place automatically as succession and adaptation would in nature."

In this view, multi-specific models will be necessary for application in predicting long-term responses to changes in nutrient loading. The resulting enhanced complexities would present no particular computational problem, but would multiply parameter estimation difficulties.

This author was involved in monitoring and modelling studies carried out by Process Research, Inc., on the Charles River Basin<sup>66,67</sup>. The goal of this work was to develop a model to predict the impact of various management programs upon swimming quality in this nine-mile-long impoundment. A diagram of the control pathways in the model is given in Figure 2.4-5. An analysis of existing and projected future conditions indicated that transparency and collform bacteria levels were the two factors most severely limiting use of the basin waters for swimming. Due to the presence of relatively 1

high levels of background turbidity and color, light was an important environmental factor, controlling both algal growth rates and coliform decay rates. An examination of the algal population data indicated significant shifts in algal species and densities with river reach and season. These shifts correlated with surface-water salinity changes induced by penetration of sea water upstream in the form of a salt wedge. It was necessary to include three algal types in the model in order to achieve adequate agreement between simulated and observed algae and transparency data. This is another instance in which a single-specie model was found to be an inadequate representation of the algal population dynamics and the resultant impact on water quality.

In some applications, the essential aspects of the complex models can be summarized into relatively simple formulations, which retain realism, but require substantially less data and effort to apply. An example of such work is that of Lorenzen and Mitchell<sup>54,55</sup>, who developed a model for predicting the theoretical effects of artificial destratification on algal production in Lakes and impoundments. In this effort, a general equation describing the rate of change of algal concentration in a lake was simplified in successive stages. Assuming that the lake was well-mixed above the thermocline, and that the primary term accounting for algal loss was respiration, the growth equation was simplified to:

$$\frac{3C}{dt} = \mu C - rC \qquad (2.4-5)$$

where

€

| μ | - | growen zace (z/uay)                     |
|---|---|-----------------------------------------|
| r | Π | respiration rate (1/day)                |
| С | Ħ | algal concentration (g/m <sup>3</sup> ) |

For eutrophic lakes, the effect of nutrient concentration on  $\mu$  was assumed to be negligible, and, accordingly,  $\mu$  was considered to be a function only of available light. Respiration rate was assumed to be constant. Both  $\mu$  and r were to be evaluated at typical, mid-summer epilimnion temperatures. Employing traditional formulations for the effects of light intensity on  $\mu$  and for the effects of algal concentration on the light extinction coefficient in the water column, Lorenzen and Mitchell obtained an expression for the peak light-limited biomass per unit area by setting the integral of equation <sup>(5)</sup> over an averaging time period,  $\Delta T$ , equal to zero and solving for CZ:

$$\mathbf{C}_{\mathbf{L}}^{\max} \mathbf{Z} = \frac{\mu_{\max}}{\mathbf{r}\beta} \mathbf{F} - \frac{\alpha}{\beta} \mathbf{Z}$$
(2.4-6)

$$F = \frac{1}{\Delta T} \int_{0}^{\Delta T} \ln (AI_{0}(t) + (1 + \overline{AI_{0}(t)}^{2})^{\frac{1}{2}}) dt \qquad (2.4-7)$$

$$I_{O}(t) = I_{O}^{\max} \frac{1}{2} (1 + \cos \frac{2\pi t}{\lambda}), -\lambda/2 < t < \lambda/2 \qquad (2.4-8)$$
  
= 0, otherwise  
$$\varepsilon = \alpha + \beta C_{L}^{\max} \qquad (2.4-9)$$

where,

.

(

(

| CL<br>L            | a | light-limited biomass $(g/m^3)$                                                              |
|--------------------|---|----------------------------------------------------------------------------------------------|
| Z                  | = | mixed depth (m)                                                                              |
| u<br>max           | = | maximum growth rate (1/day)                                                                  |
| β                  | = | incremental light extinction<br>coefficient due to algae (m <sup>2</sup> /g)                 |
| a                  | = | background light extinction<br>coefficient (m <sup>-1</sup> )                                |
| £                  | = | total extinction coefficient $(m^{-1})$                                                      |
| F                  | = | light/depth integral (dimensionless)                                                         |
| λ                  | ÷ | day length (hrs)                                                                             |
| ∆۲                 | = | time interval for averaging ( $\geq 24$ ) (hrs)                                              |
| I <sup>max</sup> o | ₽ | surface light intensity at noon (lux)                                                        |
| 1 <sub>0</sub> (t) | = | surface light intensity at time of<br>day t (lux)                                            |
| t                  | = | time of day, plus or minus from solar<br>noon (hrs)                                          |
| A                  | = | parameter describing the effect of light intensity on algal growth rate (lux <sup>-1</sup> ) |

.

It was further assumed that the peak nutrient-limited biomass was given by a concentration  $C_N^{\max}(g/m^3)$ , which depended upon the availability of nutrients in the system. The value of X could be estimated from:

$$C_{N}^{\text{max}} = N_{O}/\gamma$$
 (2.4-10)

where

y = nutrient content of algal biomass (g-nutrient/g-biomass)

This essentially treats the lake as a batch reactor, ignoring any effects of nutrient inflow or outflow between the times of spring overturn and peak biomass.

For some typical parameters values, Lorenzen and Mitchell summarized the results in graphical form by plotting peak nutrient-limited and light-limited biomass against depth (Figure 2.4-6). The nutrient-limited line has a positive slope, while the light-limited line has a negative one. Peak



**(**;

Ć

Ć



Effect of Mixed Depth upon Biomass Potential According to Lorenzen and Mitchell

{

biomass would be restricted to the region below both of these lines. Accordingly, biomass would be nutrient limited if the mixed depth were less than the Z value corresponding to the intersection of the two lines, and light-limited, otherwise.

These results were used to explain observed variations in the response of lakes to artificial mixing, which was assumed to effectively increase mixed depth. The model predicted that nutrient-limited lakes would show increases in peak biomass as a result of mixing, whereas light-limited lakes would show decreases. This was suggested as a theoretical basis for estimating whether or not mixing would be beneficial to a given lake.

A difficulty in applying this model is that associated with the estimation of the various parameters and in evaluating the light/depth integral, F, which had to be done numerically. To simplify the latter, Sykes<sup>81</sup> employed Steele's <sup>79</sup> formulation for algal growth rate as a function of light intensity:

$$\mu = \mu_{max} \frac{I_{(z,t)}}{I_{s}} e^{\left(1 - \frac{I(z,t)}{I_{s}}\right)}$$
(2.4-11)

where

I<sub>s</sub> = saturation light intensity, characteristic of algae (lux)

(

Integration of this expression over mixed depth, Z , and incorporation into the Lorenzen-Mitchell model gave the following expression for the light/depth integral:

$$F = \frac{2.718}{\Lambda T} \int_{0}^{\Lambda T} (e^{-\frac{I_{0}(t)}{I_{s}}} e^{-\epsilon Z} - \frac{I_{0}}{I_{s}}) dt \qquad (2.4-12)$$

For parameter values over a reasonably wide range, Sykes demonstrated that the first term in the integral was essentially equal to one, while the second term equalled one during the night and averaged near zero during the day. Thus, for a 24-hour period and daylength  $\lambda$ , the integral can be evaluated as:

$$F = \frac{2.716}{24} [(1-0) \lambda + (1-1)(24 - \lambda)]$$

$$= \frac{2.718 \lambda}{24}$$
(2.4-13)

Sykes neglected to include the daylength factor,  $\lambda/24$ , in his result. Despite this oversight, he made a substantial contribution by greatly simplifying the evaluation of the light/depth integral term in the Lorenzen-Mitchell model.

With this simplification, according to equations (6) and (13), application of the model can be made with estimates of the following parameters:  $\mu_{max}/r$ ,  $\alpha$ ,  $\beta$ , N<sub>o</sub>, y, and  $\lambda$ . Literature values for  $\mu_{max}/r$  are generally in the range of 10 - 20,  $\beta$  ranges from 0.2 to 0.4 m<sup>2</sup>/g<sup>54</sup>, and  $\alpha$  from 0.1 to > 1 m<sup>-1</sup> <sup>54</sup>. The first two are fundamental characteristics of the algae and would not be expected to vary much from lake-tolake. The last parameter,  $\alpha$ , is more lake-specific.

Further simplification of the parameter estimation problems for a given lake might be facilitated with the use of Secchi disc observations. A number of investigators  $^{36,84}$  have shown that the Secchi depth is inversely related to the extinction coefficient in the water column:

$$Z_{c} \varepsilon = k \qquad (2.4-14)$$

where

C

{

Holmes<sup>36</sup> has shown that the value of k is approximately 1.44. Combining this equation with equation (9):

$$\frac{1}{Z_s} = \frac{\alpha + \beta C}{k}$$
(2.4-15)

It is assumed that Secchi depth observations are available over the course of a year, and that  $\alpha$ , the residual, or non-algal portion of the optical density is independent of season. The maximum Secchi depth would be observed when algal biomass is at an insignificant level. Accordingly:

$$\frac{1}{2^{\max}_{s}} = \frac{\alpha}{k}$$
(2.4-16)

$$\alpha = \frac{k}{z_{s}^{\max}} = \frac{1.44}{z_{s}^{\max}}$$
(2.4-17)

Likewise, during peak algal biomass periods, the Secchi depth would be at a mimimum:

$$\frac{1}{z_{s}^{\min}} = \frac{1}{k} \left( \alpha + \beta C_{L}^{\max} \right) = \frac{1}{z_{s}^{\max}} + \frac{\beta C_{L}^{\max}}{k} \qquad (2.4-18)$$

If the peak biomass is not nutrient-limited, equations (6), (13), and (18) can be combined to give:

$$\frac{1}{z_{\rm s}^{\rm min}} = \frac{1}{k} \frac{\mu_{\rm max}}{r^2} F$$
 (2.4-19)

$$\frac{\mu_{\max}}{r} = \frac{Z}{k r z_s^{\min}} = \frac{12.71 Z}{\lambda z_s^{\min}}$$
(2.4-20)

Thus, this scheme permits application of the Lorenzen-Mitchell model with knowledge of the maximum and minimum Secchi depths, spring nutrient concentration  $(N_0)$ , daylength  $(\lambda)$ , and the stoichiometric parameters  $\beta$  and y. The first four are directly observable. The last two are fundamental characteristics of algae and would not be expected to vary much from lake to lake.  $\mu_{max}/r$  is also a fundamental parameter, and agreement between the predictions of equation (20) and the literature range of 10 to 20 could be viewed as a partial verification of this approach. For Onondaga Lake, New York, for example, equation (20) has been evaluated from mixed depth (Z=7.2m) and Secchi observations ( $z_s^{min} = 0.5 m$ ) to yield a value of 12.1 for  $\mu_{max}/r$ , within the range of reported values. Similarly, for this lake a value of 0.58 for  $\alpha$  has been estimated from equation (17), based upon an observed maximum Secchi depth of 2.5 m. Application

(

of this scheme to Onondaga Lake will be discussed further in Chapter 3. The above is intended as a demonstration of how simplification of the general questions describing algal dynamics can yield model formulations which have realism and which are relatively easily applied to address specific problems.

### 2.5 Analysis of Empirical Approaches

The relatively widespread application of the models of 93,95 Vollenweider and of Dillon for predicting lake trophic state has resulted primarily from their attractive simplicity, relatively low data requirements, and initial successes. A valid criticism of these models is that they have been derived from generally small and somewhat restrictive data bases. This has occurred more out of necessity than out of choice, because of the limited availability of data. The empirical nature of these models leads to questions about their validity in lake systems other than those from which they have been derived. While they are all based upon the sound concept that nutrients, in particular, phosphorus, partially control rates of eutrophication, the parameter estimates, i.e. (the slopes and intercepts of the "permissible" and "dangerous" loading lines) have been empirically, even subjectively determined and may not be appropriate for other lakes. Increased availability of data on lakes from wider geographic, morphometric, and trophic ranges has facilitated systematic evaluation of these schemes.

( .

In order to provide a basis for evaluating and comparing these models, data from 105 northern temperate lakes have been compiled (Table 2.5-1). The 1972 sampling of the National Eutrophication Survey<sup>86</sup> has served as the primary data source. The lakes in this group are located primarily in Minnesota, Michigan, Wisconsin, Maine, New Hampshire, Vermont, and New York. These data have been screened by Reckhow<sup>68</sup> to eliminate lakes with acknowledged gross uncertainties in their measured or calculated nutrient or hydrologic budgets. A discussion of the limitations and errors likely to characterize the nutrient budgets estimated for the NES lakes has been presented previously. These data have been supplemented with lake data from the general literature<sup>18,39,78,93</sup>. The only restriction placed on the selection of the data base was that all lakes have mean depths greater than one meter.

Overall, the data set consists of 22 oliogotrophic, 23 mesotrophic, and 60 eutrophic lakes. The classifications have been based upon subjective evaluations of such indicator variables as transparency, chlorophyll, and hypolimnetic dissolved oxygen. Nutrient data should not have contributed directly to these classifications, so that valid evaluations can be made of models predicting trophic state as functions of morphometric, hydrologic, and nutritional factors.

# Table 2.5-1

# Tabulation of Lake Data

| LAKÊ                     | S AREA | D AREA | 2           | . T       | QS                 | L           | R            | · C1    | 68       | TROPHIC .                             | · •       |
|--------------------------|--------|--------|-------------|-----------|--------------------|-------------|--------------|---------|----------|---------------------------------------|-----------|
|                          | K#2    | KH2    | H.          | YRS       | H/YR               | G7 M2 - Y R | -            | G7 M3   | G/M3     | STATE                                 | Reference |
| SARATOGA                 | 1.00   | 632.   | 7.94        | 0.411     | 19.295             | 1.600       | 0+530        | 0.0829  | 0.0390   | E                                     | 86        |
| SACAGANDAGA              | 7.45   | 2583+  | 7.59        | .0.455    | _16.700            | 0.180       | 0.250        | 0.0108  | 0.0081   | <u> </u>                              | ĭ         |
| RÉUKA                    | 2+89   | 424+   | 22.60       | 7.813     | 2.893              | 0.045       | 0.556        | 0.0156  | 0.0069   |                                       |           |
| LOWER ST REGIS           | 0.11   | 53.    | 5,10        | 0.301     | 16+932             | . 0.400     | <u>0.111</u> | 0+0236  | 0.0210   | F                                     | _         |
| CEUSS                    | 0.54   | -0025. | 5.50        | 0+019     | 286.458            | 33.500      | 0.226        | 0.1169  | 0.0905   |                                       |           |
| CONESUS                  | : 0.79 | 168.   | <u> </u>    | 1.701     | 5.233              | 0.380       | 0,560        | 0.0726  | 0,0320   | E                                     | _         |
| LAYUGA                   | 10.50  | 1862.  | 54.50       | 11-236    | 4.850              | 0.540       | 0.570        | 0.1113  | 0+0479   | M                                     | •         |
| CARRY FALLS              | 1.59   | 2262.  | 5.40        | 0.104     | _51.323            | 0.720       | 0,280        | 0.0139  | 0.0100   | <u> </u>                              | - · .   · |
| CANNENS FERRY            | 1-19   | 115/-  | 14.50       | 0+5/5     | 33.409             | . 4.260     | 0+710        | 0.1275  | 0+0370   | E                                     | 1.        |
| CANANDAIGUA              | 2.02   |        | 34+00       | - 14+ 423 |                    |             | 0.720        | 0.0159  | 0.0129   |                                       | -         |
| RANGELES<br>LONG         | 27+20  | 2304   | 13.41       | 2.000     | 2411B              | 0.130       | 0.443        | 0+0110  | 0.0070   |                                       | 1         |
|                          | 17.35  | 011    | 3.66        | 0.115     | 11.076             | 0-400       | 0.300        | 0.0280  | 0.0120   |                                       | — i       |
| CINCE                    | 12422  | 2434   | 3.34        | 0.010     | 314060             | A. 370      | 0.160        | 0+0109  | 0+0120   | · ·                                   | 1         |
| 1 400141                 | 0.62   | 694.   | 1.68        | 0.003     | 500.000            | 25.210      | 0.200        | 0.0450  | 0.0340   |                                       | - !       |
| ABRINARIO MT             | 3.34   | 1016.  | 3.34        | 0.010     | 314.000            | 11.260      | 0.270        | 0.0350  | 0.0262   |                                       | ļ         |
| MATERMINE AD             | 2.40   | 290.   | 12.68       | 0 227     | 45 848             | 1.340       | 0.240        | 0.0337  | A. 0192  | <u>_</u>                              | - 1       |
| MATERCOURT<br>MITERCOURT | 100.44 | 840.   | 13.11       | A_000     | 3.277              | 0.110       | 0.210        | 0.0334  | A. 0001  |                                       | 1         |
| ALTONA                   | 3.40   | 2100.  | 2,13        | 0.014     | 152.141            | 10.010      | -0.080       | 0.1308  | 0.1413   |                                       | — i       |
| SALL CLATOS              | 4.52   | 1541.  | 2.76        | 0.027     | 83.706             | 9.040       | -0.020       | 0.1080  | 0.1107   |                                       | 1         |
| G6 26(1)                 | 0,95   | 253.   | 1.22        | 0.010     | 40.467             | 8.570       | 04140        | 0,2107  | 0.1770   | e e                                   |           |
| STNISSIPPI               | 9.31   | 987.   | 1.40        | 0-090     | 15.556             | 6.350       | 0.010        | 0.4082  | 0.4041   | - E                                   | 1         |
| TUMALINE                 | 0.61   | 12.    | 3.78        | 0-693     | 5.455              | 1.400       | 0.470        | 0-2567  | D-1360   | - E                                   |           |
| WILLUA A                 | 20.78  | 847.   | 4.57        | 0.326     | 14.018             | 0.470       | 0.110        | 0.0335  | 0.0298   |                                       | 1         |
| OUTTERNUT                | . 4.07 | 119.   | 4.24        | 0_400     | 10.600             | 0.640       | 0.030        | 0.0604  | 0.0566   | E                                     |           |
| NAGAWICKA.               | 4,15   | 116+   | 9.91        | 1,500     | 6.607              | 1.330       | 0.560        | 0.2013  | 0+0886   | E                                     | î         |
| TAINTER                  | 6.85   | 4351.  | 4.05        | 0+027     | 150.000            | 20.300      | 0.120        | 0.1353  | 0.1191   | Ē                                     | — 1 ·     |
|                          | 21+97  | 14375. |             | 0.046     | 191.522            | 8.870       | 0,150        | _0.0463 | 0+0394   | <u> </u>                              | · · · ·   |
| ALBERT-LEA               | 9.93   | 382.   | 1-07        | 0.200     | 5.350              | 6.310       | 0.290        | 1.1794  | 0.8374   | E                                     | - t ·     |
| NC QUADE                 |        | _ 64.  | 2.74        | 0.156     | 17.564             | 1.200       | 0.290        | 0.0683  | _ 0.0485 | <u>'E</u>                             | ·         |
| PELICAN                  | 44.29  | 179.   | 2-41        | 3+300     | 0.730              | 0+060       | 0.500        | 0+0822  | 0.0411   | . 14                                  | — · ·     |
| SILVER                   | 1.71   | 6.     | 1.22        | 2.300     | 0.530              | 0.560       | 0.320        | 1.0934  | _0.7435  | <u> </u>                              | i         |
| ANDKUSIA                 | 6.11   | 1896.  | 7.92        | 0.130     | 60+923             | 4-026       | 0.370        | 0.9660  | 0.0416   | e E                                   | i         |
| CASS                     | 63.17  | 2927.  | <u></u> 62  | 0.856     | 8,881              | 0.326       | 0.600        | _0.0367 | 0+0147   | <u> </u>                              | <b></b>   |
| CUKATO                   | 2+20   | 417.   | 7+62        | 1.100     | 6+927              | 2.600       | 0+440        | 0+3753  | 0.2102   | E .                                   |           |
| NEST                     | 3.82   | 319.   | 4.57        | 0.521     | 6.772              | 0.800       | 0.560        | 0+0912  | 0+0401   | <u>E</u>                              | <b></b>   |
| NULF                     | 4.23   |        | 8+23        | 0-101     | 84.457             | 6.930       | 0.120        | 0.0161  | 0.00/0   |                                       | 1         |
|                          |        | 10344  | <u></u>     | 0,13      | 13.283             | 0.440       | 0,300        | 0.0331  | - 0.0212 |                                       | - :       |
| UUFFALU<br>EETEE         | 0.11   | 114.   | 4+92        | 1+400     | 3.15/              | L.400       | 0.290        | 0.0017  | 0.2040   |                                       | i .       |
| ESIES                    |        |        | 3+05        | 0.030     | 1 101.651          | A-910       | 0,050        | 0+0947  | 0.0900   | <u> </u>                              |           |
|                          | 44,40  | 13101. | - 2-13      | 0+030     | > >9+167           | 5.330       | 0.030        | 0,0439  | 0.0910   |                                       | 1         |
| CA STOWATED              | 30+04  | 231014 | <u>L+D3</u> | 0.011     | 100+304            | 12+210      | 0.030        | 0.0314  | 0.0300   | · · · · · · · · · · · · · · · · · · · | į .       |
| lernater<br>15leh        | 457.20 | 9404   | 2.10        | 1.400     | 1 3.100<br>1 3.100 | 0.040       | 0.180        | 0.1010  | 0.0270   |                                       | Į         |
| AL 1 21-AL               |        | 2007   | 3.16        | 2,240     | ) 174-314          | 31.400      | 0.240        | 0.1781  | 0-1353   |                                       | <u>:</u>  |
| FORD                     | 4.25   | 2108.  | 6.36        | 0.051     | 106.341            | 16.170      | 0.190        | 0-1521  | 0-1232   |                                       | !         |
| HIWHNS                   | 38-85  | 127-   | 14.94       | 15.600    | U 4.95A            | 0+030       | 0.670        | 0.0313  | 0.0103   | ŏ                                     |           |
| LENG INTCHIGANS          | 0.85   | 64.    | 5.14        | 0.08      | 5 60.941           | 4.620       | 0.090        | 0.0758  | 0.0690   | Ē                                     |           |
| 1130611753               | 81.10  | 494.   | 2,32        | 1.30      | 1.787              | 0.050       | 0.500        | 0.0337  | 0.0168   | ) <u>N</u>                            |           |
| STRAWBERRY               | 1.04   | 914-   | 6.74        | 0.01      | 6 189.244          | 9.190       | 0.221        | 0.0486  | 0.0376   | J Ê                                   |           |
| THALMAPPLE               | 1.00   | 930-   | 4.24        | 0.03      | 140.791            | 9,240       | 0.210        | 0.0656  | 0.051    | ) Ē                                   | <u> </u>  |
|                          | 10.41  | 1305.  | 6.86        | 0.15      | 3 44.718           | 1.960       | 0.222        | 0.0443  | 0.0344   | E                                     |           |
| 2055                     | 1.19   | 1261.  | 1.52        | 0.00      | 5 277.164          | 17-020      | 0.144        | 0.0614  | 0.0526   | E E                                   |           |
|                          |        |        |             |           |                    |             | *****        |         |          |                                       | v         |

2-80

 $\bigcirc$ 

| Table | 2.5-1 | (continued) |
|-------|-------|-------------|

.

•••

. .

,

•

| LAKE                      | S ANEA   | U AREA |        | YRS             | M/YR             | GZNZ-YR           | - <u>K</u> | 6/83    | 6/43    | STATE      | - Refe    |
|---------------------------|----------|--------|--------|-----------------|------------------|-------------------|------------|---------|---------|------------|-----------|
|                           | •        | KIIG   |        |                 |                  | ₩ <b>7 H</b> 6-1A |            |         |         |            | Vera      |
| SLATERSVILLE              | . 0+84   | 232.   | 2,44   | 0.014           | 178-029          | 5.6LO             | 0+168      | 0.0315  | 0.0262  | Ę          | 86        |
| SEBAGU                    | 116.49   | 1026.  | 30.79  | 5.430           | 5.702            | 0.080             | 0.500      | 0.0140  | 0.0070  | <u> </u>   | !         |
| DAY UP HAPLES             | 30.85    | 305.   | 4.27   | 0.071           | 59.948           | 0.520             | 0.192      | 0.0087  | 0.0070  | . 0        | 1         |
| HOUSEHEAD                 | 303.22   | 2977.  | 16.46  | 3,000           | 5.488            | 0.080             | _0.375     | 0+0146  | _0+0091 |            | - 1       |
| LENG CHILIP               | 19+71    | 295.   | 10.37  | 1.200           | 8.038            | 0.140             | 0.500      | 0+0162  | 0.0081  |            |           |
| CHANLEYUIA                | 09+88    | 787.   |        | 3.200           | 2+290            | 0.120             | 0.844      | 0.1043  | 0.0303  | <u></u>    | -         |
| COVETAL                   | 2.01     | 61+    | 4.74   | 3,300           | 1.284            | 1.229             | 0.714      | 0.0546  | 191010  | 1 5        |           |
|                           | 1.26     | 14.    | 8-63   | 4.200           | 2-054            | 0.220             | 0.636      | 0.1071  | 0.0349  |            | - 1       |
| HARRINAN RES              | ·        | 478.   | 10.37  | 0.214           | 48-507           | 0.880             | 0.170      | 0.0181  | 0.0150  | - <b>R</b> |           |
| POWDER MILL               | 1.51     | 313.   | 2.47   | 0.018           | 136.735          | 3.250             | 0.099      | 0.0234  | 0.0211  | · E        | - 1       |
| RELLY'S FALLS             | 0.52     | 565+   | 2.26   | 0.004           | 550.268          | 28.810            | 0.327      | 0+0524  | 0.0352  | Ē          | _ 1       |
| GLEN                      | 0.61     | 523.   | 3.35   |                 | 408,988          | 13.130            | 0,318      | 1560-0  | 0+0219  | 3          | .         |
| ST CROEX                  | 33.24    | 19919. | 8.78   | 0.063           | 139.373          | 8.890             | _0.326     | 0.0638  | 0+0430  | E_         |           |
| NENPHRENAGOG              | 94+62    | 1686.  | 15,55  | 1.700           | 9+146            | 0+500             | 0.420      | 0+0547  | 0.0317  |            | - 1       |
| WAGONGA                   | 6.54     |        | 1.28   | 1.400           | 0.914            | 4.000             | 0.570      | 4.3750  | 1.8912  | - E        | _ Ŧ       |
| TRUUT                     | 7+65     | 42+    | 15.24  | 17.400          | 0.876            | 0.372             | 0.920      | 0+4247  | 0.0340  | Ē          |           |
| CLEAR                     |          |        | 12.50  | 7.692           | 1.625            | 6+040             | 0.800      | 0.0246  | 0,0049  | 0          | - 18      |
| ékle .                    | -        |        | 10.00  | 2.500           | 7.200            | 1.000             | 0.840      | 0.1472  | 0.0236  | - E -      | 1         |
| ONTERIO                   |          |        | 84,00  | <u>6.579</u>    | 12-768           | 0.650             | 0.780      | 0.0509  | 0.0112  | <u>N</u>   | <b></b> } |
| SUPERION                  | Ξ        | _      | 149103 | 104+014         | 0-184            | 0.030             | 0.900      | 0.0392  | 0.0038  |            | 1         |
| <u>NICIIOA (</u>          |          |        | 4.40   | 4.202           | 24050            | 0.340             | 0.030      | 0.1747  | 0.0352  |            | -         |
| ect<br>Akaliacan          |          | =      | 75.30  | 402V2<br>58.874 | 1.280            | 0.390             | 04070      | 0.3047  | 0.0167  |            |           |
| Skalla                    |          |        | 26.50  | 1,174           | 23.585           | 2,200             | 9.650      | 0.0933  | 0.0376  |            | - 1       |
| R AL AMALY A              | -        | _      | 58.00  | 111.111         | 0. 622           | 0. 120            | 0.900      | 0.6130  | 0.0413  | Ē          |           |
|                           |          |        | 21.00  | 108.494         | 0.193            | 0.500             | 0.900      | 2.5880  | 0.2588  | F          | - 1       |
| NEXIMA                    | -        | -      | 7.80   | 1,200           | 6.437            | 2.160             | 0.710      | 0.3296  | 0.0955  | · .        | i         |
| MAUBESA                   |          |        | 4.80   | 0.300           | 15.984           | 9,930             | 0.0        | 0.6212  | 0-6712  | · Ē ·      |           |
| KEGUNSA                   |          |        | 4.60   | 0.350           | 13-154           | 6.670             | 0.090      | 0.5071  | .0.4614 |            | _ 7       |
| TAHOE                     |          |        | 303.00 | 714,286         | 0.424            | 0+042             | 0.930      | 0+0990  | 0.0069  | . 0        |           |
| AEGERISEE                 |          |        | 49.00  | <u> </u>        | 5.635            | 0.160             | 0.680      | 0.0284  | 0.0091  |            | - 93      |
| TURLERSEE                 | -        | -      | 14.00  | 2.151           | 6.510            | 0.300             | 0,800      | 0.0461  | 0+0092  | M          | -1        |
| HALLWILERSEE              | <u> </u> |        | 28.00  | 3.846           | 7,260            | 0.550             | 0.360      | 0.0756  | 0,0484  | <u></u>    | - 1       |
|                           | —        | -      | 100+00 | 4-818           | 20+500           | 1+100             | 0.650      | 0.0537  | 0-0185  | i E        | · •       |
| 2001CUC66                 | <u> </u> |        | 50.00  | 2+34            | <u> </u>         | 1+#50             | 0.110      | 0.1962  | 0.0001  | <u> </u>   | - 1       |
| PD SIE ENCLE              | _        | _      | 10.00  | 2 01 0          | 342000           | 1.320             | 0+230      | 0.0300  | 0.0473  |            | ļ         |
| MALINEACEDEE              |          |        | 1 4.00 | 2-04            | 4.310            | <u> </u>          | 0+020      | <u></u> | 0.0013  | <u> </u>   | - 1       |
| NACHCOGERJEE<br>Nachdirin |          | =      | 54.00  | 4.343           | 14980<br>6450    | 1.150             | 0.010      | 0.4744  | 0.0912  | E E        |           |
| C-AVACHERSEE              |          |        | 44-00  | 16.040          | 2.214            | 4.030             | 0.040      | 0 2017  |         | Ę          | - 7       |
| CANékaN                   | 12.72    | 3250.  | 7.10   | 0.063           | 117.498          | T. 9. 0           | 0.340      | 0.0174  | 0.0115  |            | - 72      |
| FOUS-MILE                 | 7.73     | 45.    | 9,30   | 4,310           | 2,158            | 0.108             | 0.830      | 0.0501  | 0-0085  | <u> </u>   | - 10      |
| 806                       | 2.27     | 29.    | 19.00  | 3.020           | 5.960            | 0.157             | 0.730      | 0.0263  | 0.0071  | Ö          | _         |
| 12-MILE-BUSHKUNG          | 11.65    | 933.   | 18.10  | 0.480           | 37.708           | 0.330             | 0.360      | 0.0068  | 0.0056  | 0          | -         |
| HALLS                     | 5.32     | 270,   | 27.20  | 1+120           | 24.286           | 0.205             | 0.530      | 0.0084  | 0.0040  | 0 0        | I         |
| bitCH                     | 1.36     | 515.   | 9.80   | 0.057           | 2 188.462        | 1.520             | 0+0+0      | 0.0081  | 0+0076  | 0          | — )       |
| MAPLE                     | 3,32     | 502,   | 11+60  | 0.140           | 78.378           | 0.830             | 0.320      | 0.0106  | 0.0012  | <u> </u>   | l         |
| PINE                      | 1+06     | 285.   | 7.40   | 0+05            | 8 127.586        | 1.055             | 0+020      | 0.0083  | 0+0081  | L , H      | ŀ         |
| CRANBERAY                 | 0.69     | 269,   | 3.50   | 0.01            | <u>7 205+882</u> | 1.300             | -0.040     | 0.0063  | 0.0066  | . 0        | !         |
| EAGLE-HOOSE               | 5.15     | 251.   | 12.80  | 0.52            | L. 24+568        | 0.225             | 0.320      | 0,0092  | 0.0062  | 2 0        |           |
| OBLUNG-HAL SURTON         | 11.03    | 143.   | 17.70  | 2. 601          | ) 6.á0d          | 0.124             | 0.720      | 0.0192  | 0.005   | tO .       |           |

#### 2.5.1 Lake Data Base Description (continued)

( )

As an initial step in the analysis, the statistical distributions of the variables in Table 2.5-1 were examined. It was evident that a logarithmic transformation was appropriate to promote symmetry in the distributions in all of the variables except R, the retention coefficient. Histograms of the transformed data, stratified by trophic state, were generated using the BMDP7D computer program  $^{26}$ , which also computed gross and within-group means, standard deviations, and ranges, as well as analyses of variance to test the statistical significance of variations across groups. Table 2.5-2 identifies the variables examined and displays the results.

The summary F table indicates that there were significant differences across groups for all variables, with significance levels ranging from greater than 99.99% for L, CI, CO, and L/Z to 93.02% for R. The highest F level observed was 66.2, in the case of the average outlet concentration, CO. Trophic states were most highly stratified, or, equivalently, most distinctly classified, on this variable. This is consistent with Dillon's model. The stratification with respect to depth is consistent with Vollenweider's first model, i.e., on the average, eutrophic lakes were shallower. However, the stratification with respect to surface overflow rate, QS, is contrary to Vollenweider's second model, which predicts that higher surface overflow rates should be associated with oligotrophic lakes. Because of

| Vari | able |                                         | Units                  | Transfor-<br>wation | P<br>Statistic <sup>a</sup> | Prob. F<br>Exceeded | Table |
|------|------|-----------------------------------------|------------------------|---------------------|-----------------------------|---------------------|-------|
| 1    | z    | mean depth                              | (m)                    | log <sub>10</sub>   | 10.51                       | .0001               | B     |
| 2    | т    | hydraulic residence time                | (yr)                   | <sup>109</sup> 10   | 7.82                        | .0007               | с     |
| 3    | QS   | surface overflow rate                   | (m/yr)                 | log <sub>10</sub>   | 3.35                        | .03 <del>9</del> 1  | A     |
| 4    | L    | phosphorus loading per unit area        | (g/m <sup>2</sup> ~yr) | log <sub>10</sub>   | 52 <b>.9</b> 0              | <.0001              | D     |
| 5    | L/2  | phosphorus loading per unit volume      | (g/m <sup>3</sup> -yr) | <sup>log</sup> 10   | 43.44                       | <.0001              | E     |
| 6    | cı   | average inlet phosphorus concentration  | (g/m <sup>3</sup> )    | 10 <sup>9</sup> 10  | 37.84                       | <.0001              | G     |
| 7    | со   | average outlet phosphorus concentration | (g/m <sup>3</sup> )    | امع<br>10           | 66.21                       | < 0001              | Ħ     |
| 8    | R    | phosphorus retention coefficient        | -                      | -                   | 2.73                        | .0698               | F     |

- -

...

Identification of Variables and Analyses of Variance Across Trophic Groups

Table 2.5-2

. .

a - degrees of freedom = 2,102

- .

. .

.

.

•

2-83

 $\sum_{i=1}^{n}$ 

.

.

| BULATIZIO          | · VIPLARLE       | 3 95              | STRATIFICATION                        |         | 1ABLE       | 15                                     |                                       |
|--------------------|------------------|-------------------|---------------------------------------|---------|-------------|----------------------------------------|---------------------------------------|
| •                  |                  |                   |                                       |         |             |                                        | EXCLUDED                              |
| QL1                | ¢                | ME10              |                                       |         | EUTA        |                                        | VALUES<br>ES                          |
| DEDINTS            | *****            | ************      | ****                                  | ******  | *********   | • • • • • • • • • • •                  | **********                            |
| 10003              |                  |                   |                                       |         |             |                                        |                                       |
| 2-10-11            |                  |                   |                                       | -       |             |                                        |                                       |
| 2.6033             |                  | •                 |                                       |         |             |                                        |                                       |
| 2.2001 *           |                  |                   |                                       |         |             |                                        |                                       |
| 2.0221.            |                  | •                 |                                       |         |             | •                                      |                                       |
| 1.000344           |                  |                   |                                       |         |             |                                        | •••                                   |
| 1.402100           |                  |                   |                                       |         |             |                                        | · · · · · · · · · · · · · · · · · · · |
| 1.2301             |                  |                   |                                       |         | ******      |                                        |                                       |
| 1.0001             | -                | Ņ4#               |                                       |         | 4484        |                                        |                                       |
| <u></u>            | ****             | <del>}**</del>    | ··· · · · · · · · · · · · · · · · · · |         | ******      |                                        |                                       |
| 0.0001**           |                  |                   | •                                     |         | · · ·       |                                        |                                       |
| 999999             |                  |                   |                                       |         |             |                                        |                                       |
| - U. U.1U1 P       |                  |                   |                                       |         |             |                                        |                                       |
| -4+2003 -          |                  | •                 |                                       |         |             |                                        |                                       |
| <u>-0,40,110</u>   |                  |                   |                                       |         |             |                                        |                                       |
| -U+QA91<br>-U+QA91 |                  |                   |                                       |         |             |                                        |                                       |
| WP MEATS           | LAT DENOTED BY M | S IF THEY COINCID | E WJEH **\$, #*                       | S DIHCA | WJ SE.      |                                        |                                       |
| EAN                | 0.576            | 1.0               | 16                                    |         | 1.343       |                                        |                                       |
| <u></u>            | Ca193            |                   | <u>22</u>                             |         | 40.000      |                                        | ·                                     |
|                    | c. 314           | 2.3               | 25                                    |         |             |                                        |                                       |
| *( <i>MUN</i> ) *  | 6.372            | -0+1              | 35                                    |         | -0.714      |                                        |                                       |
| L GAQUPS CO        | MANNED CCASAS EX | CLUNED IF SPECIAL | COOSS FOR EIT                         | IER VAR | TADLET      |                                        |                                       |
|                    |                  |                   | UN OF SQUARES                         | DF      | MEAN SQUARE | e RATIO                                | PROB. # EXCEEDED                      |
| LHE                | 1-174            | BETHELM           | 4.2245                                | 2       | -2.1143     | 3.3482                                 | 0.0341                                |
| 1128.              |                  | <u>METHIN</u>     | <u></u>                               | <u></u> |             | ······································ |                                       |
| N                  | 10%.             | ratal             | 84.6382                               | 104     |             |                                        | · •                                   |

B - Mean Depth

ţ

ĆE

• • • • •

;

ť

| TARULATIEN OF VARIATLE           | 1.2 HITH STRATEFICATIO                 | I ON VANJAMLE 9 TS        | •                                     |
|----------------------------------|----------------------------------------|---------------------------|---------------------------------------|
| 0116                             | MĚÍD                                   | EUTR                      | <u>EKCLUOED</u><br>VALUES<br>TS       |
| ****************                 | ****************                       | ************************* | *******************                   |
| 4.764P                           |                                        |                           |                                       |
|                                  |                                        |                           |                                       |
| 2+9601<br>                       |                                        |                           |                                       |
| 2.1003+                          | ······································ |                           |                                       |
|                                  | <u></u>                                | ·····                     |                                       |
| 1.0521+*                         | •                                      |                           |                                       |
| 1.2003**                         | •                                      | +                         |                                       |
| 1.240540000000                   |                                        |                           | ·····                                 |
|                                  |                                        |                           | ·                                     |
| 0.5035                           |                                        | ********                  |                                       |
|                                  | <del></del>                            |                           |                                       |
|                                  |                                        | *****                     |                                       |
| 0.3003<br>6-1.304                | · · · · · · · · · · · · · · · · · · ·  | *****                     |                                       |
| - 0.001                          | ·····                                  |                           |                                       |
| Chatful                          | -                                      |                           | · · · · · · · · · · · · · · · · · · · |
| CHOOP HEARTS ARE CITIKITED BY HI | 2 IN THEN COINCIDE MICH #.25 H.        | -> DIMENTIZE              | 4                                     |
| HEAN 1.279                       | 1.047                                  | 0.766                     |                                       |
| 2. 1/1 Ye (7.559                 | 0.411                                  | 0.597                     |                                       |
| MARINET CANES                    |                                        | P+900                     |                                       |
| NEGENUA 64344                    | 0.345                                  | 0. 929                    |                                       |
| ALL GADU'S CONDINED COASES EN    | CLUDED IF SPECIAL CODES FOR EIT        | HER VAFCAULES             |                                       |
|                                  | SUN OF SQUARES                         | OF HEAN SQUARE F MATIO    | PPOB. # EXCEEDED                      |
| Meail U.+31<br>S. Urv. D.+31     | REINECH 7+3123                         | 2 2.1561 10.5046          | 0.0201                                |
| M 105+                           | FOTAL 25.2463                          | 104                       |                                       |
|                                  |                                        |                           |                                       |
|                                  |                                        |                           |                                       |

• •
| C - Mean Hydraulic | Residence | Time |
|--------------------|-----------|------|
|--------------------|-----------|------|

( ÷

۰.

ſ

.

•

-. .

.

| AUULATION OF WARTABLE                                                                                          | ZT. HITH STAL         | TIFECATION ON VA  | <u>e anle 9</u>    | <u></u>  |                                        |
|----------------------------------------------------------------------------------------------------------------|-----------------------|-------------------|--------------------|----------|----------------------------------------|
|                                                                                                                |                       |                   |                    |          | EXCLUDED.                              |
| 3.15                                                                                                           | MESO                  |                   | EVTR               |          |                                        |
| ************************                                                                                       | ********************  | ***********       |                    | ******** | ******************                     |
| 3.3003                                                                                                         |                       |                   |                    |          |                                        |
| 4.0001*                                                                                                        |                       |                   |                    |          |                                        |
| 2.40014                                                                                                        |                       |                   |                    |          |                                        |
| 2.1000                                                                                                         | -                     |                   | 1.00               |          |                                        |
| 1.3001                                                                                                         |                       |                   |                    |          |                                        |
| 4+1035**                                                                                                       | <u> </u>              |                   |                    |          |                                        |
| 9.49001 FP<br>M.AU.11 6496449                                                                                  | ****                  |                   | ****               |          |                                        |
| 4.30UJH+                                                                                                       | **                    |                   | ******             |          |                                        |
| - 0, 0000 -                                                                                                    |                       | ······            |                    |          |                                        |
| -0.4001                                                                                                        |                       |                   |                    |          |                                        |
| -0.40034                                                                                                       | **                    |                   |                    | -        |                                        |
| -1.5020                                                                                                        |                       |                   | ******             |          | ······································ |
| -1-40024                                                                                                       |                       |                   | ****               | <u> </u> |                                        |
| -221441                                                                                                        | •<br>•                |                   |                    |          | •                                      |
| OUP HEATS AFE ULHUTED BY Nº                                                                                    | E EF THEY COINCIDE WI | TH ++5+ N+5 OTHER | W 16               |          |                                        |
| IE AIJ 0.315                                                                                                   | 0.029                 |                   | -0+311             |          |                                        |
| _VLV                                                                                                           |                       |                   |                    |          |                                        |
| x [HuH2.255                                                                                                    |                       |                   | Z_046              |          |                                        |
| NIHUN -1.770                                                                                                   | -2.000                | •                 | -2.925             |          |                                        |
| L GOUPS COMMENTS COASES EX                                                                                     | LUCED IF SPECIAL COD  | ES FOR ESTHER VAL | 1                  |          | • • • •                                |
|                                                                                                                | 504 0                 | F SUUARES OF      | HEAN SOUARE        | F AAFID  | PROB. F EXCELOED                       |
| -9.243<br>DFV. 1.110                                                                                           |                       | 1.1019 102        | N. 5129<br>F. 0493 | 7,8150   | 0.0001                                 |
| ومحجب والقابة والمتحدث والمتح | 10141 42              | 0.134b 184        |                    |          |                                        |
| M 10 <b>F</b> +                                                                                                | 101/6 10              |                   |                    |          |                                        |

÷ .

.....

.

•

.

D - Phosphorus Loading Per Unit Area

|                         |                                       |                  | - 'EXCLUDED            |
|-------------------------|---------------------------------------|------------------|------------------------|
| CL15                    |                                       | cute             | VALUES                 |
| ***********             | **********************************    | ••••••••••       | ******************     |
| 4 - (1,1,1,3)           |                                       |                  |                        |
| 1.0005                  |                                       |                  |                        |
| 4-0403                  |                                       |                  |                        |
|                         |                                       |                  |                        |
| 1.6.0                   | •                                     | ********         |                        |
| U.6031                  |                                       | *****            |                        |
| C-4031                  |                                       |                  |                        |
| 0.434                   |                                       | P++              |                        |
| *u_dup1+                |                                       | 44449            |                        |
| -0.2441+                |                                       |                  | -                      |
| -0,-001+                | N+ **                                 | *******          |                        |
|                         |                                       |                  |                        |
| -1.44499.000            |                                       |                  |                        |
| -1 .2                   |                                       |                  | A                      |
| *****                   | <b>#</b> *                            |                  |                        |
| *1+0JJJ **              |                                       |                  |                        |
| MUP HEAVE ANE DENETED   | ET MIS IF THEY COLACIDE WITH + * 5. H | *S OTHERWISE     |                        |
| LAN +0.533              | -0.144                                | 0.482            |                        |
| <u>_QEY0.462</u>        |                                       | 2.592            |                        |
| 4 62,000<br>61894 0.343 | 23.000                                | 10.000           |                        |
| -1.525                  | -1.578                                | -4.65\$          |                        |
| - GROUPS CEMBERED LEAS  | ES EXCLUDED LE SPECIAL CODES FOR EL   | THER VARIABLES   | <u></u>                |
|                         | SUN OF SUUTRES                        | DP MEAN SQUARE F | RATIO PROD. F EXCLEDED |
| 14 U.UO4                | 861ween #3.7347                       | 2 10.0014 54     | 0.0300                 |
| 14 605.                 | <u>411HIH 32-4425</u>                 | Q, <u>1104</u>   |                        |
|                         | 10/AL 00+6273                         | 144              |                        |

,

## Table 2.5-2 (continued)

| Е- | Phospl | orus | Loading | Per | Unit | Volume |
|----|--------|------|---------|-----|------|--------|
|    |        |      |         |     |      |        |

C

t

;

į

i

i

2

| TABULATICA OF VAPTAPLE          | 5 L/E               | ATTERCATIO | 1 34 VI | APLE 9      | 15       |                   |
|---------------------------------|---------------------|------------|---------|-------------|----------|-------------------|
|                                 |                     |            | <u></u> |             |          | EKCLUDED          |
| 011G                            | MESO                |            |         | EUTA        |          | VALUES<br>TS      |
| *********************           | ****************    | *****      | ******  | ********    | ******** | ***************** |
| Leave                           |                     |            |         |             |          |                   |
| 1.5001                          |                     |            |         |             |          | <u></u>           |
|                                 | :                   |            |         |             |          |                   |
| 0.6001                          |                     | · • • • •  |         |             |          | · · · ·           |
|                                 |                     |            |         | ******      | -        |                   |
| -0.3301*                        |                     |            |         | <b>H+++</b> |          |                   |
| +0.6001                         | *****               |            |         | *****       |          |                   |
| -1.2003-                        | •                   |            |         |             | -        |                   |
| -1.5001                         |                     | -          |         |             | <b>·</b> | ·                 |
| -2.1001/1070484                 |                     |            |         |             |          |                   |
| -2.43.[+4++                     |                     | •          |         | •           |          |                   |
|                                 |                     |            |         |             |          |                   |
| -3.3(3)                         |                     |            |         |             | · ·      |                   |
| - 3 = 5(y) =<br>- 3 = 3(y) =    | •                   |            |         |             |          |                   |
| Shuup HEARS ARE GENDILD BY M'   | TF THEY CULNCLOB W  | 1TH        | s uther | WISE        |          |                   |
| MCAN =2.042                     | +1.441              |            |         | =0.244      | · ,      |                   |
| S. UFY. 0.613                   |                     |            |         | 0.900       |          |                   |
| 13 225004<br>MENTSAIN           | 23+000              | +          |         | 40.000      |          |                   |
| MEHEHUK -3,051                  | -2.191              |            |         | -2.250      |          |                   |
| ALL GRILIPS COMPINIE ICASES BIC | LUDED OF SPECIAL CO |            | IFA VAR | (ÅRCE)      |          |                   |
|                                 |                     |            |         |             |          |                   |
|                                 | SUM                 | JP SUUARES | 0F      | NEAN SUUARE | r sateo  | PROB. P ENCELOED  |
| MEAN -0.927                     | DETYPEN             | 61-9401    | ?       | \$0.9958    | 43.4413  | 0.0000            |
| N 105.                          | FUTAL 1             | 34.7663    |         | <u></u>     |          |                   |
|                                 |                     |            |         | <u> </u>    |          |                   |
| MEGEMUM -3.458                  |                     |            |         |             |          |                   |

.

E - Retention Coefficient

•

.

| TABULATER OF VARIAN         | LE                                     | STRATERICATION                        |                     | ADLE        | 15-     |                  |
|-----------------------------|----------------------------------------|---------------------------------------|---------------------|-------------|---------|------------------|
|                             |                                        |                                       |                     |             |         | EXCLUDED         |
|                             |                                        |                                       | _                   | EUTR        | •       | VALUES<br>TS     |
| MEDPUTRITS                  | ********                               | •••••                                 | • • • • • • • • • • | *********** | ******  |                  |
| 6.1201                      | •                                      | •                                     |                     |             |         |                  |
| 6. 7aDł                     | •                                      | · · · · · · · · · · · · · · · · · · · |                     |             |         |                  |
|                             | · · · •                                | · · · ·                               |                     | **          |         |                  |
| 0.77010                     | ••                                     | ••••                                  |                     | **          |         |                  |
| (064+D                      |                                        | ···                                   |                     |             |         |                  |
| 4+5003+94<br>9+59019        |                                        |                                       |                     | **          |         |                  |
| 0.4230-                     | ***                                    |                                       |                     | *           |         |                  |
| 0.2801                      |                                        |                                       |                     | ****        |         | •                |
| V- 14-1                     | ÷++                                    |                                       |                     | ******      |         |                  |
| -0.000)                     | ······································ |                                       |                     | *****       |         |                  |
| -0.0701+<br>                |                                        | · · · ·                               |                     | •           |         |                  |
|                             |                                        |                                       |                     |             |         |                  |
| GROOP MEATS ARE DENO.       | TED BY NºS IF THEE CONCED              | E WITH **>/ M*                        | S UTHER             |             |         |                  |
| NEAN 0+54E<br>              | 0.44                                   | 63<br>NI                              | · ·                 | 0.384       |         |                  |
| 4 22-000<br>ML (1909) C-930 | 33,00                                  | 00                                    |                     | 60.000      |         | •                |
| MIHIMUN -0.010              | 010                                    | 20 .                                  |                     | -0.030      |         |                  |
| ALL GAOUPS COMPINED         | GASES EXCLUDED IF SPECIAL              | CODES FOR ELTH                        | IER VAPS            | IOLES .     |         |                  |
|                             |                                        | IN OF SQUARES                         | C14                 | HEAN SOUARE | F RATIO | PAOB. P EXCELOED |
| MEAN 0.4                    | A SE FALEA                             | 0.4200                                | 102                 | 0.2100      | 2.7321  | 0.0598           |
| 10 103.                     | TOTAL                                  |                                       | 104                 |             |         |                  |
| MINIMUM ~0.0                | 10                                     |                                       |                     |             |         |                  |
|                             |                                        |                                       |                     |             |         |                  |

(

| TANULATINA UP VARIABLE           | 6 GI               | AATIFICATION  | UH VAP   | LABLE 9     | 75      |                  |
|----------------------------------|--------------------|---------------|----------|-------------|---------|------------------|
|                                  |                    |               |          |             |         | Excluden         |
| QU¢                              | MESO               |               |          | #J2TR       |         | ¥4LUES<br>TS     |
| MivEulota                        | ***************    |               |          | *********** |         | *********        |
|                                  |                    |               |          |             | •       |                  |
| . U.Suur                         | •                  |               |          |             |         |                  |
|                                  |                    |               |          | *           |         |                  |
|                                  |                    | <u> </u>      |          |             |         |                  |
| -2,241                           | ·                  |               |          |             | •       |                  |
| -J.5433                          | •                  |               |          | *****       |         |                  |
|                                  | ***                |               |          | Hettett     |         |                  |
| -1.2000                          | ***                |               |          | *********   |         |                  |
| -1,6003****                      | N******            |               |          |             |         |                  |
| -1.6003 1000                     |                    |               |          |             |         |                  |
| -2,2000                          |                    |               | -        |             |         |                  |
| -2-+001                          |                    |               | •        |             |         | ·····            |
| GROUP HEANS APE DEMOTED BY MIS   | IF THEY COINCLOS   | WBTH +*S, 4*  | S OTHERN | it se       |         |                  |
| NEAN -1+70"                      | -1.442             | •             |          | -0.441      |         | ·····            |
| N 22.000                         | 23.000             |               |          | 60.000      |         |                  |
|                                  | -0.516             |               |          | - 0.611     |         |                  |
|                                  |                    |               |          |             |         |                  |
| ALL GLUUPS CORNIN' & SCADES BACK | .UCED IF SPECTAL C | DOES PER RATA | IER VAR  | 40(C)       |         |                  |
|                                  | \$14               | OF SQUARES    | OF.      | MEAN SQUARE | FRATIO  | PROB. P EXCEEDED |
| HLAN -1.171                      | BETHEEN            | 14.0/03       | 2        | 7.0382      | 37.8343 | 0+0000           |
| 4 105.                           | FUTAL              | JJ.0+65       | 10+      |             |         |                  |
|                                  | ······             |               |          | • •         |         |                  |

H - Average Outlet Phosphorus Concentration

| TABULATION OF VARIABLE       | 7 CD WETH S         | TASTIFICATION                         | OH YAR   | 140LE        | <u>TS</u> |                  |
|------------------------------|---------------------|---------------------------------------|----------|--------------|-----------|------------------|
|                              |                     |                                       |          |              |           | EXCLUDED         |
| QL16                         | . NESO              |                                       |          | EUTR         |           | TS VALUES        |
| NIDPUTNIS                    | **********          | **********                            |          | **********   | *******   | **************** |
| G.1541 ·                     |                     |                                       |          | - •          |           |                  |
| 0.1501                       |                     |                                       |          |              |           |                  |
|                              |                     |                                       |          |              |           |                  |
| -4.1921                      |                     |                                       |          | •            |           |                  |
| -u.4501<br>u.4501            |                     |                                       |          |              |           |                  |
| -0.75.75                     |                     |                                       |          | ***          |           |                  |
| -1.0231                      |                     |                                       |          | *****        |           |                  |
| •1.203)<br>•1.303            |                     | · · · · · · · · · · · · · · · · · · · |          | <u>Heter</u> |           |                  |
|                              |                     |                                       |          | *********    |           |                  |
| -1.6503<br>                  |                     |                                       | •        |              |           |                  |
| -1.4533 4988                 | ****                |                                       |          |              |           |                  |
| -2.2.2.4                     |                     |                                       |          |              |           |                  |
| GROUP ILLING ARE DEMOTED BY  | PS IF THEY CUINCIDE |                                       | S OTHERI | IL SE        |           |                  |
| 1244 -2.827                  | -1.01               | 6                                     | -        | -1.190       |           | · · · · ·        |
| 4                            | 23,00               | 3                                     |          | Q <u>152</u> |           |                  |
| KallHum -1.000               |                     | <u> </u>                              |          | 0,274        |           |                  |
| -2+413                       | -2-10               | •                                     |          | ~1.749       |           |                  |
| ALL GROUPS CONSIMED FRASES E | XCLUDED IA SPECTAL  | CODES FOR ELT                         | HFR VAR  | AULES        |           |                  |
|                              | \$U                 | H OF SQUARES                          | DF       | MEAN SQUARE  | F AATEO   | PROD. F EACEEDED |
| NC 44 -1.501<br>Se ULY 0.556 | QETHEEN<br>WITHEN   | 14.2640                               | 102      | 9,1343       | 66,2120   | <b>0.</b> 0000   |
| N 105+                       | FOTAL               | 32.3407                               | 104      |              |           |                  |
| Minimum -2.412               |                     |                                       |          |              |           |                  |

2-87

• • • • • •

G - Average Inlet Phosphorus Concentration

intercorrelations among these variables, this univariate analysis should be viewed only as a preliminary description of the data, and not as a basis for evaluating models.

In order to further characterize the data, the productmoment correlation matrix of the variables is presented in Table 2.5-3. A high degree of association among the variables is indicated. Accordingly, a univariate or bivariate analysis would be virtually meaningless in establishing cause-effect relationships. Using the BMDP2R program<sup>26</sup>, stepwise linear regressions were performed to determine the proportion of the variance of each variable which could be explained by those remaining. In this analysis, the variables were grouped into two categories: "independent" and "dependent". Trophic state, outlet concentration, and retention coefficient were included in the latter because they are viewed as response variables, whose values depend upon the lake's nutrient dynamics. Only "independent" variables were allowed to enter into the regression analyses, subject to significance and tolerance constraints. Table 2.5-3 shows that between 55 and 87% of the variance in the independent variables could be explained. A linear model including T, L, and CI could explain 90% of the variance in the outlet concentration data. Of the three linearly\*-related variables 2,

\* On logarithmic scales, QS = Z - T.

Product-Moment Correlation Coefficients :

( ,

ł

í

|     | Variable <sup>a</sup> |       |       |       |       |       |       |       |                       |  |
|-----|-----------------------|-------|-------|-------|-------|-------|-------|-------|-----------------------|--|
|     | Z                     | Т     | QS    | L     | L/Z   | CI    | CO    | R     | <u>TS<sup>b</sup></u> |  |
| Z   | 1.000                 |       |       |       |       |       |       |       |                       |  |
| Т   | .745                  | 1,000 |       |       |       |       |       |       | ľ                     |  |
| QS  | 411                   | 915   | 1.000 |       |       |       |       |       |                       |  |
| L   | 529                   | 788   | .755  | 1.000 |       |       |       |       |                       |  |
| L/Z | 804                   | 875   | .707  | .930  | 1.000 |       |       |       | 1                     |  |
| CI  | 157                   | . 203 | 372   | .327  | . 293 | 1.000 |       |       |                       |  |
| CO  | 512                   | 242   | .020  | .618  | .655  | .847  | 1.000 |       | י                     |  |
| R   | .631                  | .827  | 747   | 598   | 692   | .230  | 289   | 1.000 |                       |  |
| TS  | 412                   | 361   | . 244 | .701  | .670  | .640  | .741  | 226   | 1.000                 |  |

Results of Stepwise Linear Regressions :

| Resul             | ts of Stepwise Linear Regressions : | •              |       | Variables Allowed <sup>C</sup> |
|-------------------|-------------------------------------|----------------|-------|--------------------------------|
| Equat             | ion                                 | R <sup>2</sup> | SEE   | to Enter                       |
| Z                 | = .690 + .360 T281 CI               | .654           | . 293 | T,L,CI,QS                      |
| Z                 | = 1.011 + .331 T                    | .555           | .330  | T,L,QS                         |
| Т                 | = .951 - 1.331 L + 1.016 CI         | .858           | .442  | <b>ℤ</b> ,L,CI,QS              |
| Т                 | ≂-1.196 + 1.027 Z760 L              | .770           | .537  | Z,L,QS                         |
| L                 | = .690640 T + .719 CI               | .868           | . 293 | Z,T,CI,QS                      |
| L                 | □134566 T                           | .620           | .495  | z,t,Qs                         |
| C1                | ≖ -1.024 + .616 T + .907 L          | .666           | . 329 | Z,T,L,QS                       |
| CO                | =371310 T145 L + 1.029 CI           | .901           | .178  | Z,T,L,CI,QS                    |
| 1-R               | = "515210 T                         | .684           | .159  | Z,T,L,CI,QS                    |
| log <sub>10</sub> | [1-R] =401292 T119 L                | .680           | .177  | Z,T,L,CI,QS                    |

a - all variables transformed to log<sub>10</sub>, except R and TS
b - TS = trophic state = 1,2,3, for oligotrophic, mesotrophic, and eutrophic lakes, respectively
c - subject to F≥3.0 and tolerance ≥ .01

.

X

T , and QS , hydraulic residence time alone entered significantly in all of the regressions.

The regressions in Table 2.5-3 serve as means of characterizing the multivariate distribution of the data. Because of the high degree of collinearity it will be difficult to establish specific cause-effect relationships based upon these data. Accordingly, application of any of the analysis below to other lakes should be done with extreme caution. A gauge of the applicability of the analysis to a given lake would be how well it conforms to the multivariate distribution characterized in Tables 2.5-2 and 2.5-3. If a lake's depth, residence time, loading, and average inlet concentration are within the ranges of the distributions in Table 2.5-2 and if they can be predicted reasonably well by the relationships in Table 2.5-4, as gauged by the respective standard errors of estimate, then the lake could be considered a member of the same "population" and the results discussed below could be applied with increased confidence. Applicability is also subject to geographical constraints.

2~90

Dillon's model suggests that the retention coefficient for total phosphorus is an important factor required to estimate the response of a lake's trophic state to a given phosphorus loading. In order to predict such response a priori, i.e., without taking any measurements on the lake or on its outlet, a means of estimating R from known morphometric or hydrologic factors is required. Kirchner and Dillon suggested an empirical formulation which permitted estimation of R from the surface overflow rate, QS (equation (2.4-3)). The next step in the analysis is to examine some more theoretically-based models for prediction of the phosphorus retention coefficient and to compare them with the strictly empirical models in Table 2.5-3 and with Dillon and Kirchner's equation. One primary objective is to estimate the extent to which the sedimentation of phosphorus can be represented as a first-order reaction. As discussed previously, this is one of the principal assumptions made in the application of Dillon's model to predict the ultimate response of a given lake to a reduction or increase in phosphorus loading.

The qualified success of Vollenweider's simple, first-order model for phosphorus removal suggested that it serve as a basis for the present analysis:

$$1 - R = \frac{1}{1 + KT}$$
 (2.5-1)

where,

(

í

```
K = effective first-order phosphorus
    decay coefficient (1/yr)
```

A more general form of this model is proposed in which K is allowed to vary with depth and residence time:

$$K = a [1/T]^{b} Z^{C}$$
 (2.5-2)

The inclusion of these effects can be considered as a test of the simple first-order model, in which K is assumed to be constant. Evaluation of the extent to which K, or, equivalently, the parameters a, b, and c, vary with concentration can provide a basis for evaluating first-order kinetics.

Using the above model, the parameters a, b, and chave been estimated for each of the five subsets of the original 105-lake data base: one for each of the three trophic states; one containing all lakes; and one containing lakes with average inlet concentrations less than 0.1 g/m<sup>3</sup>. The data set was stratified to examine the stability of the parameters and hence the validity of the model across groups. The last group was included to eliminate those lakes with extemely high phosphorus

**(**];

concentrations, which had generally higher standard errors of estimate and for which trophic classification would be obvious in any case. The BMDP3R nonlinear regression program <sup>26</sup> was employed to estimate the parameters.

Results are given in Table 2.5-4. The parameter expressing depth-dependence, c, was not significantly different from zero in any of the groups. Accordingly, the model was simplified by setting c equal to zero and new estimates for parameters a and b were derived. t-Tests indicated no significant differences of the parameter estimates across groups. Thus, the firstorder model appeared to be stable, with optimal parameter estimates independent of concentration. The standard errors of estimate were 0.110, 0.155, and 0.164 for oligotrophic, mesotrophic, and eutrophic lakes, respectively.

Subsequent analyses have been performed using the parameter estimates derived from the last data set in Table 2.5-4, i.e. a = 0.824, b = 0.546, and c = 0. Normality in the residuals is indicated by the probability plot in Figure 2.5-1. Figures 2.5-2 and 2.5-3 plot observed versus estimated retention coefficients and average outlet concentrations, respectively. The symbols 0, M, and E have been used to represent the various trophic states. Residuals patterns generally reflect higher standard errors in the eutrophic groups, but no trend. No trends

## Table 2.5-4

Parameter Estimates for Phosphorus Retention Model

Model:
$$1 - R = \frac{1}{1 + T K}$$
T = hydraulic residence time (yrs) $K = a \begin{bmatrix} \frac{1}{T} \end{bmatrix} Z^C$ T = hydraulic residence time (yrs) $K = a \begin{bmatrix} \frac{1}{T} \end{bmatrix} Z^C$ R = effective first-order sedimentation coefficient  
a,b,c = estimated parameters

- -

| Statistic         | Oligotrophic | Mesotrophic | Eutrophic        | All Lakes   | CI<.1g/m <sup>3</sup> |
|-------------------|--------------|-------------|------------------|-------------|-----------------------|
| N                 | 22           | 23          | 60               | 105         | 71                    |
| a                 | 1.383 ± .785 | .711 ± .301 | .832 ± .249      | .866 ± .188 | .803 ± .227           |
| ь                 | .445 ± .092  | .531 ± .111 | .471 ± .083      | .500 ± .054 | .549 ± .058           |
| c                 | 204 ± .212   | .040 ± .176 | .114 ± .133      | .023 ± .091 | .010 ± .111           |
| SEE               | .113         | .159        | .165             | .150        | .137                  |
| R <sup>2***</sup> | .849         | .704        | .693             | .723        | .727                  |
| (c = 0.)          | .817 ± .105  | .778 ± .118 | $1.050 \pm .124$ | -912 ± .068 | .824 ± .067           |
| Ъ                 | .489 ± .080  | .517 ± .092 | 434 ± .076       | .492 ± .046 | .546 ± .046           |
| SEE               | .110         | .155        | .164             | .150        | .136                  |
| R <sup>2</sup>    | .849         | .704        | .693             | .723        | .727                  |
|                   |              |             |                  |             |                       |

Data Set

÷.,

2-94

.





Ć



Ć



Observed and Estimated Phosphorus Outlet Concentrations

(

were evident in plots of residuals versus other independent variables, including Z, T, QS, L, lake surface area, or drainage area.

Table 2.5-5 compares the standard errors of estimate in predicting the retention coefficient and average outlet concentration with the respective standard errors of the empirical models in Table 2.5-3 and of Dillon and Kirchner's equation. The derived model performs about as well as the empirical linear regressions and generally performs better than the latter. Advantages of the non-linear model over the linear regressions are its partial theoretical basis, relative parsimony (2 parameters), and property that the computed retention coefficient is restricted to the range of 0 to 1. Using only the eleven lakes which were in common with Dillon and Kirchner's data set, the standard errors of the retention coefficient estimates were 0.137 and 0.111, for the non-linear model and Dillon and Kirchner's equation, respectively.

Parameter estimates indicate that the effective first-order decay coefficient in these lakes is roughly inversely proportional to the square root of the mean hydraulic residence time. This dependence is further examined in Table 2.5-6 and Figure 2.5-4. Table 2.5-6 presents analyses of variance on the decay coefficient in the original form, K, and on a form corrected for residence

| Comparisons of Phosphorus Retention Models                                            | U                           | Predicted V        | ariable                     |                     |  |
|---------------------------------------------------------------------------------------|-----------------------------|--------------------|-----------------------------|---------------------|--|
| Mode1                                                                                 | Retention<br>R <sup>2</sup> | Coefficient<br>SEE | Average (<br>R <sup>2</sup> | Outlet Conc.<br>SEE |  |
| Dillon and Kirchmer (equation 2.4-3)<br>R = .426 exp(271 QS) + .574 exp(00949 QS      | .512                        | .197               | .807                        | .245                |  |
| Linear Regressions (Table 2.5-3)<br>1 - R = .515210 T<br>CO =371310 T145 L + 1.029 CI | .684                        | .159<br>-          | .901                        | -<br>.178           |  |
| Nonlinear Regression (Table 2.5-4)<br>1 - R = 1/[1 + .824 T <sup>.454</sup> ]         | .720                        | .151               | .906                        | .171                |  |

۰.

**---**.

4

Table 2.5-5

ι

a - base 10 logarithm

b - 105 lakes included in analysis

.

.

C

Ċ

#### Histograms of Observed and "Corrected" First-Order Sedimontation Coefficients for Total Phosphorus

A - Observed FABULATION OF WARTABLE 38.8 VITH STRATIFICATION ON VARIABLE 9.75 . . ...... -----TRCLUCES τř 0116 **EVID** 10.00 \*\*\*\*\*\* ATTA 18 ATTA 18 ATTA 18 . 
 #10P00H15

 3.3039

 3.4031

 3.4031

 3.4031

 3.4031

 3.4031

 3.4031

 3.4031

 3.4031

 3.4031

 3.4031

 3.5021

 3.5021

 3.5021

 3.5021

 3.5021

 3.5021

 3.5021

 3.5021

 3.5021

 3.5021

 3.5021

 3.5021

 3.5021

 3.5021

 3.5021

 3.5021

 3.5021

 3.5021

 3.5021

 3.5021

 3.5021

 3.5021

 3.5021

 3.501

 3.501

 3.501

 3.501

 3.501

 3.501

 3.501

 3.501

 3.501

 3.501

 3.501

 3.501

 3.501 ----.... TANUATIONS AND CONVERTINGS UNION FOLLOW EXCLUDE VALVES LISTED ANDVE . -1.2031+ 4.5 UP PEARS ALE OINCILE BY A'S JETAIR COINCIDE ALBACES W TO AND AND A TO AND AND A TO A -0.114 0.357 23.300 3.247 -0.928...... 0.114 0.714 \$7.000 **16 AN** . .. Der. • ्र कर ALAINUM ALAINUM 2,014 -1-168 \*\* = ==\*\* ALL GROWS CONSTREE ICASES ENGLASED IF SPECIAL CODES FOR EXAMINA VALIABLES SUR OF SQUARTS OF , FLAX SQUARE & BATED PROF. F EXCERDED. . . . . . --- 0.071 14 AN ... . . NATING Mating -----... -- --+-B.-. Corrected. TABULATION OF VARIABLE SE REFERENCE WERE STRATEFICATION ON VALLAGE .9 15 E BELLIOSEO 0.66 ••••• tuft \*\*\*\*\*\*\*\*\*\*\*\*\*\* V 68 88 ----••• • ... -+ Aug. 10. 10. 10. 10. 1 • \*\*\*\* EJELOPID TALUIS • TABULATEONS AND COMPUTATIONS IMIGH FALLON EXCLUDE VALUES LISTED AND 21410401W 6604.5 6606.0 -----0.1034 0.303 0.4031 0.4031 0.1335 0.4031 0.1335 0.1335 0.1335 0.1335 0.1335 0.1335 0.1335 0.1335 0.1335 0.1335 0.1335 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 0.1355 -..... \*\* \*\* ...... . ..... ... ٠ ٠ .. : . .... • • ••• ••• erone where we dematte by with the late coincide with the "unit vise" -0.655 0.236 21.000 6.732 -0.613 -8.845 8.393 23.308 0.613 -1.138 0.003 0.468 0.000 0.003 0.000 0.013 etan. 1. DIT. N Khiddhud HLHEmun -----ALL CACUPS CONSIMICS SCALEDS EXCLUDED 34 SPECELL CODES FOR REPART VILLAGES ... .... 12249 S- DEV, -----M MAJICIN MENENIA -0-48ZA . .....



### 2.5.2 Phosphorus Retention Models (continued)

time effects, K\_ :

$$K = \frac{R}{1 - R} \frac{1}{T}$$
 (2.5-3)

$$K_c = K T^b$$
,  $b = 0.546$  (2.5-4)

The  $\log_{10}$  transformation has been applied to these variables, and the four lakes in the data set with negative retention coefficients have been excluded. Table 2.5-6 indicates significant stratification of the uncorrected decay coefficient, K, with trophic state, but insignificant stratification of the coefficient after it has been corrected for the effects of hydraulic residence time. The standard deviations of the corrected decay coefficients are 0.224, 0.391, and 0.468 for oligotrophic, mesotrophic, and eutrophic lakes, respectively. This may reflect a corresponding increase in noise levels in the systems or in the data. The effect of T on K is also illustrated in Figure 2.5-4. Again, greater deviations are observed for eutrophic lakes.

The parameter estimates indicate that lake mean depth alone has little influence on phosphorus sedimentation rate, except insofar as depth may influence residence time. The final twoparameter model is similar to that proposed independently by

C :

( :

Larsen and Mercier<sup>45</sup>, whose estimates for a and b were 1.12 and 0.49, respectively, as derived from a sample of 20 oligotrophic and mesotrophic lakes. The apparent dependence of K on T could be explained in part by deviations from the completely-mixed assumption inherent in the first-order model formulation. Equations (1) and (2) can be combined to give:

$$1 - R = \frac{1}{1 + KT} = \frac{1}{1 + aT^{1-b}}$$
(2.5-5)

If the true sedimentation coefficient were independent of T, then the same results would be reached if T were replaced by an effective hydraulic residence time, given by:

$$T_{ef} = T^{1-b}$$
 (2.5-6)

Substitution of  $T_{ef}$  for  $T^{1-b}$  in equation (5) gives the standard first-order equation:

$$1 - R = \frac{1}{1 + a T_{of}}$$
 (2.5-7)

## 2.5.2 Phosphorus Retention Models (continued)

In terms of lake volume, V , and mean flow, Q :

$$T_{ef} = \left[\frac{v}{Q}\right]_{ef} = \left[\frac{v}{Q}\right]^{1-b}$$
 (2.5-6)

Conceivably, thermal stratification could reduce the effective volume of the lake and partially account for the observed dependence of K on T. Vollenweider attempted to account for this on one of his earlier modifications of the first-order model <sup>15</sup>. He suggested that the effects of stratification could be incorporated into the model by employing the following expression for  $T_{ef}$ :

$$T_{ef} = \frac{V}{Q} + \frac{V}{V_{a}} - 1$$
 (2.5-9)

where  $V_e$ , "mean exchange epilimnion", was defined as that portion of the lake's volume taking part in the washout process. He did not propose a general model for  $V_e$ , however.

Temperature effects may also account in part for higher mean sedimentation coefficients observed in lakes with lower residence times. The temperatures of low residence time lakes

**(**)

()

would be expected to respond faster to seasons, because of their higher advective thermal energy exchange rates. The response time would not depend upon lake mean depth, because of the apparent linear increase of vertical transport rates with depth, as noted by Snodgrass<sup>78</sup> and Blanton<sup>7</sup>. This dependence would cancel the effects of higher surface-area-to-volume ratios in shallower lakes. Accordingly, one would expect that the peak, volume-averaged temperature would decrease with increasing hydraulic residence time. Assuming that the reactions involved in phosphorus sedimentation are temperature-dependent, this could partially account for the observed dependence of K on T.

A third explanation is that the reactions for phosphorus deposition may be dependent upon the presence of other substances, for example, other nutrients for biologically mediated reactions, iron for chemically mediated reactions, or sediment, for adsorptionsedimentation reactions. Lakes with higher flushing rates would have greater supplies of these supplementary materials, which may, in turn, increase the net removal rates of phosphorus from the systems.

The above explanations should be qualified with a note that multicolinearity in the data (Table 2.5-3) generally makes it difficult to attribute variations in the dependent variable to specific factors. Theoretical interpretation of the parameter

## 2.5.2 Phosphorus Retention Models (continued)

**{** 

(

estimates is thus rather tenuous. This re-emphasizes the need to insure that a given lake conforms to the correlation structure of these data before applying any of the previous or forthcoming results to that lake.

The stability of the parameter estimates across trophic states has not provided a basis for rejecting the validity of a first-order model in which the retention coefficient is independent of concentration. Multicolinearity and/or errors in the data may have effectively crippled the ability of the analysis to detect significant parameter differences across groups, however. The first-order assumption could best be tested with time series data from a number of lakes. The implications of first-order behavior are particularly important in two respects: in predicting system response to changes in phosphorus loading and in predicting response to changes in hydrologic regime. The former has implications in the allocation of acceptable nutrient loadings. The latter has implications in the interpretation of data obtained from sampling under a given set of hydrologic conditions. This involves the conversion or extrapolation of a lake's nutrient budget, measured during a given year, to long-term-average conditions.

The higher standard errors of estimate characteristic of the eutrophic group may be attributed to a number of factors. From a

( )

theoretical point of view, this could be due to certain characteristics of eutrophic lakes which profoundly affect nutrient dynamics, such as anaerobiosis or zero-order dependence of algal growth rates on phosphorus concentrations. It may also be attributed to higherfrequency or higher-amplitude variations of phosphorus concentrations in eutrophic lakes. Such variations would increase the errors in the retention coefficients estimated from a fixed number of lake outlet samples per year (typically, 12, for the NES lakes). Lower stability is a general characteristic of eutrophic ecosystems. In addition, in this particular data set, the geometric mean hydraulic residence time for the eutrophic group is 0.25 years, as compared to 2.42 and 1.07 years for the oligotrophic and mesotrophic groups, respectively (Table 2.5-2). A lower residence time would cause less attenuation of the relatively high-frequency variations characteristic of lake inflows. This, in turn, would increase the sampling frequency required in order to quantify average lake (or outlet) conditions to within a given standard error.

C

í

The residuals obtained in estimating the retention coefficient according to the above model result from errors in measurement (input/output estimates), as well as from model errors. In order to disect the residual variance into these two components, some assumptions must be made about their forms. The actual and measured values of 1-R are assumed to be given by:

$$Y = 1 - R = \frac{CO}{CI}$$
 (2.5-10)

$$Y_{m} = \frac{CO (1 + e_{o})}{CI (1 + e_{i})}$$
(2.5-11)

where,

The variables  $e_0$  and  $e_1$  represent measurement errors in outlet and inlet concentration estimates. Their standard deviations equal the coefficients of variation of the respective estimates:

$$S_{e_o} = CV_{com}$$
 (2.5-12)

$$s_{e_i} = CV_{ci_m} \qquad (2.5-13)$$

According to the first-order model, the estimated value of

Y is given by:

.

$$Y_e = \frac{1}{1 + K_e^T} = \frac{1}{1 + KT (1 + d_k)}$$
 (2.5-14)

where, ĸ = estimated first-order decay coefficient (l/yr) K = actual first-order decay coefficient (1/yr)

- $d_k = model error variable with mean$ 
  - zero and standard deviation skk

A calculated residual is given by:

$$r_{y} = Y_{m} - Y_{e} = \frac{CO(1 + e_{0})}{CI(1 + e_{1})} - \frac{1}{1 + KT(1 + d_{k})}$$

$$= \frac{(1 + e_{0})}{(1 + e_{1})} \frac{1}{1 + KT} - \frac{1}{1 + KT(1 + d_{k})}$$
(2.5-15)

1

(

## 2.5.3 Error Analysis (continued)

(

Since the error variables are assumed to be independent, the variance of the residuals can be calculated from the following expected value formula<sup>5</sup>:

$$s_{ry}^{2} = \left(\frac{\partial r_{y}}{\partial e_{o}}\right)^{2} s_{e_{o}}^{2} + \left(\frac{\partial r_{y}}{\partial e_{i}}\right)^{2} s_{e_{i}}^{2} + \left(\frac{\partial r_{y}}{\partial d_{k}}\right)^{2} s_{d_{k}}^{2} \qquad (2.5-16)$$
$$= x^{2} \left(s_{e_{o}}^{2} + s_{e_{i}}^{2}\right) + x^{2} \left(1 - x\right)^{2} s_{d_{k}}^{2} \qquad (2.5-17)$$

The first term in this equation essentially represents measurement error, and the second, model error. The corresponding expression for the residual variance obtained in estimating the logarithm of the outlet concentration is given by:

$$s_{r_c}^2 = s_{e_0}^2 + s_{e_i}^2 + (1 - Y)^2 s_{d_k}^2$$
 (2.5-18)

Equation (18) indicates that the measurement error component of the concentration residuals is constant, while the model error component decreases to zero as Y approaches 1, or as T approaches zero. This property is used below to quantify the two components, via further analysis of observed residual

### patterns.

(

For any range of Y values, a value of  $S_{r_c}^2$  can be computed directly from the residuals. Equation (18) can be viewed as a linear model for the residual variance, in which the independent variable is  $(1-Y)^2$ , the slope is measurement error,  $(S_{e_0}^2 + S_{e_1}^2)$ , and the intercept is model error,  $(S_{d_k}^2)$ . In order to estimate the slope and intercept, the lake data have been divided into five groups of approximately equal sample size, based upon residence time (or Y values). For each group, the mean  $(\overline{Y})$ , variance  $(S_Y^2)$ , and mean squared residual have been calculated. In order to account for variation of Y within each group, the expected value of  $S_{r_c}^2$  has been computed:

$$E (S_{r_{c}}^{2}) \approx S_{e_{0}}^{2} + S_{e_{1}}^{2} + (1 - \overline{Y})^{2} S_{d_{k}}^{2} + \frac{1}{2} (\frac{\partial^{2} S_{r_{c}}^{2}}{\partial Y^{2}}) S_{Y}^{2}$$

$$= S_{e_{0}}^{2} + S_{e_{1}}^{2} + [(1 - \overline{Y})^{2} + S_{Y}^{2}] S_{d_{k}}^{2}$$
(2.5-19)

Using the values computed for the various data groups, a linear regression of mean squared residual on the variable  $(1 - \overline{Y})^2 + S_Y^2$  has been done. The results (Figure 2.5-5) indicate a high correlation between these two variables



Outlet Concentration Residual Variance for Various Residence Time Intervals

Figure 2.5-5

(:

ť

:

( :

(r = 0.977). The least squares estimates of the slope and intercept are given by:

slope ≂ S<sup>2</sup>d<sub>k</sub> = 0.299 <u>+</u> 0.038

intercept =  $s_{e_1}^2 + s_{e_0}^2 = 0.0178 \pm 0.010$ 

The slope, or model error, is well-determined relative to the intercept, which is significantly different from zero at between the 80% and 90% confidence levels. Thus, according to this analysis, the residual variance of Y and ln (CO) are given by:

$$s_{r_y}^2 = 0.018 y^2 + 0.30 y^2 (1-y)^2$$
 (2.5-20)

$$s_{r_c}^2 = 0.018 + 0.30 (1 - Y)^2$$
 (2.5-21)

In each equation, the first term represents measurement error, and the second, model error.

Figures 2.5-6 and 2.5-7 illustrate the abilities of these equations to depict residual patterns in the data. The variance



( )

t

( :

( .

components have been multiplied by four, to represent approximate 95% confidence regions. These figures permit further interpretation of the residuals as originating from either of the two sources of error for various values of residence time. In both cases, measurement error dominates at low values of T, while model error dominates at high values. The best estimate of the measurement error variance can be interpreted in terms of the coefficients of variation of the inlet and outlet phosphorus concentration estimates:

$$s_{e_i}^2 + s_{e_o}^2 = 0.018 = cv_{ci_m}^2 + cv_{co_m}^2$$
 (2.5-22)

for  $CV_{cim} = CV_{com} = CV_m$ ,

$$CV_{=} \sqrt{0.018/2} = 0.095 = 0.1$$
 (2.5-23)

2

Thus, the estimated measurement error term corresponds roughly to coefficients of variation on the order of 0.1. Based upon the standard error the intercept, the 95% confidence range for this statistic is from 0 to 0.15.

In applying the model developed above to predict the

# 2.5.3 Error Analysis (continued)

C

(

Ć.

phosphorus retention coefficient for a given lake, three sources of error would be present -- model error, parameter error, and independent variable error:

$$Y = 1 - R = (1 + aT^{1-D})^{-1}$$
 (2.5-24)

$$s_{\rm Y}^2 = s_{\rm e}^2 + s_{\rm p}^2 + s_{\rm i}^2$$
 (2.5-25)

Model error,  $s_e^2$ , is estimated from the above analysis:

$$s_e^2 = 0.30 Y^2 (1-Y)^2$$
 (2.5-26)

The second error term,  $s_p^2$ , is attributed to uncertainty in the parameter estimates:

$$s_p^2 = \left(\frac{\partial Y}{\partial a}\right)^2 s_a^2 + \left(\frac{\partial Y}{\partial b}\right)^2 s_b^2 + 2 \left(\frac{\partial Y}{\partial a}\right) \left(\frac{\partial Y}{\partial b}\right) s_a s_b \rho_{ab}$$
 (2.5-27)

# 2.5.3 Error Analysis (continued)

where,

$$a = 0.824$$
  
 $b = 0.546$   
 $\rho_{ab} = -0.0064$   
 $s_a = 0.067$   
 $s_b = 0.046$ 

Evaluation of the expression gives:

$$s_p^2 = 0.001 \, y^4 \, T^{-908} \, (4.49 + 1.44 \, \overline{\ln T}^2 + 0.032 \, \overline{\ln T}) \, (2.5-28)$$

The third term in equation (25),  $S_{\underline{i}}^2$ , is associated with possible error in the estimate of the independent variable, T :

$$s_{1}^{2} = \left(\frac{\partial x}{\partial T}\right)^{2} s_{T}^{2}$$
  
= 0.140 x<sup>4</sup> T<sup>\*908</sup>  $\left(\frac{s_{T}}{T}\right)^{2}$  (2.5-29)  
= 0.140 x<sup>4</sup> T<sup>\*908</sup>  $\overline{CV}_{T}^{2}$ 

í

ľ

**(** )

These three error components are plotted against T on logarithmic scales in Figure 2.5-8 To evaluate the last term,  $\overline{CV}_{T}$  is assumed to be 0.10, corresponding to an approximate 95% confidence region of  $\pm$  20% in the estimate of T. The plot shows clearly that the model error term dominates over the other terms by about one and two orders of magnitude, respectively, for all residence time values.

Further reduction of the parameter error term would be achieved by increasing the sample size, i.e., by adding additional lakes to the data set. However, if these lakes were drawn from the same general population, the effect on the total projection error would be small, since the parameter error term is already insignificant compared with the model term, which would not be expected to change much with sample size. Thus, for this type of data and model, sample size appears to have been adequate for parameter estimation purposes.

The dashed line in Figure 2.5-8 represents measurement error, estimated according to equation (20):

$$s_m^2 = 0.018 y^2$$
 (2.5-30)

As noted previously, the estimate of this variance component

()

1

Variance Components of Retention Coefficient Prediction



( .

ſ

Ĺ

is not very accurate, with a 95% confidence range of 0 to 0.045  $y^2$ . In addition, it is specific to this data set, so that it would apply only in cases where the nutrient budgets are estimated using data of the same quantity and quality, and techniques similar to those employed in the National Eutrophication Survey. For lakes with residence times less than about 0.15 years, measurement errors dominate over other types. In this region, any errors in the model projection would be difficult to detect with this type of data.

If a projection of the retention coefficient were made for a given lake and subsequently compared with a measured value, model error would only be detectable if the following criterion were satisfied:

$$t = \frac{|Y_m - Y_e|}{S_m} \ge t_{\alpha,n}$$
 (2.5-31)

where,

 $Y_m$  = measured value of 1-R  $Y_e$  = estimated value of 1-R  $S_m$  = standard error of measured value  $t_{\alpha,n}$  = t statistic at significance level  $\alpha$  and n degrees of freedom in  $Y_m$  estimate

## 2.5.3 <u>Brror Analysis</u> (continued)

()

ť

6 :

If this procedure were repeated for a large number of lakes at a given residence time, the first and second moments of t would be given by:

E (t) = 0 (2.5-32)  

$$s_t^2 = 1 + \frac{s_e^2}{s_m^2}$$
 (2.5-33)

where

$$s_e^2 \approx \text{model error variance}$$
  
 $s_m^2 \approx \text{measurement error variance}$ 

The fraction of lakes in the population for which equation (31) would be satisfied would be given by:

$$\beta = 1 - \int_{-t_{\alpha,n}}^{+t_{\alpha,n}} f_t dt$$
 (2.5-34)
where,

( )

- $\beta$  = fraction of lakes satisfying equation (31)
- ft = normal probability density function
   t with mean zero and standard
   deviation S<sub>t</sub>

At a given residence time, the value of  $S_t^2$  can be estimated from equations (33) and (21):

$$s_{t}^{2} = 1 + \frac{s_{e}^{2}}{s_{m}^{2}} = 1 + \frac{(1-Y)^{2} s_{d_{k}}^{2}}{s_{e_{1}}^{2} + s_{e_{0}}^{2}} = 1 + 16.7 (1-Y)^{2}$$
 (2.5-35)

Equations (34) and (35) have been evaluated as a function of hydraulic residence time, for  $t_{\alpha,n} = t_{.05,\infty} \approx 2$ , and for various values of the variance ratio  $S_{dk}^2/(S_{ei}^2 + S_{eo}^2)$  ranging from 5 to 40. The results (Figure 2.5.9) depict the detectability of model errors as a function of residence time and model error to measurement error ratio. The nominal value of 16.7 for the latter represents the best estimate of the curve for data of the type used in estimating the model. At a given residence time, the fraction increases with variance ratio. As T approaches zero, model error vanishes and the





Figure 2.5-9

· ٦

۰.

( .

statistic approaches 0.05, corresponding to the selected significance level of  $t_{\alpha,n}$ . All of the curves approach upper asymptotes as T increases, reflecting an increasing detectability of projection errors. The curve corresponding to the nominal variance ratio has been mapped onto the frequency distribution of residence times in the data set (Table 2.5-2). The result indicates that significant model errors would be detectable in approximately 31% of the lakes, if the data base were similar to that provided by the National Eutrophication Survey. In Figures 2.5-6 and 2.5-7, twenty-four residuals lie outside the estimated measurement error curves. This corresponds to an observed model error detection frequency of 34%.

In applying the model to predict outlet concentration, another error term must be added:

$$co = CIY$$

$$\frac{s_{co}^{2}}{co^{2}} = \frac{s_{ci}^{2}}{cI^{2}} + \frac{s_{Y}^{2}}{Y^{2}}$$

$$cv_{co}^{2} = cv_{cim}^{2} + \frac{1}{Y^{2}} (s_{e}^{2} + s_{p}^{2} + s_{i}^{2}) \approx cv_{cim}^{2} + \frac{1}{Y^{2}} s_{e}^{2}$$

$$= cv_{cim}^{2} + 0.30 (1 - Y)^{2} \qquad (2.5-36)$$

#### 2.5.3 Error Analysis (continued)

(

Because of their relative insignificance, the parameter and residence time error terms have been neglected. The added term,  $CV_{ci_m}^2$ , accounts for uncertainty in the inlet concentration, or phosphorus loading estimate.

Equation (36) has been plotted in Figure 2.5-10 for various assumed values of  $\ensuremath{\operatorname{CV}_{\text{ci}_{\mathfrak{m}}}}$  . The dashed line again represents measurement error, assuming  $CV_{CO_m} = 0.10$ , as estimated above. The relative importance of the various error components can be used as a basis for allocating additional effort to model development or to monitoring in the interest of reducing the variance of the projection. Model error dominates over other various components at high residence times. Accordingly, development and/or application of a more sophisticated model may be justified in this region, providing that its model and parameter errors are lower. As residence time decreases, potential measurement errors dominate, and the prediction becomes increasingly sensitive to errors in the loading or inlet concentration estimate. At lower residence times, this model would be considered adequate and additional effort would be most effectively applied in monitoring (or source modelling) in order to develop a better loading estimate.

In a specific application, the variance of the inlet

C

phosphorus concentration estimate can be calculated according to the techniques discussed in Section 2.3.2, if the estimate is based upon direct measurements. If such are not available, the estimate can be based upon land use data, using the loading factors in Table 2.3-1. The ranges of these loading factors can be used to estimate variance. If a rectangular distribuis assumed, the standard deviation is given by 0.29 times the range<sup>77</sup>. For the ranges and mean values for total phosphorus loading given in Table 2.3-1, a coefficient of variation of about 1 is indicated. Comparison with the other error components in Figure 2.5-10 indicates that the accuracy of an outlet phosphorus concentration projection would be limited most by potential errors in such indirect loading estimates. This type of error could presumably be reduced by considering some of the other factors influencing nutrient export discussed in Section 2.3.1. However, the current state-of-the-art of watershed models is such that the accuracy of the lake model derived above would probably not limit the accuracy of an outlet phosphorus concentration projection. Thus, the model would be considered adequate for use with indirect loading estimates.

A final aspect to consider is the potential errors involved in predicting chlorophyll concentration in lakes based upon predictions of the phosphorus retention model.

9

Dillon and Rigle<sup>14</sup>, have examined the relationship between measured total phosphorus concentration at spring overturn and mean summer chlorophyll-a concentration based upon data from 19 lakes in southern Ontario and from 27 other lakes in North America, as reported in the literature. All of these lakes were phosphorus-limited systems, with total nitrogen to total phosphorus ratios greater than 15 at spring overturn. They fitted the following regression line to their data (Figure 2.5-11) :

| in (Ch1) = | 7.393 + 1.449 ln (CS) |
|------------|-----------------------|
| r =        | 0.95                  |
| SEE =      | 1.01                  |

where,

- Chi = mean summer chlorophyll-a concentration (mg/m<sup>3</sup>)
- CS = total phosphorus concentrationof spring overturn (g/m<sup>3</sup>)
- SEE = standard error of estimate

(2.5 - 37)



Figure 2.5-11

Relationship Between Mean Summer Chlorophyll a and Total Phosphorus Concentration at Spring Overturn According to Dillon and Rigler<sup>14</sup>

#### 2.5.3 Error Analysis (continued)

G

Coupling of this model with the phosphorus retention model would permit estimation of chlorophyll concentrations from loading, hydraulic residence time, and depth estimates.

In order to do this, a means of relating the spring overturn to mean outlet phosphorus concentration would be required. It is assumed that, if the data were compiled, a model would be estimated for this purpose. Independent variables in such a model would probably include mean outlet concentration, inlet concentration, residence time, and possibly some indicator of seasonal flow variations. Alternatively, the chlorophyll equation above could be re-estimated based upon outlet phosphorus concentration. At spring overturn, lakes are completely mixed, so that CO = CS would not be a bad assumption during this period. Since flows are generally highest during this period, the annual flow-weighted-average outlet concentration would be a strong function of the spring concentration, and vice-versa. For the sake of the following arguments, it is assumed that the spring overturn concentration can be equated to the average outlet concentration.

This permits evaluation of potential errors involved in predicting summer chlorophyll from loading, residence time, and depth estimates: C

$$\ln(Ch1) = 7.393 + 1.449 \ln(CO)$$
 (2.5-38)

$$s_{\ln (\text{chl})}^2 = s_i^2 + s_p^2 + s_r^2 \approx \overline{cv}_{\text{chl}}^2$$
 (2.5-39)

The independent variable error in this case is calculated from the total error in predicting the outlet phosphorus concentration (equation (36)):

$$s_{i}^{2} = \left(\frac{\partial \ln (Ch1)}{\partial \ln (C0)}\right)^{2} s_{k(C0)}^{2} \approx 1.449^{2} cv_{c0}^{2} = 2.10 cv_{c0}^{2}$$
 (2.5-40)  
= 2.10  $\left(cv_{cim}^{2} + 0.30 (1 - Y)^{2}\right)$  (2.5-41)

This error results from uncertainty in the inlet concentration estimate and from model error. The parameter error is due to uncertainty in the slope and intercept of the regression line in Figure 2.5-11 :

$$s_{p}^{2} = \overline{SEE}^{2} \left( \frac{1}{n} + \frac{(\ln CS) - \ln (CS)}{n s_{\ln (CS)}^{2}} \right)$$
 (2.5-42)

For values of 
$$\overline{\text{SEE}}^2$$
, n ,  $\overline{\ln(\text{CS})}$  , and  $s^2_{\ln(\text{CS})}$  derived from

С

( )

the Dillon and Rigler article<sup>14</sup>, and assuming that CO  $\simeq$  CS :

$$s_{p}^{2} = 1.02\left(\frac{1}{46} + \frac{\left[\ln\left(C0\right) + 3.950\right]^{2}}{46\left(1.016\right)}\right)$$

$$= 0.022\left(1 + \frac{\left[\ln\left(C0\right) + 3.950\right]^{2}}{1.016}\right)$$
(2.5-43)

.1

The residual error term, also derived from the Dillon and Rigler article is given by:

$$s_r^2 = \overline{SEE}^2 = 1.02$$
 (2.5-44)

The four terms of the total chlorophyll prediction error are plotted versus hydraulic residence time in Figure 2.5-12. The independent variable term,  $S_i^2$ , has been evaluated according to equation (41), assuming  $CV_{Cim} = 0.1$ . The parameter error term has been evaluated for various values of CO.

Figure 2.5-12 shows clearly that the residual error term dominates over phosphorus prediction and parameter error terms for all values of hydraulic residence time. It is unlikely that the negative distribution of the error components would change ſ

e .

## Variance Components of Mean Summer Chlorophyll a Predictions



much if equation were re-estimated in terms of CO instead of CS, i.e. this distribution is probably not very sensitive to the assumption that CS = CO. In addition, the Dillon and Rigler equation was derived from a fairly narrow range of lakes. An analysis of data from the National Eutrophication Survey lakes<sup>87</sup> indicates higher standard error of estimate in the chlorophyllphosphorus relationship than that found by Dillon and Rigler.

These results indicate that the limiting factor in the accuracy of a chlorophyll prediction would probably be in the chlorophyll-lake phosphorus relationship and not in the lake or outlet concentration prediction. Thus, the phosphorus retention model could be considered adequate in this application, and future modelling efforts would best be focused upon the chlorophyll-phosphorus relationship. A chlorophyll prediction would be useful in that it could be related directly to criteria and to such trophic state indicators as productivity, transparency, and hypolimnic dissolved oxygen deficit. The next step in the analysis is to examine the feasibility of predicting lake trophic state from the kinds of data described above. The models of Vollenweider 93,95 and Dillon<sup>16,17</sup> will serve as a primary basis for the analysis. Each of these models is viewed as a set of coordinate axes. Parameter estimates define the locations of the lines delimiting trophic state. The goal of this work is to evaluate both the models and their respective parameter estimates. Accordingly, the ability of each model to predict trophic state has been examined using two sets of parameter values: that originally proposed by the respective authors and an "optimal" set estimated for this data base. Optimal parameter values have been found by posing the classification problem as a linear discriminant analysis and utilizing the BMDP7M discriminant analysis program  $\frac{26}{10}$  to determine the parameter estimates which best separate the trophic states on the coordinate axes defined by each model.

Vollenweider's first and second models are essentially bivariate classification functions employing L and Z, L and QS, respectively, while Dillon's model is a univariate function employing CO. In estimating optimal coefficients for these models, the corresponding variables have been forced into the classification scheme and all other variables have been excluded. Using the stepwise feature of the BMDP7M program, an optimal linear classification model has also been estimated. In the stepwise mode, the program selects the variables which provide the most discrimination power in order of decreasing power, until some specified lower level of significance is reached. Reckhow<sup>68</sup> has employed stepwise discriminant analysis in developing a model for classifying oxic and anoxic lakes.

Homogeneity of variance across groups is one of the principal assumptions of the analysis, when pooled estimates of the covariance matrix of the entered variables are employed<sup>77</sup>. In order to eliminate the apparent skewness in the distribution of the log-transformed average outlet concentration data for the eutrophic group (Table 2.5-2) and to promote homogeneity of variance across groups, lakes with average outlet concentratione greater than 0.2 g/m<sup>3</sup> have been excluded from the discriminant analysis. These lakes are highly eutrophic and would not be misclassified by any of the models. The analysis have been performed on the remaining 22 oligotrophic, 23 mesotrophic, and 50 eutrophic lakes.

Table 2.5-7 summarizes the group means and standard deviations for each of the variables in the analysis. The definitions and transformations of these variables have been previously defined in Table 2.5-2. The variable "COEST" is the average outlet concentration, as estimated using the model for phosphorus retention derived previously:

## • Table 2.5-7

· }

٩,

:

;

;

Summary of Distributions and Initial Discriminating Powers of Variables in Discriminant Analysis

| MEANS                                                                                                                                              | *                                                                                             |                                                                                              |                                                                                           |                                                                                      |
|----------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------|
| GROUP -                                                                                                                                            | OLIC                                                                                          | NESO                                                                                         | EUTA                                                                                      | ALL OPS.                                                                             |
| 37                                                                                                                                                 | 1.25845                                                                                       | 1.04731                                                                                      | 0.01312                                                                                   | 0,97334                                                                              |
| 4 T                                                                                                                                                | 0.38777                                                                                       | 0.02902                                                                                      | -0.4R016                                                                                  | -0.26652                                                                             |
| 5 L                                                                                                                                                | -0.83315                                                                                      | -0+44404                                                                                     | 0.46511                                                                                   | -0.04512                                                                             |
| _6_02                                                                                                                                              | -2.12684                                                                                      | -1.81557                                                                                     | -1.31076                                                                                  | -1.62197                                                                             |
| 7 CI                                                                                                                                               | -1.70923                                                                                      | -1.46233                                                                                     | -1.01017                                                                                  | -1.28468                                                                             |
| a chest                                                                                                                                            | -2.11365                                                                                      | +1.77567                                                                                     | -1.21675                                                                                  | -1.55977                                                                             |
| ~~ 9 q\$`` <b></b>                                                                                                                                 | 0,87600                                                                                       | " 1.01829 T                                                                                  | 1.50128                                                                                   | 1.23956                                                                              |
| 10 TS                                                                                                                                              | 1.00000                                                                                       | 2.00000                                                                                      | 3.00000                                                                                   | 2.29474                                                                              |
| COUNTS                                                                                                                                             | 22.                                                                                           | 23.                                                                                          | 50.                                                                                       | 95.                                                                                  |
|                                                                                                                                                    |                                                                                               |                                                                                              |                                                                                           |                                                                                      |
| STANDARD DEV                                                                                                                                       | TATIONS                                                                                       |                                                                                              |                                                                                           |                                                                                      |
| GROUP .                                                                                                                                            | OLIG                                                                                          | HESO                                                                                         | SUTA                                                                                      | ALL GPS.                                                                             |
| GROUP -                                                                                                                                            | OLIG                                                                                          | HESO                                                                                         | SUTA                                                                                      | ALL GPS.                                                                             |
| GROUP =<br>VARIABLE<br>3 2                                                                                                                         | 014110AS                                                                                      | 0+47320                                                                                      | EUTA                                                                                      | ALL GPS.                                                                             |
| GROUP -                                                                                                                                            | OLIG                                                                                          | 0.47320<br>0.94547                                                                           | EUTA<br>C. 44015<br>1.0P116                                                               | ALL GPS.                                                                             |
| <u>GQOUP =</u><br>GQOUP =<br>VARIABLE<br>3.2<br>4.7<br>.5.L                                                                                        | OLIG<br>0.44866<br>1.07009<br>0.48175                                                         | HFS0<br>0.47320<br>0.94547<br>0.96076                                                        | EUTR<br>C. 44015<br>1.09116<br>0.62004                                                    | ALL CPS.<br>0.45019<br>1.04772<br>0.57770                                            |
| STANDARD DEN<br>GQUIP =<br>VARIABLE<br>-3.2<br>-5.L<br>-5.L<br>-5.C                                                                                | 0.44866<br>0.46866<br>1.07009<br>0.48175<br>0.15507                                           | <u>MFSO</u><br>0.47320<br>0.94547<br>0.56076<br>0.26253                                      | <u>EUTR</u><br><u>C. 44015</u><br>1.0P116<br>0.62034<br>0.26423                           | ALL GPS.<br>0.45019<br>1.04772<br>0.57730<br>0.24322                                 |
| STANDARD         DEV           GQOUP         •           VARIABLE         3.2           4.7         •          5         C.0          7         CI | OLIG<br>OLIG<br>0.44866<br>1.07009<br>0.48175<br>0.15507<br>0.31525                           | HESO<br>0.47320<br>0.94547<br>0.56076<br>0.26253<br>0.36882                                  | <u>SUTR</u><br><u>C. 44015</u><br><u>1.00116</u><br><u>0.62034</u><br>0.26423<br>C. 33414 | ALL GPS,<br>0.45019<br>1.04772<br>0.57730<br>0.24322<br>0.33854                      |
| STANDARD DEV           GROUP           VARIABLE           4           5           6           7           7           8           2000             | OLIG<br>0.44866<br>1.07009<br>0.48175<br>0.15507<br>0.31525<br>6.17629                        | MESO<br>0.47320<br>0.94547<br>0.56076<br>0.26253<br>0.36882<br>0.26098                       | EUTA<br>                                                                                  | ALL GPS.<br>0.45019<br>1.04772<br>0.57730<br>0.24322<br>0.33854<br>0.24131           |
| STANDARD DEV           GQOUP           VARIABLE           3 2           - 5 L           - 5 L           - 7 CI           - 8 COEST           9 QS  | 0.44866<br>0.44866<br>1.07009<br>0.48175<br>0.15507<br>0.31525<br>0.1529<br>0.1529<br>0.75936 | MFS0<br>0.47320<br>0.94547<br>0.56076<br>0.26253<br>0.26253<br>0.24098<br>0.24098<br>0.24098 | EUTR<br>0.44015<br>1.08116<br>0.62004<br>0.26423<br>0.33414<br>0.26531<br>0.77709         | ALL GPS.<br>0.45019<br>1.04772<br>0.57770<br>0.24322<br>0.3354<br>0.24131<br>0.75777 |

| ***********************                                                                                                  | *******             |            | ******                    | **************                        |     |
|--------------------------------------------------------------------------------------------------------------------------|---------------------|------------|---------------------------|---------------------------------------|-----|
| STEP NUMBER O                                                                                                            |                     |            |                           | · · · · · · · · · · · · · · · · · · · |     |
|                                                                                                                          | - <b>*</b>          | VARTABLE   | F TO<br>Enter<br>F= 2 52  | FORCE TOLEASING                       | · . |
| · · · · · · · · · · · · · · · · · · ·                                                                                    |                     | 3 2<br>4 T | 7.901                     | 1.00000                               |     |
|                                                                                                                          | <u>.</u>            | 6 CO       | . 95.614<br>              |                                       |     |
| ۵۵ ه و سي هادامه اليه متينان المار <u>من و</u> رو و واد الموجود و المراكبين <u>و من و </u> | •••• <b>*</b> ••••• |            | 117.140<br>6. <b>49</b> 4 | 1 1.000000                            |     |

1

$$COEST = (1 - RE) CI$$
 (2.5-45)

$$1 - RE = \frac{1}{1 + 0.824 \text{ m}^{454}} \qquad (2.5-46)$$

The base 10 logarithmic transformation has been applied to COEST.

The second part of Table 2.5-7 summarizes the initial discrimination power of the variables. The F statistics represent univariate analysis of variance statistics, analogous to those found in Table 2.5-2. Table 2.5-7 indicates that COEST had the highest discrimination power, followed by CO, L, CI, T, Z, and QS. The F values were all significant at the 95% level or greater. As discussed previously, however, this does not indicate that there were seven independent and significant factors, because of intercorrelations among these variables (Table 2.5-3).

Tables 2.5-8 to 2.5-12 present the results of the discriminant analyses. In each case, the table contains the significance levels of the entered and remaining variables, summary statistics characterizing the degree of group separation, classification functions, classification matrix, and a jackknifed classification []

matrix. Table 2.5-13 summarizes and compares the statistics obtained for the various models.

To estimate optimal parameters for Vollenweider's first model, Z and L have been forced into the classification scheme (Table 2.5-8). The low F level for Z (0.28) indicates that it did not add significantly to the classification power in L alone (F = 33.6). Appreciable power remained, however, in the excluded variables, which had F levels ranging from 33 to 48, after both L and Z had entered. Thus, this model apparently did not make use of all of the available discriminating power in the data. Wilk's lambda and the overall F are multivariate analysis of variance statistics<sup>12</sup>, the interpretation of which will be discussed later. An F matrix characterizes the separation of each of the three possible pairs of groups and indicates that the distinction between oligotrophic lakes and eutrophic lakes was relatively high (F = 39.6), while that between mesotrophic and eutrophic lakes was fairly low (F = 2.8).

The classification functions correspond to the parameter estimates. Each function consists of a constant plus a coefficient for each variable. In classifying a given observation, a function value is computed for each group, and the observation is estimated to have the highest probability of belonging to the group with the highest corresponding function value. In order to derive (

the lines delimiting the mesotrophic from the eutrophic states on the L versus Z axes, for example, the corresponding classification functions would be subtracted from one another and solved for L as a function of Z.

The classification matrix summarizes the ability of the model to classify the lakes correctly. Observed states correspond to rows, and estimated states, to columns. If the model worked perfectly, all of the lakes would be on the upper left to lower right diagonal. In Table 2.5-8, 72.7%, 52.2%, and 78.0% of the oligotrophic, mesotrophic, and eutrophic lakes, respectively, were classified correctly. The "jackknifed" classification matrix provides a somewhat more robust estimate of the probability of correctly classifying a lake. To derive this matrix, each case (lake) was excluded from the analysis one lake at a time, the classification functions were re-calculated, and the excluded lake was classified based upon the derived functions. Thus, the overall percent correct for the jackknifed classifications, 66.3%, is a measure of the probability of correctly classifying a new lake which has not been used to estimate the parameters of the discriminant model.

Vollenweider's second model (Table 2.5-9) compared favorably with his first. Both L and QS entered significantly, although some discrimination power still remained in the excluded variables. Table 2.5-8

# Table 2.5-9

٠.

Discriminant Analysis - Vollenweider's First Model Discriminant Analysis - Vollenweider's Second Model

٠

| \$tep munita '2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | STEP WINNER Z<br>VANTABLE ENTERED. 9 QS                                                                                          |
|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------|
| VANIANL®         6 TO PONCE         0 YANIANL®         F TO PONCE         TO FONCE                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | VARIABLY F TO FORCE - VAPIABLE F TO FORCE TH PALACE                                                                              |
| 5 C 04.200 0 0.009707<br>5 C 33.595 C 5 60 7.51510 0 0.009777<br>5 C 0.000777<br>5 C 0.009777<br>5 C 0.0097777<br>5 C 0.009777<br>5 C 0.009777<br>5 C 0.009777<br>5 C 0.0097777<br>5 C 0.0097777<br>5 C 0.0097777<br>5 C 0.0097777<br>5 C 0.0097777<br>5 C 0.009777777<br>5 C 0.0097777<br>5 C 0.00977777<br>5 C 0.009777<br>5 C 0.0097777<br>5 C 0.0097777<br>5 C |                                                                                                                                  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 8 COFST 7-205 0 .0-211177                                                                                                        |
| U-STATISTIC OR WILES' LAMOOA 0,4909303 DEGREES OF FAFEDON 2 2 92<br>APRORIMATE F-STATISTIC 19,458 DEGREES OF FREEDON 4,000 102,00                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | U-STATUSTIC ON WILKS' LANODA 0,2721270 DEGATES DE FARFODA 2 2 92<br>APRIDIMATE E-STATISTIC 41.722 DEGATES DE RAREDA 4.00, 182.00 |
| F + HATAIR DEGREES OF FREEDOR + 2 92                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | F - NATALX OUGHERS OF FREEDON + 2 91                                                                                             |
| alto0LTGESO                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | DLIG MESO                                                                                                                        |
| FUTP \$9.55 20-29                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | EUTR 105.00 49.45                                                                                                                |
| CLASSLESCATION FUNCTIONS                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                  |
| GRAUP - OLIG NESO FUTR                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | GROUP & OLIG MESO FUTR                                                                                                           |
| 5 L -0.15362 0.25645 3.11757                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | • Q5 18.19369 15.06960 9.00136                                                                                                   |
| ECHISTANT -5.08875 -3.81347 -6.10934                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | CONSTENT -14.13276 -13.04422 -5.43044                                                                                            |
| LLASSIFICATION MATRIX                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | CLASSIFICATION NETTIX                                                                                                            |
| Ganup PERCENT NUMBER OF CASES CLASSIFIED INTO GOTUP -<br>CORNECT                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | CADUP PERCENT NUMPER OF CASES CLASSIFIED INTO GAOUP -<br>CARLECT                                                                 |
| 02.16 HESO EVITH                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | 01.16 WFSO EUTH<br>01.18 #540 EUTH                                                                                               |
| 9150 5242 6 12 3<br>FUTR 76.0 1 10 39                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | FUL:                                                                                                                             |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | T07AL78.92545                                                                                                                    |
| SACKENITED CLASSIFICATION                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | JACKENJPED, CLASSEFJCATION                                                                                                       |
| CORRECT CONTRACTOR OF CASES CLASSIFIED INTO CROUP -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | GAOLO PERCENT MIRAER OF CASES CLASSIFIED INTO GROUP ~                                                                            |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 0116 1850 Eura                                                                                                                   |
| eura 7640 1 tt 38                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                  |
| 10T4L 44.3 27 24 44                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | .TOTAL 74.8 25 25 45                                                                                                             |

2-140

. .

 $C_{\lambda}$ 

The F statistics and Wilk's lambda indicate that the trophic states were separated considerably better on the L versus QS axes, as compared with the L versus Z axes. Overall, the model classified 78.9% of the lakes correctly, an increase of 8.4% over the first model. For the oligotrophic group, the percent correct increased from 72.7% to 86.4%, corresponding to an approximate halving in the percent misclassifications.

Table 2.5-10 indicates that Dillon's model generally performed somewhat less well than Vollenweider's second model, but better than his first. Significant discrimination power remained in COEST (F = 9.05) and in L (F = 6.25), after CO had entered. The mesotrophic lakes were classified poorly relative to either of the two previous models. The overall percent correct was 73.7%. As in the case of Vollenweider's second model, but not of his first, no eutrophic lakes were misclassified as oligotrophic or vice-versa.

In the etepwise analysis (Table 2.5-11), both COEST and L entered at F levels of 49.0 and 8.4, respectively, leaving no significant discrimination power in the remaining variables. Thus, these two variables alone constitute an optimal linear classification model for these lakes and data. As expected, this model performed better than any of those examined previously. 84.2% of the lakes were classified correctly, with most of the

## Table 2,5-10

٠

## Table 2.5-11

Discriminant Analysis - Dillon's Model

Discriminant Analysis - Stepwise

۰. .

| STEF NUMBER L<br>VARTABLE FINTERED & CO                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | STEP WUMER 2<br>VARIALE ENTERED S L                                                                                               |
|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------|
| VARTARLE F TO FORCE + VANTARLE # TO FORCE TOLERANCE                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                   |
| 05- 3 92 •                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | REMOVE LEVEL . ENTER LEVEL.                                                                                                       |
| 6 C2 95.614 1 = 5 2 0.018 0 0.936412 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | - 5 L 8-413 1 4 5 2 0.249 1 0.346772<br>a corst 49.474 1 4 4 7 0.371 1 0.34744                                                    |
| • 5 L •                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | • 4 CO 0.371 L 0.412862 .<br>• 7 CT 0.438 L 0.192687                                                                              |
| B COFST \$2,045 0 2,444116     • • • • • • • • • • • • • • • • •                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | • • QS 0.638 1 0.039461                                                                                                           |
| U-STATISTIC ON WILES' LANDOA 0.3240276 DEGUESS OF FREEDOM 1 2 42                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | U-STATISTIC DE VILKS" LANDA G.ZS78966 DEGATES DE SEFEDON Z 2 92<br>APPROXIMATE F-STATISTIC 67.787. DEGETES DE BEFEDON 6.00 182.00 |
| all and the second | P. + MATRIE DECREES OF PREEDOM + 2 93                                                                                             |
| P = MATHER DEGREES OF PREECON + 1 92                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                   |
| 14.10 MESA<br>MESAD 18.42                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 4837 12.18<br>EUTO 128.16 55.87                                                                                                   |
| futa 192.00 67.46                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | CLASSIFICATION FUNCTIONS                                                                                                          |
| CLASSIFICATION WHETIONS                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                                                                   |
| vatante                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | Valiante                                                                                                                          |
| • cn -33. +53.40 -30.6+153 -22.151+1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | # CDEST -34.01045 -30.51033 -21.70244                                                                                             |
| ÊGIGTANT -\$9.33212 -28.95992 -15.82055                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | CONSTANT -37.33014 -28.09104 +14.95984                                                                                            |
| CLASSIFICATION WANTER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | CLASSIFICATION NATRIX                                                                                                             |
| GOUD SEACENT NUMBER OF CASES CLASSIFIED INTO GROUP -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | GROUP PEACENS NUMBER OF CASES CLASSIFIED INTO GROUP -                                                                             |
| 0(16 NESO FUTP                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                                                                                   |
| 4553 43.5 8 10 5<br>FUTR #2.0 0 9 41                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | ₩37 73.0 0 17 2<br>FUI9 50.0 7 43                                                                                                 |
| TOTAL 73.7 27 44                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | TOTAL 01-2 24 45                                                                                                                  |
| JACKRUJFED CLASSI PICATION                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                   |
| GROUP PEPCTUT NUMBER OF CASES CLASSIFIED INTO GROW -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                   |
| CONSIGN OLIG MESO FUTH                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                   |
| 0110 04.4 10 5 0<br>43.5 0 10 5                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                   |
| EUTA \$2.0 0 \$ 41                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | EUTA \$1.0 0 7 43                                                                                                                 |
| F714L 73.7 21 22 46                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | TUTAL 8442 24 26 45                                                                                                               |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                   |

•...

•

2-142

1

X

(

improvement in the mesotrophic group.

As a final step in the discriminant analysis, the BMDP7M program performs a principal component analysis of the classification functions. The principal components are linear functions of the entered variables which are independent of each other. The components are determined so as to maximize the variance (or discrimination power) of the first component, or canonical variable. Canonical correlation coefficients computed for all principal components characterize their relative discriminating powers.

The stepwise analysis (Table 2.5-11) indicated that the rate of phosphorus input, L, in addition to the estimated mean outlet concentration, COEST, were the best variables to use in a linear model to classify the lakes. These two variables are obviously correlated, since the latter was calculated from:

COEST = 
$$(1 - RE) CI$$
 =  $(1 - RE) \frac{LT}{2}$  (2.5-47)

The two principal components of these variables had canonical correlation coefficients of 0.87 and 0.07, respectively. This indicated that essentially all of the classification ability was located in the first canonical variable. This result

( :

effectively reduced the bivariate classification model to a univariate one. The analysis suggested that the lakes could best be classified along the axis of the following canonical variable:

$$\log x = 0.1853 \log L + 0.8147 \log COEST$$
 (2.5-48)

$$= 0.1853 \log L + 0.8147 \log [L(1-RE)/QS] \qquad (2.5-49)$$

 $= \log L - 0.8147 \log \left[ \frac{QS}{(1-RE)} \right]$  (2.5-50)

The BMDP7D program<sup>26</sup> was again employed to examine the stratification of the trophic states along this axis. The analysis of variance indicated that the logarithm of X had more discrimination power (F = 119) than any of the variables previously examined (Table 2.5-2). However, the standard deviations of  $\log_{10} X$  in the three groups were 0.16, 0.23, and 0.37, respectively. Bartlett's test<sup>77</sup> indicated that the hypothesis of homogeneity of variance could be rejected at the 99.5% level. In order to stabilize the variance across groups, the following transformation was employed:

$$xT = -1/x^{-25}$$
 (2.5-51)

$$X = L \left[ \frac{QS}{(1-RE)} \right]^{-.8147}$$
(2.5-52)

( ]

The transformed variable had standard deviations of 0.28, 0.34, and 0.33, respectively. Bartlett's test indicated that the hypothesis of homogeneity of variance across groups could not be rejected at the 75% level. When the 10 lakes with outlet concentrations greater than 0.2  $g/m^3$  were excluded from the analysis, the transformation in equation (51) was still found to be necessary to stabilize the variance across groups.

The discriminant analysis program was re-run, allowing XT alone to enter. Table 2.5-12 shows that no significant discrimination power remained in the excluded variables. 86.3% of the lakes were classified correctly along this axis. The analysis of variance in the second part of Table 2.5-12 illustrates the distribution of this variable across groups, using the entire data set. The overall F level of 162.3 can be compared with the levels for the original variables, which ranged from 3.3 to 66.2 (Table 2.5-2). It seemed apparent that this canonical variable incorporated essentially all of the linear discriminating power of the variables in the analysis.

Table 2.5-13 summarizes and compares the statistics obtained for the various discriminant runs. An important statistic characterizing the overall separation of the groups is Wilk's lambda, the multivariate analysis of variance statistic, which has a possible range of zero to one<sup>12</sup>. The analog of lambds in

#### Table 2.5-12

Discriminant Analysis - Canonical Variable Derived from Stonuice Analysish

| arehaise anatypis.                                              |         |
|-----------------------------------------------------------------|---------|
| SIFF AUNTER IN XT                                               |         |
|                                                                 |         |
| YATTOLE PARTONE AND TANK A THE PARTONE                          |         |
| 6fe 2 92                                                        |         |
|                                                                 |         |
| • 3 6 0.8 19 0 4.643 bar                                        |         |
|                                                                 | •       |
|                                                                 |         |
| and a second a second a                                         |         |
|                                                                 |         |
| U-STATESTIC D' attas Lineba d.2180336 OrGiers OF Paterde L 2 92 | • - • • |
|                                                                 |         |
| - P MATRAN DEGATES OF FEFTORS + 1 42                            |         |
| L                                                               |         |
| WEG3   97.13<br>  Pure 201.36 (02.6)                            |         |
|                                                                 |         |
|                                                                 |         |
|                                                                 |         |
| 14 C1                                                           |         |
|                                                                 |         |
|                                                                 |         |
| CLASSIF_CAFINA WITAIN                                           |         |
| . STOUR                                                         | •       |
| COMARGET OLTA MENO FUTA                                         |         |
| alia to.t 20 2 0                                                |         |
| CUTR 00.0 0 3 45                                                |         |
|                                                                 |         |
|                                                                 |         |
| aucusserer coussererien                                         |         |
| CROJE                                                           |         |
|                                                                 |         |
|                                                                 |         |
| PESO 43.2 5 15 3                                                |         |
|                                                                 |         |
| YV(4)YJ+E,E,E,                                                  |         |
|                                                                 |         |

.



Ċ

()

X = L [ QS/(1-RE)]\*.815

| Model                              | Variables | Wilk's<br>Lambda, A | Degrees<br>of<br>Freedom | 8 V - T. | Overal1<br>F | Degrees.<br>of<br>Freedom | W/0    | F Matrix | 0/E   | Degrees<br>of<br>Freedom | Total<br>Percent<br>Misclassified | Average<br>Percent<br>Misclussified |
|------------------------------------|-----------|---------------------|--------------------------|----------|--------------|---------------------------|--------|----------|-------|--------------------------|-----------------------------------|-------------------------------------|
|                                    |           |                     |                          |          |              |                           |        |          |       |                          |                                   |                                     |
| Vollenweider I                     | L,Z       | 0.491               | 2,2,92                   | 0.509    | 19.4         | 4,182                     | 2.8    | 20.3     | 39.6  | 2,91                     | 29.5                              | 32.4                                |
| Vollenweider II                    | l'óz      | 0.272               | 2,2,92                   | 0.728    | 41.7         | 4,182                     | 8.1    | 49.7     | 105.0 | 2,91                     | 21.1                              | 22.9                                |
| Dillon                             | 60        | 0.325               | 1,2,92                   | 0.675    | 95.6         | 2,92                      | 18,4   | 67.9     | 172.0 | 1,92                     | 26.3                              | 29.4                                |
| Stepwise Model                     | l,coest   | 0.238               | 2,2,92                   | 0.762    | 47.8         | 4,182                     | 12.2   | 54.4     | 128.2 | 2,91                     | 15.8                              | -16.4                               |
| Stepwise, First<br>Canonical Varia | x x       | 0.225               | 1,2,92                   | 0.775    | 158.4        | 2,92                      | 37.3 . | . 102.1  | 291.2 | 1,92.                    | 14.7                              | 15.1                                |

# Table 2.5-13 Summary of Statistics Characterizing Discriminant Analyses

<sup>a</sup>Analog of R<sup>2</sup> in a multiple regression analysis. <sup>b</sup>X = L[QS/(1-RE)]<sup>\*,815</sup>

#### 2.5.4 Trophic State Prediction (continued)

(

ſ

6 .

the univariate case is the within-group sum of squares over the total sum of squares. The smaller the value of lambda, the greater the separation of the groups. One minus lambda, also listed in Table 2.5-13 is an analog of  $R^2$  in a multiple regression analysis, in the sense that it is the multivariate version of the explained sum of squares over the total sum of squares. The "Total Percent Misclassified" has been computed from the total number of misclassifications and the total number of lakes, while, the "Average Percent Misclassified" represents the average of the percents misclassified for each group. The latter effectively normalizes the statistic for differences in the numbers of samples across groups. The statistics in Table 2.5-13 provide a means of ranking the various models. They generally indicate that the L versus QS axes (Vollenweider's second model) provide the most discrimination power of the three established models, while the functions derived from the stepwise analysis are superior overall.

Using the entire sample of lakes, the functions derived above and Vollenweider's and Dillon's models, with original and optimal parameters, are compared in Table 2.5-14 and displayed in Figures 2.5-13 through 2.5-18. For each model and parameter set, Table 2.5-14 gives the loading equation used to classify the groups, the classification matrix, and percentage misclassifications by group and overall. Two forms of Dillon's model have

## Table 2,5-14

Comparisons of Model Performance in Classifying 105 Lakes

| Logoing<br>Model Equations                  | Observed<br>Trophic<br>State | Eg<br>Tr<br>O | timat<br>ophic<br><u>N</u> | ted<br>2 State<br><u>E</u> | Percent    | Totel<br>Percent<br>Error | Average<br>Percent<br>Error |
|---------------------------------------------|------------------------------|---------------|----------------------------|----------------------------|------------|---------------------------|-----------------------------|
| Volleveider I <sup>B</sup>                  |                              | 12            | E                          |                            | AC C       | 75.7                      | 59.0                        |
| Low = 0.025 z0.602                          |                              | 44            | 9<br>4                     | 18                         | 69.6       | 1.5.7                     |                             |
| L = 0.050 z <sup>0.602</sup>                |                              |               | - ;                        | 13<br>80                   | 1.7        |                           | 1                           |
| MP - CICCO D                                | 1 -                          | •             | •                          |                            | <b>.</b> . |                           |                             |
| Vollenweider I <sup>b</sup>                 | 0                            | 16            | 4                          | 2.                         | 27.3       | 28,6                      | 32.3                        |
| $L_{0H} = 0.054 \ z^{0.548}$                | н                            | 8             | 12                         | <b>S</b> .                 | 47.8       | 1                         |                             |
| L <sub>MB</sub> = 1.268 Z <sup>-0.089</sup> | B                            | 1             | 12                         | 47                         | 21.7       |                           | Ì                           |
| Vollonweider II <sup>a</sup>                | 0                            | 21            | 1                          | 0                          | 4.5        | 16.2                      | 21.1                        |
| LON = 0.100 QS <sup>0.500</sup>             | м                            | 8             | 11                         | 4                          | 52.2       |                           | <b>.</b>                    |
| LHE = 0.200 QS <sup>0.500</sup>             | [ 2                          | 0             | 4                          | 57                         | 6.7        |                           |                             |
| Vallewaiden tib                             |                              | 10            |                            | •                          |            | 10.0                      | 22.0                        |
| Law = 0.057.050,639                         | <b>1</b>                     | 19            | 3<br>14                    | τ                          | 30 1       |                           | · · · ·                     |
| 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1       |                              | 0             | 14<br>A                    | 52                         | 13.3       | 1                         | }                           |
|                                             |                              | - <b>-</b>    |                            |                            |            |                           |                             |
| Dillon (R Observed) <sup>a</sup>            | 0                            | 18            | 4                          | 0                          | 18.2       | 18.9                      | 27.3                        |
| L <sub>OH</sub> = 0.010 QS/(1-R)            | н                            | 7             | 9                          | 7                          | 60,3       |                           |                             |
| L <sub>ME</sub> = 0.020 QS/(1-R)            | в                            | . 0           | 2                          | 58                         | 3,3.       |                           |                             |
| Dillon (R Observed) <sup>b</sup>            | 0                            | 19            | 3                          | 0                          | 13.6       | 23.8                      | 28.3                        |
| $L_{OM} = 0.0107 \text{ QS}/(1-R)$          | н                            | 8             | 10                         | 5                          | 56.5       |                           |                             |
| LME = 0.0274 QS/(1-R)                       | B                            | 0             | 9                          | 51 ·                       | .15.0      |                           |                             |
| Dillon (R Estimated) <sup>®</sup>           |                              | 20            | 2                          | 0                          | 9.1        | 19.0                      | 26.4                        |
| $L_{mi} = 0.010 \text{ OS}/(1-\text{RD})$   | Г.                           | 8             | 8                          | 7                          | 65.2       |                           |                             |
| Lang = 0.020 Q5/(1-RD)                      | B                            | 0             | 3                          | 57                         | 5.0        |                           |                             |
|                                             | · .                          |               |                            |                            |            |                           |                             |
| Dillon (R Estimated)                        | 0                            | 22            | 0                          | 0                          | 0.0        | 17.1                      | 20.7                        |
| LOM = 0.0107 QS/(1-RD)                      | м                            | 9             | 11                         | 3                          | 52.2       |                           |                             |
| $L_{\rm ME} = 0.0274  \rm{QS}/(1-RD)$       | В                            | 0             | 6                          | 54                         | 10.0       |                           |                             |
| Stepwise Discriminant Analysis              | 0                            | 20            | 2                          | 0                          | 9.1        | 14.3                      | 15.6                        |
| Low = 0.0170 [OS/(1-RE)]0.866               | Ň                            | 4             | 17                         | 2                          | 26.1       |                           |                             |
| Lun = 0.0655 [QS/(1-RE)]0.794               | E                            | 0             | 7                          | 53 <sup>·</sup>            | 11.7       |                           |                             |
|                                             |                              |               | _                          |                            |            |                           |                             |
| Stepwise Discriminant Analysis              | 0                            | 20            | 2                          | 0                          | 9,1        | 12.4                      | 14.5                        |
| LOM = 0.0190 [QS/(1-RE)]                    | м                            | 4             | 17                         | 2                          | 26.1       | l                         |                             |
| $L_{ME} = 0.0549 [QS/(1-RE)]^{0.015}$       | E                            | 0             | 5                          | 55                         | 8.3        |                           |                             |

<sup>a</sup>Original paramotors.

C

(

()

bOptimal parameters.

٩,

CFirst canonical variable.



C

Ċ

Lake Classification According to Vollenweider's First Model







**(**),

()



4

Lake Classification According to Dillon's Model with Estimated Retention Coefficients

**(**:

been evaluated: one employing the observed retention coefficient and one employing the retention coefficient estimated according to Dillon and Kirchner's equation for R as a function of QS(equation (2.4-3)).

Comparing the percents misclassified by group shows that the major effect of estimating optimal parameters for each model has been to reduce the errors in the mesotrophic group and to distribute misclassifications more evenly among the three trophic states. Generally, little effect on the total or average percents correct is observed as a result of estimating optimal parameters for each model, except in the case of Vollenweider's first model, for which the shift in line distinguishing eutrophic from mesotrophic lakes was relatively dramatic (Figure 2.5-13). Optimization of the parameters for the established models generally had more effect on the "dangerous" than on the "permissible" loading lines. The effect has been to raise the former. For instance, the mean outlet concentration separating mesotrophic from eutrophic lakes according to Dillon's model was raised from 0.020  $g/m^3$  to 0.027  $g/m^3$ . The latter figure is closer to the value of 0.025  $g/m^3$ recommended by the Environmental Protection Agency as a water quality criterion "to maintain conditions free of nuisance algal blooms"

The average percents misclassified ranged from 14.5% for

C

the model based upon the first principal component derived from the stepwise analysis to 38.9% for Vollenweider's first model with its original parameter values. Misolassification by more than one trophic state occurred only in the case of the latter. The analysis suggests that the L versus Z axes are clearly inferior to the other classification schemes examined. It is more difficult to distinguish among the remaining models, however.

Using the estimated, as opposed to the observed retention coefficient, seems to have improved the performance of Dillon's model somewhat. Likewise, the stepwise discriminant analysis selected COEST instead of CO in deriving the optimal linear classification model for the data. If average outlet concentration, as an indicator of average lake concentration, is an important factor in determining lake trophic state, then the above results suggest that the estimated outlet concentrations (according to Dillon's equation or the model derived in the previous section) may be superior indicators of actual lake conditions than the reported values. As discussed previously, it is likely that significant errors exist in the NES estimates of the retention coefficients, because of the assumptions employed in converting the phosphorus balance observed for the year of sampling to an "average hydrologic year". The phosphorus retention models may act as filters by removing errors in the

( )

**(** :

reported retention coefficient (or outlet concentration) data. This suggests that the evaluation of Dillon's model based upon observed retention coefficients may have been hampered somewhat by the quality of the data.

The results of the stepwise analysis are presented on two sets of coordinate axes: COEST versus L (Figure 2.5-17) and L versus QS/(1-RE) (Figure 2.5-18). The fact that L entered significantly into the classification scheme after COEST is reflected by the slight downward slope of the classification lines in Figure 2.5-17. If these lines were horizontal, concentration alone would be significant. The apparent importance of both concentration and loading suggests that both the amount of nutrient available and the rate at which it is applied determine trophic response, although the latter is of relatively minor importance. The axes have been transformed in Figure 2.5-18 to isolate loading on the y-axis and to display the classification scheme in a manner similar to Figures 2.5-13 through 2.5-16, representing the other models evaluated. The fact that the discriminant lines in Figure 2.5-18 ars nearly parallel explains the success of the first principal component in capturing most of the discrimination power of these varaibles. The horizontal axis in Figure 2.5-18 can be interpretated as the areal removal rate of phosphorus per unit of concentration:

<sup>\*</sup> Another interpretation is that the canonical variable is more closely related to spring overturn phosphorus concentration than is outlet concentration alone; the former has been shown to correlate with mid-summer chlorophyll concentrations (se Figure 2.5-11)



0610 (<u>COEST</u>

()





Lake Classification According to Model Derived from Stepwise Discriminant Analysis on L vs. QS/(1-RE) Axes

2-155

$$\frac{QS}{1 - RE} = QS (1 + KT) = QS + KZ = QS + U \qquad (2.5-53)$$

where

M

(

K = offective first-order decay coefficient (1/yr) = 0.824 T<sup>-.546</sup>

U = effective phosphorus settling velocity (m/yr)

KΖ

The settling velocity is a term first introduced by Vollenweider<sup>94</sup>. The total removal rate is comprised of a flushing term, QS , and a sedimentation term, U .

Subsequent analysis will deal with the discriminant model based upon the first canonical variable derived from the stepwise analysis. This model represents the best summary of these particular data. No notion of the general superiority of this model over the others investigated is implied, with the possible exception of Vollenweider's first model. The techniques employed below to interpret and apply the discriminant model could be applied equally as well to the other models or other data bases.

One of the advantages of posing the lake trophic state estimation problem as a formal discriminant analysis is that the computed classification functions can be used to estimate the posterior probability that a given lake belongs to a given class. The following formula is useful is this regard <sup>26</sup>.

$$p_{ij} = \frac{e^{f_{ij}}}{n_k e^{f_{ik}}}$$
(2.5-54)
$$k=1$$

where,

( )

n<sub>k</sub> = number of groups
f<sub>ik</sub> = classification function value for
lake i and group k
p<sub>ij</sub> = posterior probably that lake i
belongs to group j

Table 2.5-15 lists the jackknifed probabilities for the misclassified lakes in the analysis based upon the first canonical variable derived from the stepwise analysis (Table 2.5-12). The probabilities provide means of assessing the severity of a given misclassification. For instance, Cayuga, reportedly mesotrophic, was misclassified as eutrophic by the discriminant model, but the computed 2-157

à
| Table | 2.5-15 |
|-------|--------|
|       |        |

|                  | Ś    |     |                                              |      |       |  |  |  |  |
|------------------|------|-----|----------------------------------------------|------|-------|--|--|--|--|
|                  | E.   | T.S | Classification<br>Probabilities <sup>a</sup> |      |       |  |  |  |  |
| Lake             | Obs. | Est | 0                                            | M    | E     |  |  |  |  |
| Winnepesaukee    | 0    | М   | .456                                         | .542 | .002  |  |  |  |  |
| Crystal          | 0    | М   | .144                                         | .830 | .026  |  |  |  |  |
| Sacandaga        | м    | ο   | . 890                                        | .110 | .000  |  |  |  |  |
| Keuka            | м    | 0   | .996                                         | .004 | .000  |  |  |  |  |
| Cayuga           | м    | E   | .010                                         | .493 | .497  |  |  |  |  |
| Clyde            | м    | Е   | .003                                         | .256 | .741  |  |  |  |  |
| Leech            | м    | 0   | .504                                         | .494 | .002  |  |  |  |  |
| Long             | м    | 0   | .797                                         | .203 | .000  |  |  |  |  |
| Okanagan         | м    | E   | .002                                         | .205 | . 794 |  |  |  |  |
| Pine             | м    | 0   | .618                                         | .382 | .001  |  |  |  |  |
| Lower St Regis   | Е    | м   | .135                                         | .840 | .026  |  |  |  |  |
| Conesus          | Е    | м   | .009                                         | .501 | .490  |  |  |  |  |
| Bemidgi          | Ē    | м   | .073                                         | .854 | .074  |  |  |  |  |
| Powder Mill      | E    | М   | .001                                         | .547 | .442  |  |  |  |  |
| Bodensee-Obersee | Е    | м   | .003                                         | .771 | .197  |  |  |  |  |
| Zurichsee        | Е    | М   | .031                                         | .756 | .213  |  |  |  |  |

Misclassified Lakes

a - Jackknifed

C.

(

Ć

### 2.5.5 Discussion of Misclassified Lakes (continued)

probability levels for the mesotrophic and eutrophic states were 0.493 and 0.497, respectively. Thus, this misclassification was not a severe one. A more serious error occurred in the case of Sacandaga, a reportedly mesotrophic lake which was estimated to have probabilities of 0.89 and 0.11 for the oligotrophic and mesotrophic states, respectively.

Patterns in the lake misclassifications have been compared with other residual patterns in the analysis, including those derived from the independent variable regressions in Table 2.5-3 and from the phosphorus retention models. No consistent patterns were evident. Possible sources of misclassification errors include erroneous data, model errors, and non-steadystate conditions.

Errors could occur in both the dependent and the independent variables. Mistakes in the originally-reported classifications represent dependent variable errors. The most severe misclassification occurred in the case of Keuka, classified by the EPA as mesotrophic, but which was assigned to the oligotrophic class with a posterior probability of 0.995. Examination of the NES working paper on this lake indicated no basis for the mesotrophic classification. The lake had relatively high transparency, low chlorophyll levels, and no evidence of hypolimnetic dissolved oxygen depression. It is not surprising that most of the

( )

classification errors involved mesotrophic lakes, since a lake in this class could be misclassified in either of two directions. The misclassifications arising from errors in the originallyreported trophic states might be traced to errors in data or in its interpretation. Errors of the latter type could be reduced by developing and applying a more objective classification scheme, similar to Shannon and Brezonik's Trophic State Index<sup>73</sup>. Another source of misclassifications is errors in the independent variables, i.e. in the estimates of phosphorus loading, hydraulic residence time, or mean depth.

Model errors would occur in systems with particular characteristics which would tend to alter their behavior relative to other lakes. One such case is Powder Mill, a eutrophic pond with a residence time of 6.6 days and with extensive growths of aquatic macrophytes.<sup>86</sup>The misclassification of the pond as mesotrophic may be due to the relatively rapid flushing of nutrients having less of a regulating effect upon the productivity of rooted vegetation than upon the productivity of suspended phytoplankton. As noted in Section 2.2, lakes with aquatic weed problems generally do not conform to classification schemes developed for lakes dominated by algal productivity. Possible effects of nitrogen limitation in some systems may be another source of model error.

### 2.5.5 Discussion of Misclassified Lakes (continued)

1

( :

A third type of error would arise from non-steady-state conditions existing in systems whose nutrient budgets and/or biota had not fully responded to changes in trends in phosphorus loadings at the time of sampling. For instance, the underclassification of New York State lakes Sacandaga, Keuka, and Conesus may have been due in part to the effects of detergent legislation which was being implemented in parts of New York at about the time of the NES sampling (1972). This might have influenced the phosphorus loadings, but the trophic states would not have had time to respond fully to any such changes. The under-classification of Bodensee-Obersee may have been due to a similar effect. Loading values of 4 and 1.1 grams/m<sup>2</sup>-yr are given for this lake in Vollenweider's 1968 <sup>93</sup> and 1973<sup>95</sup> papere, respectively. If the difference in loading is not due to an error, the lack of response time may explain this misclassification, since the latter loading value was used.

According to Uttormark and Wall's <sup>92</sup> dynamic lake classification scheme (Figure 2.2-1), the water quality of the underclassified lakes would be improving and that of over-classified lakes would be degrading. Accordingly, the misclassifications in Table 2.5-15 would serve as a basis for selecting likely candidates for lake protection and renewal programs. Such decisions would be made, however, only after other possible sources of classification error had been eliminated.

(:

The following section demonstrates the application of the discriminant model in a management context. The particular problem addressed is that of rationally selecting design value for an "acceptable" phosphorus loading for a given lake under conditions of uncertainty in the information used to formulate the problem. The particular management objective selected for the analysis is that the lake of concern be "non-eutrophic", i.e. mesotrophic or oligotrophic. The approach developed below could be applied equally as well to other possible objectives, such as achieving or preserving oligotrophic status. When there is uncertainty in the information or models employed in this type of design, probabilistic considerations become important. Under such conditions, the rational design basis is to satisfy a given probability level of achieving the management objective.

The property that the discriminant model can be used to generate classification probabilities is particularly useful in this application. The principal component analysis of the stepwise discriminant model has reduced the classification problem to a single dimension. Using the results in Table 2.5-12 and equations (51), (52), and (54), a probability can be assigned to any value of the canonical variable, X, according to: ())

$$p(e:X) = \frac{exp(f_e)}{exp(f_o) + exp(f_m) + exp(f_e)}$$
(2.5-55)

$$f_{e} = -18.508 - 20.487 \text{ XT}$$

$$f_{m} = -36.773 - 29.327 \text{ XT}$$

$$f_{o} = -53.804 - 35.646 \text{ XT}$$

$$\text{XT} = -x^{-.25}$$

$$\text{X} = \text{L} \left[ \text{QS} / (1 - \text{RE}) \right]^{-.8147}$$

$$\text{L} - \text{RE} = 1./[1. + .824 \text{ T}^{.454}]$$

Substitution of the corresponding exponential into the numerator of equation (55) would generate a probability for either of the other two trophic states from estimates of L , Z , and T . A plot of these probabilities against the base 10 logarithm of X is given in Figure 2.5-19. This plot clearly shows the separation of the trophic states according to the discriminant model. Figures 2.5-20 and 2.5-21 contain the same information plotted on log L versus log (QS/[1-RE]) axes. The lines in Figures2.5-20 and 2.5-21 correspond to lines of constant probability of eutrophy and oligotrophy, respectively. In a management context, the probabilistic interpretation of these lines renders them more useful than the "permissible" and







2-164

Ĉ





0,25

Ë

0.25

0,25

0,50

**q 10** 

0.7

0,75

0,00

100100

()

0,50

0.75

Figure 2.5-21



ı.

X

"dangerous" lines employed in the Vollenweider plots.

Suppose that the design basis were to be 95% sure that a given lake would be non-eutrophic. If there were no uncertainty in the variables needed to compute the value of X as a function of management strategy, i.e. in L, Z, or T, the rational loading allocation would correspond to the intersection of the 0.05 line in Figure 2.5-20 and a vertical line at the value of QS/(1-RE) appropriate for the particular lake. An equivalent method would be to find the value of log X in Figure 2.5-19 corresponding to a 0.05 eutrophic probability and to solve for the equivalent value of L. This has been done for eutrophic probabilities of 0.01, 0.05, and 0.10:

$$L_{.01} = 0.0237 [QS (1 + 0.824 T.454)].815 (2.5-56)$$

 $L_{.05} = 0.0310 [Qs (1 + 0.824 T<sup>454</sup>)]<sup>815</sup>$  (2.5-57)

$$L_{.10} = 0.0354 [Qs (1 + 0.824 T454)]815 (2.5-58)$$

These equations could be considered design equations for cases in which there is no uncertainty in the loading, residence time, or mean depth estimates. <u>(</u> .

When X cannot be specified exactly, however, the problem is somewhat more complicated. Uncertainty in X arises from sampling, measurement, and/or estimation errors in the estimates of L, T, and Z, all of which refer to mean quantities. This uncertainty is more precisely defined as the inability to exactly specify the X value which would result from the implementation of a particular management strategy. For instance, it would not be possible to exactly estimate the phosphorus removal efficiency of a future tertiary treatment plant. Existing or future non-point sources of phosphorus may be poorlydefined and thereby contribute substantial uncertainty to the estimate of total phosphorus loading. The effects of changes in land use on non-point sources would likewise be diffic plt to predict with much accuracy. Uncertainty in X would also rise from errors in the estimates or measurements of hydrology (groundwater and surfacewater flows) and lake morphology. The total uncertainty in X measures the relative ability of the planner or designer to match the target value of X specified by the discriminant model, given the available information and control technologies.

A first-order error analysis (Table 2.5-16) can be used to roughly estimate the error distribution of X from the error distributions of L, T, and Z. Clean estimates of the covariance matrix of L, T, and Z are difficult to obtain.

First-Order Error Analysis for Estimates of Canonical Variable X

Original definition :

 $X = L [QS / (1 - RE)]^{-.815}$ 

In terms of CI, QS, and Z :  $X = CI QS^{-185} [1 + .824 Z^{-454}QS^{-.454}]^{-.815}$ 

From Expected Value Theory :

$$\operatorname{Var}(X) \stackrel{3}{=} \stackrel{3}{\underset{i=1}{\overset{3}{\overset{3}{\overset{}}}} \stackrel{3}{\underset{j=1}{\overset{3}{\overset{}}}} \left( \frac{\partial X}{\partial y_{i}} \right) \left( \frac{\partial X}{\partial y_{j}} \right) \operatorname{Cov}(y_{i}, y_{j})$$

where:

$$y_1 = CI$$
$$y_2 = QS$$
$$y_3 = Z$$

Assuming  $Cov(y_i, y_j) = 0$  for  $i \neq j$ :

$$\frac{\operatorname{Var}(X)}{X^{2}} = \frac{\operatorname{Var}(CI)}{CI^{2}} + \alpha^{2} \frac{\operatorname{Var}(Z)}{Z^{2}} + (\alpha + .185)^{2} \frac{\operatorname{Var}(QS)}{QS^{2}}$$

$$\frac{2}{CV_{X}} = \frac{2}{CV_{CI}} + \alpha^{2} \frac{2}{CV_{Z}} + (\alpha + .185) \frac{2}{CV_{QS}}$$

where:  

$$\alpha = \frac{.305 \text{ z}^{-454} \text{ gs}^{-.454}}{1 + .824 \text{ z}^{-454} \text{ os}^{-.454}}$$

( )

(

. .

(

ſ

The problem can be simplified considerably in cases in which the off-diagonal elements of the matrix can be assumed to be negligible. In order to reduce the importance of the offdiagonal elements, the algebra in Table 2.5-16 has been done with X expressed in terms of CI, QS, and Z. This roughly partitions the total error into a concentration, a hydrologic, and a morphometric component. Since each of these variables represents a mean value, the standard error of the population mean can be computed directly if the estimate is based directly upon independent observations. Consideration of measurement errors and subjective assessments of uncertainty by people familiar with the respective sampling and measurment problems can also be incorporated into the error estimates. Since QS and Z represent relatively easily-observable quantities, errors in these estimates would be expected to contribute relatively little to the total uncertainty in X . Accordingly, most of the uncertainty would generally be attributed to the average inlet concentration estimates. Because the second deviation of X with respect to CI vanishes, higher-order terms in the expected value of X are insignificant and have been neglected.

Techniques for estimating the error distribution of a loading or average inlet concentration estimate from continuous flow and grab-sample concentration measuremente have been

### 2.5.6 Application Strategies (continued)

đ.

**(**)

discussed previously (Section 2.3.2). If loading has been estimated based upon tributary land use or population data, the distributions of unit loading factors (e.g.  $g-P/m^2$  Urban Area-year) found in the literature could serve as a basis to approximate the variance of the average inlet concentration estimate.

Derived from the equations in Table 2.5-16, an estimate of the coefficient of variation of X can provide a means of assessing the effects of this type of uncertainty upon the trophic state projection and upon the rational loading allocation. Two forms are suggested for the error distribution of X: normal and log-normal. The former seems appropriate in the sense that X represents an estimate of a mean quality, the distribution of which would tend toward normality as sample size increases. The latter has the desirable and realistic property of postivity and probably more adequately reflects the distributions of the measurements employed in estimating L and QS. The calculations outlined below have been done for each of these assumed distributions.

Assume that the error distribution of X can be described by completely by the two parameters  $\hat{X}$  and  $S_{\hat{X}}^{*}$ . The following multiplicative rule can be employed to combine the trophic state and error distributions:

$$p(e: \hat{X}, S_{\hat{X}}) = p(e: \hat{X}) \cdot p(\hat{X}: \hat{X}, S_{\hat{X}})$$
 (2.5-59)

The probability of a eutrophic state classification, given X and  $S_X^{*}$ , equals the probability of the classification, given X (equation (55)), times the probability of X, given  $\hat{X}$ and  $S_X^{*}$ . This equation can be evaluated by substituting the expression for the probability density function of X according to the assumed distribution and integrating over all possible X :

For normal X :

$$p(e:\hat{X},S_{\hat{X}}) = \int_{0}^{\infty} p(e:X) \frac{1}{S_{\hat{X}}\sqrt{2\pi}} e^{-\frac{1}{2}(\frac{X-X}{S_{\hat{X}}})^{2}} dX$$
 (2.5-60)

For lognormal  $\hat{X}$ :

$$p(e:\hat{x}, S_{\hat{x}}) = \int_{-\infty}^{\infty} p(e:X) \frac{1}{S_{10gX}^{\sqrt{2\pi}}} e^{-\frac{1}{2}(\frac{\log x - \log x}{S_{10gX}})^2} d\log x \quad (2.5-61)$$

From the theory of the lognormal distribution :

$$S_{logX} \simeq \sqrt{\ln(1 + CV_X^2)/2.303}$$
 (2.5-62)

$$\widehat{\log X} \simeq \widehat{\log x} - \frac{1}{2} s_{1 \circ g X}^2 \qquad (2.5-63)$$

$$\overline{CV}_{X} = s_{X} / \hat{X}$$
 (2.5-64)

### 2.5.6 Application Strategies (continued)

(

Equations (60) and (61) have been integrated numerically using Simpson's rule and the results are presented in Tables 2.5-17 and 2.5-18. Tables 2.5-19 and 2.5-20 present corresponding results for the oligotrophic state. Mesotrophic probabilities can be determined by difference. Essential aspects of these tables are summarized in the contour plots of Figure 2.5-22. The probabilities have been evalued for  $\hat{X}$  values ranging from 0.006 to 0.25 and  $CV_{\hat{X}}$  values from 0. to 0.8. At  $CV_{\hat{X}} = 0.$ , the results correspond to the curves in Figure 2.5-19. As the coefficient of variation of the X estimate increases, these curves spread out and the trophic state distinctions become less clear, as is evident in Figure 2.5-22.

At low probabilities of eutrophy, the effect of uncertainty in X is more pronounced for the lognormal distribution as compared with the normal distribution. This reflects the skewness of the former toward high X values. Because of this aspect, the lognormal tables and contour lines are more conservative than the normal ones. The normal tables and lines are probably more appropriate for use when the estimate is based upon a large number of direct measurements and is relatively well-defined.

The above results can be used to estimate the effects of uncertainty in X on the design value which satisfies a given

# Eutrophic Probabilities for Normal Error Distribution of X Coefficient of Variation of X

();

Ċ

1 :

| log <sub>10</sub> X | <b>X</b> .      | 0.0         | .05               | .10     | .15        | .20               | .30      | .40                  | . 50               | .60              | • 70                 | .80      |
|---------------------|-----------------|-------------|-------------------|---------|------------|-------------------|----------|----------------------|--------------------|------------------|----------------------|----------|
| -2.2000             | V.00631         | 0.0000      | 0.0000            | 0.0000  | 0.0000     | 0.3333            | 3.339)   | 3.3333               | 3.3333             | 1.3333           | 3.3333               | 3. 3331  |
| -4+1603             | 0.00661         | 0.0008      | 0.0000            | 0,0000  | 0.0000     | 0,0000            | 0.0000   | 0.0000               | 0.0000             | 0.0000           | 0.0000               | 0.0001   |
| -2.1000             | 0.00692         | 0.0000      | 0-000             | 0.0000  | 0+0000     | 0.0000            | 0.0000   | 0.0000               | 0.0000             | 0.0000           | 0.0001               | 0.0001   |
| -2-1433             | <u>J. 11724</u> | 1,1010      | 2.3.1.1           |         | 1.0000     | 1.000             | 0.0000   | 0.0000               |                    | 0.0001           | 0.0001               | 240092   |
| -2.10dd             | 0.00794         | 0.0000      | 0.0000            | 0.0000  | 0.0000     | 0.0000            | 0.0000   | 0.0000               | 0.0000             | 0.0001           | 0.0002               | 3.3333   |
| -2. 3033            | 3.33832         | 3.3333      | 3.3333            | 3.3333  | 3.0000     | 0.0000            | 0.0000   | 0.0000               | 0.0001             | 0.0001           | 0+0003               | 0.0005   |
| -2.0000             | 0+06871         | 0.0000      | 0.0000            | 0+0000  | 0.0000     | 0.0000            | 0.0000   | 0.0000               | 1000.0             | 0.0002           | 0-0004               | 0.0007   |
| -2.0400             | 0.60912         | 0.0000      | 0.0000            | 0.0100  | 343333     | 3,3333            | 913333   | 3*3331               | 3.3331             | 1.3133           | 1.1113               | 3* 30 34 |
| -Z.LUUJ             | 1.01000         | 0.0000      | 0.0000            | 0.0000  | 0.0000     | 0.0000            | 0.0000   | 0.0001               | 0.0003             | 0.0004           | 0.0010               | 0.0017   |
| -1-9633             | 3.31347         | 3.3333      | 2.2222            | 3,3333  | 3,3333     | 2.2222            | 3.3331   | 3.3332               | 0.0004             | 3.3338           | 0.0014               | 0.0022   |
| -1.9603             | 0.01096         | 0.0000      | 0.0000            | 0.0000  | 0.0000     | 0.0000            | 0.0001   | 0.0003               | 0.0006             | 0.0011           | 0.0018               | 0.0029   |
|                     | 0.01148         | 0.000       | 0.0000            | 0.0000  | 0.0000     | 0.0001            | 0.0002   | 0.0004               | 0.0011             | 0.0019           | 0.0033               | 0.0050   |
| -1.9030             | 0.01259         | 0.0000      | 0.0000            | 0.0001  | 0.0001     | 0.0002            | 0.0004   | 0.0008               | 0.0015             | 0.0027           | 0.0043               | 0.0065   |
| -1.0800             | 3.31318         | C.0001      | 0.0001            | 3,3,9,1 | 1.111      | 3. 1115           | 2.3372   | າມມີ                 | J.0321             | 3.3335           | 3.3356               | 3.3383   |
| -1.3000             | 0.01380         | C.CCUL      | 0.0001            | 0.0002  | 0.0002     | 0.0003            | 0.0008   | 0+0015               | 0.0028             | 0.0047           | 0.0072               | 0.0106   |
| -1.8400             | U-01445         | 0.0002      | 0.0002            | 1 1114  | 1 1116     | 1. 1117           | 1.3115   | 1.1124               | 1. 1:169           | 3.0079           | 0.0093               | 0.0135   |
| -1.8000             | 0.01505         | C.0004      | 0.0004            | 0.0006  | 0.0000     | 0.0011            | 0.0021   | 0.0038               | 0.0065             | 0.0101           | 0.0149               | 0.0206   |
| ~ L. 1400           | 0.01660         | _6.0006     | 0.0007            | 0.0008  | 0.0011     | 0.0015            | 0,0029   | 0,0051               | Q+0084             | J. 3129          | 3.3186               | 3.3254   |
| -1.1012             | 3.)1738         | 2.1016      | 3+0010            | 0.0012  | 0.0016     | 1200.0            | 0.0039   | 0.0068               | 0.0109             | U-0163           | 0.0231               | 0.03101  |
| -1.1400             | 0.01020         | 0.0014      | 0.0015            | 0,0010  | 0.0023     | 0.0030            | 0.0053   | 0.0089               | 0.0139             | 1 1265           | 1. 1367              | 3.0451   |
| -L.7200             | 0.01995         | 0.0020      | 0.0030            | 0.0035  | 0.0044     | 0.0056            | 0.0093   | 0.0149               | 0.0223             | 0.0315           | 0.0420               | 0.0536   |
| -1.0003             | 0.02089         | 0.0041      | 0.0043            | 0.0049  | 0.0060     | 0.0076            | 0.0123   | C.0190               | 0.0279             | 0.0385           | 0.0505               | 0.0634   |
| -1 .00 A            | 3. 32168        |             | 3, 3363           | 3,3368  | "2 PCC + C | 2.21.22           | 3. 3159  | 3, 3291              | J. C346            | 0.0467           | 0.0601               | 0.0743   |
| -1.0400             | 0.02291         | 0.0079      | 0-0083            | 0.0094  | 5.5110     | 0.0135            | 0.0206   | 0.0304               | 6.0425             | 0.0562           | 0.07[]               | 0+0865   |
| -1.0200             | 1. 12512        | 1.0100      | 0.0114            | 0.0167  | 0.0194     | 0.0231            | 0.0203   | 0.0468               | 0.0025             | 0.0795           | 0.0970               | 0.1144   |
| -1.9800             | 0.02630         | 0.0145      | 0.0202            | 0.0221  | 0.0254     | 0.0298            | 0.0419   | 0.0573               | 6.0748             | 0.0933           | 0.1119               | 0.1302   |
| -1.3033             | J. 32754.       | 3. 3258     | 3.3204            | 1.3293  | 3.3329     | 3.0382            | 0.0522   | 0.0695               | 0.0880.0           | 0+1086           | 0.1282               | 0.1471   |
| -1-2400             | J. U2684        |             | 0.0347            | J.0376. | 0.0422     | .0.0484.,         | U. 0645. | 0.0838               | 0+1040             | 0+1255           | 0.1458               | 0+1651   |
| -1.5200             | 0.03020         | 0.0430      | 0.057.            | 0.0483  | 0.0000     | 1. 1010           | 0.0957   | 0.1363               | 0.1414             | 0.1637           | 0.1046               | 0.2039   |
| -1.60.00            | 0.02311         | 3.3717      | 3. 3732           | 1. 1776 | 3.3844     | 3. 3933           | 3.1149   | 3.1387               | 3,1624             | 3.1849           | 3.2356               | 3.2245   |
| -1.4000             | 0.03460         | 0.0935      | 0.0921            | 1.0909  | 0-1014     | 0.1140            | 0.1367   | 0.1612               | 0.1850             | 0.2072           | 0+2275               | 0.2458   |
| - 6+4450            | 0.03031         | 0.1110      | 9.1148            | 4-1199  | 0.1278     | 0.1378            | 0.1611   | 0.1857               | 0.2072             | 0.2303           | 0.2502               | 0.2675   |
|                     | 21 824          | 1. J. 139 ( | J,1912.<br>3.1727 | J.1773  | J+1247     | 1+1042            | 0.2176   | 0.2403               | U.2616             | 0.2805           | 0.2973               | 0.3122   |
| -1.3000             | 0.04167         | 0.2067      | 0.2084            | 1615.0  | 0.2202     | 0.2291            | 0.2494   | 3,2733               | 3.2892             | 3.3363           | 3.3214               | 3.3347   |
| -1.3433             | 3. 34365        | 0.2471      | 0.2485            | 0.2523  | 0.2583     | 0.2658            | 0.2832   | 0.3010               | 0.3177             | 0.3326           | 0.3450               | 0.3573   |
| -1.3430             | J. 04571        | 0-2918      | 0.2926            | 0.2953  | 0.2996     | 0.3052            | 0+3107   | 0.131                | 0.3468             | 0.3592           | 0.3701               | 0.3/98   |
| -1.5203             | J. J4787        | J.3473      | 3.3433            | 3.3414  | 1,3096     | 3.3907            | 0.3930   | 0.3003               | 0.4059             | 0.4124           | 0.4185               | 0.4240   |
| -1.2000             | 0.05248         | 0.4439      | 0.4428            | 0.4401  | 0.4370     | 0.4343            | 0.4319   | 0.4327               | U. 4354            | 0.4387           | 0.4422               | 0,4456   |
| -1.2013             | J. J5456        | J.4973      | 3.4952            | 3.4438  | 3.4849     | 3.4790            | 8+4704   | 0.4661               | 0.4045             | 0.4647           | J. 4655              | 0,4667   |
| -1.2433             | 0.05755         | 0.5500      | 0.547c            | 0.5411  | 0.5325     | 0.5235            | 0.5087   | 0.4991               | 0,4714             | 0.4901           | 0.4683               | 0.4073   |
| -1.2230             | 0.06025         | 0.6609      | 0.5980            | 0.5901  | 0.5791     | 0.6094.           | J. 7403  | 0.5629               | J 7217<br>0.5489   | 0.5390           | 0.5319               | 0.5266   |
| -1.1600             | 0.06607         | 0.6940      | 0.6902            | 0+6908_ | 0.4004     | 0.6494            | 0.6181   | 0.5914_              | 0.5754             | 0.5624           | 0.5527               | 0.5453   |
| -1.1633             | 3.36919         | 3.7353      | 3.7315            | 3.7214  | 3,7361     | 1.6879            | 3.6518   | 3+6227               | 3.6039             | 9.5848           | 3.5727               | 0.5633   |
| -L-1400             | J.67245         | C.171J      | 0.7684            | 0.7584  | 0-7427     | 0.7234            | 0+6838   | 0.4777               | 0.6254             | 0.6004           | 0.0103               | 3.5971   |
|                     | 1. 17966        | 1.8112      | 3.83.11           | 1.8211  | 3.8060     | 0.7862            | 0.7419   | 0.7023               | 0.6710             | 0.0210           | G.6279               | 0.6130   |
| -[.3630             | 4.08318         | J.8581      | 0.8553            | 0.0471  | 0.8328     | 0.8134            | 0.7680   | 0.7260               | 0.6920             | 0.6654           | 0.6446               | 0.6281   |
| -Lougue             | U.C6710         | 0.8740      | 0.8772            | 0.8697. | 1.850.5    | 3.0376_           | 3.7923   | 3.7481_              | 3.7119.            | 2.6832           | 3.6636               | J. 6425  |
| -1.04-04            | 0.05121         | C-3980      | 0.8959            | 0.8692  | 0.8772     | 0.8595            | 0.8141   | 0+7657               | 4.7300             | 0.7000           | 0.60037              | 0.6697   |
| -1.0200             | 0.09551         | 0.9137      | 1.4254            | 1.9234  | 1.91 14    | 1.0456            | 3.8524   | 3.8356               | 3.7646             | 3.7339           | 3.7037               | 0.6816   |
| -0.9000             | U.10472         | 0.9384      | 0.4370            | 0.9326  | 0.9241     | 0.9103            | 0.8089   | 0.6220               | 4.7799             | 0.7449           | 0.7165               | 0.6933   |
| -0.9600             | 0.L0965         | 0.9479      | 0.9467            | 0.9429  | 0,9356     | 0.9232            | 0.8838   | U.837L               | 6.7942             | 0.7542           | 3. 7286              | 3.7344   |
| -3,9433             | 3.11462         | 3+9959      | 3.9542            | 3.9517  | 0.9454     | 0.9343            | 0.8971   | 0.8510               | C.8076             | 0.7823           | 0,7400               | 0.7248   |
| -6.9200             | 0+12043         | 0.9627      | 0.9617            | 0.9664  | 1.4517     | 1.0421            | 1.9196   | J. 4753              | J. 8315            | 1.7931           | 3.7639               | 3.7342   |
|                     | 0.13183         | 10.5722     | 0.9726            | 0.9707  | 0.9607     | 0.9591            | U.9290   | 0.8859               | 0.8421             | 0.8033           | 3.7735               | 3.7433   |
| -3.8633             | 3.13835         | 1.5773      | 1.9768            | 3.9751  | 0.9717     | 0.9652            | 0.9373   | 0.8955               | 0.8520             | 8518+0           | 0.7794               | 0.7514   |
| -0.8400             | 0.14455         | 0.5807      | 0-9902            | 0.9789  | 0.9760     | 0.9703            | 0.9447   | 0.9043               | 0.8611             | 0.8217           | 0.7879 (<br>9 7050   | 1.744    |
| -0.82.00            | Q.15177         | 0.9836      | ؕ9832.<br>0.4⊬54  | J.982J. | J. 4 / 4 6 | J. 9/97           | 0.9470   | 0.9195               | 0.8773             | 112              | 0.8032 (             | 5.7737   |
| -0.0000             | 0.12547         | 0.4000      | 0.9877            | 0.99469 | 0.9452     | 0.9810            | 0,9521   | 0.9261               | 0.8845             | 0.8450           | 0.8101               | .7804    |
| -3.7033             | 3.17379         | 2.9898      | 3.9895            | 3.9388  | 3.9574     | 1.9843            | 3.9665   | 2.9321               | 2.8912             | 0.8517           | 0,8166               | 0.7866   |
| -1.7400             | 0.10198         | 0.9912      | 0.9910            | 0.9104  | 0-9022     | 0.9866            | 0.9105   | 0.9375               | 0.8973             | 0.0540           | V+8228 (<br>1.830≤   | D.7925   |
| -0.7.00             | 0.19050         | 0.9925      | 0.9923            | 0.9718  | 0.7908     | 0.9882            | 0+9739   | U+9424 (<br>A.8648 - | 0.0042             | 0.8/01 -         | 0.8340 ·             | 0.0033   |
| 1-3:365-            | 1.2CH42         | 0.4945      | 0,9943<br>0,9943  | 0.4430  | 0.9932     | 0.7902_<br>0.7916 | 0.9146   | 0.9509               | 9.9130             | 0.8744           | 0.8391               | 5008.0   |
| -4.6643             | J.21879         | 0.9952      | 3.9951            | 3.9940  | 1+9942     | 1.9928            | 3.9819   | 3.9545               | 3.9174             | J.8792           | J.8439               | .8126    |
| -0.4400             | 0.2710          | 0.9959      | 0.9951            | 9.9955  | 0.9950     | 0.9938            | 0-9840   | 0.9578 1             | 0.9216             | 0.8836           | 0.8485 (             | D.BL71   |
|                     | 0+21990         | 0.5964      | 0.9962            | 0,9901  | 0.9957     | 1-005-            | 0.9358   | 0.9408 (<br>0.9476 ) | 1127273<br>1227293 | U-34/8<br>U-84/8 | 0,0547 4<br>0,0567 4 | ).#24A   |
|                     | 3465161 1       | 143703      | 4+2269            | *****   | 4+7782     | *****             | 442019   | ******               |                    |                  |                      |          |

-

•

# Eutrophic Probabilities for Lognormal Error Distribution of X

(

( ,

(;

 $m_{\rm e}^2$ 

Coefficient of Variation of X

|   | log         | ۲ n                      | C          | x                    | 0.0       | .05                         | .10                  | .15                  | .20                  | .30                  | •40                  | .50                  | .60                  | .70                  | .80                |
|---|-------------|--------------------------|------------|----------------------|-----------|-----------------------------|----------------------|----------------------|----------------------|----------------------|----------------------|----------------------|----------------------|----------------------|--------------------|
|   |             |                          | _          |                      |           |                             |                      |                      |                      |                      |                      |                      |                      |                      |                    |
|   | 15          | (+∠40)<br>               | 0          | 0+00631              | 0.0000    | ) <b>3,3303</b><br>) 0,8808 | J.3373               | 3.3333               | 3.3333               | 0.0000               | 0-0000               | 0.0001               | 0.0004               | 3.3339               | 0.0019             |
|   | -2          | .160                     | Ŭ          | 0.00492              | 0.0000    | 0.000                       | 0.0000               | 0.0000               | 0.0000               | 0.0000               | 0.0000               | 0.0002               | 0.0006               | 0.0014               | 0.0027             |
| • |             | -1-1.                    | <u>ر</u>   | J. JJ724             | 3.3333    | 3.3303                      | 3.3333               | 3-3333               | 7.7273               | 3.3333               | 0.0000               | 0.0002               | 0.0007               | 0.0017               | 0.0033             |
|   | 1.2         | + 1 ÚÜ1                  | 5          | 0.00794              | U.CCUU    | 0.0000                      | 0.0000               | 0.0000               | 0.0000               | 0.0000               | 0.0001               | 3.3334               | 3.3312               | 3.3326               | 0.0014             |
|   | -2          |                          | <b>)</b>   | J. JJ832             | 0.0000    | 0.0000                      | 0.0000               | 0.0000               | 0.000                | 0.0000               | 0.0001               | 0.0005               | 0+0015               | 0.0031               | 0.0055             |
|   | 1           | -060                     |            | 6.CG871              | 1 0.0000  | 0.0000                      | 0.0000               |                      | 0.0000               | 0,0000               | 0.000Z               | 0.0007               | 0.0018               | 0.0038               | 0.0065             |
|   | -2          | .020                     | i i        | u. CC955             | 0.0000    | 0.0000                      | 0.0000               | 0.0000               | 0.0000               | 0.0001               | 0.0004               | 0.0012               | 0.0028               | 0.0055               | 0.0090             |
|   | 124         |                          | j i        | 0.01000              | 6.0000    | 0.0000                      | 0+0000               | 0.0000               | 0.0000               | 0.0001               | 0.0005               | 0.0015               | 0.0035               | 0.0066               | 0.0105             |
|   | E           | 1900                     | j (        | 0.01046              | 0.0000    | 0.0000                      | 0.0000               | 0.0000               | 0.0001               | 0.0003               | 0.0009               | 0.0025               | 0.0053               | 0.0093               | 0.0142             |
|   | -+          | 044J                     | <u>)</u>   | 0.01140              | 0.0000    | 0.0000                      | 0.0000               | 0.0000               | 0.0001               | 1,1114               | 7-1215               | 3.3331               | 3.3364               | 3.3113               | J. J164            |
|   | 1-1         | *7530<br>*7530           | , ,<br>, , | 0.01202<br>0.01259   | 0.0000    | 0.0000                      | 10000.0              | 0.0001               | 0.0001               | 0.0005               | 0.0010               | 0.0039               | 0.0078               | 0.0130               | 0.0190             |
|   | -ī          | .443.                    | j j        | 2.21310              | 1.1231    | 3.3331                      | 3.3331               | 3+3332               | 3.3333               | 3, 3313              | J.JJ27               | 3.3361               | J. 1113              | 3.3178               | 3.0251             |
|   | 124         | , 1000 a                 | 1          | 0.01380<br>0.01446   | 0.0001    | 0.0001                      | 0.0002               | 0.0003               | 0.0004               | 0.0013               | 0+0035               | 0.0076               | 0.0136               | 0.0200               | 0.0287             |
|   | -i          | .823                     | 5_         | J. J1514             | 1.1313    | 3.3333                      | 3.3334               | 1.3336               | 3.0002               | 0.0024               | _0.0050_             | 0.0115               | 0.0192               | 0.0280               | 0.0372             |
|   | -1          | .464                     | 1          | 0.01545              | 1-0.0004  | 3.0004                      | 0.0006               | 0.0006               | 0.0013               | 0.0033               | 0.0074               | 0.0141               | 0.0227               | 0.0324               | 0-0422             |
|   | 121         | 1000                     | 5          | 0.01736              | C.JCJ9    | 0.0007                      | 0.0004               | 0.0017               | 0.0025               | 0.0057               | 0.0117               | C.0206               | 0.0113               | 0.0427               | 0.0538             |
|   | -1          | . 1400                   |            | 0.01820              | 0.0014    | 0.0015                      | 0.0018               | 0.0024               | 0,0035               | 0.0075               | 0.0147               | 0+0248               | 0.0366               | 0.0487               | 0.0604             |
|   | 121         | +12JJ<br>.Judi           |            | ), )1936<br>5.81996  | 1.1323    | 3.3321                      | J.JJ26               | 3.3334               | 3.3348               | 3, 3397              | J.J[82<br>0.0224     | 0.0352               | J. 3445              | J+1555               | 0.0677             |
|   |             |                          |            | 0.02089              | 0.00+1    | 0.0043                      | 0.0051               | 0.0064               | 0. 1086              | 0.0160               | 0.0275               | 0.0417               | 0.0567               | 0.0712               | 3. 3843            |
|   | 121         | .463                     |            | 3. 32180             | 3.3357    | 1.3363                      | J.3373               | 9.0087               | 0.0114               | 0.0205               | 0.0334               | 0.0491               | 0.0651               | 0.0802               | 0.0937             |
|   | [-1         | +0200                    |            | 1.02399              | 0.0108    | 0.0113                      | 0.0128               | J.)155               | 3. 3195              | 3. 3319              | 3.3480               | 3.3669               | 3. 3047              | 3.1338               | 3.1147             |
|   | 1-1         | .063                     | 2          | 0.02512              | 0.1140    | 0.0152                      | 0.0171               | 0.0204               | 0.0252               | 0.0395               | 0+0580               | 0.0776               | 0.0960               | 0.1124               | 0-1264             |
|   | ··          |                          |            | J. J2754             | 1.1258    | J.)267                      | 3.3295               | U.U242               | 3.3413               | 3. 3595              | 3.3812               | 3.1327               | J.122)               | 3.1384               | 0.1521             |
|   | -1          | . 2434                   | i i        | .02384               | 0-0338    | 0.0349                      | 1.0182               | 0.0438               | 0.0516               | 0.0721               | 0.0952               | 0.1173               | Catle7               | 0.1529               | 0.1662             |
|   | [ ]]        | 。シェンゼ<br>。シュンショ          |            | J. 03020<br>J. 13162 | 1. 0+0458 | 0.0451                      | J-0471               | J-0556               | 0.0644               | 0.1036               | 0+1109               | 0.1333               | 0.1525               | J.1683<br>0.1847     | 0.1968             |
|   | -i          | •4dvu                    | i i        | 1.03211              | 0.0711    | 0.0735                      | 0.0786               | 0.0869               | 0.0976               | 0.1227               | 0.1477               | 0.1696               | 0.1876               | 0.2020               | 0.2133             |
|   | <u>_</u>    | • 44 () (                |            | 1.634 <u>68</u>      | 9.6905    | 0-0924                      | 1.1982               | 21172                | <u>.).1105</u> .     | 2.1942               | 2.1689               | 3.1433               | J.2369               | J,22J3 '             | J.23 J6            |
|   | 1-1         | .4200                    | ŭ          | 1.03031              | 0.1357    | 0.1151                      | 0.1213               | 0.1581               | 0.1425               | 0.1945               | 0.2168               | 0.2349               | 0.2489               | 0.2594               | 0.2674             |
|   | -1          | .4333                    | 1          | ). J398L             | 3.1739    | 3.1732                      | 3.1796               | 3.1891               | 3.2333               | J.2233               | 3.2435               | J.2094               | 0.2714               | 0.2803               | 0.2868             |
|   | 121         | د بان کانا<br>د بان کانا |            | 1+Q+164<br>1-04365   | 0.2067    | 0.2689                      | 0.2544               | 0.2238               | 0.2340               | 0.2344               | 013016               | 0.3119               | 0.2944<br>J.3192     | 3.3242               | 3.3275             |
|   | 1-1         | ډ د و.                   | <u>,</u>   | 1. 14571             | 2.4918    | 0.293.                      | 0.2975               | 0.2035               | 0.3102               | 0,3220               | 0.5327               | J 3398               | 0.3444               | 0.3472               | 0.3486             |
|   | Ett         | • 22vù                   | U          | ).04787<br>).04787   | 0.3400    | 0.3410                      | 1.4425               | 0.34/6               | 0.3518               | 0+3555               | 0.3651               | 0.3085               | 0.3702               | 0.3707               | 0.3703             |
|   | ['-i]       | .2800                    |            | 1.05240              | U.4439    | 0.4435                      | J.4427               | 0.4414               | 0.4400               | 0+4363               | 0+4322               | 0 4279               | 0.4234               | 0,4190               | 0.4147             |
|   | 1-21:       | 2611                     | 0          | 1. 05496             | 0.4973    | 0.4962                      | 0.4934               | 0.4896               | 0.4851               | 0.4756               | 0.4666               | 0.4582               | 0+4506<br>0-4740     | U.4437               | 0.4373             |
|   | 1.21        | .2230                    | 0          | 05020                | 0.6009    | 0.5987                      | 0.5927               | 0.5843               | 0.5745               | 0.5541               | 0.5355               | y, 5193              | 0.5054               | 0.4935               | 0.4831             |
|   | =           |                          | Ũ          |                      | 0.6492    | 0.6465                      | 0.0395               | 0.6293               | U. 5174              | 0.5924               | 3.5695               | J.5496               | 3.5320               | J.5184               | 3.5361             |
|   | Et          | 1000                     | 1          | 1.06919              | 0.7350    | 0.7320                      | 0.7238               | 0.0719               | 0.6971               | 0.6653               | 0.6353               | 0,5091               | 0.5860 (             | 0.5679               | 0.5518             |
|   | -1          | 1400                     | ō          | +ú7245               | 3.7718    | J. 7686                     | 3.7637               | 0.7483               | 3.7332               | 3.6993               | 3.0067               | 3.6378               | J.6132               | 3.5923               | 3.5745             |
|   | 171         | 1200                     | ี<br>ม     | ,07506<br>,07944     | 0.8045    | 0.8305                      | 0+7937<br>N-8231     | 0.78[6 -             | 0.7663 -<br>0.7965 - | 0+7314 -<br>0-7613 - | 0.7254               | 0.6657  <br>0.6926   | 0.6390 (<br>8.6641 ( | 0.6397  <br>1.6397   | 0.5968 :           |
| • | 1-11        | ננשר.                    | Ĵ          | . 18318              | 3.8501    | J.8556                      | 3.8488               | 3.0385               | 3.8237               | 3.7890               | 0.7525               | 0.7185               | u005 (               | D.6627               | 0.4405             |
|   | l-≣t:       | 10000                    | . 0        | -CUTIO               | 0.4530    | 0.3774                      | 0.8712               | 0.8613               | 0.8480               | 0.8144 -<br>1.8176   | 0.7779 (<br>\        | 0+7431 4             | 0.7120 (             | 3.6850 (<br>).7366   | 0.6616             |
|   |             | 3233                     | ŏ          | .09551               | 0.9137    | 0.9120                      | 0.9072               | 0.8973               | 0.8882               | d.8503 -             | 0.0234               | 0.7885               | 0.7562 (             | 1.7275               | 0.7023             |
|   | -1.         | 6000                     | Ģ          | 10001                | 0.9271    | 0.9256                      | 0-9214               | 0.9144               | 0.9045               | 0.8770               | 0.8436               | 0.8091               | 0.7767 (             | D.7475 (             | 0.7218             |
|   | -c.         | ,9077                    | U          | .10765               | C.9479    | J. 9371<br>U. 9468          | J.9339<br>U.9438     | J.9274<br>0.9385 (   | J.9187<br>0.9309 (   | 0.0930<br>0.9083 (   | J.#02J (<br>0.8787 ( | 0.0463 (             | J.1902 .<br>D.8145 ( | ).7851 (             | 0.7506             |
|   | <u>=</u> 9, | 9400                     | <u>_</u> u | -11402               | 0.9959    | 0.9556                      | 0.9524               | 0.9479               | 0.9413               | 0.9212               | 0.8937               | 0.8628               | 0.8317 (             | .9025                | .7759              |
|   |             | 9083                     | j<br>J     | 12543                | J.9627    | 3.9615                      | 3,9597               | 1.9559 (<br>1.9626 ( | 0.9502 (<br>0.9577 ( | U.9324  <br>G.4427   | 0.9072 (<br>0.9193 ( | 0.8779 (<br>0.8518 ( | 0.8478 (<br>D.8628 ( | ),8190 (<br>),8146 ( | 0.7925             |
|   | - J.        | 6600                     | ū          | .13193               | 0.97.12   | 0.9126                      | 0.9711               | 0.9683               | 0.9642               | 1.9516               | 1.9311               | J.9344 .             | 5.8767               | .8492                | . 8233             |
|   | <u>:</u>    | 1000                     | ᆤ          | - 1 29 25.           | 0.9173    | 0.9766                      | 0.9755               | 0.9731               | 0.9696_              | 0.9579               | 0.9394 (             | 0.9158 (<br>0.9250 - | 0.88995 (            | ).8629 (<br>1.8767 ( | 0.8509             |
|   |             | 4233                     | ្រ័        | .15137               | J.\$850   | J.9832                      | 3.9823               | J.9836 -             | 3.9781               | J. 9695              | 3.9553               | 3.9352               | .912)                | .8876                | . 8635             |
|   | -4.         | 6JUJ                     | 0          | -15850               | 0.9860    | 0.9820                      | 0.9849               | 0.9835               | 0.9814               | 0.9740               | 0.9613 (             | 0.9434               | 9218                 | 6.8986 C             | .8754              |
|   | 듣           | 7433                     | о          | +17379               | J.409A    | 0.9877  <br>).9845          | )"884) ·             | 77089 (<br>77089 (   | 0,9845 (<br>0,9845 ( | 0.9779 0             | 0.9008 (<br>0.9714 / | 0.9571 (             | 0.9388 C             | ).9182 (             | ).896 <b>8</b>     |
|   | -0.         | 1-00                     | Ŭ          | .10198               | 0.9912    | 0.9910                      | 0.9906               | 0.9897               | 0. 4985              | 0.9840               | 0.9757 (             | 0,9628.0             | .9460 0              | .9248                | .9065.             |
|   | -3.         | 7200                     | 0.         | .15056               | J.9925    | 0.9923 (                    | 0.9919<br>1.4448     | J.9912".             | 3.9931 (<br>1.9914 4 | J.9463               | 3.9792 J             | 3.9678 .<br>1.9722 - | ].9526 ]<br>].9584 A | 1.9346 )<br>.9614 (  | 1.9154  <br>1.9234 |
| 1 | -0.         | 6999                     | Ŭ          | .20894               | 0.9945    | 0.794.1                     | 1.9940               | 0.9935 (             | 0.9928               | 0.9903 (             | 6.9848 (             | 0.9760               | .9636 0              | .9482 (              | 9312               |
|   |             | COJJ.                    | 3          | .21679               | 3.5952    | J.9453                      | .9948                | J.9944               | ,9938                | 3.9915               | 3.9873               | .9794                | .9682 0              | .7541                | .9382              |
|   | -0.         | 0200<br>0200             | 0-<br>0-   | *53880               | 0.9559    | 0,9957 (<br>0,9963          | }•9955 (<br>),9961 ( | 0.9952 (<br>0.9958 / | 0,9954 (<br>),9954 ( | 0.9927 1<br>0.9937 1 | 0.9905 (             | J.9648 4             | 3.9723 Q<br>3.9759 J | ).9642 (             | .9534              |
|   |             | 6111                     | Ĵ,         | .25121               | 3.9964    | 1.9964                      | .9966                | 0.9964 (             | . 9960               | . 4746 (             | <b>1.9</b> 919 (     | 9669                 | . 4791 0             | . 9684 0             | .9557              |

٢.

# Oligotrophic Probabilities for Normal Error Distribution of X

( )

 $(\cdot)$ 

÷

Coefficient of Variation of X

|   | log <sub>10</sub> X | x                | 0.0           | •05                 | .10               | .15                 | .20          | . 30     | • 40             | .50                | .60         | .70                | .80      |
|---|---------------------|------------------|---------------|---------------------|-------------------|---------------------|--------------|----------|------------------|--------------------|-------------|--------------------|----------|
|   | 1-2.2000            | 0.00641          | 0.0055        | 0.9953              | 0.9944            | 0.9942              | 0.9931       | 6.4699   | 0.9000           | 0.9610             | 0.9247      | 0.4869             | 0.8511   |
|   | -2-1833             | 0.00661          | 0.9941        | 0.9919              | 0.9334            | 0.9925              | 0.9914       | 0.9871   | 0.9745           | n. 4685            | 3.9231      | 3.4743             | 3.6431   |
|   | -2.1033             | 3.33692          | 3.9925        | 0.9922              | 0.9916            | 0.9945              | 0.4869       | 0.9862   | 0.9723           | 0.4511             | 0.9163      | 10.6700            | 0.8340   |
|   | -2.1400             | 0.00724          | 0.9903        | 0.9901              | 0.4842            | 0.9879              | 0.9860       | 0.9804   | 0.9673           | 0.9447             | 0.9086      | 0.8610             | 0.8219   |
| • | -2.1200             | 0.00759          | 0.9877        | 3.9873              | 3.9861            | 3.9847              | 3.9824       | 3.9756   | 3.9612           | 3.9372             | 3.8997      | 3.85.19            | 1.8126   |
|   | -2.1000             | 0.00794          | 0.5843        | 0.9839              | 0.9827            | 0.9807              | 0.9779       | 0.9649   | 0.9540           | 0.9285             | 0.8896      | 0.8395             | 0. 8002  |
|   | -2.0000             | 0.00832          | C.9801        | 0.9795              | 0.9781            | 0.9757              | 0.9724       | 0.9630   | 0.9456           | 0.9185             | 0.0702      | 0.8269             | 0.7867   |
|   | -2.0033             | 0.33871          | 3.9748        | J.9742              | 3.9724            | 3.9696              | 3.9457       | 3.9548   | 3.9357           | 0.9071             | 0.8654      | 0.8131             | 0.7720   |
|   | -2+6400             | G.CC912          | 0.9683        | 0.9679              | 0.9654            | 0.9621              | 0.9575       | 0.9448   | 0.9243           | 0.8941             | 0.8512      | 0.7979             | 0.7562   |
|   | -2.0230             | 0.00955          | C.9602        | 0.9593              | 0.9909            | 0.9510              | D+9477       | 0.9336   | 0.9112           | 3.8796             | J.8356      | 3.7816             | 3.7394   |
|   | -2.3333             | 3.91333          | C.9504        | 0.9494              | U.9466            | 0.9421              | 0.9360       | 0.9202   | 0.6963           | 0,8635             | 0.8186      | 0.7640             | 0.7216   |
|   | - t • aaoo          | 0.01047          | U.9384        | 0,9373              | .0.9341,          | 0.9289              | 0.9222       | 0.9048   | 0.8795           | 0.0457             | 0.8002      | 0.7454             | 0.7029   |
|   | -1-2000             | 0+01046          | 3.9243        | J.9227              | 3,9191            | 3.9134              | 3.4191       | 3.8873   | <b>J.8</b> 6J8   | J. N2 64           | 3.7835      | ).7257             | J.6833   |
|   | -1.9400             | 0.01148          | 0,9067        | 0,9453              | 0.9014            | 0.4952              | 0.4972       | _0.8674. | _0.4040Z         | 0,8054             | 0,7596      | 0.1050             | 0.6631   |
|   | -1.9200             | 0.01202          | 0.8863        | 0.0648              | 0+8806            | 0.8740              | 0.8650       | 0-8452   | 0.8177           | 0.7830             | 0.7376      | 0.6835             | 0.6422   |
|   | -1.9000             | 0+01259          | 0.8023        | 1-8008              | 0.8564            | 0.8497              | 0.8411       | 0.8206   | 0.1934           | 0.7592             | 0.7145      | 0.6613             | 3.6239   |
|   | -1.00JJ             | J. J[310         | 3.8346        | 1.0331              | J-8588            | 3.8221              | 0.8137       | 0.7937   | 0.1614           | 0.1342             | 0+6906      | 0.6384             | 0.5992   |
|   |                     | 0.01300          | . 0.4030      | 0.8015              | 0.1413            | 0.7913              | 0,7034       | 0.7047   | 0.1340           | 0.1081             | 040039      | 0.0172             | V-2112   |
|   | -L+0400             | 0.01442          | 0.7014        | 0.7862              | 0.7041            | 0,1912              | 3.1333       | J./330   | 3.11.30          | 340131             | 1.0411      | 3+3410             | 342551   |
|   | -1.0200             | U. OLSRA         |               | 0                   |                   | 0.6406              | -84,574      | 000140   | 1.494            | 4 4 250            | 0.6011      | 0 5660             | U+2330   |
|   | -1-7434             | 3. 31663         | 1.6497        | 1.7302              | 1.6196            | 1.4396              | 1.6.171      | 1.6311   | J.A178           | 3.6044             | 1.8632      | 1.5212             | 0.4800   |
|   | -1.7000             | 0.01738          | 0.5919        | 1.5974              | 0.6017            | 0.5951              | 0.5958       | 0. 5947  | 0.5056           | 0.5597             | 0.5373      | 1.4966             | 0.4674   |
|   | -1-1400             | 0.01820          | 0.5429        | 0.5439              | 0-5468            | 0.5504              | 0.5539       | 0.5579   | 0.5533           | 0.5390             | 0.5115      | 0.4734             | 3.4462   |
|   | -1.7233             | 3. 31936         | 2.4935        | 3.4951              | 3.4996            | 0.5055              | 0.5116       | 0.5210   | 0.5211           | 0.5105             | 0.4862      | 0.4505             | 0.4254   |
|   | -1.7000             | J. 01995         | 0.4448        | 0.4469              | 0.4527            | 0.4638              | 0.4695       | 0.4844   | 0.4892           | 0.4425             | 0.4612      | 0.4281             | 0.4051   |
|   | -1.0000             | U. 02C89         | 0.39/5        | 0.4000              | 0.4073            | 3.4173              | 1.4282       | 3.4483   | J.4578           | 3 4553             | J. 4369     | 3.4363             | 3.3053   |
|   | -1.0000             | J. J2106         | 0.3524        | 0.3552              | 0.1032            | 0.3748              | 1864.0       | 0.4131   | 0.4272           | 0.4282             | 0.4132      | 0.3850             | 0.3662   |
|   | -1.0400             | 0.02291          | 0.3102        | 0.3132              | 0.3217            | 0.3346              | 0.3496       | 0. 1791  | U+3976           | 0.4022             | 0.3902      | 0.3645             | 0.3476   |
|   | -1+0233             | J. J2399         | J.2712        | J.2742              | J.203L            | 1.2947              | ).3131       | 3.3465   | 3.3093           | 3.3771             | 3.3080      | 0.3447             | 0.3298   |
|   | -1.4000             | 0+02512          | 0+2357        | 0.2386              | U.2475            | 0.2614              | 0.2788       | 0.3154   | 0-3415           | 0.3530             | 0+3466      | 0.3256             | 0.3126   |
|   | - 6.02903           | .y. 02630        | 9+2036        | , U+ 2 045 <u>.</u> | _) <u>, 4</u> 151 | 0.2290              | 0.2469       | 0.2840   | 0,3154.          | 0.3299             | 0.326L      | 3.3373.            | J+296L   |
|   | -1+2033             | 3. 32754"        | 3.1753        | 2-1777              | 1.1928            | 0.1995              | 0.2175       | 0.2584   | U+2906           | 0.3079             | 0.3000      | 0.2898             | 0+2803   |
|   | -1+5460             | 0.02864          | 0.1497        | 0.1522              | 0.1598            | 0.1728              | 0.1936       | 0.2326   | 0.2671           | 0.2070             | 0-2879      | 0.2731             | 0.2652   |
|   | 1-1-5400            | 0.03020          | 0+1275        | 0.1247              | 1.1361            | J+1484              | 3+1082       | 3.2367   | 3.2471           | 3.2012             | 3+2132      | 3.2312             | 3.25.26  |
|   | 1.5000              | 0.03102          | 0.1001        | 0.1101              | 14,1104           | 0.1211              | 0.1942       | 0+1900   | U+2243           | 0+4402             | 0.2754      | 0.2420             | 0.2311   |
|   | -1.44000            | 84441            | 3. 3766       | 1. 1742             | 1 3432            | 1. 1926             | 3.1.17.1     | 1.1474   | 1.1874           | 3.2164             | 0.2225      | 0.2141             | 0.2118   |
|   |                     | 0.04641          | 0.0640        | 0.0654              | 0.0698            | 0.0782              | 0.0915       | 0.1310   | 0-1708           | 0.1910             | 0.2084      | 0.2012             | 0.2000   |
|   | -1.4203             | 0.33802          | 0.0532        | 0.0544              | 0.3283            | 0.0057              | 0.0740       | 0.1150   | 0.1556           | 0.1046             | J. 1951     | 2.1891             | 3.1893   |
|   | -1+4333             | 3. 33991         | 3.3439        | 0.0450              | 0.0484            | 0.0550              | 0.0661       | 0.1041   | 0.1415           | 0.1711             | 0.1826      | 0.1777             | 0-1785   |
|   | -1-3430             | 0.04169          | 0.0360        | 0.0369              | 0.0199            | 0.0457              | 0.1557       | 0+0448   | <b>U.1286</b>    | 0+1587             | 0.1710      | 0.1669             | 0+1686   |
|   | -1.JoUd             | 0.04365          | 0.0293        | 7:3331              | 75EC+C            | 3.3378              | J. )468      | 3.3768   | 3.L148           | J.1471             | J. 16 JL    | 3.1568             | 3.1592   |
|   | -1-2944             | <u>0. 045 71</u> | <u>0.0237</u> | 0.0244              | 7.4260.           | 0-0311.             | Qa1390       | 0,0690   | <b>D</b> _1061   | .0,1303.           | _0.1498     | 0+1472.            | 0.1504   |
|   | -1-200              | 0-04787          | 0.6190        | 0.0196              | 0-0215            | 0.0254              | 0-0324       | 0.0603   | 0+0962           | 0.1204             | 0.1403      | 0.1383             | 0-1421   |
|   | -1+3333             | J. J5312         | 3.3151        | J. J156             | 3.3113            | 3. 32 36            | 3+ 32 68     | 7.7250   | 3.3673           | 0.1172             | 0.1314      | 0+1299             | 0-1343   |
|   | -1-2800             | 0.05240          | 0.0118        | 0.0123              | 0+0137            | 0.0100              | 040220       | 0.0450   | 0.0/92           | 0+ 038             | U.123L      | 0.1220             | V.1269   |
|   | -1.2000             | 0.03490          | 0.0071        | 0.0076              | 0.0108            | 0.0133              | 0.0107       | 0.0399   | 343719           | 0 1032             | 0.1007      | 3.1077             | 0.1197   |
|   | 1                   | 1-06076          | 0.0071        | 0.0057              | 0.0009            | 0.0100              | 0.3110       | 0.0345   | 0.0594           | 0.0471             | 0.1315      | 0.1013             | 0.1071   |
|   | -1-21013            | 4,06313          | 1.1142        | 3- 3346             | 3-1351            | 1.3365              | 1. 1 196     | 3.3261   | 3.353.           | 3.3810             | 3. 1953     | 3. 1952            | 5.1513   |
|   |                     | 0.06607          | 0.001         | 0.0034              | 0-0019            | U-0051              | 0.0077       | 0.0227   | 0.0490           | 0.0754             | 0.0895      | 0.0895             | 0.0958   |
|   |                     | 0-06919          | 0.0024        | 1.0025              | 0.0030            | 0.0019              | 0.0061       | 0.0197   | 0.0445           | 0.0702             | 0.0841      | 0.0841             | 0.0907   |
|   | -1-1433             | J. 37245         | 3.3310        | 3.3319              | 3.3322            | 3.3333              | 3.3349       | 0.0171   | 0.0400           | 0.0655             | 0.0791      | 0.0791             | 0.0859   |
|   | -1.1200             | 0.07586          | 0.0013        | 0.0014              | 0.0017            | 0.0023              | 0.0038       | 0.0149   | 0.0370           | 0.0611             | 0.0744      | 0.0743             | 0.0814   |
|   | -1-10-10            | 0.07944          | 0.0010        | 0.0010              | 0.0313.           | 0.0016              | 0.003)_      | 3. 31.29 | 2.J338_          | 3, 3571            | 3. 37 33    | 3.3699             | 2, 3772  |
|   | -1.0000             | 0.04318          | 0.0007        | 0.0005              | 0.0309            | 0.0013              | 0.0024       | 0.0113   | 0.0309           | 0+0534             | 0.0659      | 0.0657             | 0.0732   |
|   | -1-0000             | 0-06710          | 0.0005        | 9.0006              | 0.0007            | 0.0010              | 0.0013       | 0.0098   | 0.0283           | 0.0200             | 0.0621      | 0.0618             | 0.0696   |
|   | -1-3433             | J. J9121         | 3.3334        | 3.3334              | 3.3335            | 1.1118              | 3.3314       | 1.2386   | 2.3259           | 3. 3469            | 3.3586      | J.J582             | 3.3661   |
|   | - 1,0200            | 0.09551          | 0.0003        | 0.0003              | 0.0004            | 0.0009              | 0.0011       | 0.0075   | 0.0238           | 0.0440             | 0.0553      | 0.0548             | 0.0629   |
|   | -1.000              | 0.10001          | 0.0012        | 0.0002              | 0.0003            | J+0004              | 0.0009       | 0.0066   | 0.0219           | 0.3413             | 0.0522      | 0.0516             | 0.1244   |
|   | -9.9833             | J. 1 3472        | 3+3332        | 3.3332              | 3-3335            | 3-3003              | 0.0007       | 0.0058   | 0.0201           | 0.0389             | 0.0493      | 0.0486             | 0+0571   |
|   | -0.9000             | 0.16903          | 0.0001        | 0.0001              | 0.0002            |                     | 0.0009       | 0.00001  | 0.0100 ·         | 0.0300             | 1442        | 0.0499             | 1 7610   |
|   |                     | U. 11402         |               | 0.0001              | 0.0001            | J. J. J. J. Z.      | 14 J J J J 7 | 0.000    |                  | J.J.J.47<br>0.0176 | 2. J742     | 3.3610             | 0.0103   |
|   |                     | 0.12606          | 3-03-4        | 4.0001              | 020001            | 0.0001              | 0.0003       | 0.0016   | 0.0147           | 0.0100             | 0.0100      | 0.0348             | 0.0469   |
|   | 1-1.000             | 1.12183          | . 1.3131      | 3. 13 33            | 1.3331            | 2.2731              | 3. 1132      | 1.0133   | 3,3136           | 3,3/91             | 1. 1316     | 0.0368             | 0.0445   |
|   | with Hoth           | 0.13805          | 8.0000        | 0-0000              | 0.0000            | 0.0001              | 0.0001       | 0.0027   | 0.0126           | 0-0275             | 12.0361     | 0.0349             | 0.0422   |
|   | -0-0400             | 0.14455          | 0.0000        | 0.0400              | 0.0000            | 0.0005              | 0.0001       | 0.0020   | 0.0117           | 0.0261             | 0.0345      | 3. 3332            | 3. 33991 |
|   | -1.021              | 3. 151 37        | 3.3113        | 1.111               | 1.1000            | 0.0000              | 0.0001       | 0.0023   | 0.0109           | 0.0241             | 0.0129      | 0.0316             | 0.0377   |
|   | -0.8000             | U. 15850         | 0.0000        | 3.0000              | 0.0300            | 0.0030              | 0.0001       | 0.0019   | D.0102           | 0+0214             | 0.0115      | 0.0301             | 0.0355   |
|   | -0.7800             | 4.16297          | 0.0000        | 0.0000              | 0.0333            | 1.111               | 1.1111       | 3. 3319  | 3.3395           | 3. 3222            | 3.3332      | 3. 2286            | 3, 3334  |
|   | -0. 7000            | 0.17379          | C.COUU        | 0.0000              | 0.0000            | 0.0000              | 0.0000       | 0.0015   | U+0089 (         | 0+0210             | 0+0249      | 0.0272 (           | 0+0314   |
|   | -0.7400             | 0+18195          | _ີຕະຈາກຕໍາ    | 4+0400              | 0,1030 (          | <b>0.000</b> 0      | 0.0000       | 0.0016   | 0,0083_0         | U+Q_UU             | 0.0277      | 0.0259 (           | 0.0294   |
|   | -3.12)3             | 3.19336          |               | <b>31333</b> 2,     | 3-3333            | 1.1111              | 1.1100       | 0.112.   | 2.2277           | 2.3133             | 3.3265      | 0.0246             | 0.0276   |
|   | - 9. 7903           | u. 19954         | 0.0000        | 040000              | 6-9984            | 0.0000              | 0.0000       | 0.0311 ( | 0.0073 1         | 0+0101             | 0.0254      | 0.0233 (           | 9.0259   |
|   | -3.0800             | U-20394          | 0.0000        | 0.0000              | 0.0000            | u.uaua (            | 0.0000       | 0.0012   | 0.0068           | 0.0172             | 0+0242      | J. J22 J           | J+9244   |
|   | - 3-0633            | 3+21879          | 7+2231        | 3,3333              |                   | 0.0000              | 0.0000       | 0+0012 ( | 3 <b>.0064</b> ( | U+6764             | 0.0231 (    | 0.0206 (           | .0230    |
|   | 1-0.0400            | N. 2 .800        | 0.0000        |                     | , 10000 (         | 0.00000 (<br>1.11+1 |              | 0+00II ( |                  | 1. 1141            | V 1 V 2 V 1 | 7.VL73 (<br>1 1194 | 1.33.4   |
|   |                     | 1 2 51 21        | 0.0000 ·      | 0,0000              | 0.0300            | 0.0000              | 0.0000       | 0.0000   | 0.0054           | )144               | 0.0198      | 0.0165 4           | 1.01671  |
|   |                     |                  |               |                     |                   |                     |              |          |                  |                    |             |                    |          |

2~175

....

# Oligotrophic Probabilities for Lognormal Error Distribution of X Coefficient of Variation of X

| log <sub>lo</sub> x | Х,                 | 0.0             | .05                | .10                               | .15       | .20              | . 30                | -40                   | .50                 | •60               | .70                   | .80     |
|---------------------|--------------------|-----------------|--------------------|-----------------------------------|-----------|------------------|---------------------|-----------------------|---------------------|-------------------|-----------------------|---------|
| -2+2000             | 0.00631            | 0.9955          | 0,9951             | 0.9946                            | 0,9937    | 0.9925           | 0.9677              | 0.9798                | 0.9693              | 0.9574            | 0.9453                | 0.9335  |
| -2.1800             | 0.00661            | 0.9941          | 0.9937             | 0.9931                            | 0.9920    | 0.9904           | J.9847              | 3.9756                | 3,9639              | 2,9510            | 3.9382                | 0.9261  |
| -2.1030             | 0.00692            | 0.9925          | 0.9920             | 0.9912                            | 0.9899    | 0.9879           | 0.9811              | 0.9706                | 0.9577              | 0.9438            | 0+9303                | 0-9178  |
| -2-1400             | 0.00729            |                 | 0.9870             | 0.9009                            | 0.9874    | 0.9840           | 0.9101              | 0.0501                | 0.0474              | 0.0260            | 0.0122                | 0.9089  |
| -2.1000             | 0.00794            | 0.5843          | 0.9846             | 0.9422                            | 0.9798    | 0.9762           | 0.9651              | 0.9502                | 0.9420              | 0.9171            | 0.9021                | 0.0992  |
| -2.0800             | 0.00832            | 0.9801          | 0.9793             | 0.9776                            | 0.9746    | 0.9734           | 0.9577              | J.9413                | 3.9235              | 0.9364            | 3.8913                | 3.8776  |
| -2.0600             | 0.00871            | 0.9740          | 0.9739             | 0.9718                            | 0.9683    | 0.9634           | 0,9490              | 0.9310                | 0.9122              | 0.8946            | 0.8790                | 0.8657  |
| -2.0400             | 0.00415            | 0:4993          | 0.9673             | 0.9648                            | 0.9607    | 0.9549           | 0.9387              | 0.9194                | 0.8998              | 0.6818            | 0.8062                | 0.8530  |
| -2-3233             | J. J. 955          | 0.9632          | 1.9591             | 3,9562                            | J,9514    | 0.9447           | J. 9269             | 0.9363                | 0.8863              | J. 8679           | J.0524                | 0+8395  |
| =1.9600             | 0.01000            | 0.9384          | 0.9369             | 0.9331                            | 0.9268    | 0.9185           | 0.8976              | 0.875                 | 0.8546              | 0.0329            | 0.8221                | 0.8252  |
|                     | 3. 31 396          | 3.9243          | 1.9223             | 3-9[8]                            | 3.9110    | 0.9020           | 0.8800              | 0.8574                | 0.8369              | 0.8196            | 0.8056                | 0.7943  |
| -1.9400             | 0.01148            | 0.90bT          | 0,9049             | 0.9031                            | 0.8925    | 0,8829           | 0-8602              | 0.8377                | 0.8177              | 0.8013            | 0.7881                | 0.7777  |
| -1.9200             | 0.01202            | 0.9861          | 0.8843             | 0-8791                            | 0.8711    | 3.8611           | 0.8303              | 3.8162                | J. 7972             | 0./819            | 1.7698                | 3.7633  |
| -1-9000             | 0.01259            | 0.8023          | 1048.0             | 0.8548                            | 0.8400    | 0.0365           | 0.8140              | 0.1931                | 0.7754              | 0.7614            | 0.7506                | 0.7423  |
| -1.0000             | 0.01310            | 0.0440          | 1 9 779            | 0.8210                            | 1 7977    | 1.7745           | 1.7501              | 1.7614                | 0.7523              | 1.7176            | 0.700a                | 0.7235  |
| -148400             | 0-01443            | C.7L74          | 0.7655             | 0.7606                            | U.75.5    | 0.7453           | 0.7284              | 0.7139                | 0.7026              | 0.6942            | 0.6883                | 0. 6862 |
|                     | 0.01514            | 0.7281          | 0.7264             | U.7222                            | 0,7163    | 0.7095           | 0.6959              | 0+6846                | 0.0701              | 0.6701            | 3.6601                | 3.6636  |
| -1-8333             | "J" J 1585"        | 3.6853          | J.684J             | "J.6849"                          | 0.6765    | 0.6715           | 0.6610              | 0.6541                | 0.6487              | 0.6453            | 0.6434                | 0.6426] |
| -1.7800             | 0.01660            | 0.6397          | 0.6389             | 0.6370                            | 0.0344    | 0.6316           | 0.6263              | 0.6225                | 0.6205              | 0.6148            | 0.6201                | 0.6211  |
| -1.7003             | 0.01730            | 0.5919          | 0.5412             | 0.5912                            | J.59J0    | 1.5432           | J4389/              | J-59J2                | 0.5422              | J.5938            | 345704                | 3.5993  |
| -1.7/00             | 0+01010            | 616970          | 0.9452             | 0.4469                            | 0.5007    | 0.5051           | 0.5246              | 0.5240                | 0.5327              | 0.5407            | 0.5480                | 0.5548  |
| -1.7337             | J. J1995           | J.4448          | J.4461             | 3,4533                            | 3.4558    | 3.4626           | J.4769              | J.4936                | 3.5329              | 0.5138            | 0.5235                | 0.5322  |
| -1.0000             | U. 02089           | 0.3975          | 0.3992             | 0.4043                            | 0.4118    | 0.4207           | U. 4395             | 6.4573                | 0.4731              | 0.4869            | 0.4990                | 0.5095  |
| -1.660J             | 0.02160            | 0.1524          | 0.3545             | 0.3605                            | 0.1694    | 0.3801           | 0.4028              | 0.4244                | 0.4435              | J. 46 J1          | J.4744                | J.4868  |
| [· -1.6433          | 3. 32291           | 21111           | 3.3125             | 0,3191                            | 0.3291    | 0.3411           | 0.3671              | 0.3922                | 0.4143              | 0.4335            | 0.4500                | 0+4642  |
| -1.0200             | 0+02377            | 0.2357          | 1.2481             | 0.2000                            | 1.2559    | 1.2404           | 1,2094              | 3.3312                | 1.3576              | 1.3813            | 1.4318                | 1.6196  |
| -1.9400             | 0.02630            | . 0.20.6        | 0.2060             | 0.2129                            | 0.2236    | 0.2371           | 0.2689              | 0.3010                | 0.3303              | 0.3560            | 0.370Z                | 0.3973  |
| -1+5000             | 0.02754            | 051750          | 0.1772             | Č. 1936                           | 0.1943    | 0.2077           | 0.2397              | 0.2710                | 0.3040              | 0.3314            | 0.3551                | 0.3756  |
| -1.59333            | 3. 32804           | 3.1497          | 1.1518             | J-1224                            | J.1078    | 2*1929           | J.2125              | 3-2403                | 0.2787              | 0.3075            | 4.3326                | 0.3543  |
| -1-5579             | 0-010-0            | 0.127>          | 0.1294             | 0+1120                            | 0+1442    | 0.1566           | 0.1875              | 0.2216                | 0.2545              | 0.2844            | 0.3106                | 0-3335  |
| -1-2000             | 1.14211            | 0.1001          | 0.1070             | 0.1147                            | 0+1233    | 0.1155           | 0.1043              | 0.1766                | J+2310              | J. 2622           | J.2689J               | 1.3131  |
| 1 -1.9000           | 0.03468            | 0.0746          | 0.0779             | 0.0420                            | 0.04889   | 0.0985           | 0.1247              | 0.1566                | J. 1695             | 0.2207            | 0.2490                | 0.2742  |
| -1.4400             | 0.03631            | 3. 364 3        | 1. 3652            | 3. 3088                           | 3. 3749   | 3. 1830          | 3.1378              | 0.1382                | 3.1734              | 3.2315            | 3.2331                | 3.2557  |
| - 1.4203            | 50820.0            | 0.0532          | 0.0242             | 0-0374                            | 0.0628    | 0.0705           | 0.0927              | 0-1214                | 0-1526              | 0.1833            | 0.21ZQ                | 0-2379  |
| -1.4000             | 0-03981            | 0.1439          | 0.0448             | 0,0476                            | 0.0524    | 0.0392           | 0.0794              | 0.1062                | 0.1.62              | 0.1663            | 0-1947                | 0.2207  |
| -1.0011             | 3. 39169           | 1,3363          | J. J368            | J+ J192                           | J+J434    | J. J499          | J. J070             | 0.0924                | 0+1-10              | 0.1303            | 0.1784                | 0+2044  |
| -1.3030             | 0.04571            | 0.0293          | 0.0300             | 0.0321                            | 0.0293    | 0.0411           | 0.0483              | 3.3692                | 3.3945              | 3.1216            | 3-1485                | 3.1739  |
| -1.3233             | 3.3.787            | 0.0195          | 0.0195             | 0.3211                            | 0.0239    | 0.0279           | 0.0405              | 0.0594                | 0.0130              | 0.1088            | 0.1349                | 0.1599  |
| -1.3000             | 0.05012            | 0.0151          | 0.0155             | 0.0109                            | 0.0193    | 0.0220           | 0.0339              | 0.0508                | 0+0736              | 0.0971            | 0.1222                | 0.1466  |
| · -1.20v0           | 0.49248            | 0.0119          | 0.0122             | 0.6134                            | 3.3155    | 3. 3185          | 3.3282              | 3.3433                | 3.36.3              | 3. 3463           | 3-1134                | 2-1340  |
| 1.2400              | 0.05490            | 0.0071          | 0.0096             | 0.0100                            | 0.0123    | 0.0149           | 0.0233              | 0.0367                | 0.0:50              | 0.0475            | 0.0994                | 0.1223  |
| -1.2233             | J. 36 124          | 3, 3, 355       | J. 3357            | 3.3364                            | 3.3376    | 3.3395           | 3. 31 57            | J. 5261               | 3.3413              | 3.0594            | D.0799                | 0.1010  |
| -1.2003             | 01130.0            | 0.0042          | 0.0043             | 0.0049                            | 0.0050    | 0.0075           | 0.0128              | 0.0218                | 0.0.52              | 0.0521            | 0.0714                | 0.0915  |
| -L+LOUJ"            | 0.06607            | 0.0031          | 0.0033             | 0.0038                            | 0.0046    | 0.0059           | 0.0103              | 0.0182                | 1010.0              | 3.3456            | 3. 3635               | 3. 3626 |
| -1-1633             | 7.20418            | 3.3324          | 3.0025             | 0.0029                            | 0.0015    | 0.0046           | 0.0383              | 0.0151                | 0.0256              | 0.0397            | 0+0564                | 0-0744  |
| -1,1400             | U.07245            |                 | 0.0019             | 0.0022                            | 0.0027    | 0.0036           | 0.0067              | 0.0125                | 0.0217              | 0+0345            | 0.0499                | 0.0009  |
| -1.1400             | 4.67944            | 0.0000          | 9-3010             | 0.0015                            | 0.0016    | 0.0021           | 0.0042              | 0.00044               | 0.0255              | ジャジビフタ<br>パッロノンガー | 14747<br>1477         | 0.0536  |
| -1.0803             | U.08318            | 0.0007          | ง้างอย์ส์          | 0.0009                            | 0.0012    | 0.0016           | 0.0031              | 0.0048                | 0.0130              | 0.0221            | 0.0340                | 0.0470  |
| 1. 3633             | J. J4713           | , <u>),</u> ))) | 3*7379             | 3.3337                            | 1.1316    | 3.3312           | 3.3326              | 0.3355                | 0.0108              | 0.0190            | 0.0297                | 0.0425  |
| -1+0460             | J.09121            | 0.0004          | 0-0004             | 0-0305                            | 0.0007,   | 0.0004           | 0.0020              | 0.0045                | 0-00-00             | 0.0165            | 0-0259 -              | 0.0377  |
| -1.0200             | 0,09551            | 0.0003          | 0.0003             | 0+0004                            | 0.0005    | 0.3337           | J.JJ16.             | J+3336                | J.0375<br>0.0545    | 3.3138            | 3.3228                | 3. 3334 |
| -6-9804             | 0.10472            | 0.0002          | 0.0002             | 0.0003                            | 0.0004    | 0.0005           | 0.0012              | 3.332%                | 3.3351              | 3.3399            | 1.0190                | 3. 3259 |
| -3.9033             | 0.10465            | 0.0001          | 0.0001             | 0.3302                            | 0.0002    | 0.0003           | 0.0007              | 0.0018                | 0+0042              | 0.0083            | 0.0146                | 0.0220  |
| -0.9400             | 0.11482            | 0.0001          | 0.0001             | 1000 0                            | 0.0001    | 0.0002           | 0+0005              | 0.0014                | 0.0034              | 0.0070            | 0,0125                | 0.0199  |
| -6.9203             | J-12323            | 3.3331          | 2.3331             | 3.3331                            | 1.3331    | 3.3332           | 3.3334              | 2.2311                | 3.3328              | 3.3358            | 3.3137                | 2-0174  |
| -C.9003             | 0.12590            | G. DCJL         | 0-0001             | 0.0001                            | 0.0001    | 0.0001           | 0.0003              | 0.0009                | 0.0022              | 0.0049 (          | 1.J092                | 0.0152  |
| -1.8611             | 1.138103           | 3.3133          | 1.111              | 3.3133                            | 1.3333    | 1.0001           | 0.0002              | 0.0005                | 0.0010              | 0.0041 0          | 1-0066                | 0.0114  |
| -0.0400             | Ũ.14455            | - 0.0000 ·      | J.0COU             | 0.0000                            | 0.0030    | 0.0000           | 0.0001              | 0.0004                | 0.0012              | 0.0028            | 0.0056                | 0.0097T |
| -0.8200             | U.15137            | C.CL00          | 4.0000             | 0.0000                            | 0.000.0   | 3.3335           | 3.3331              | 3.0000                | 3.3339              | 3.3323            | 3.3341                | 5. 3385 |
|                     | 0.15350            | 0.000           | 0.0000             | 0.0000                            | 0.0000    | 0.0000           | 0.0001              | 0.0002                | 0.0007              | 0.0019 (          | 0.0040                | 0.0073  |
|                     | U.16597            | 0.0000          | 0.0000             | 0.0000                            | 0+20000   | 0.0000           | 0.0001              | 0.0002                | 0.0006              | 0-0015 (          | 0.0033 (              | 3.0063  |
| - J+ (6 J J         | J+1/3/9<br>J+1/3/9 | 0.0000          | 3433333<br>0.(1101 | ال ( ال ال وال<br>ال ال ال ال وال | 0.00030L- | 1,1111           | i EEEEeE<br>hadddar | 0.0091 ·              | 1+11004             | -01010 ·          | )'UUSJ (<br> -  20 (  | 1 AAAA  |
| -0.7200             | 0.19056<br>0.19056 | 0.0000          | <b>0.000</b>       | 0.0100.0                          | 0.0000    | 0.0000<br>0.0000 | 0.00000             | 0.0001                | 3.0004              | 0.0010 0          | 1.0710                | 1000    |
| -3.7333             | 3.19954            | . 3. 3333       | 3.3333             | 3+3333                            | 0.0000    | 0.0000           | 0.0000              | p.0001                | 0.0602              | 0.0007            |                       | 5.0033  |
| -0.6800             | U+ 2C894           | 0.0000          | 0.0000             | 0.0000                            | 0.0000    | 0.0000           | 0.0000              | 0.0000 0              | 2-001-2             | 0.0005            |                       | 8100.0  |
|                     | 0.21874            | _ C.JUJU 1      | .0000              | 6.012                             | "         | 2.2222           | ຳ:ວາງງ              | 2.2273                | 14 14 16            | 3.3334            |                       | 1. 2324 |
| -0.0400             | U. 22910           | 0.0000          | 0.0000             | 0.0000                            | n*9000 (  | 0.0000           |                     | 0.00000               | 140C01              | 0.0003 (          | , <b>0009</b> (       | 05001   |
| -1.0200 0           | <u>1.25</u> 1.11   |                 | 7°7711             | 0.0000,0<br>1.111                 | 7.3334    | <u>, 1777</u>    | 3.3332              | 1 DDDD <sup>6</sup> 0 | 7*77'+F<br>1*AA1E ( | 3.3332 4          | ) 10001 (<br>140001 ( | 0.0014  |
|                     |                    | ******          |                    |                                   |           |                  |                     |                       |                     |                   |                       |         |

.

( .

()

: ,

•••

C:



×.

(

(



Contour Plots of Trophic State Probabilities vs. X and Coefficient of Variation of X for Normal and Lognormal Error Distributions ( )

probability of eutrophic classification. Tables 2.5-21 and 2.5-22 have been derived by interpolating Tables 2.5-17 and 2.5-18, for the normal and log-normal distributions, respectively. Rows in each table represent constant probability levels, ranging from 0.01 to 0.2, as given in column one. The second column indicates the values of X corresponding to the various probabilities, assuming that X can be specified exactly, i.e.  $CV_{\chi} = 0$ . The remaining columns indicate the percent reduction in the design values of X which would be necessary to account for the increasing uncertainty levels. The tables indicate that the cost of this uncertainty generally increases with decreasing design probabilities. Uncertainty costs are higher for the lognormal distribution.

### Effect of Uncertainty in X on Rational Design Values for Lognormally Distributed Errors

| 8      | h     |      | COEI | IIClent | OI VALIA | CION OF 1 | K Estima | Cê    |             |       |       |
|--------|-------|------|------|---------|----------|-----------|----------|-------|-------------|-------|-------|
| p(e) ~ | X     | .05  | .10  | .15     | .20      | .30       | .40      | .50   | <b>.</b> 60 | .70   | .80   |
| .01    | .0237 | -0.7 | -2.7 | -5.7    | -9.7     | -19.2     | -29.1    | -38.2 | -46.1       | -52,8 | -58.4 |
| .02    | .0264 | -0.6 | -2.4 | -5.2    | -8.8     | -17.4     | -26.3    | -34.7 | -42.1       | -48.4 | -53.7 |
| .05    | .0309 | -0.5 | -2.0 | -4.4    | -7.4     | -14.5     | -21.9    | -28.9 | -35.2       | -40.7 | -45.4 |
| . 10   | .0354 | -0.4 | -1,7 | -3.5    | -5.9     | -11.5     | -17.3    | -22,9 | -28.0       | -32.5 | -36.4 |
| .20    | .0414 | -0.3 | -1.0 | -2.3    | -3.8     | -7.2      | -10.8    | -14.3 | -17.5       | -20.3 | -22.8 |

Coefficient of Variation of X Estimate

### Table 2.5-22

### Effect of Uncertainty in X on Rational Design Values for Normally Distributed Errors

| _                 | 1.    |                   |      |      |      |       |       |       |       |       |       |  |  |
|-------------------|-------|-------------------|------|------|------|-------|-------|-------|-------|-------|-------|--|--|
| p(e) <sup>a</sup> | x     | .05               | .10  | .15  | .20  | . 30  | .40   | . 50  | .60   | .70   | .80   |  |  |
| .01               | .0237 | -0.6 <sup>C</sup> | -2.3 | -4.9 | -8,0 | -14.9 | -21.7 | -27.8 | -33.3 | -38.2 | -42.4 |  |  |
| .02               | .0264 | -0.5              | -2.1 | -4.4 | -7.3 | -13.7 | -20,1 | -26.1 | -31.5 | -36.2 | -40.4 |  |  |
| .05               | .0309 | -0.5              | -1.8 | -3.7 | -6.2 | -11.8 | -17.6 | -23.1 | -28.1 | -32.6 | -36.7 |  |  |
| .10               | .0354 | -0.4              | -1.4 | -3.0 | -5.0 | -9.7  | -14.7 | -19.6 | -24.1 | -28.3 | -32.2 |  |  |
| .20               | .0414 | -0.2              | -0.8 | -1.8 | -3.1 | -6.3  | -9.9  | -13.7 | -17.4 | -20.9 | -24.2 |  |  |

Coefficient of Variation of X Estimate

a - p(e) = probability of eutrophic classification

b - X<sub>o</sub> = design value of X, assuming it can be specified without error; X = L [QS (1 + .824T<sup>.454</sup>)]<sup>-.815</sup> c - percent change in X<sub>o</sub> required to account for given level of uncertainty

();

Quantifying the effects of uncertainty in the decision variable upon the rational design provides a means of rational- . izing the need for additional data acquisition. These effects can be expressed in economic terms, i.e. in terms of the cost associated with the additional nutrient source abatement required to account for the uncertainty. Typically, nutrient abatement entails substantial economic investment (e.g. pipelines and treatment plants) or losses (e.g. restrictions in land development). The marginal costs attributed to the uncertainty can be compared with the monitoring and analytical investments required in order to reduce that uncertainty, i.e. to obtain a better loading, residence time, or mean depth estimate. Because the abatement costs are generally on a much higher scale than monitoring costs, an overall optimization would probably justify additional source monitoring or other measurements, until the uncertainty in the decision variable, X , is low enough so that its effect on the design is low, or until the flat portions of the countour lines in Figure 2.5-22 are reached. In this region, the coefficient of variation of X is about 0.15 or less, and the effect of the assumed error distribution is small. This approach toward rationalizing the need for monitoring by expressing it in terms of the design costs of uncertainty has been explored by Meta Systems, Inc.<sup>61</sup> in a study for the Department of Transportation, which involved an examination of the

**X** •

ſ

potential effects of additional streamflow gauging sites upon the rational design of highway culverts.

Tables 2.5-21 and 2.5-22 indicate that if X can be defined such that its coefficient of variation is less than about 0.15, uncertainty costs would generally be less than 5% in the application of the discriminant model for loading allocations at the given design probability levels. Usually, most of the uncertainty in X would be attributed to the loading or average inlet concentration estimate. It would be of interest to consider the practical implications of this variability objective upon actual stream monitoring requirements.

To provide a partial basis for assessing the potential impacts of the Cross-Florida Barge Canal, Meta Systems<sup>62</sup> developed nutrient budgets on various reaches of two major drainage basins in northcentral Florida: the Oklawaha and the Withlacoochee. For each sampling station an attempt was made to quantify both the mean and the variance of the nutrient flux estimate on yearly and long-term-average time scales. The techniques, assumptions, and limitations of the estimates have been discussed previously (Section 2.3.2).

Figure 2.5-23 is a log-log plot of the coefficients of variation of the total phosphorus flux estimates against the numbers of concentration samples employed in deriving the estimates.

Pigure 2.5-23

( )

Ċ

**(**):

Coefficient of Variation of Mean Phosphorus Plux Estimate Against Sample Size for Various Stations in the Cross-Plorida Barge Canal Study<sup>62</sup>



Sample Size

2-182

())

(°,

Regardless of the particular technique employed to estimate the average flux from the continuous flow and grab-sample concentration data, the standard error of the mean varies inversely as the square root of the effective concentration sample size (Table 2.3-2). Because sampling frequencies were generally less than one per month, the effect of autocorrelation upon the effective sample size was assumed to be negligible (Figure 2.3-1). The lines in Figure 2.5-23 have slopes of -1/2 and, accordingly, are lines of constant variability in the sampled stations. The relative variability is defined as k , the product of the coefficient of variation of the mean flux estimate and the square root of the sample size. In the figure, three types of stations have been distinguished: springs, river stations, and impoundment stations. Their respective locations on the plot reflect generally increasing variability. In order to define the mean flux to within a given coefficient of variation, fewer samples are required for the spring stations than for the impoundment stations. Solid symbols were derived from long-term-average flux estimates and open symbols, from yearly estimates. Because of sampling error in the variance, as well as the mean, the yearly estimates. generally exhibit more scatter.

A k value of about 0.5 seems to characterize most of the open river and impoundment stations. At this level of variability, about 11 samples would yield a coefficient of variation of 0.15 in C

( )

the mean. A similar analysis of NES tributary or other stream data would be required in order to determine whether the variabilities and resulting sampling requirements based upon the Florida data are generally applicable. Of particular interest would be the relationships between flux variability and such factors as flow, flow variability, and drainage basin characteristics.

The coefficient of variation of a total loading estimate derived from independent flux estimates can be calculated from:

$$\overline{CV}_{T}^{2} = \sum_{i=1}^{M} \frac{\overline{CV}_{i}^{2} \ell_{i}^{2}}{\left(\sum_{i=1}^{M} \ell_{i}\right)^{2}}$$
(2.5-64)

If the individual fluxes are estimated with the same accuracy,  $\overline{cv}_{\star}$ :

$$\overline{CV}_{T}^{2} = \overline{CV}_{*}^{2} \frac{\Sigma \ell_{\pm}^{2}}{(\Sigma \ell_{\pm})^{2}}$$
(2.5-65)

Accordingly, the following bounds can be set upon the ratio of  $\overline{CV}_{T}$  to  $\overline{CV}_{*}$ :

( ·

(

$$\frac{1}{\sqrt{m}} \leq \frac{\overline{CV}_{T}}{\overline{CV}_{+}} \leq 1 \qquad (2.5-66)$$

The lower limit corresponds to the case in which all  $L_i$  are equal. This analysis indicates that if the estimates of the individual fluxes all satisfy the maximum variability requirement, then the total loading estimate will at least satisfy that requirement.

Assuming that an average variability of 0.5 is typical, a monthly sampling frequency appears to be adequate to define a loading or inlet concentration estimate for application of this particular model. This applies only to tributaries not dominated by point sources. This result may reflect the fact that the model itself was derived primarily from nutrient balances which had been estimated based upon monthly sampling. This re-emphasizes the need to appreciate that these results are both model- and dataspecific.

In view of these results, the interpretation of the probabilities generated directly from the discriminant model (i.e. p(e: X)in equation 2.5-54) warrants additional discussion. To some extent, the effects of uncertainty in X have already been included in this function, since errors were present in the X estimates of the data set employed to estimate the parameters of the classification model. A more accurate definition of p (e: X)would be: "the probability of a eutrophic classification, given an X estimate which had been generated using the same types and amounts of data and the same techniques employed in developing the phosphorus budgets in the original data set". This may account for the apparently small effect of additional uncertainty in X on the classification probabilities for coefficients of variation less than 0.15. In this range, the added uncertainty may be small compared to that already present. The results of the error analysis conducted on the phosphorus retention model (Section 2.5.3) indicated that coefficients of variation in the range of 0.10 may be typical of the inlet and outlet phosphorus concentration estimates in the data set.

The major question that remains is: what would happen to the shape of p(e: X) if better estimates of X were available? The separation of the trophic states along the X axis in Figure 2.5-19 may become more distinct and the design levels of X to achieve a given probability of eutrophic classification may become higher. Hence, uncertainty costs may already be present in the  $X_0$  values of Tables 2.5-21 and 2.5-22. These costs could only be quantified by refining the data or using other lakes with more accurate nutrient budget estimates. The net result is that the actual

()

probability of eutrophic classification for a given X value may be lower than that calculated according to the above scheme, if the model were applied to a lake with a loading estimate which is more accurate than those of the original data set. Hence, in such a case, the scheme would yield a conservative design value for phosphorus loading. Because of the potential economic costs associated with this conservatism, a thorough re-examination and refinement of the data and techniques employed in estimating nutrient budgets may be justified. This would reduce the errors in the independent variables employed in the discriminant analysis, and any resultant uncertainty costs. ()

The availability of phosphorus balance and trophic state data on a variety of lakes has permitted analysis and comparison of existing empirical models for predicting phosphorus retention coefficients and lake trophic states. A preliminary analysis of the data revealed significant stratification of all morphologic, hydrologic, and phosphorus-related variables considered across trophic states. A high degree of multicollinearity was apparent, indicating that cause-effect relationships would be difficult to establish and that application of the results could only be properly done to lakes which conformed to the multivariate distribution of the data analyzed.

A theoretically based model for phosphorus retention was proposed and estimated. The effective first-order decay coefficient of phosphorus was found to be roughly inversely proportional to the square root of lake hydraulic residence time and independent of depth. Effects of incomplete mixing, temperature variations, and the coupling of phosphorus dynamics with various, flushingdependent, physical, chemical, and biological processes were suggested as possible explanations for this behavior. The two parameters of the model were found to be stable across trophic states. The standard errors of estimate were substantially higher in eutrophic lakes, however. This model compared favorably with strictly empirical ones derived from linear regression and with the equation suggested by Kirchner and Dillon<sup>42</sup>.

An error analysis indicated that residual errors dominated over potential parameter and independent variable errors in predictions of the retention coefficient. On this basis, it was concluded that, for this type of model, the size of the data set was adequate for parameter estimation purposes. An attempt was made to disect the residual error into model and measurement errors. The former was shown to increase with hydraulic residence time; the latter, to decrease. The results indicated that coefficients of variations of about 0.1 may characterize measurement errors in the average inlet and outlet phosphorus concentration It was also shown that, as hydraulic residence time estimates. decreased, the accuracy of outlet concentration predictions became increasingly dependent upon the accuracy of the inlet concentration (or loading) estimate, and therefore, that refinement of the model would have little impact upon the total prediction error at low residence times. At high residence times, application of a more sophisticated model would be justified, providing that its model and parameter errors were lower. Evidence was presented which suggested that the factor limiting the accuracy of a chlorophyll prediction from phosphorus loading, residence time, and depth would be in the empirical chlorophylllake phosphorus relationship, and not in the lake phosphorus prediction. Thus, it was concluded that the accuracy of the phosphorus retention model was sufficient for application with state-of-the-art chlorophyll models.

\* model and measurement errors

Discriminant analysis was employed to compare existing models for trophic state prediction and to derive an optimal linear classification model for this data set. With the parameters of the models optimized, the average percentages of misclassified lakes were 32.3%, 28.3%, and 22.0% for Vollenweider's first model ( L versus Z )<sup>93</sup>, Dillon's model ( CO )<sup>18</sup>, and Vollenweider's second model ( L versus QS )<sup>95</sup>, respectively. Optimization of the parameters for this data set had little effect on the classification errors obtained when the originally-reported parameter values of the respective models were used. Evidence suggested that the performance of Dillon's model may have been hampered by systematic errors in the retention coefficient estimates of the National Eutrophication Survey Lakes.

Stepwise discriminant analysis selected the outlet concentration estimate based upon the phosphorus retention model discussed above and the total phosphorus loading as components of an optimal linear classification model for these lakes. After these variables had entered, no significant discrimination powers remained in the other variables. This model misclassified an average of 15.9% of the lakes, with most of the errors centered in the mesotrophic group. A principal component analysis was employed to reduce the classification model to a univariate one, which further reduced average classification

### 2.5.8 Summary (continued)

C

error to 14.5%. Misclassifications were discussed as possibly relating to errors in the dependent and independent variables and to unsteady-state conditions in the lakes.

In applications, it was shown that the discriminant model could be used to generate probabilities of trophic state classification as a function of a single variable, which could be computed from phosphorus loading, residence time, and mean depth estimates. Means of incorporating the effects of uncertainty in this "decision" variable on the classification probabilities and upon the rational loading allocation for a given lake were derived. In real applications, uncertainty in the loading estimates would generally contribute most to the uncertainty in the decision variable. The effect of this uncertainty upon the rational loading allocation which satisfies a given probability of eutrophic classification was demonstrated as a means of justifying additional monitoring effort. These effects were found to be small, provided that the coefficient of variation of the decision variable estimate was less than about 0.15. This, in turn, was shown to be possibly typical of a flux estimate derived from monthly tributary sampling, as employed by the NES. Because of possible effects of errors in the original data set, it was suggested that application of the classification model to lakes with nutrient budgets which had been more accurately determined than those in the original

data set would yield conservative design values for phosphorus loadings.

The empirical modelling approach taken in this section has been based upon five principle assumptions : (1) mass balance; (2) completely mixed conditions; (3) phosphorus limitation of lake ecosystems; (4) first-order kinetics for phosphorus sedimentation; (5) the possible influence of lake morphometric and hydrologic factors upon nutrient dynamics and trophic state response. The accuracy and value of the models developed have been shown to be limited by various features of the data base, including measurement or estimation errors in the independent and dependent variables, multicollinearity, nonsteady-state conditions existing in the lakes during sampling, and the subjectivity involved in the original lake classifications. In this context, it is difficult to identify model deficiencies due to effects of other controlling factors or aspects of system behavior which have been ignored, such as vertical stratification, sediment releases, or limitation by light of other nutrients. In short, data of this sort have not provided a very good basis for model discrimination. Because of these aspects, the general superiority of the developed models cannot be claimed. The primary emphases are upon the approaches taken and techniques employed in their development and evaluation and upon the strategies proposed for their application. In these regards, the underlying theme is in the development and use of error estimates.

### 2.5.9 Suggestions for Future Work

This section has illustrated some potentially useful approaches to empirical modelling. One general concept to consider is that these models are at most as good as the data used to derive them. Improvement in the quality of the data could improve model accuracy and, in applications, enhance design efficiencies. "Data errors" do not result from measurement errors alone. The selection of techniques for data reduction can also be critical. Use of an improved set of techniques could conceivably permit improvement in data quality without having to resort to additional measurements.

One potential area for technique improvement is in the development of lake nutrient budgets. The assumptions employed in estimating mass fluxes from grab-sample concentration data have been discussed in Section 2.3.2. The likely introduction of errors due to the methods employed by the National Eutrophication Survey in converting these budgets to normal hydrologic years has also been considered in Section 2.3.4. Essentially three independent and indirect means have been used to estimate the extent of data errors inherent in the lake nutrient budgets. Analysis of the phosphorus retention coefficient residuals in Section 2.5.3 indicated that coefficients of variation of about 0.10 may be typical of errors in the mean inlet and outlet phosphorus
## 2.5.9 <u>Suggestions for Future Work</u> (continued)

(

concentration estimates. The results of the Cross-Florida Barge Canal Study<sup>62</sup> indicated that monthly tributary sampling would yield annual flux estimates with coefficients of variation of about 0.15, for streams of the same general characteristics as those included in that study. It was also shown that the coefficient of variation of a total loading estimate derived from the sum of individual stream fluxes would generally be less than that of the individual flux estimates. Finally, trophic state probabilities generated by the discriminant model were found to be relatively insensitive to errors in the phosphorus loading estimates for coefficients of variation of about 0.15 or less. Direct analysis of concentration and flow data on a stream-by-stream basis would be required in order to develop prior, direct estimates of these errors. Such an analysis, combined with an improved strategy for hydrologic year corrections, could conceivably improve the quality of the nutrient budget data without resorting to additional measurements.

Another area for improvement is in the lake trophio state classification. In the above analysis, trophic state classifications had been originally given by somewhat subjective interpretations of lake response data. Errors at this stage of the analysis could conceivably be reduced by developing and employing a more objective classification scheme. The general

(

ĺ

approach would be to develop a continuous response index, similar to Shannon and Brezonik's "Trophic State Index"73. The index would include variables which could be related directly to established beneficial-use criteria. One approach would divide the lakes into discrete states corresponding to various beneficial uses, based upon the index and the established criteria. Discriminant analysis could then be used to separate the beneficial-use states based upon stimulus variables. Alternatively, the continuous index could be preserved and directly related to the stimulus variables using regression techniques, as employed by Shannon and Brezonik<sup>74</sup>. Either approach would provide a model which could be used as a basis for probabilistic designs, analogous to those discussed in Section 2.5.6. This work would provide a more objective model basis and reduce dependent variable (classification) errors. Again, these improvements would be possible without additional data acquisition.

 $\left( \right)$ 

Policy decisions in lake water quality management are reached through a complex process involving a variety of environmental, economic, political, and technologic factors. This chapter has focused upon some aspects of the technologic functions which can be employed to provide water quality impact assessments. In making such projections, the engineer or planner must make numerous decisions, often ranging from data acquisition to model selection and parameter estimation. The variety of methods discussed indicates that there is considerable latitude in this area and that there is no uniquely correct path from the definition of the analytical problem to its solution. One hopes that the policy recommendations reached by the planner are independent of his analytic decisions, i.e. of the particular set of data and methods employed. This is subject to the constraint that the monitoring and modelling are properly implemented in each case. The robustness of a particular analysis can be tested by comparing results derived from alternative, but apparently equally-valid approaches.

Some of the factors discussed in this chapter can be used as a basis for making analytic decisions. A proper balance between monitoring and modelling effort is a key objective in this regard. In an ideal situation, the costs of the total analytic effort might be balanced by the design costs associated with uncertainty in the analysis, as discussed in Section 2.5.7.

( )

More often, total analytic (monitoring and modelling) effort is constrained. In this case, a strategy would be to try to allocate monitoring and modelling resources so as to minimize projection uncertainty or to minimize expected regret associated with projection errors. Still more often, the analyst does not have control over monitoring, i.e., the data are given. In this case, the selection and/or development of models would again be guided by data availability.

Two general types of models have been discussed in this chapter as relating to analysis of lake water quality problems: source models and lake models. Source models can be used to relate various watershed or point-source characteristics to nutrient loadings. Some of these characteristics may be uncontrollable; others may be decision variables. If the various types of errors associated with the projections of these models were sufficiently small, they could be used as a substitute for direct monitoring data. As discussed in Section 2.3.2, the state-of-the-art of these models is such that direct estimation of nutrient fluxes via monitoring is generally preferable. The development of these models is essential to understanding and eventual control of non-point sources and their effects on lake water quality.

Lake models predict the response of lake water quality

#### 2.6 General Comments and Conclusions (continued)

( )

**(** )

to a given set of boundary conditions. They have been roughly classified as "empirical" and "theoretical" in this chapter. Some bases for model selection are discussed below.

The applicability of a given model may be governed in part by the management criteria of concern. The more empirical models tend to predict long-term responses and, as such, apply to long-term criteria. Some of the theoretical models can be used to predict system response under short-term, critical conditions, which are often of management concern. An example would be the response of hypolimnic dissolved oxygen levels to a critical set of meteorologic conditions. The short- and longterm responses are generally correlated, in that oligotrophic and eutrophic lakes each behave in characteristic ways. Thus, the more empirical models can be used to predict short-term responses indirectly, while theoretical system models can predict such directly.

Another basis for model selection is accuracy. The quantification of various error components involved in an empirical model projection has been demonstrated in Section 2.5.3. Such an analysis can be used straightforwardly to balance monitoring and modelling efforts. The relative complexities of the "theoretical" models would render a corresponding error analysis difficult, though not impossible.

One of the factors limiting such an analysis would probably be the accuracy of the estimates of the parameter covariance matrix. This matrix would be used along with the sensitivity coefficient matrix to estimate the effects of parametric uncertainty upon projection errors, according to the methods and equations discussed in Chapter 1. The covariance matrix could be estimated straightforwardly if the parameter estimates were derived directly from system response data, as in the case of the phosphorus retention model developed and analyzed in Sections 2.5.2 and 2.5.3. In more complex models, parameter estimates have generally been derived from independent laboratory studies. Estimation of parameter covariances in these cases would be more subjective. Even if the experiments provided parameter distributions, there would be uncertainty as to whether the experimental conditions adequately represented "field" conditions. Nevertheless, it would be possible to approximate the various error components of a projection derived from a theoretical model through a systematic assessment of parameter distributions and comparisons of estimated and observed system responses. This would permit straightforward comparisons of models with regard to accuracy.

A key result obtained from the analysis of the phosphorus retention coefficient residuals in Section 2.5.3 is that model errors would be detectable in only about one-third of the lakes,

()

}

if measurement strategies and data reduction techniques similar to those employed by the National Eutrophication Survey were used. Thus, for about two-thirds of the lakes in the analysis, there would be little benefit from employing a more accurate model than the one developed, providing that the objective were to predict outlet phosphorus concentration. Another key result of this analysis is that potential model errors increase with hydraulic residence time. This would indicate a corresponding increase in the importance of accuracy as a criterion for model selection. A similar error analysis would indicate whether this were true in the cases of other water quality components.

Another basis for model selection is generality. A model should be general enough to be valid for both present and future, or "design" states of a system under study. The former is necessary for calibration and verification, the latter for application purposes. The parameter estimates and, in some cases, the functional forms of empirical models are based directly upon system response data. These models are at most as general as the data used to derive them. Thus, the nature of the data base and characteristics and dependencies of model performance both inside and outside of the data base are keys to assessing empirical model generality. The more elaborate theoretical models are supposedly independent of particular system response data, because both their formulations and their

2

parameter estimates are usually derived independently. Thus, if the formulations and parameters estimates are correct, a theoretical model could be constructed to have high generality. However, as discussed in Section 2.4.2, this generality has not as yet been demonstrated. It has not been shown that a single model with a unique set of parameter values can be an equally valid representation of lakes at different trophic stages. The evidence presented in Section 2.4.2 suggested that this lack of generality might be reduced by increasing complexity, specifically to incorporate many species at each level of the aquatic food chain. However, this would exponentially increase parameter estimation difficulties which are already formidable.

The net result of these considerations is that theoretical models still need to be tailored to specific applications. This creates an empirical link between the "theoretical" model and system response data and reduces the supposed gap in generality between the "empirical" and "theoretical" types of models. Thus, while there are certainly some general aspects of the behavior of thase systems, analysis of lake water quality problems is still a relatively empirical exercise, entailing both monitoring and modelling of the systems and their boundary conditions.

### REFERENCES - CHAPTER 2

()

[

- 1. Aitchison, J. and J.A.C. Brown, <u>The Lognormal Distribution</u>, Cambridge University Press, 1963.
- Ashton, P.M. and R.C. Underwood, ed., <u>Non-Point Sources of Water</u> <u>Pollution</u>, Proceedings of a Southeastern Regional Conference, Virginia Water Resources Center, September 1975.
- Bannister, T.T., "Production Equations in Terms of Chlorophyll Concentration, Quantum Yield, and Upper Limit of Production", <u>Limnology and Oceanography</u>, Vol. 20, No. 3, pp. 342-364, May 1975.
- 4. Bayley, G.J. and J.M. Hammersly, "The Effective Number of Independent Observations in an Autocorrelated Time Series", <u>Journal of the</u> Royal Statistical Society, Vol. 8 (1B), pp. 184-197, 1946.
- 5. Benjamin, J.R. and C.A. Cornell, <u>Probability</u>, <u>Statistics</u>, <u>and</u> <u>Decision for Civil Engineers</u>, McGraw-Hill Book Co., 1970.
- Bierman, V.J., F.H. Verhoff, T.L. Poulson, and M.W. Tenney, "Multinutrient Dynamic Models of Algal Growth and Species Competition in Eutrophic Lakes" in <u>Modeling the Eutrophication Process</u>, Ann Arbor Science, 1974.
- Blanton, J.A., "Vertical Entrainment into the Epilimnion of Stratified Lakes", <u>Limnology and Oceanography</u>, Vol. 18, No. 5, pp. 697-704, September 1973.
- Bormann, F.H., G.E. Likens, T.G. Siciama, R.S. Pierce, and J.S. Eaton, "The Export of Nutrients and Recovery of Stable Conditions Following Deforestation of Hubbard Brook", <u>Ecological Monographs</u>, Vol. 44, No. 3, pp. 255-277, Summer 1974.
- 9. Cahill, T.H., P. Imperato, and F.H. Verhoff, "Evaluation of Phosphorus Dynamics in a Watershed", <u>Journal of the Environmental</u> <u>Engineering Division</u>, <u>American Society of Civil Engineers</u>, Vol. 100, No. EE2, pp. 439-454, April 1974.
- Chen, C.W. and G.T. Orlob, "Ecologic Simulation for Aquatic Environments", Final Report, OWRR C-2044, Water Resources Engineers, Walnut Creek, California, 1972.
- Cooke, G.E., J.N. Bhargara, M.R. McCemas, M.C. Wilson, and R.J. Heath, "Some Aspects of Phosphorus Dynamics of the Twin Lakes Watershed", in <u>Modeling the Eutrophication Process</u>, Ann Arbor Science, 1974.

()

- 12. Cooley, W.W. and P.R. Lohnes, <u>Multivariate Data Analysis</u>, John Wiley & Sons, New York, 1971.
- 13. Correll, D.L., J.W. Pierce, and M.A. Faust, "A Quantitative Study of the Nutrient, Sediments and Coliform Bacterial Constituents of Water Runoff from the Rhode River Watershed", in ref. 2, 1975.
- Dillon, P.J. and P.H. Rigler, "The Phosphrous-Chlorophyll Relationship in Lakes", <u>Limnology and Oceanography</u>, Vol. 19, No. 1, pp. 767-773, September 1974.
- Dillon, P.J., "A Critical Review of Vollenweider's Nutrient Budget Model and Other Related Models", <u>Water Resources Bulletin</u>, <u>American</u> <u>Water Works Association</u>, Vol. 10, No. 5, pp. 969-989, October 1974.
- 16. Dillon, P.J. and F.H. Rigler, "A Test of a Simple Nutrient Budget Model for Predicting the Phosphorus Concentration in Lake Water", <u>Journal of Fisheries Board Research of Canada</u>, Vol. 13, pp. 1771-1778, 1974.
- Dillon, P.J., "A Manual for Calculating the Capacity of a Lake for Development", Limnology & Toxicity Section, Water Res. Branch, Ontario Ministry of the Environment, October 1974.
- Dillon, P.J., "The Phosphorus Budget of Cameron Lake, Ontario: The Importance of Flushing Rate to the Degree of Butrophy of Lakes", <u>Limnology and Oceanography</u>, Vol. 20, No. 1, pp. 28-45, January 1975.
- Dillon, P.J. and W.B. Kirchner, "The Effects of Geology and Land Use on the Export of Phosphorus from Watersheds", <u>Water Research</u>, Vol. 9, pp. 135-148, 1975.
- DiToro, D.M., D.J. O'Connor, and R.V. Thomann, "A Dynamic Model of Phytoplankton in Natural Waters", Environmental Engineering and Science Program, Manhattan College, New York, June 1970.
- 21. DiToro, D.M., D.J. O'Connor, and R.V. Thomann, "A Dynamic Model of the Phytoplankton Population in the Sacramento-San Joaquin Delta", in <u>Nonequilibrium Systems in Natural Waters</u>, Adventures in Chemistry Series, No. 106, American Chemical Society, pp. 131-180, 1971.
- DiToro, D.M., D.J. O'Connor, R.V. Thomann, and J.R. Mancini, "Preliminary Phytoplankton-Zooplankton-Nutrient Model of Western Lake Erie", <u>Systems Analysis and Simulation in Ecology</u>, Vol. 3, Academic Press, 1973.

- DiToro, D.M., D.J. O'Connor, and R.V. Thomann, "Preliminary Model of Potomac Estuary Phytoplankton", <u>Journal of the Environmental</u> <u>Engineering Division</u>, <u>American Society of Civil Engineers</u>, Vol. 100, No. EE2, pp. 699-715, June 1974.
- DiToro, D.M., D.J. O'Connor, R.V. Thomann, and R.P. Winfield, "Mathematical Modeling of Phytoplankton in Lake Ontario 1. Model Development and Verification", EPA-660/3-75-005, March 1975.
- 25. DiToro, D.M. and W.F. Matystik, "Phytoplankton Biomass Model of Lake Huron and Saginaw Bay", in ref. 90, 1976.
- Dixon, W.J., ed., <u>BMDP Biomedical Computer Programs</u>, University of California Press, 1975.
- Enviro Control, Inc., "National Assessment of Trends in Water Quality", National Technical Information Service, Springfield, Virginia, June 1971.
- 28. Frink, C.R., "Septic Tanks as Non-Point Sources", presented at the Lakes Region Planning Commission Workshop on Lake Management and Related Non-Point Source Pollution Control, Weirs Beach, New Hampshire, May 1976.
- Grenney, W.J., D.A. Bella, and H.C. Curl, "A Theoretical Approach to Interspecific Competition in Phytoplankton Communities", <u>American</u> <u>Naturalist</u>, Vol. 107, No. 955, May-June 1974.
- 30. Grenney, W.J., D.A. Bella, and H.C. Curl, "Effects of Intracellular Nutrient Pools on Growth Dynamics of Phytoplankton", <u>Journal of the</u> <u>Water Pollution Control Federation</u>, Vol. 46, No. 7, pp. 1751-1760, July 1974.
- 31. Guy, H.P., "An Overview of Non-Point Water Pollution from the Urban-Suburban Area", in ref. 2, 1975.
- 32. Harms, L.L., J.N. Dornbush, and J.R. Andersen, "Physical and Chemical Quality of Agricultural Land Runoff", <u>Journal of the Water Pollution</u> Control Federation, Vol. 46, No. 11, November 1974.
- Hetling, L.J. and R.M. Sykes, "Sources of Nutrients in Canadarago Lake", <u>Journal of the Water Pollution Control Federation</u>, Vol. 45, No. 1, pp. 145-156, January 1973.
- Hobbie, J.E. and G.E. Likens, "Output of Phosphorus, Dissolved Organic Carbon, and Fine Particulate Carbon from Hubbard Brook Watershed", <u>Limnology and Oceanography</u>, Vol. 18, No. 4, pp. 734-742, July 1973.

1

-}

- 35. Hoel, P.G., <u>Introduction to Mathematical Statistics</u>, Fourth Edition, Wiley & Sons, 1971.
- 36. Holmes, R.W., "The Secchi Disc in Turbid Coastal Waters", Limnology and Oceanography, Vol. 15, No. 5, pp. 688-694, September 1970.
- 37. Huck, P.M. and G.J. Farguhar, "Water Quality Models Using the Box-Jenkins Method", <u>Journal of the Sanitary Engineering Division</u>, <u>American Society of Civil Engineers</u>, Vol. 100, No. EE3, pp. 735-753, June 1974.
- Hutchinson, G.E., "Eutrophication: The Scientific Background of a Contemporary Practical Problem", <u>American Scientist</u>, Vol. 61, pp. 269-279, May-June 1973.
- Imboden, D.M., "Phosphorus Model of Lake Eutrophication", <u>Limnology</u> and Oceanography, Vol. 19, No. 2, pp. 297-304, March 1974.
- Johnson, N.M., G.E. Likens, F.H. Bormann, D.W. Fisher, and R.S. Pierce, "A Working Model for the Variation in Stream Water Chemistry at the Hubbard Brook Experimental Forests, New Hampshire", <u>Water</u> Resources Research, Vol. 5, pp. 1353-1363, 1969.
- Keup, L.E., "Phosphorus in Flowing Waters", <u>Water Research</u>, Vol. 2., pp. 373-386, 1968.
- 42. Kirchner, W.B. and P.J. Dillon, "An Emprical Method of Estimating the Retention of Phosphorus in Lakes", <u>Water Resources Research</u>, Vol. 11, No. 1, pp. 182-183, February 1975.
- Kirchner, W.B., "An Examination of the Relationship Between Drainage Basin Morphology and the Export of Phosphorus", <u>Limnology and Oceano-</u> <u>graphy</u>, Vol. 20, No. 2, pp. 267-270, March 1975.
- 44. Kilhus, S.P., J.D. LaPerriere, and R.W. Bachman, "Nutrients and Algae in Some Central Iowa Streams", <u>Journal of the Water Pollution</u> <u>Control Federation</u>, Vol. 47, No. 7, pp. 1870-1874, July 1975.
- Larsen, D.P. and H.T. Mercier, "Lake Phosphorus Loading Graphs: An Alternate", US EPA, National Eutrophication Survey, Working Paper No. 174, 1975.
- 46. Lassiter, R.R. and D.K. Kearns, "Phytoplankton Population Changes and Nutrient Fluctuations in a Simple Aquatic Ecosystem Model" in <u>Modeling the Eutrophication Process</u>, Ann Arbor Science, 1974.
- 47. Lassiter, R.R., "Modeling Dynamics of Biological and Chemical Components of Aquatic Ecosystems", National Environmental Research

Center, Office of Research and Development, EPA-660/3-75-012, May 1975.

- 48. Lee, G.F., "Factors Affecting the Transfer of Materials Between Water and Sediments", Literature Review No. 1, University of Wisconsin, Water Resources Center, 50 pp., July 1970.
- Lehman, JJ., D.B. Bodkin, and G.E. Likens, "The Assumptions and Rationales of a Computer Model of Phytoplankton Population Dynamics", <u>Limnology and Oceanography</u>, Vol. 20, No. 3, pp. 342-264, May 1975.
- Lettenmaier, D.P., "Detection of Trends in Water Quality Data from Records with Dependent Observations", <u>Water Resources Research</u>, Vol. 12, No. 5, pp. 1037-1046, October 1976.
- 51. Likens, G.E., F.H. Bormann, N.M. Johnson, D.W. Fisher, and R.S. Pierce, "Effects of Forest Cutting and Herbicide Treatment on Nutrient Budgets in the Hubbard Brook Watershed-Ecosystem", <u>Eco-</u> <u>logical Monographs</u>, Vol. 40, No. 1, pp. 23-47, Winter 1970.
- 52. Likens. G.E. and R.H. Borman, "Linkages Between Terrestrial and Aquatic Ecosystems", <u>Bioscience</u>, Vol. 24, No. 8, pp. 447-456, August 1974.
- Lehr, R.C., "Characteristics and Comparative Magnitudes of Non-Point Sources", <u>Journal of the Water Pollution Control Federation</u>, Vol. 46, No. 8, pp. 1849-1872, August 1974.
- 54. Lorenzen, M.W. and R. Mitchell, "Theoretical Effects of Artificial Destratification in Impoundments", <u>Environmental Science and Tech-</u>nology, Vol. 7, No. 10, pp. 939-944, October 1973.
- 55. Lorenzen, M.W. and R. Mitchell, "An Evaluation of Artificial Destratification for Control of Algal Blooms", <u>Journal of the</u> American Water Works Association, Vol. 67, p. 373, 1975.
- 56. Lorenzen, M.W., D.J. Smith, and L.V. Kimmel, "A Long Term Phosphorus Model for Lakes: Application to Lake Washington", presented before the Division of Environmental Chemistry, American Chemical Society, Philadelphia, April 1975.
- 57. Matalas, M.C. and W.B. Langbein, "Information Content of the Mean", <u>Journal of Geophysical Research</u>, Vol. 67, No. 9; pp. 3441-3448, 1962.
- 58. Mauleg, K.W., D.P. Larsen, D.W. Schultz, and H.T. Mercier, "A Six-Year Water, Phosphorus, and Nitrogen Budget for Shagawa Lake",

()

Journal of Environmental Quality, Vol. 4, No. 2, pp. 236-242, 1975.

- McMichael, F.C. and J.S. Hunter, "Stochastic Modeling of Temperature and Flow in Rivers", <u>Water Resources Research</u>, Vol. 8, No. 1, pp. 87-98, February 1972.
- Mehta, B.M. and R.C. Ahlert, "Stochastic Variation of Water Quality of the Passaic River", <u>Water Resources Research</u>, Vol. 11, No. 2, pp. 300-308, April 1975.
- 61. Meta Systems Inc., "Assessment of National Small Rural Watersheds Program", prepared for the U.S. Department of Transporation, Federal Highway Administration, 1975.
- 62. Meta Systems Inc., "Cross-Florida Barge Canal Project Nutrient Budget", for U.S. Army Corps of Engineers, Jacksonville District, Contract 17-75-C-0077, March 1976.
- 63. O'Melia, C.R., "An Approach to the Modeling of Lakes", <u>Schweitz</u> <u>Zeitschrift fur Hydrologie</u>, Vol. 34, pp. 1-33, 1972.
- 64. Park, R.A. et al., "A Generalized Model for Simulating Lake Ecosystems", Eastern Deciduous Forest Biome, U.S. International Biological Program, Contribution No. 152, May 1974.
- Patten, B.C., "Need for Ecosystem Prospective in Eutrophication Modeling", in <u>Modeling the Eutrophication Process</u>, Ann Arbor Science, 1974.
- 66. Process Research, Inc., "The Impact of the Federal Water Pollution Control Act on the Charles River and Boston Harbor", Report to the National Commission on Water Quality, Washington, D.C., June 1975.
- Process Research, Inc., "Final Report on the Storrow Lagoon Demonstration Project", for the Metropolitan District Commission, Commonwealth of Massachusetts, 1975.
- 68. Reckow, K., Environmental Systems Program, Harvard University, unpublished results, 1975.
- Resource Planning Associates, "208 Modeling Approach Working Paper", prepared for Lakes Region Planning Commission, Meredith, New Hampshire, August 1975.
- 70. Scavia, D., B.J. Eadie, and A. Robertson, "An Ecological Model for the Great Lakes", in ref. 90, 1976.

( 1

ſ

- Schindler, D.W., "Eutrophication and Recovery in Experimental Lakes: Implications for Lake Management", <u>Science</u>, Vol. 184, pp. 897-899, May 1974.
- 72. Schofield, W.R. and R.G. Krutchkoff, "Deterministic Model of a Dynamic Eutrophic Estuary", <u>Journal of the Environmental Engineering</u> <u>Division</u>, <u>American Society of Civil Engineers</u>, Vol. 100, No. EE4, pp. 979-996, August 1974.
- 73. Shannon, E.E. and P.L. Brezonik, "Eutrophication Analysis: A Multivariate Approach", Journal of the Sanitary Engineering Division, <u>American Society of Civil Engineers</u>, Vol. 98, No. SAl, pp. 37-57, February 1972.
- 74. Shannon, E.E. and P.L. Breznok, "Relationships Between Lake Trophic State and Nitrogen and Phosphorus Loading Rates", <u>Environmental</u> <u>Science and Technology</u>, Vol. 6, No. 8, pp. 719-735, August 1972.
- 75. Sheldon, A.L., "A Quantitative Approach to the Classification of Inland Waters", in <u>Natural Environments</u>, Krutilla, J.V. (ed.), Johns Hopkins University Press, Baltimore, Maryland, pp. 205-261, 1972.
- 76. Smith, F.E., "Effects of Enrichment in Mathematical Models", in <u>Eutrophication: Causes, Consequences, Correctives</u>, National Academy of Sciences, Washington, 1969.
- 77. Snedecor, G.W. and W.G. Cochran, <u>Statistical Methods</u>, Sixth Edition, Iowa State University Press, 1967.
- 78. Snodgrass, W.J. and C.R. O'Melia, "A Predictive Model for Phosphorus in Lakes - Development and Testing", Ph.D. Thesis, Department of Environmental Sciences and Engineering, University of North Carolina, 1974.
- 79. Steele, R.J., "Environmental Control of Photosynthesis in the Sea", Limnology and Oceanography, Vol. 7, pp. 137-150, 1962.
- 80. Stumm, W. and J.A. Leckie, "Phosphate Exchange with Sediments; Its Role in the Productivity of Surface Waters", <u>Advances in Water</u> <u>Pollution Research</u>, San Francisco and Hawaii, Vol. 2, Paper III-26, 16 pp., 1970.
- Sykes, R.M., "The Prediction of Lacustrine Trophic Status", Civil Engineering Department, Ohio State University, unpublished manuscript, 1975.

)

- 82. Thomann, R.V., "Variability of Waste Treatment Plant Performance", <u>Journal of the Sanitary Engineering Division</u>, <u>American Society of</u> Civil Engineers, Vol. 96, No. SA3, pp. 819-837, June 1970.
- Thomann, R.V. and R.P. Winfield, "On the Verification of a Three-Dimensional Phytoplankton Model of Lake Ontario", in ref. 90, 1976.
- Tyler, J.E., "The Secchi Disc", <u>Limnology and Oceanography</u>, Vol. 13, No. 1, January 1968.
- 85. US EPA, "Proposed Criteria for Water Quality", Washington, D.C., 1973.
- US EPA, Pacific Northwest Environmental Research Laboratory, National Environmental Research Center, Corvallis, Oregon, National Eutrophication Survey, Working Paper Series, 1972-75.
- 87. US EPA, "The Relationships of Phosphorus and Nitrogen to the Trophic State of Northeast and North-Central Lakes and Reservoirs", National Eutrophication Survey, Working Paper No. 23, December 1974.
- US EPA, "An Approach to a Relative Trophic Index System for Classifying Lakes and Reservoirs", National Eutrophication Survey, Working Paper No. 24, 1974.
- 89. US EPA, "Relationships Between Drainage Area Characteristics and Non-Point-Source Nutrients in Streams", National Eutrophication Survey, Working Paper No. 25, Pacific Northwest Environment Research Lab, Corvallis, Oregon, August 1974.
- 90. US EPA, <u>Environmental Modeling and Simulation</u>, Proceedings of a Conference, Cincinnati, Ohio, April 1976.
- 91. Uttormark, P.D., J.D. Chapin, and K.M. Green, "Estimating Nutrient Loadings of Lakes from Non-Point Sources", Office of Research and Monitoring, US EPA, EPA-660/3-74-020, August 1974.
- 92. Uttormark, P.D. and J.P. Wall, "Lake Classification- A Trophic Characterization of Wisconsin Lakes", National Environmental Research Center, Office of Research and Development, US EPA, Ecological Research Series, EPA-660/3-75-003, June 1975.
- 93. Vollenweider, R.A., "Scientific Fundamentals of the Eutrophication of Lakes and Flowing Waters, with Particular Reference to Nitrogen and Phosphorus as Factors in Eutrophication", Organization of Economic Cooperation and Development, Paris, September 1970.
- Vollenweider, R.A., "Possibilities and Limits of Elementary Models Concerning the Budget of Substances in Lakes", <u>Arch. Hydrobiol.</u>, Vol. 66, No. 1, pp. 1-36, April 1969.

- 95. Vollenweider, R.A., "Input-Output Models", unpublished manuscript, 1973.
- Wang, W.C. and R.L. Evans, "Dynamics of Nutrient Concentration in the Illinois River", <u>Journal of the Water Pollution Control</u> <u>Federation</u>, Vol. 42, p. 2117, 1970.
- 97. Weibel, S.R., R.J. Anderson, and R.L. Woodward, "Urban Land Runoff as a Factor in Stream Pollution", <u>Journal of the Water Pollution</u> <u>Control Pederation</u>, Vol. 36, No. 7, pp. 914-924, 1964.
- 98. Welch, P.S., Limnology, First Edition, McGraw-Hill, New York and London, 1935.

3.0 EXPLORATORY ANALYSIS OF ONONDAGA LAKE WATER QUALITY DATA

## 3.1 Introduction

 $\bigcap$ 

### 3.1.1 General Description

Onondaga Lake is situated at the northern edge of the City of Syracuse in Central New York State (Pigure 3.1-1). It is a relatively small lake with a surface area of 11.7 km<sup>2</sup> and with average and maximum depths of 12 and 20.5 meters, respectively. The lake's 600 km<sup>2</sup> drainage basin (Pigure 3.1-2) contains approximately 325,000 people and essentially all of Onondaga County's 140 industries<sup>24</sup>. It discharges to the Seneca River at an average rate of 19.3 cubic meters per second (W.Y. 1968-74) and has a mean hydraulic residence time of 84 days.

The lake's primary use has long been as a receptacle for municipal and industrial wastes. About 14% of the average outflow volume consists of effluent from Onondaga County's primary sewage treatment facility. A steel mill and a solvay process plant discharge thermal and chemical wastes directly into the lake or its immediate tributaries. Land use in the watershed is about 15% urban and 34% agricultural. About 35% of the urban land is served by combined sewers. Thus, the impact of diffuse sources may be appreciable, although such effects are currently screened by the predominance of point sources.

The location of the lake renders it a potentially valuable



·\_\_\_)



(\_\_\_)

# Figure 3.1-2

2

# Onondaga Lake/Watershed



# 3.1.1 General Description (continued)

recreational asset to Onondaga County's residents. However, like most urban lakes and rivers, severe water quality problems have precluded realization of this potential. In addition to waste assimilation, the current beneficial uses of the lake are restricted to industry (cooling and process waters), transportation (the lake discharges to the New York State barge canal system), and some recreation (boating and picnicking along nothern shores). Fishing has been outlawed since 1970. The future use of Onondaga Lake for contact recreation is one of Onondaga County's major environmental goals and an issue of great public concern.

# 3.1.2 Water Quality Issues

The quality of Onondaga Lake water is unique<sup>24</sup>. A mean dissolved solids concentration of 3500 ppm and chloride content of 1700 ppm, corresponding roughly to 10% seawater, render it one of the most saline lakes in the Northeast. Most of the salinity is due to waste discharges from the Allied Chemical Company solvay process plant, although there is historical and geological evidence indicating that natural sources of salinity may also be appreciable. This aspect of the lake's water quality is thought to have important physical, chemical, and biological effects on structural and functional aspects of the Onondaga Lake ecosystem. Physically, during stratified periods,

#### 3.1.2 Water Quality Issues (continued)

density gradients are markedly enhanced, and vertical mixing hindered, as a result of vertical salinity gradients. The lake does undergo two periods of vertical circulation yearly, normal for lakes of this depth and at this latitude. Calcium and sodium, the major cations in the lake, are present in about equal mass concentrations. Chemical precipitation of calcium salts is marked, the lake being supersaturated with respect to calcium carbonate and calcium phosphate most of the year<sup>24</sup>. Biologically, salinity is thought to be an important factor in determining and regulating the dominant species of biota at various trophic levels, ranging from phytoplankton to fish.

Onondaga Lake has been classified as extremely eutrophic. The abundance of primary nutrients has permitted the development of algal blooms to objectionable levels. In 1968-69, the average concentration of total phosphorus in the epilimnion was  $2.34 \text{ mg/l}^{24}$ , over one hundred times that considered "dangerous" from a eutrophication standpoint<sup>39</sup>. During the same period, time- and volume-averaged dissolved oxygen was 30% of saturation and the hypolimnion of the lake was anaerobic between early March and late October.

Potential toxic effects due to extreme levels of ammonia, chromium, copper, and mercury have also been of concern. The problem of mercury contamination in Onondaga Lake fish has been sufficiently prominent to merit a paragraph of discussion in the 1975 Report of the National

## 3.1.2 Water Quality Issues (continued)

Council on Environmental Quality<sup>5</sup>. The geometric mean fecal coliform density was twice the New York State bathing standard in the epilimnion during Onondaga County's baseline monitoring period, 1968-69, reflecting primarily both dry- and wet-weather overflows from the Syracuse combined sewer system<sup>24</sup>. The existence of extensive sludge deposits in the vicinity of the sewage treatment plant outfall site has raised questions about the feasibility of substantially improving lake water quality by reducing pollution sources alone<sup>24</sup>.

## 3.1.3 Plans for Pollution Abatement

The plan for abating the pollution of Onondaga Lake centers on the construction of a facility to provide tertiary treatment of Onondaga County's sewage. The facility, currently under construction and scheduled for completion in 1980, is expected to remove 95 percent of the influent BOD and 90 percent of the influent total phosphorus. The unique aspect of this plan is that the  $0.29 \text{ m}^3$ /sec waste bed overflow from the Allied Chemical plant will be used as a precipitating agent to effect phosphorus removal from the municipal waste stream (design flow 3.78 m<sup>3</sup>/sec). The industrial stream is currently the major source of the lake's salinity, which is not expected to be influenced by this plan. Allied's waste beds will be used in turn for disposal of digested sludge.

## 3.1.3 Plans for Pollution Abatement (continued)

Reduction of the municipal sewage phosphorus loadings alone is not generally anticipated to bring about a reversal of the eutrophic conditions in the lake<sup>38</sup>. Although diffuse nutrient sources (combined sewer overflows and urban and agricultural runoff) have not been accurately quantified, they are considered potentially of sufficient magnitude to support abundant algal activity in the lake after reduction of municipal sewage loadings. The EPA<sup>38</sup> has concluded: "The best reason for instituting phosphorus removal at the MSSTP is protection of Lake Ontario". Considerably more measurements and/or analysis may be required in order to quantify nonpoint phosphorus discharges and determine the degree of control necessary to significantly abate Onondaga Lake's eutrophic symptoms.

Another aspect of the basin plan is that the existing submerged sewage outfall will be abandoned during dry weather periods and a shoreline, surface outfall constructed. Since the combined municipal/ industrial effluent is expected to have a chloride concentration in excess of 7000 g/m<sup>3</sup>, it will be much more dense than ambient lake waters (averaging 1700 g/m<sup>3</sup>). The surface outfall has been proposed "to provide maximum mixing of the plume with ambient water and some degree of bio-degradation before sinking into the hypolimnion", and "to prevent the tendency to create a permanent stratification within the lake"<sup>21</sup>.

**(** :

### 3.1.3 Plans for Pollution Abatement (continued)

The surface outfall plan has been criticized with regard to possible water quality and aesthetic implications<sup>28,38</sup>. The saline industrial stream currently discharges into Ninemile Creek, the lake's largest tributary, prior to entering the lake. This creek affords about twice as much dilution of the saline stream as will the municipal effluent. The physics of the situation dictate that the greater the difference in density between the effluent and ambient waters, the greater will be the tendency for stratification. Implementation of the plan may result in enhanced vertical density gradients in lake and concommitant water quality impact. These considerations suggest, minimally, that the design of the outfall to permit maximum dilution of the effluent may be critical. In addition, the mixing of the calcium-rich effluent with ambient lake waters is expected to induce calcium carbonate precipitation and result in a visible white plume, which may be aesthetically objectionable<sup>38</sup>.

With the exception of salinity, industrial discharges have reportedly been brought under control<sup>38</sup>. In 1974, Crucible Inc., a steel mill formerly contributing substantial quantities of oil and grease, chromium, copper, and iron, installed treatment facilities to limit these discharges. In 1970, Allied Chemical was legally forced to reduce mercury discharges from its chlor-alkalai plant from 10 kg/ day to 0.4 kg/day<sup>38</sup>. The problem of mercury contamination in Onondaga Lake fish has persisted, however, according to an ongoing NYDEC fish

# 3.1.3 Plans for Pollution Abatement (continued)

( )

monitoring program<sup>38,5</sup>. No continuing program for monitoring marcury discharges or ambient lake levels is known to the author.

The EPA<sup>37</sup> has recognized that the solvay process in general presents special waste treatment problems. Existing plants have been granted special exemption from future zero discharge limitations. Accordingly, there is no plan to reduce the dissolved solids content of Onondaga Lake. Allied Chemical has indicated that the expense of eliminating the saline discharge would entail closing of its solvay plant and thus a substantial economic loss to Onondaga County. The plan to utilize the process waste for phosphorus removal from the municipal waste stream represents a rather unique, compromising alternative.

Another aspect of the pollution abatement program concerns the control of combined sewage discharges during dry and wet weather. Onondaga Country has undertaken an extensive sewer maintenance program and most of the dry-weather overflows into tributary creeks are reported to have been eliminated<sup>14</sup>. An EPA Demonstration Grant has been awarded to O'Brien and Gere to investigate high-rate processes for bacteria, ammonia, and suspended solids removal from sewer overflows. As yet, no final solution to the problem has been adopted, although "general abatement" of the overflows is required by the year 1985.

## 3.1.4 Scope of Chapter

 $\bigcirc$ 

Following is a preliminary analysis of available data relevant to the water quality of Onondaga Lake and its tributaries. This work is intended to provide a basis for more detailed modelling studies aimed at policy evaluation. The levels and variations of major water quality components are summarized and displayed along temporal and spatial dimensions in order to elucidate trends and seasonal variations and to determine the extents of vertical and horizontal mixing in the lake. Considerable effort has been spent in investigating or developing methods of summarizing and displaying the data. Information obtained from the U.S. Geological survey is used to develop a hydrologic balance, which is combined with tributary, point source, and lake water quality data to formulate mass balances on major water quality components. Trends in these mass balances and in ambient lake concentrations, and shifts in phytoplankton populations observed over the study period are interpreted mechanistically. The mass balances are used in combination with some of the models discussed and developed in Chapter 2 in a preliminary assessment of the potential for controlling the eutrophication problem through point and non-point source nutrient controls or through lake mixing. The results of this analysis, coupled with considerations of the water quality issues and management policies outlined above, serve as a basis for the modelling efforts described in Chapter 4.

### 3.2 Data Base

É

## 3.2.1 Lake and Tributary Water Quality

In 1967, the Onondaga County Department of Public Works obtained an FWQA Research and Development Grant (11060FAE) to determine the feasibility of a cooperative municipal industrial approach to the wastewater problems. As part of the contract, in 1968, Onondaga County initiated an extensive monitoring program of Onondaga Lake and its tributaries. The survey, performed by O'Brien & Gere Engineers, Inc., was the first systematic and comprehensive investigation of water quality in the area. The stated objectives of the initial phase of the program (1968-69) were to<sup>24</sup>:

- "1. Ascertain the present trophic status of the lake;
- 2. Evaluate the impact of engineering programs;
- Provide baseline data for ongoing evaluations;
- Establish a program for continuous monitoring of the lake."

The baseline study emphasized that the most important objective was "to evaluate the impact of various pollution abatement programs with respect to the best uses of the lake."

# 3.2.1 Lake and Tributary Water Quality (continued)

This program has been continued on a more or less uniform basis through 1976<sup>20</sup>. The ongoing study has been "designed to gauge the reaction of the lake to various changes in activity surrounding the lake." The monitoring report for the 1970 survey also suggests that it is "on the basis of accumulated information that a predictive model can be developed and the water quality resulting from planned pollution abatement programs can be determined." Thus, the aims of this program appear to be to monitor ambient trends and to provide basic data for development and calibration of predictive planning models. The essential features of this program are discussed below.

Table 3.2-1 presents the sampling frequencies for each station and year. The stations can be located on Figure 3.1-1. Lake station 2 was abandoned in 1971, after enough data had been obtained to indicate that the concentration differences between stations 1 and 2 were relatively small, and thus that the lake was well-mixed on horizontal planes. This assumption is examined further in Section 3.4.1. A total of ten influent tributary stations have been established, seven of which have been sampled regularly on a biweekly basis. The stations on the major tributaries (Onondaga Creek, Ninemile Creek, Ley Creek, and Harbor Brook) have been located at USGS gauge sites to facilitate flow measurements. Grab samples of the three major point-source discharges (Allied Chemical, Crucible, and Metro STP) have also been taken regularly.

# Table 3.2-1

٠

,

# Onondaga Lake Survey

# Numbers of Samples by Station and Year

| ·     | Station                           | Depth                 | 1968      | 1969 | 1970 | 1971 | 1972 | 1973 | 1974 | 1975 | Total |
|-------|-----------------------------------|-----------------------|-----------|------|------|------|------|------|------|------|-------|
|       | Station 1                         | 0~18 m<br>0 3m interv | 23<br>als | 40   | 25   | 11   | 19   | 17   | 16   | 15   | 168   |
| 3     | Station 2                         | 0-16 m<br>8 3m interv | 21<br>als | 41   | 11   | 0    | 0    | 0    | 0    | 0    | 73    |
|       | Loy Creek                         | surface               | 0         | 15   | 24   | 24   | 24   | 23   | 25   | 23   | 158   |
|       | Onondaga Creek                    | surface               | 0         | 17   | 24   | 24   | 25   | 24   | 25   | 23   | 168   |
|       | llarbor Brook                     | surface               | 0         | 16   | 22   | 24   | 25   | 24   | 25   | 23   | 159   |
|       | Ninemile Creek                    | surface               | 0         | 16   | 24   | 24   | 25   | 24   | 25   | 23   | . 161 |
|       | Bloody Drook                      | surface               | 0         | 15   | 22   | 0    | 0    | 0    | 0    | 0    | 37    |
| ន     | Sawmill Creek                     | surface               | 0         | ?    | 18   | 0    | Q    | Û    | 0    | 0    | 18    |
| UTARI | Sanitary Landfill <sup>a</sup>    | surface               | Û         | 7    | 16   | 0    | 0    | 0    | 0    | 0    | 16    |
|       | Allied Chemical                   | effluent              | O         | 15   | 24   | 24   | 23   | 24   | 25   | 23   | 158   |
|       | Crucible                          | effluent              | 0         | 19   | 24   | 24   | 24   | 24   | 25   | 23   | 163   |
| 1     | Metro STP                         | effluent              | 0         | 15   | 24   | 24   | 25   | 24   | 25   | 23   | 160   |
| ļ     | Outlet (in) <sup>b</sup>          | surface               | Û         | 0    | Û    | 0    | 21   | 24   | 23   | 23   | 91    |
| 1     | Outlet (out) <sup>b</sup>         | bottom                | 0         | 0    | 0    | 0    | 24   | 24   | 23   | 23   | 94    |
|       | Outlot (unspecified) <sup>b</sup> | 2 ·                   | 0         | D    | 20   | 22   | 0    | 0    | 0    | 0    | 4Z    |

٠

a - discharges into Ley Creek

b - see text

.

3-13

· ·

.

٠

-----

#### 3.2.1 Lake and Tributary Water Quality (continued)

Grab samples of tributaries and point sources may not have provided a sufficient basis for accurately estimating average mass fluxes of materials into the lake. In the cases of tributaries, periodic grab sampling has not taken into account the effects of intervening storms. This is a particular problem in streams which are subject to combined sewer overflows, as are Onondaga Creek and Harbor Brook. Periodic grab sampling of point sources at these frequencies has not accounted for probable weekly and diurnal periodicities in these sources, which are coupled with domestic and industrial activities.

Sampling the outlet of Onondaga Lake is difficult due to peculiar hydrodynamic conditions. Seneca River waters have been frequently observed to flow into the lake at the surface, while the dense lake waters flow out in the lower layers. The Seneca River is part of the New York State barge canal system and its levels are regulated. The pattern of bilaminar flows at the outlet may become more distinct in dry seasons when river levels may intermittently exceed lake levels. In 1970 and 1971, the outlet was sampled regularly, apparently without distinction as to flow direction. In 1972-75, attempts were made to sample both outflowing (bottom) and inflowing (surface) waters at the outlet station. This flow pattern renders the outlet data difficult to interpret.

#### 3.2.1 Lake and Tributary Water Quality (continued)

Table 3.2-2 summarizes the components monitored in the lake and tributary surveys. A total of thirty chemical species have been included. The study has provided meteorological data on wind speed, direction, and air temperature over the lake. In latter years, an increasing emphasis has been placed on bacteriological measurements. The phytoplankton population has been characterized by identifying, counting, and sizing the dominant species. "Biomass" represents an aggregate estimate of the total phytoplankton cell volume per volume of sample, as calculated from algal counts and sizes. In the latter years, measurements of chlorophyll have been used to characterize total algal biomass. Zooplankton have been counted and identified as to general class (rotifers, copepods, cladocerans) throughout the survey period and as to species in the baseline study, which also included sampling and identification of fish and benthic organisms.

Two comments are in order relative to the selection of components. The omission of mercury from the list prior to 1975 seems unfortunate, considering that mercury toxicity has been a predominant water quality issue. Phosphate detergent legislation in 1971 and future tertiary treatment are measures aimed at controlling eutrophication by limiting phoephorus supplies. However, the monitoring program has not included any measure of organic phosphorus in the sources or lake. "Total inorganic phosphorus" has been substituted

Table 3.2-2

. ..

. .

. --

Onondaga Lake Survey Components Wonitored

|                                                | enoisese<br>enoisese |                     | Onondega<br>Component | Lake Sur<br>s Monito | Vey<br>Tod |                       |        |              |
|------------------------------------------------|----------------------|---------------------|-----------------------|----------------------|------------|-----------------------|--------|--------------|
| ວິນອຸກອງ                                       | ;2832 5:86           |                     |                       | YEA                  | £          |                       |        |              |
| ±07                                            | Xs.1<br>I T E İ      | (1968-69<br>1968-69 | 1970                  | 1971                 | 2261       | 1975                  | 1974   | 1975         |
| Temperature                                    | **                   | ×,                  | ж,                    | **                   | **         | **                    | × •    | **           |
| Alkalinity<br>pH                               | **                   | ××                  | < ×                   | < ×                  | < ×        | * >< 3                | ~~>    | (×)          |
| Total Inorganic Carbon<br>Total Organic Carbon | × ×                  | •                   |                       | м                    | ×          | * *                   | ~ ~ :  | - <b>-</b>   |
| 8                                              | ч н<br>              | **                  | ×                     | ×                    | ×          | ×                     | **     | < ×          |
| Dissolved Oxygen                               |                      |                     | ; <b>H</b> 5          | **                   | нн         | жн                    | ××     | ××           |
| BUD-<br>Conduct tvity                          | ~×<br>~×             | <×                  | <b>e 5</b> -e :       |                      |            |                       |        | **           |
| Total Inorganic P                              | **                   | × ×                 | ж ж                   | * *                  | ××         | м ж                   | * *    | ~ ×          |
| Drganic N                                      |                      |                     |                       | **                   | × >        | × >                   | 24 )×  | ж н          |
| N-52X                                          | * ×                  | **                  | ~ ×                   | ×                    | < ×:       | <ul> <li>H</li> </ul> | с эс : |              |
| N-192                                          | **                   | **                  | **                    | жж                   | ××         | * *                   | M N    | ĸĸ           |
| sulfate                                        | < ><<br>< >          | < ×                 | (×                    | ~                    | ×          |                       | **     | *            |
| Sulfide                                        |                      | ××                  | <b></b>               | X                    | ×          | ××                    | ~ ~    |              |
| Hagnesium                                      | <                    |                     | : >< 1                |                      | **         |                       | **     | **           |
| · Sodium                                       | × >                  | ××                  | ж ж                   | ни                   | < ×        | ~ ~                   | ~ ×    | < ×          |
| forassium<br>Iron                              | < \<br>< \           | < ×                 | ~                     |                      | -          |                       | ж      | *            |
| Copper                                         | × ×<br>× ×           | ж ж<br>             | мн                    | ××                   | * *        | * *                   | ~ ×    | < <b>x</b> . |
| zinc                                           | •                    | ×                   | ж                     |                      |            |                       |        | ×            |
| Mercury<br>Manganese                           | **                   | ×                   | ×                     |                      |            |                       |        | :            |
| Fluoride<br>Constat Calida                     | ×                    | ××                  | ми                    |                      |            |                       |        |              |
| Dissolved Solids                               | < ><                 | .×                  |                       | :                    | :          | 1                     | ;      | ,            |
| Transparency                                   | ×                    | ו                   | <b>,</b>              | ~                    | -          | ~                     |        | ł            |
| Wind Speed                                     | × ×                  | к <b>н</b>          | < ><                  | <b>~</b> ×           | ~          | ~                     |        |              |
| Air Temperature                                | X                    | ×                   | ×                     |                      | -          |                       | ,      | ,            |
| Total Collforms                                | ×*                   | ,                   | ,                     |                      |            | ×                     | < >    | ~ >:         |
| Fecal Streptococci                             | < ×                  | •                   | r i                   |                      |            |                       |        | +            |
| Blomass                                        | ×                    | ×                   | X                     | ×                    | * *        | **                    | Я      | ×            |
| Chlorophy11<br>; Zooplankton                   | < ><                 | ×                   | ×                     | ×                    |            | :×:                   |        | *            |
| Phytoplankton                                  | ×                    | ×                   |                       |                      |            |                       |        | -            |

٠

### 3.2.1 Lake and Tributary Water Quality (continued)

for the "total phosphorus" measurements usually included in lake surveys. In the lake, the organic fraction of total phosphorus is likely to become dominant, should phosphorus supplies become limiting to algal growth. Hydrolysis of organic phosphorus discharged into the lake may serve as an additional nutrient source which would not be accounted for in a source control program based upon inorganic phosphorus data alone.

## 3.2.2 Metro Sewage Treatment Plant Operating Data

To supplement the O'Brien and Gere grab-sample data on the Metro STP discharge, the Onondaga County Department of Drainage and Sanitation has provided copies of treatment plant operating records for the years 1968 to  $1974^{23}$ . These reports have provided information on minimum, mean, and maximum daily flows and on daily mean concentrations of suspended solids, settleable solids, pH, and BOD<sub>5</sub>, based upon 24-hour flow-weighted, composite samples of the plant influent and effluent for 1972-72, and upon 8-hour composite samples for 1968-71. Sampling was generally not done on weekends, so that mean lake loadings derived directly from these data may be somewhat biased.

Analysis of 24-hour composite samples for total inorganic phosphorus was done at approximately weekly intervals from 1972 to 1974. Comparison of these data with the O'Brien and Gere morning grab-sample

#### 3.2.2 Metro Sewage Treatment Plant Operating Data (continued)

data on the plant effluent over the eame period has provided a means of calibrating the latter measurements and correcting them for timeof-day. Both surveys have used Onondaga County's laboratories for analysis and have sampled on weekdays, so that differences in the phosphorus data can likely be attributed to diurnal variations. The data from each survey have been aggregated and averaged at one-month intervals and compared on a month-by-month basis. The mean ratio of the composite to morning grab-sample concentration was found to be 1.43 (standard deviation 0.28). This is in general agreement with the analyses of Mauleg et al.<sup>17</sup> and Shannon<sup>29</sup>, both of whom have observed marked diurnal periodicities in total phosphorus concentrations of sewage, with concentrations tending to be lowest in mid-morning hours.

The Metro STP operating records have provided information of the incidence of plant upsets due to operating or equipment problems, potentially of use in interpreting residuals calculated in future modelling studies. The records also note the incidence of major storms requiring bypass of sewage around the plant directly into the lake. No information on the quantity or quality of the bypassed flow is available, however.

## 3.2.3 Hydrology

ş

The US Geological Survey has provided tributary flow data for the development of a hydrologic balance. Table 3.2-3 summarizes the availability of streamflow data for a total of six gauging stations. Only two stations (Onondaga Creek at Dorwin Ave. and Ninemile Creek at Camillus) were operated for the entire period of interest (1968-74). In order to estimate missing flow data, relationships have been developed relating flows at stations with missing data to the flows at either of the stations with a complete record. The regression models employed and parameter estimates are given in Table 3.2-4.

No evaporation data for the region and period have been located. To estimate evaporation from the lake surface, regional monthly-average values given by Meyer<sup>18</sup> have been used. While year-to-year climatological variations would be expected to influence the evaporation rates, the overall hydrologic balance of the lake is not sensitive to direct precipitation or evaporation, since the lake's mean surface overflow rate is about 52 meters/year, as compared with mean precipitation and evaporation rates of 1.13 meters/year and 0.67 meters/year, respectively.

The above information has been used to formulate a seven-year hydrologic balance on Onondaga Lake on a monthly basis (Table 3.2-5, Figure 3.2-1). Since no continuous gauging at the outlet has been done, the total outflow for each month has been estimated by summing
# Table 3.2-3

# Availability of Flow Data for Onondaga Lake Hydrologic Balance

|         | USGS     |                                 | Drainage                |      |      | . Wa | ter Ye   | ar   |      |          |
|---------|----------|---------------------------------|-------------------------|------|------|------|----------|------|------|----------|
| Station | Code     | Location                        | Area (km <sup>2</sup> ) | 1968 | 1969 | 1970 | 1971     | 1972 | 1973 | 1974     |
| A       | 04239000 | Onondaga Creek<br>9 Dorwin Ave. | 229.2                   | X    | x    | X    | x        | X    | χ.   | x        |
| B       | 04240010 | Onondaga Creek<br>@ Spencer St. | 282.3                   |      |      |      | <b>X</b> | X    | x    | X        |
| С       | 04240105 | · Harbor Brook<br>@ Hiawatha    | 29.3                    |      |      |      | X        | ́х   | X    | <b>X</b> |
| D       | 04240120 | Ley Creek<br>@ Syracuse         | 77.4                    |      |      |      |          |      | x    | X        |
| E       | 04240200 | Ninemile Creek<br>@ Camillus    | 218.3                   | x    | X    | X    | x        | x    | X    | x        |
| F       | 04240300 | Ninemile Creek<br>@ Lakeland    | 297.8                   |      |      |      | x        | Х.   | x    |          |

X = flow data available

1

# Table 3.2-4

# Regression Models Used to Estimate Missing Flow Observations

General Model :

| <sup>log</sup> 10 <sup>Q</sup> i | $= a + b \log_{10} Q_j + e$                          |
|----------------------------------|------------------------------------------------------|
| Q <sub>i</sub> ,Q <sub>j</sub>   | = mean monthly flow for stations i and j $(m^3/sec)$ |
| a, b                             | = regression parameters                              |

|   |       | -    |  |
|---|-------|------|--|
| = | error | term |  |

e

÷.

| Station i                         | Station j                                               | Number of<br>Observations | a    | b    | r <sup>2</sup> | SEE          |
|-----------------------------------|---------------------------------------------------------|---------------------------|------|------|----------------|--------------|
| Onondaga Creek<br>@Spencer Street | Onondaga Creck<br>@Dorwin Avenue                        | 49                        | .253 | .834 | .986           | •037         |
| Ninemile Creek<br>@Lakeland       | Ninemile Creek<br>@Cemilus                              | 34                        | .422 | .773 | .974           | <b>.</b> 046 |
| Ley Creek<br>ØSyracuse            | Onondaga Creek<br>@Dorwin Avenue                        | 21                        | 174  | •555 | .773           | .108         |
| Harbor Brook<br>@Hiawatha Blvd.   | Ninemile Creek<br>@Camilus<br>+ .91 m /sec <sup>a</sup> | 48                        | 863  | .827 | <b>.</b> 832   | .098         |

a - .91 m<sup>3</sup>/sec added to Camilus flow to account for upstream drinking water diversion from Otisco Lake and to linearize relationship between Harbor Brook and.Ninemile Creek flows.

| Table | з. | 2-5 |
|-------|----|-----|
|-------|----|-----|

### Onondaga Lake Water Balance

Mean Flows in Cubic Meters Per Second

| Drainage        |                    |       | Water Year |       |       |       |       |       |       |
|-----------------|--------------------|-------|------------|-------|-------|-------|-------|-------|-------|
| •               | (km <sup>2</sup> ) | 1968  | 1969       | 1970  | 1971  | 1972  | 1973  | 1974  | Mean  |
| Onondaga Creek  | 282.3              | 5.42  | 6.67       | 4.82  | 5.72  | 6.56  | 6.51  | 5.86  | 5.94  |
| Harbor Brook    | 29.3               | .47   | .55        | .43   | .45   | .55   | .60   | .56   | .52   |
| Ley Creek       | 77.4               | 1.37  | 1.56       | 1.26  | 1.37  | 1,51  | 1.73  | 1.44  | 1.46  |
| Ninemile Creck  | 297.8              | 6.96  | 8.35       | 6.18  | 8.06  | 7.62  | 8.79  | 6.95  | 7.57  |
| Ungauged Inflow | 39.6               | .82   | .99        | .73   | .90   | .94   | 1.02  | .85   | .89   |
| Metro STP       | -                  | 2.10  | 2.19       | 2,60  | 2.68  | 2.95  | 2.99  | 3.49  | 2.72  |
| Precipitation   | 11.7               | . 38  | .35        | . 35  | .40   | .46   | .50   | .52   | .42   |
| Evaporation     | ~                  | .25   | .25        | .25   | .25   | .25   | .25   | .25   | .25   |
| Outflow         | 738.1              | 17.28 | 20.42      | 16.14 | 19.33 | 20.34 | 21.88 | 19.42 | 19.26 |

3-22

.



# Figure 3.2-1

Monthly Water Balance

#### 3.2.3 Hydrology (continued)

ġ.

Ć

()

the inputs and subtracting evaporation. The ungauged portion of the inflow has been estimated by multiplying the total gauged tributary inflow by the ratio of ungauged to total gauged drainage areas (0.058).

Annual outflow ranged from 16.1 cubic meters per second in 1970 to 21.9 cubic meters per second in water year 1972, corresponding to a range of 100 to 74 days in mean hydraulic residence time. The Metro STP discharge averaged 14% of the mean outflow rate. The two industrial discharges, Crucible and Allied Chemical, representing exchange flows of 0.29 and 3.62 m<sup>3</sup>/sec, respectively, did not influence the net water balance of the lake.

#### 3.2.4 Meteorology

Meteorologic data has been obtained from the Local Climatological Data publication of the US Department of Commerce<sup>36</sup> for the weather station at Hancock Airport, Syracuse, located about four miles northeast of the lake. Table 3.2-6 summarizes relevant data on precipitation, sumshine, cloud cover, and wind speed and direction over the period 1968-74.

# Table 3.2-6

Summary of Meteorologic Data 1968-74

Hancock Airport, Syracuse, New York

| Calendar Year | Precipitation (cm) | Percent of Possible Sunshine | Average Wind Speed (km/hr) | Resultant Wind Speed (km/hr) | Resultant Wind Direction <sup>a</sup> | Maximum Wind Speed (km/hr) |
|---------------|--------------------|------------------------------|----------------------------|------------------------------|---------------------------------------|----------------------------|
| 1968          | 112.3              | 45.3                         | 15,1                       | 5.1                          | 26.0                                  | 69.4                       |
| 1969          | 81.4               | 46.8                         | 14.2                       | 4.2                          | 26.0                                  | 75.8                       |
| 1970          | 97.1               | 48.8                         | 15.0                       | 3.7                          | 25.2                                  | 69.4                       |
| 1971          | 99.5               | 48.5                         | 15.4                       | 4.7                          | 25.4                                  | 75.8                       |
| 1972          | 140.7              | 42.3                         | 14.7                       | 3.7                          | 25.5                                  | 83.9                       |
| 1973          | 133.7              | 44.1                         | 15.0                       | 4.2                          | 25.3                                  | 72.6                       |
| 1974          | 127.6              | 42.4                         | 15.5                       | 4.8                          | 25.8                                  | 96.8                       |

a - 0 = 36 = True North

 $(\cdot)$ 

()

3.3 Methods

<u>( )</u>

The techniques discussed below have been applied to summarize and display the water quality data along spatial and temporal dimensions. This work is essentially exploratory in nature. It is intended to elucidate essential temporal and spatial variations and associations within data. Results are contained in Appendices A through D.

#### 3.3.1 t-Tests for Horizontal Mixing

Paired t-tests (Appendix A) have been made in order to test the statistical significance of the differences in the mean concentrations of various components between stations 1 and 2 (see Figure 3.1-1). Tests have been applied to data from each of the seven depths sampled and to epilimnion (0., 3 meters, 6 meters) and hypolimnion (12 meters, 15 meters, 18 meters) averages. The objective of this work is to determine the degree of horizontal mixing in the lake at each vertical level.

For each vertical level and component, all pairs of concentration measurements taken simultaneously have been selected from the general data matrix. For each station, the mean, standard deviation, and standard error of the mean have been calculated. A paired t-statistic has been computed according to:

## 3.3.1 t-Tests for Horizontal Mixing (continued)

$$t_{k} = \frac{\overline{c}_{1k} - \overline{c}_{2k}}{\sqrt{\sum_{i=1}^{N_{k}} (c_{1ki} - c_{2ki})^{2} / N_{k} (N_{k} - 1)}}$$
(3.3-1)

where,

£.

C

C

$$C_{jki} \approx \text{concentration of station } j, \text{depth } k,$$
  
and date i  
 $N_k \approx \text{number of paired observations at depth } k$   
 $\overline{C}_{jk} \approx \text{mean concentration at station } j$  and  
depth  $k$   
 $t_k \approx \text{t-statistic at depth } k$ 

The t-statistic, with  $N_k$  - 1 degrees of freedom, has been used to test the significance of the differences in means between the two stations. Since station 2 was abandoned after 1970, the data for these tests spans the 1968-70 period, a total of 88 sampling dates.

## 3.3.2 GRID Displays

GRID<sup>30</sup>, a program from the Harvard Laboratory for Computer Graphics and Spatial Analysis, has been used to display water quality

### 3.3.2 GRID Displays (continued)

()

( );

variations as a function of time and depth (Appendix B). Variations in concentration have been represented at six levels using symbols of increasing visual intensities to indicate increasing concentrations. For each water quality component, the original matrix of concentration observations (~ 168 dates × 7 depths) has been transformed into a 96  $\times$  19 matrix by aggregating and averaging the data at monthly intervals and interpolating linearly with depth to provide estimates at one meter intervals from 0 to 18 meters. Accordingly, a one-month by one-meter GRID cell size has been used. Months with missing profiles have been estimated by interpolating between adjacent months. Profiles missing for more than three successive months have been left blank in the displays. The six concentration levels have been selected at equal intervals for components which appeared to be distributed normally and at geometrically increasing intervals for components which tended toward lognormal distributions. This type of display has been found to be particularly helpful in identifying seasonal stratification patterns.

## 3.3.3 Line Plots

Seasonal and long-term changes in concentration have been illustrated by plotting raw and smoothed volume-averaged concentrations as a function of time (Appendix C). For each sample profile,

#### 3.3.3 Line Plots (continued)

the volume-averaged concentration has been computed as:

$$\overline{C} = \frac{0^{\int_{z_{\text{max}}}^{z_{\text{max}}} C_{z} a_{z} dz}{0^{\int_{z_{\text{max}}}^{z_{\text{max}}} a_{z} dz}}$$
(3.3-2)

where,

 $(\cdot)$ 

CI

$$a_z = 1$$
 ake surface area of depth z  
C\_ = observed concentration at depth z

The above integration has been performed using the trapezoidal rule, with surface areas derived from Figure 3.3-1.

In order to estimate seasonal and long-term components of the variations in each time series, a variety of "smoothing" techniques have been tried. One of the most flexible and generally satisfactory methods found was that of "cubic splines", due to Reinsch<sup>27,41</sup>. This technique consists of fitting piecewise cubic polynomials to each set of four successive observations in the time series. The coefficients of the polynomials are selected to maximize the smoothness of the total curve subject to a specified constraint on the mean squared deviation of the fitted curve from the observations. The measure of smoothness employed is minus the integral of the squared second

# ONONDAGA LAKE

# Surface Area and Volume Vs. Depth



C

(

# 3.3.3 Line Plots (continued)

۰.

derivative. Mathematically, the problem is formulated as:

maximize: 
$$F \simeq -f [S''(t)]^2 dt$$
 (3.3-3)  
tmin

subject to:  

$$\begin{array}{ccc}
N & S(t_{1}) - y_{1} \\
\Sigma & (-----) \\
\underline{i=1} & \delta y_{1}
\end{array}$$

$$\begin{array}{ccc}
Z \\
\underline{i=1} & \delta y_{1}
\end{array}$$

where,

X

(

C

$$S(t) = a_i + b_i(t-t_i) + c_i(t-t_i)^2 + d_i(t-t_i)^3$$
,  $t_i \le t \le t_{i+1}$ 

$$a_i, b_i, c_i, d_i = coefficients for the time interval,
 $t_i \leq t \leq t_{i+1}$   
 $S(t) = "smoothed" value at time t$   
 $y_i = observed value of time t_i$   
 $\delta y_i = estimate of the standard deviation
of observation  $y_i$   
 $N = total number of observations$$$$

Coefficients at the beginning and ends of the time series are subject to the following constraints:

( ];

$$c_1 = c_N = d_N = 0$$
 (3.3-5)

The computations of the coefficients have been performed using a subroutine in the IBM SLMATH Program Library<sup>12</sup>. This method selects the smoothest possible curve that fits the data to within the specified accuracy. By varying the specified levels of the  $\delta y_i$ , varying degrees of smoothness can be achieved. After some experimentation, most satisfactory results have been reached by first transforming the data to render variance independent of level, which, in many cases required log transformation. In such cases, a small number has been added to each observation before taking logarithms to allow inclusion of zeroes. The standard deviation of the transformed data has been computed and each element in the

 $\delta y_1$  vector set equal to a specified fraction, f, of the computed standard deviation. In most cases f level of 1.0 has been found to give smooth curves which seem to follow trend, while an f level of 0.5 has been found to give curves which follow seasonal variation, as well as trend. In a series with a large trend, such as phosphorus, there was little difference in the smoothed curves at f levels of 1.0 and 0.5, and f levels of 0.5 and 0.25 have been used to depict trend and seasonal variations, respectively. Plotting the raw data as points and each of the two smoothed curves on the same graph has

#### 3.3.3 Line Plots (continued)

been found to give a reasonable display of the various components of the time series variation. This technique is similar to that employed by Spirtas and Levin<sup>31</sup> in examining air pollution data.

The smoothing techniques described above are a form of exploratory data analysis and have been employed only as an objective means of summarizing the data and elucidating essential aspects. Since second moments have not been considered, no notion of the statistical significance of trend or seasonal components has been implied. The method has been employed as a partial means of suggesting time series model formulations, for which parameters might be estimated and tested for significance. The cubic spline method selected for smoothing compares favorably with other techniques, such as moving averages with various weighting schemes. The latter methods generally require equally spaced observations and give smooth curves whose variation tends to lag behind variations in the data, unless optimal filtering coefficients are used.

#### 3.3.4 Mass Balances

**(** - 1

Lake and tributary quality data have been combined with hydrologic data in order to formulate mass balances on major water quality components for water years 1970 through 1974 (Appendix D). As discussed in Section 3.2, only approximate estimates of mass fluxes can be derived,

#### 3.3.4 <u>Mass Balances</u> (continued)

£

because only grab-sample data was available for the tributaries and point sources. Estimation of confidence limits for the mass fluxes according to the methods outlined in Section 2.3.2, would be essential to an engineering evaluation of potential impacts of source control measures, but has been excluded from this preliminary analysis.

For each tributary or point source, continuous stream flow and grab-sample concentration data have been used to estimate average mass fluxes at one-month intervals. The algorithm employed to perform these calculations is based upon the following equality:

$$\overline{W}(t_2 - t_1) = \int_{t_1}^{t_2} W_t dt = \int_{t_1}^{t_2} q_t c_t dt = \int_{t_1}^{v_2} c_t dv \quad (3.3-6)$$

$$v_2 - v_1 = \int_{t_1}^{t_2} q_t dt$$
 (3.3-7)

where,

 W = mean mass flux between time t<sub>1</sub> and time t<sub>2</sub> (mass/time)
 w<sub>t</sub> = instantaneous flux at time t (mass/time)
 q<sub>t</sub> = instantaneous flow at time t (vol/time)

#### 3.3.4 Mass Balances (continued)

(

C

v = integrated flow (vol)

According to the scheme employed, for a given set of concentration and flow data, the coefficients of variation of sample concentration and sample flux are computed and compared. If concentration is found to be more variable than flux, calculations are performed on the flux versus time scale, otherwise they are performed on the concentration versus volume scale. In either case, the series is first smoothed by fitting piecewise linear least-squares segments to each set of three successive observations. This smoothing operation is performed three times. The series is then integrated numerically using the trapezoidal rule and the integral is evaluated at monthly intervals by linear interpolation. Generally, concentration has been found to be less variable than flux in the tributary and lake data and more variable than flux in the point source data.

The fundamental equation employed in the mass balance is:

Input - Output = Change in + Accumulation (3.3-8) Storage

# 3.3.4 Mass Balances (continued)

ď

 $( \cdot )$ 

(:)

For each component and month, an estimate of the total input flux has been derived by summing the individual tributary and point source contributions. Ungauged inputs have been estimated based upon gauged tributary inputs and drainage area. Due to the difficulty in sampling the lake outlet (see Section 3.2.1), concentration data from the epilimnion (average of 0, 3, and 6 meter samples at station 1) have been used to estimate outflow. The change in storage term in equation (8) has been determined from the volume-averaged concentration data at station 1. Finally, the accumulation term has been calculated by difference.

The mass balances have been plotted in cumulative form in Appendix D. For each component, the cumulative input plot depicts the integrals of the mass flux from each source. The cumulative balance plot depicts the integrals of the total input, output, change in storage, and accumulation terms. Changes in the slopes of these curves reflect flux rate changes. 3-36

÷,

#### 3.4 Results and Discussion

#### 3.4.1 Horizontal Mixing

**(**);

()

Table 3.4-1 summarizes the results of the tests for significant differences in means between stations 1 and 2 in the eplimnion and hypolimnion. The oalculations are tabulated in detail in Appendix A. In Table 3.4-1, components have been grouped according to whether the tests indicated that the mean for station 1 was higher, that there was no significant difference between the stations, or that the mean for station 2 was higher. In addition, the differences in means have been computed as percents of the grand means.

Figure 3.1-1 shows that station 1 is located in the south basin of the lake, closer to Syracuse and the Metro STP discharge. Results of these tests indicate that the eplimnion at station 1 is generally a more reduced environment, significantly higher in  $BOD_5$ , ammonia, nitrogen, and organic nitrogen, and lower in dissolved oxygen and  $NO_3$ -N than the epilimnion at station 2. Higher rates of primary productivity at station 2 are indicated by higher levels of blomass, dissolved oxygen, and pH and lower levels of alkalinity. The slightly higher epilimion mean temperature at station 1 may reflect the industrial cooling water discharges at the south end of the lake.

Differences between stations in the hypolimnion are less easily interpreted. Alkalinity and pH show the same pattern as in the epilimnion. Higher levels of  $BOD_5$  at station 2 may reflect sedimentation

| Table | 3.4 | <b>-1</b> |
|-------|-----|-----------|
|-------|-----|-----------|

# Summary of Results of E-Tests for Horizontal Mixing

|          | STATION 1 <sup>e</sup><br>SIGNIFICANTLY <sup>®</sup><br>HIGHER<br>COMPONENT Δ <sup>b</sup> |           | NO SIGNI<br>DIFFER<br>BETWEEN S<br>Conpone | FICANT <sup>®</sup><br>ENCB<br>TATIONS<br>NT Δ <sup>b</sup> | STATION 2 <sup>0</sup><br>SIGNIFICANTLY <sup>D</sup><br>HIGHER<br>COMPONENT A <sup>D</sup> |                  |  |
|----------|--------------------------------------------------------------------------------------------|-----------|--------------------------------------------|-------------------------------------------------------------|--------------------------------------------------------------------------------------------|------------------|--|
|          | BOD <sub>5</sub>                                                                           | (28.581)  | N02-N                                      | ( .76%)                                                     | Dissolved O                                                                                | rygen (-19.52%)  |  |
|          | Temperature                                                                                | ( .731)   | Calcium                                    | (60%)                                                       | Biomass                                                                                    | ( -2.69%)        |  |
|          | Alkalinity                                                                                 | ( 2.634)  | 510 <sub>2</sub>                           | ( 3.29%)                                                    | NO3-N                                                                                      | ( -8.50%)        |  |
| υ        | Ammonia N                                                                                  | (9.104)   | Chloride                                   | (14%)                                                       | н                                                                                          | ( -1.27%)        |  |
| NION     | Organic N                                                                                  | (11.85%)  | Ortho-P                                    | (10%)                                                       | }                                                                                          |                  |  |
| JLIH     |                                                                                            |           | TIP                                        | ( -7.95%)                                                   |                                                                                            |                  |  |
| 8        |                                                                                            |           | Iron                                       | ( 2.70%)                                                    |                                                                                            |                  |  |
|          |                                                                                            |           | Chromium                                   | (62%)                                                       | l .                                                                                        |                  |  |
|          |                                                                                            |           | Copper                                     | ( 4.16%)                                                    |                                                                                            |                  |  |
| -        | Alkalinity                                                                                 | ( 1.92\$) | Dissolved<br>Oxygen                        | ( 4.24%)                                                    | BODS                                                                                       | (-11.64%)        |  |
|          | sio <sub>2</sub>                                                                           | ( 6.651)  | Biomass                                    | ( 8.04%)                                                    | рH                                                                                         | (53%)            |  |
|          | Chloride (                                                                                 | (2.103)   | Temporature                                | ( -2.31%)                                                   | í                                                                                          |                  |  |
|          |                                                                                            |           | N02-N                                      | ( 20.9 \)                                                   |                                                                                            |                  |  |
|          |                                                                                            |           | <sup>N0</sup> 3-N                          | (-20.914)                                                   |                                                                                            |                  |  |
| ۳y<br>يو |                                                                                            | 1         | Calcium                                    | ( .201)                                                     |                                                                                            |                  |  |
| DINH     |                                                                                            |           | Annonia-N                                  | ( 4.27%)                                                    |                                                                                            |                  |  |
| DCI      |                                                                                            | ľ         | Organic-N                                  | ( 6.36%)                                                    | 1                                                                                          |                  |  |
| Ħ        |                                                                                            |           | Ortho-P                                    | ( 1.84%)                                                    | [                                                                                          | ľ                |  |
|          |                                                                                            |           | TIP                                        | ( 9.93%)                                                    |                                                                                            |                  |  |
|          |                                                                                            |           | Iron                                       | ( -5.52%)                                                   |                                                                                            |                  |  |
|          |                                                                                            | Í         | Chromium                                   | ( -5.37%)                                                   | Į                                                                                          |                  |  |
|          |                                                                                            |           | Copper                                     | ( 7.05%)                                                    | [                                                                                          | •                |  |
|          | a-95% Confide                                                                              | nco level | ·                                          | d-hypolimi                                                  | on = mean of depth                                                                         | s 12m, 15m, 18m. |  |
|          | b-6 =                                                                                      | x 1001    |                                            | e-for stati                                                 | on locations, see                                                                          | Figuro 3,14      |  |

$$b = b = \frac{x_1 + x_2}{x_1 + x_2} - x$$

C

(

Ċ

c-epilimnion - mean of depths 0m, 3m, 6m;

...

#### 3.4.1 Horizontal Mixing (continued)

(

( :)

of sewage or algal solids. Growth and sedimentation of diatoms may account for slightly higher SiO<sub>2</sub> concentrations at both levels of station 1, although only hypolimnion differences are statistically significant. The slightly higher chloride levels in the hypolimnion of station 1 are unexplained.

Because prevailing winds are from the west and the lake's orientation is northwest to southeast (see Figure 3.1-1), one would expect considerable backmixing in the system in the direction from station 2 to station 1, the opposite direction from net advective outflow. These results indicate that there is some "plug flow" behavior of the system, or that, on the average, backmixing rates do not totally dominate over advection and reaction rates. The "reactions" involved include oxidation or settling of BOD, nitrification, and algal nutrient uptake and growth. Despite the fact that statistically significant differences can be discerned, the percentage differences between the stations are generally small, especially in comparison to vertical differences. It is also possible that some of the differences between the stations can be accounted for by exchange flows with Seneca River waters at the outlet.

#### 3.4.2 Vertical Mixing

(

( :

GRID displays in Appendix B illustrate marked vertical stratification patterns in the lake during summer months. Table 3.4-2 indicates the components tending to concentrate in the epilimnion, those exhibiting no marked vertical stratification, and those tending to concentrate in the hypolimnion. The pattern generally reflects the effects of atmospheric exchange and primary production in the epilimnion, settling of algal and non-algal particulates, and subsequent decomposition and release soluble compounds in the reduced conditions of the hypolimnion.

The tendencies of  $BOD_5$ ,  $SiO_2$ , ortho-P, TIP, and  $NH_3$ -N to concentrate in the hypolimnion are consistent with the above mechanisms. Higher levels of  $NO_2$ -N and  $NO_3$ -N in the epilimnion reflect higher dissolved oxygen levels there and possible denitrification reactions in the hypolimnion. The pH and alkalinity patterns are consistent with the following reactions<sup>33</sup>:

(alkalinity) (calcite)  

$$Ca^{++} + 2 HCO_3^{-} + CaCO_3^{+} + CO_2^{-} + H_2O$$
 (3.4-1)

$$co_2 + H_2 O_{\ddagger} + H_2 CO_3 + H^{\dagger} + HCO_3^{-}$$
 (3.4-2)

# Table 3.4-2

C

:

(

()

# Summary of Mid-Summer Vertical Stratification Patterns in Onondaga Lake, by Component

| Components Concentrating<br>in Epilimnion | Components Showing<br>No Stratification | Components Concentrating<br>in Hypolimnion |
|-------------------------------------------|-----------------------------------------|--------------------------------------------|
| Dissolved Oxygen                          | Chromium                                | BOD5                                       |
| Organic N                                 | Copper                                  | 510 <sub>2</sub>                           |
| <sup>NO</sup> 2 <sup>-N</sup>             | Iron                                    | TIP                                        |
| NO3-N                                     |                                         | Ortho-P                                    |
| рН                                        |                                         | NH3-N                                      |
| Temperature                               |                                         | Alkalinity                                 |
| Chlorophy11                               |                                         | Calcium                                    |
|                                           |                                         | Chloride                                   |
|                                           |                                         |                                            |

#### 3.4.2 Vertical Mixing (continued)

C

 $\cap$ 

( ;

Carbon dioxide is removed from the epilimnion as a result of photosynthesis and added to the hypolimnion as a result of respiration and organic matter decay. Removal of  $CO_2$  from the epilimnion would tend to drive reaction (1) to the right, resulting in increased removal of alkalinity through calcium carbonate precipitation.  $CO_2$ removal would also drive reaction (2) to the left, leading to a decrease in hydrogen ion concentration and an increase in pH. The temperature and chloride stratification patterns are chiefly density effects. The metals chromium, copper, and iron do not seem to stratify, possibly indicating that they are not involved heavily with the chemical and biological processes described above.

The lake does not remain vertically stratified year-round. Density gradients caused by chloride stratification are apparently not sufficiently strong to prevent fall and spring overturns induced by thermal variations. The density structure of Onondaga is further examined in Figures 3.4-1 to 3.4-5.

A GRID display of density is shown in Figure 3.4-1. For each cell in the grid, density has been computed from the corresponding temperature and chloride value using the following relationship $^{13,9}$ :

$$\rho = (1 - k_1 (T - 4)^2) (1 + k_2 C) \qquad (3.4-3)$$

where,

<sup>p</sup> = density (g/cm<sup>3</sup>)



C

·()

()

()

**(** )

```
f = temperature (degrees C)
```

$$C = chloride concentration (g/m3)$$
  

$$k_{1} = 6.57 \times 10^{-6} [(g/cm3)/(deg C)2]$$
  

$$k_{2} = 1.2 \times 10^{-6} (g/m3)^{-1}$$

A GRID display of vertical density gradient is shown in Figure 3.4-2. For each month (column), a fourth degree polynomial ( p versus z ) has been fit to the seven density values computed from the corresponding seven temperature and chloride observations. The polynomial has been differentiated and evaluated to provide estimates of the vertical density gradient at one-meter intervals. Because of the relative numerical instability of the derivatives of fitted polynomials, estimates of density gradient obtained in this way are only approximate, but sufficient for display purposes. The objective of these calculations is to illustrate the location, strength, and seasonal variation of the pycnocline, the center of which corresponds to the depth of the maximum density gradient. The magnitude of the maximum density gradient may be taken as an indication of the degree of resistance to vertical mixing.

Figure 3.4-2 illustrates the downward movement of the pycnocline from near the surface in late spring to around 12 meters in late

()

63

summer, prior to overturn. Such migration is typically attributed to evaporative cooling of surface waters and input of kinetic energy at the surface due to wind action<sup>3,11,32</sup>. Density gradients appear to be decreasing in strength somewhat during this period, particularly in comparing the early spring months of successive years. In 1974, the pycnocline was considerably weakened relative to other years.

Figure 3.4-3 displays raw and smoothed variations of the vertical temperature and chloride gradients. For each sampled profile and component, the volume-weighted gradient has been computed as:

$$\frac{\overline{dx}}{dz} = \frac{0^{\int_{-\infty}^{z_m} (x_z - \overline{x}) (z - \overline{z}) a_z dz}}{0^{\int_{-\infty}^{z_m} (z - \overline{z})^2 a_z dz}}$$
(3.4-4)

$$\overline{z} = \frac{0^{\int_{-\infty}^{2m} z a_z dz}}{0^{\int_{-\infty}^{2m} a_z dz}}$$
(3.4-5)

$$\overline{\mathbf{x}} = \frac{\int_{0}^{z_{m}} \mathbf{x}_{z} \, \mathbf{a}_{z} \, \mathrm{d}z}{\int_{0}^{f_{m}} \mathbf{a}_{z} \, \mathrm{d}z}$$
(3.4-6)

#### 3.4.2 Vertical Mixing (continued)

where,

C

( -

 $( \cdot )$ 

 $x_z = \text{concentration at depth } z \quad (g/m^3)$   $z = \text{surface area of lake at depth } z \quad (m)$  $z_m = \text{maximum depth } (m)$ 

The trapezoidal rule has been used to perform the above integration numerically. The computation of the gradient according to equation (4) is analogous to the computation of a linear regression coefficient of  $x_z$  on z, weighted according to surface area. Figure 3.4-3 indicates that both temperature and chloride gradients were less pronounced in the later years of the survey. The absolute values of the thermal gradients generally appear to peak slightly after mid-year, corresponding to peak lake temperatures (Appendix C). The absolute values of chloride gradients generally appear to peak in early spring.

The relative contributions of thermal and chemical gradients to the total density gradient are plotted in Figure 3.4-4. The volumeaverage density gradient has been approximated as:

 $\mathbb{E}_{\mathcal{L}}$  .



đ,

**(**)

(\*

Line Plots of Temperature and Chloride Gradients



. -



Figure 3.4-4

Chemical and Thermal Components of Total Density Gradient

## 3.4.2 Vertical Mixing (continued)

$$(\frac{\overline{dp}}{dz}) \stackrel{s}{\underset{tree}{=}} (\frac{\overline{\partial T}}{\partial z}) (\frac{\overline{\partial p}}{\partial T}) + (\frac{\overline{\partial c}}{\partial z}) (\frac{\overline{\partial p}}{\partial c})$$
(3.4-7)  
total thermal chemical  
density portion portion  
gradient

From equation (3):

ě.

(

(

$$(\overline{\frac{\partial \rho}{\partial T}}) = -2 k_1 (\overline{T} - 4) (1 + k_2 \overline{C}) \qquad (3.4-8)$$

$$\frac{\overline{\partial \rho}}{\partial c}) = [1 - k_1 (\overline{T} - 4)^2] k_2 \qquad (3.4-9)$$

The means and gradients of chloride and temperature have been calculated according to equations (4), (5), and (6). The two lines in Figure 3.4-4 represent variations of the total density gradient and of the chemical portion of the total density gradient. The distance between the two lines represents the thermal portion. It is evident that thermal effects dominate during mid-summer and chemical effects are most important during other periods. The figure also indicates that the decreasing trend in early spring density gradients noted previously is associated primarily with

# 3.4.2 Vertical Mixing (continued)

C

decreasing chloride stratification.

An essentially equivalent representation of the average density gradient is shown in Figure 3.4-5. The "buoyant potential energy deficit" represents the theoretical energy requirement to completely mix the lake vertically<sup>32</sup>. It has been computed from each sample profile according to:

$$PE = 9.8 \times 10^9 [_0^{f} \rho_z z a_z dz - \overline{\rho z}_0^{f} a_z dz] \qquad (3.4-10)$$

$$\overline{\rho} = \frac{0^{\int_{z_{m}}^{z_{m}} \rho_{z} a_{z} dz}}{0^{\int_{z_{m}}^{z_{m}} a_{z} dz}}$$
(3.4-11)

$$\overline{z} = \frac{0^{\int_{0}^{z_{m}} z a_{z} dz}}{0^{\int_{0}^{z_{m}} a_{z} dz}}$$
(3.4-12)

where,

( i

PE = buoyant potential energy deficit
 (joules)

 $\rho_{\tau}$  = density at depth z (g/cm<sup>3</sup>)



 $\bigcirc$ 

3-51

. . . .

S)

#### 3.4.2 Vertical Mixing (continued)

ď.

( :

()

Again, the trapezoidal rule has been used to perform the above numerical integration. A PE value of zero or less indicates instability in the water column, i.e. well-mixed conditions. A decreasing trend in PE is apparent, with yearly maximum values decreasing and longer periods of instability occurring between maxima.

The above analysis suggests that Onondaga Lake underwent a change in density stratification during the study period. Calculations in Appendix D do not indicate any change in the mean rates of chloride flux into the lake over this period. It seems most likely that the observed changes can be attributed to hydrologic and/or climatologic variations. While mean annual flows did not vary much from year to year (Table 3.2-5), maximum mean monthly flows were generally higher in later years (Figure 3.2-1). Possible climatologic variations affecting thermal or kinetic energy exchanges at the lake surface may also have influenced the degree of density stratification. Modelling studies will be required to determine whether the decreasing trend in density stratification can be attributed to such factors.

#### 3.4.3 Phytoplankton and Nutrients

( )

The vertical and seasonal distributions of nutrients in Onondaga Lake have been discussed previously (Section 3.4.2). The pattern has been shown to be consistent with algal growth in surface regions and subsequent settling and release of available nutrients into the hypolimnion. During the study period, it is unlikely that nutrients kinetically limited algal growth rates, although they may have played a role in determining dominant species. Figure 3.4-6 depicts variations of the epilimnion- and hypolimnion-average chlorophyll concentrations between 1972 and 1975. Annual cycles appear to be bimodal and no trend in peak chlorophyll concentrations can be identified.

 $SiO_2$  depletion in the epilimnion, particularly in 1974 and 1975 (Appendix B), may have controlled diatom populations. The transition from spring diatom populations to summer greens, as mediated by  $SiO_2$ , temperature, and possibly other factors, may be reflected in the bimodal chlorophyll cycles.

The patterns of epilimnion- and hypolimnion-average concentrations of total inorganic P and ortho-P are shown in Figures 3.4-7 and 3.4-8. The response to detergent phosphate legislation in 1971 is evident. The relatively low degree of vertical stratification with regard to phosphorus in the earlier years of the study may reflect the fact that phosphorus levels were too high during this period to have been



Pigure 3.4-6



Figure 3.4-7


Figure 3.4-8

3-56

 $\bigcirc$ 

#### 3.4.3 Phytoplankton and Nutrients (continued)

()

influenced significantly by algal uptake. In 1975, ortho-phosphorus averaged 0.01 g/cm<sup>3</sup> in the epilimnion on three mid-summer sampling dates, the lowest levels observed throughout the study period. Thomann et al.<sup>35</sup> have used a half-saturation constant of 0.005 g/cm<sup>3</sup> F in simulating the response of algal growth rate to phosphorus concentration. Using this value and the Michaelis-Mentem rate expression, a level of 0.01 g/cm<sup>3</sup> F would result in a 33% reduction in maximum algal growth rate due to phosphorus stress. Acknowledging the variability of the 0.005 g/cm<sup>3</sup> parameter, it seems possible that available phosphorus levels may have partially controlled algal growth rates in 1975.

Levels of ammonia nitrogen and carbon dioxide remained well above rate-limiting levels throughout the study period. It is evident that light was the primary rate-limiting resource. Mid-summer transparencies were generally on the order of one half to one meter.

The abundance of primary nutrients has supported a diverse collection of phytoplankton. Figure 3.4-9 presents semi-quantitative data on the phytoplankton and zooplankton populations between 1968 and 1974, as derived from the Onondaga Lake Study reports<sup>20,24</sup> and Sze<sup>34</sup>. Because of the wide variations in cell size from specie to specie, the cell number data in Figure 3.4-9 cannot be used alone to determine biomass distribution. In 1968, species were recorded as

|                                                 |                   |                |                                                 |                         | cht thiế hình từ t                                    | $\{ , [], [], [], [], [], [], [], [], [], [$    |                  | Les. |
|-------------------------------------------------|-------------------|----------------|-------------------------------------------------|-------------------------|-------------------------------------------------------|-------------------------------------------------|------------------|------|
| 2HY LIPL HNK LIN                                | 1968              | <u> </u>       |                                                 | <u>J¥/1</u>             | 1972                                                  | <u>    1973                                </u> | 1974             | Ξ    |
| CHROFDERIA SETIGERA                             |                   |                |                                                 |                         | <b>6</b>                                              |                                                 |                  | ļ    |
| PEDIASTAUM DUPLEX                               |                   |                |                                                 | ·                       |                                                       | <b></b>                                         | I                |      |
| ALDRELLA VULGARIS                               |                   |                |                                                 | · · · · · · · · · · · · |                                                       |                                                 | ·                | ļ,   |
| INKISTAGGESNUS FALCATUS                         |                   |                |                                                 | ***** ** ******         |                                                       |                                                 |                  |      |
| CENEDESHUS OBLIQUUS                             |                   | . <u> </u>     | <b>.</b>                                        |                         | <b></b>                                               | <b></b>                                         |                  |      |
| LOSIERIUN GARCILE                               |                   |                | -*                                              | ******                  | • • <b>• •</b> • • • •                                |                                                 |                  |      |
| <u>SIAUASSIBLE PARADOXIIN</u>                   | 1                 | -              |                                                 |                         |                                                       |                                                 |                  | £.#  |
| STAURA UYELLA<br>ELIOSIRA GRANULATA             | · ·               |                |                                                 |                         |                                                       | -                                               |                  | Ϋ́.  |
| cyclotella gloherbta<br>Cyclotella neneghiniana |                   | <b>b.b. a</b>  |                                                 | • · •                   | - <b>J</b>                                            |                                                 | ··· · • •        | 1    |
| COSCINCOISCUS SUBTILIS                          |                   |                |                                                 |                         |                                                       |                                                 | ·                | ۶ı   |
| FAAGILARIA CAOTONENSIS<br>STNEORA ULNA          |                   | · · · · ·      | -                                               |                         |                                                       |                                                 | · ·              |      |
| asterionella formosa<br>Anphiprorá albtr        |                   |                | -                                               | -                       |                                                       | - •                                             | · ·              |      |
| <u>NITZSCHIM PALEA</u>                          |                   |                |                                                 |                         |                                                       |                                                 |                  | ί.   |
| GLENCOINTUM PULVISCULLS                         | -                 | -              | -                                               |                         | •                                                     |                                                 |                  | []   |
| CATPIONONAS AVATA                               |                   |                |                                                 | 47                      |                                                       |                                                 |                  | 1 M  |
| ANABAENA CIACINALIS                             |                   |                | -                                               |                         |                                                       |                                                 |                  | γn   |
| APHANIZOPENON FLOS-ADURE                        |                   |                |                                                 |                         |                                                       |                                                 |                  | ľ    |
| SOLITERS                                        | -                 |                |                                                 |                         |                                                       |                                                 |                  |      |
| LCPEPGDS<br>CLADOCEAANS                         |                   |                |                                                 |                         |                                                       |                                                 |                  |      |
| YORTICELLA                                      | 1117500111        | 11111111111111 |                                                 | TT PJ POTTT             | ····                                                  |                                                 | THE THE PARTY OF | ļ    |
|                                                 | 1968              | 1959           | <u>1970                                    </u> | 1971                    | 1972                                                  | 1 1973                                          | 1 1974           | 1    |
|                                                 | STMEDI, LOGIO (AL | IMALS/160 LJ   | STHERE LOCIO                                    | ICITERNIT               | <sup>a</sup> parilin<br>I Chlorophyt<br>II Euglenophy | a (Greens)<br>ta                                |                  |      |
|                                                 | - 0.0 -           | 0.9<br>1.9     | - 2.0                                           | - 6.9                   | III Crysophyce<br>IV Bacillario                       | phyceae (Distons)                               |                  |      |
|                                                 | 2.0<br>3.0        | 2.9            | • 3.U<br>• 4.C                                  | - 4.9                   | V Dinophycos<br>VI Cryptomona                         | le (Dinofis)<br>Idinese                         | COLLECCE)        |      |
|                                                 | • 9.0 •           | 4.9            | <ul> <li>5.0</li> </ul>                         | - 3.3                   | VIT Nerrobyca                                         | r (Blue-Gt                                      | Nens)            |      |

Phytoplankton and Zooplankton Populations

Pigure 3.4-9

### 3.4.3 Phytoplankton and Nutrients (continued)

present/absent only.

()

The phytoplankton population is shown to consist mainly of greens and diatoms. From 1968 to 1971, blue-greens (chiefly <u>Aphanizomenon flos-aquae</u>) dominated biomass in late summer, a fact not obvious in Figure 3.4-9 because of the greater size of bluegreen cells relative to other types. The near-disappearance of blue-greens in latter years has been interpreted as a "positive" sign that conditions in the lake have been improving. Mechanistically, this has been attributed to lower levels of available phosphorus encountered after phosphate detergent legislation in 1971<sup>20</sup>, the general hypothesis being that blue-greens were less able to compete for lower levels of phosphorus than were other algal types. Alternative explanations are discussed below.

The apparent decreasing trend in SiO<sub>2</sub> levels may reflect an increase in diatom activity. Two factors may have contributed to this. Diatoms have been shown to be particularly sensitive to chromium<sup>10</sup>, the levels of which decreased markedly in 1973 (Appendices B and C). The lowering of chromium levels may have given the diatoms a competitive advantage over other algal types. Secondly, simulation studies by Bella<sup>2</sup> have shown that the degree of vertical mixing in a lake may influence interspecific competition among algal types. In particular, enhanced mixing tends to favor

### 3.4.3 Phytoplankton and Nutrients (continued)

ĩ

( :

faster-sinking algae (diatoms) over slower-sinking varieties (bluegreens). Thus, the increasing diatom activity and the near-disappearance of blue-greens may also be explained by the apparent decrease in vertical stratification, as discussed previously (Section 3.4.2).

Aside from the disappearance of blue-greens, no trend in algal species diversity is evident. On the order of twelve algal species were generally present in mid-summer samples throughout the monitoring period. In view of the abundance of nutrients and peculiar aspects of the lake (high heavy metal concentrations and solinity), the diversity of the phytoplankton population is considered surprisingly high<sup>24,38</sup>. Fish populations are likewise surprisingly diverse<sup>24,38</sup>. O'Brien and Gere<sup>24</sup> examined the temporal sequences of phytoplankton and zooplankton densities during the 1968-69 baseline study, noting that major shifts or drops in the phytoplankton population could not be explained by concomitant zooplankton increases. It was concluded that factors other than predation (possibly toxicity) were chiefly controlling the algal populations.

### 3.4.4 Dissolved Oxygen

The GRID display of dissolved oxygen in Appendix B shows the effects of aeration and primary production in the surface waters and

# 3.4.4 Dissolved Oxygen (continued)

respiration and oxygen depletion in the hypolimnion. The oxygen data have been re-plotted as percent of saturation in Figure 3.4-10. For each GRID cell, the oxygen saturation level has been computed from the corresponding temperature and chloride level using the following relationship derived from data in Fair, Geyer, and Okun<sup>8</sup>:

$$(DO)_{SAT} = \frac{448.4}{T+30.04} + \frac{3.8 \times 10^{-3}C}{T+22.8}$$

where,

$$T = temperature (degrees C)$$
  
C = chloride concentration (g/m<sup>3</sup>)

The time- and space-averaged dissolved oxygen concentration during the period amounted to 31% of saturation, reflecting the dominance of heterotrophic activity induced by external and internal BOD sources. Super-saturated levels of DO in the mid-summer epilimnion indicate regions of intense photosynthesis.

The data indicate a general trend toward improved conditions in the hypolimnion, represented by shorter annual periods of anaerobic conditions. At 18 meters, oxygen levels were below



COMPONENT : DISSOLVED OUTGER . PERCENT OF SATURATION

100.00

4.00 AND

PERTICAL SCALE I DEPTH IN NETERS ABOVE LOVEST SAMPLING POINT (18 - SURFACE)

MINISTANTAL SCALE & TIME IN MORTHS FROM JAMUARY 1960

DHONDAGA LAKE DATA PLOT - ENVIRONMENTAL SYSTEMS PROGRAM - MAPYARD UNIVERSITY



Pigure 3.4-10

#### 3.4.4 Dissolved Oxygen (continued)

 $\left[ \right]$ 

1  $g/m^3$  for ten months of the year in 1970, compared with a five-month anaerobic period in 1974. This trend can be most easily seen by comparing hypolimnic dissolved oxygen levels in the early spring of successive years in Figure 3.4-10.

Line plots of volume-averaged dissolved oxygen levels are shown in Figure 3.4-11, smoothed according to the method outlined in Section 3.3.3. Oxygen levels plotted as DO,  $DO/DO_{SAT}$ , and  $DO_{SAT}^{-}-DO$  all indicate an improving trend. Seasonal patterns are also evident and especially regular when the data are plotted as DO deficit ( $DO_{SAT}^{-}$ -DO). The regular periodicity of these data may reflect the importance of photosynthetic oxygen sources, since minimum deficits generally occur during seasons of peak solar intensity and photosynthetic rates. Temperature effects on  $DO_{SAT}^{-}$  could also partially account for this periodicity.

Dissolved oxygen levels are generally determined by the balance, or imbalance between oxygen sources and sinks. Trends in the DO data should therefore result from trends in one or more of the oxygen sources or sinks. In general, no trends can be seen in the ambient levels of BOD or in the external loading rates. It is unlikely that the internal sources and sinks of DO have been altered much, since photosynthetic nutrient supplies generally remained in excess, except, possibly, in the case of ortho-phosphorus in 1975. The DO trend can Figure 3.4-11

Line Plots - DO, DO/DO sat, and DO sat - DO

C

**C**<sup>2</sup>

**(**)



### 3.4.4 Dissolved Oxygen (continued)

best be explained by improved aeration of the bottom waters accompanying the decreasing trend in vertical stratification. Figure 3.4-12 compared the GRID displays of vertical density gradient and dissolved oxygen, as percent of saturation. The inverse relationship between density gradient and hypolimnic DO levels is apparent.

### 3.4.5 Mass Balances

( )

The results of mass balance calculations performed on major water quality components are displayed in Appendix D, and summarized in Table 3.4-3. In the latter, the four terms of each mass balance (input, output, change in storage, and accumulation) are expressed in units of grams per square meter of lake surface per year. The retention coefficient ( = accumulation/total input) and the fraction of the total input attributed to the Metro STP discharge are also given. Based upon examination of the mass balance figures in Appendix D, the component balances which exhibited trends have been broken into two or more time periods. "Trende" are indicated in the balance plots by changes in slope of the cumulative input, output, and/or accumulation lines. Division has been done at yearly intervals to incorporate seasonal effects.

As noted in Section 3.3.4, the epilimnion-averaged concentrations



Pigure 3.4-12

b - see Figure 3.4-2 for scale

# Table 3.4-3

### Hass Balances - Onondaga Lake

### Fluxes in grams per square meter of lake surface per year

| Component                           | Nater<br>Years          | Total<br>Input        | Total<br>Output             | Change in<br>Storage  | Accumulation          | Retention <sup>a</sup><br>Coefficient | Fraction of<br>Input due<br>to Metro STP |
|-------------------------------------|-------------------------|-----------------------|-----------------------------|-----------------------|-----------------------|---------------------------------------|------------------------------------------|
| Chloride                            | 1970-74                 | 72814.                | 70753.                      | 322.                  | 1739.                 | .024                                  | .055                                     |
|                                     | 1970-72                 | 47.4                  | 25.7<br>(28.0) <sup>b</sup> | .2                    | 21.5                  | .453                                  | .120                                     |
| Nitrate and Nitrite N               | 1973-74                 | 24.5                  | 20.8<br>(22.7)              | 3                     | 4.0<br>. (2.1)        | .163<br>(.086)                        | .150                                     |
| Total Fieldabl N                    | 1970-71                 | 273.8                 | 267.3                       | 2                     | 6.7<br>(32,2)         | .025                                  | .434                                     |
| IVEAL NJOLUDNI N                    | 1972-74                 | 262.3                 | 189.1<br>(171.1)            | +.8                   | 72.4<br>(90.4)        | .227<br>(.345)                        | .516                                     |
| Total Nitzagan                      | 1970-71                 | 321.5                 | 288.7<br>(263.8)            | 1.3                   | <u> </u>              | .098<br>(.175)                        | . 387                                    |
| iotal artiogen                      | 1972-74                 | 294.9                 | 214.9<br>(196.4)            | 0.0                   | 80.0<br>(98.5)        | .271<br>(.334)                        | .471                                     |
| Silics (Si0 <sub>2</sub> )          | 1970-74                 | 394.2                 | 235.8                       | -6.3                  | 164.7                 | .417                                  | .247                                     |
| Alkalinity (as Ca CO <sub>3</sub> ) | 1970-74                 | 10500.                | 9038.<br>(8799.)            | . 37.                 | 1425.<br>(1659.)      | .136<br>(.158)                        | .162                                     |
| Total Inorganic P                   | 1970<br>1971<br>1972-74 | 156.7<br>68.2<br>36.3 | 79.4<br>70.3<br>20.9        | -20.8<br>10.1<br>-7.6 | 98.1<br>-12.2<br>23.0 | .626<br>179<br>.634                   | .431<br>.545<br>.743                     |
| Ortho-P                             | 1970<br>1971<br>1972-74 | 48.0<br>32.5<br>23.7  | 39.5<br>40.0<br>15.7        | -14.3<br>11.7<br>-5.2 | 22.8<br>-19.2<br>13.2 | 477<br>592<br>.560                    | .613<br>.592<br>.697                     |
| Condensed Inorganic P               | 1970<br>1971<br>1972-74 | 108.7<br>35.7<br>12.6 | 39.9<br>30.3<br>5.2         | -6.5<br>-1.6<br>-2.4  | 75.3<br>7.0<br>9.8    | .693<br>.196<br>.778                  | .351<br>.502<br>.831                     |
| 5-Day BOD                           | 1970-74                 | 1389.                 | 289.                        | -5.                   | 1105.                 | .796                                  | .792                                     |

3-67

a Recention Coefficient = Accumulation/Total Input

b Corrected for lack of complete mixing; multiplied by ratio of Station 2 to Station 1 epilimnion - averaged concentrations (Table 7)

#### 3.4.5 Mass Balances (continued)

劉

at station 1 (see Figure 3.1-1) have been used to estimate lake outputs. Tests for horizontal mixing have indicated, however, that the lake is not completely mixed with regard to all components. The output fluxes in parenthesis in Table 3.4-3 have been corrected for this lack of complete horizontal mixing by multiplying the station 1 estimate by the ratio of the mean concentration at station 2 (closer to the lake outlet) to the mean concentration at station 1, as given in Table 3.4-1. This was not done for silica or the phosphorus components, since the differences between station 1 and 2 were not found to be statistically significant in these cases.

For each component, the accumulation term in the balance has been computed by subtracting output and change in storage from total input. This term thus represents the net sum of all fluxes which have not been measured or estimated directly, including sediment or atmospheric exchanges and chemically- or biologicallymediated transformations within the lake. The accumulation term also reflects possible errors in the input flux estimates due to sampling strategy relative to periodicities in point sources or intervening storm events in tributaries, as discussed in Section 3.2.1.

Over the five-year period, estimated chloride inputs balanced estimated outputs to within 2.4%. Since no mechanisms for chloride

#### 3.4.5 Mass Balances (continued)

Ċ

trapping within the lake are known to exist, the salt balance tends to verify the assumptions, techniques, and data used in developing the hydrologic balance and in estimating mass fluxes from grabsample concentration data. Ninemile Creek, which accepts the waste bed overflow from the Allied Chemical Company's solvay plant, accounted for 76.1% of the total chloride input to the lake over the five-year period. Assuming that the background concentrations of chloride in Ninemile Creek were the same as those in Onondaga Creek (averaging 275 g/m<sup>3</sup>), about 68.3% of the total chloride flux can be attributed to the solvay plant discharge. No trends in the chloride balance are evident.

Fluxes of reduced nitrogen forms (Kjeldahl) dominate over those of oxidized forms (nitrate and nitrite) by about an order of magnitude. Trends in the balances of each nitrogen form are evident. Nitrate and nitrite inputs were reduced by about a factor of two in 1973 and 1974, attributed primarily to reductions in loadings from Onondaga Creek and Ninemile Creek. Nitrate and nitrite may be produced in aerobic regions of the lake through nitrification and lost in anaerobic regions through denitrification. Algal uptake may be another important nitrate sink. The reduced net accumulation rates of the oxidized nitrogen forms in the latter years may reflect increased oxygen levels and enhanced nitrification.

3.4.5 <u>Mass Balances</u> (continued)

Trend in the Kjeldahl nitrogen balance is toward reduced output rates and increased net accumulation rates. During water years 1970 and 1971, outputs balanced inputs to within 11.8%. During the 1972-1974 period, however, 34.5% of the influent Kjeldahl nitrogen was trapped with the lake or converted to oxidized forms. Enhanced aeration could have accounted for this increase. The disappearance of blue-green algae over this same period may also have eliminated an important internal Kjeldahl nitrogen source.

The trends and relationships in the total nitrogen balance are similar to those of the Kjeldahl nitrogen balance. The increase in the percent of total nitrogen loading due to the Metro STP discharge from 38.7% to 47.1% may be partially attributed to the Onondaga County combined sewer maintenance program, which reportedly eliminated several dry-weather sewer overflows into Onondaga Creek and Harbor Brook during this period<sup>14</sup>. In addition, the Ley Creek Sewage treatment plant discharge was diverted from Ley Creek to the Metro STP at the end of 1969<sup>24</sup>.

The silica balance was relatively constant throughout the five-year period. An estimated 43.7% of the influent silica was trapped within the lake. Mechanisms for silica removal include uptake by diatoms and subsequent deposition, and formation and sedimentation of silica-containing minerals<sup>24</sup>.

C

The alkalinity balance likewise appears to be relatively stable. On the average, 15.8% of the influent alkalinity was removed. The primary reaction accounting for this is:

$$Ca^{++}$$
 + 2 HCO<sub>3</sub><sup>--</sup>  $\ddagger$  CaCO<sub>3</sub> + + CO<sub>2</sub> + H<sub>2</sub>O (3.4-13)  
alkalinity calcite

The high levels of calcium in the lake drive this reaction to the right in nearly all regions and seasons, except where pH falls below  $7^{24}$ .

Table 3.4-3 presents balances on three phosphorus forms: TIP, ortho-P, and condensed inorganic P ( = TIP - ortho-P). The response to phosphate detergent legislation in 1971 is evident. The time period has been broken into three parts: pre-legislation (1970), transition (1971), and post-legislation (1972-74). Comparison of pre- and post-legislation periods indicates a reduction of 77% in the TIP loadings\*. The TIP retention coefficients, however, remained

:

<sup>\*</sup> This 77% reduction is surprisingly high, in view of the amount of total phosphorus which has been attributed to detergents in sanitary wastes  $(\sim 56\%)^{29}$ . Some of this reduction could have been due to the county's combined sewer maintenance program which was being carried out simultaneously with the detergent legislation. In addition, the pre- and post-legislation phosphorus measurements do not include the organic phosphorus fraction, which would not be influenced by detergent formulation.

ĩ

relatively constant, at values of 62.6% and 63.4%, respectively. During the transition period, internal sources of TIP and ortho-P are apparent, in that the computed retention coefficients are negative. This could have resulted from exchange from sediments or with organic phosphorus forms in the water column during this period. This type of transient response is in tune with a model proposed by Lorenzen et al.<sup>16</sup> for simulating the response at Lake Washington to phosphorus diversion. Mechanisms for phosphorus removal include algal uptake and deposition and chemical precipitation. The lake is supersaturated with calcium phosphate, and formation and precipation of hydroxyapatite and fluorapatite are expected to be important mechanisms for phosphorus removal<sup>24</sup>. The relative stability of such minerals to changes in redox potential (unlike iron-phosphorus compounds) tends to indicate that release of soluble phosphorus from the sediments in significant amounts would be unlikely. The fact that the pre- and post-legislation TIP retention coefficients are similar indicates first-order behavior of the phosphorus removal mechanism and suggests that chemical precipitation may be more important than algal uptake and eedimentation. The former mechanism is first-order and the latter, zero-order in phosphorus concentration, under conditions in which phosphorus is not limiting algal growth.

The BOD, balance appears to have been relatively stable over

#### 3.4.5 Mass Balances (continued)

( )

the five-year period. The Metro STP discharge, determined from the Onondaga County plant performance records<sup>23</sup>, accounted for 79.2% of the total external loading of  $BOD_5$ . Average outflow amounted to 15.3% of the external loading. Consideration of internal loadings resulting from primary production can provide a much more complete picture of the organic matter and oxygen balances of the lake.

Gross primary production has been estimated using data provided by O'Brien and Gere<sup>20</sup>. Light/dark bottle studies were done on four days between May and September in 1974. The average daily gross producitivity was estimated at 6.1 grams of carbon per square meter per day (observed range  $3.8 - 7.4 \text{ g-C/m}^2$ -day). Assuming that 85% of the yearly production occurred in the 150-day May - September period<sup>4</sup>, an annual gross production of 1080 g-C/m<sup>2</sup>-year is estimated. Using conversion factors given by Byrlinsky and Mann<sup>4</sup>, this corresponds to 2900 g-O<sub>2</sub>/m<sup>2</sup>-yr and 10150 Kcal/m<sup>2</sup>-yr. At an annual average visible solar radiation intensity of 3200 kcal/m<sup>2</sup>-day<sup>22</sup>, this corresponds to a photosynthetic efficiency of 0.9%.

Brylinski and Mann<sup>4</sup> have studied the relationships among phytoplankton standing crop, gross productivity, and photosynthetic efficiency using the International Biological Program data base. The results of the above calculations have been compared with those of Brylinski and Mann in Figure 3.4-13. While Onondaga Lake is shown to be highly productive, the relationships among phytoplankton standing

# Relationships Among Phytoplankton Standing Crop, Gross Photosynthesis, and Photosynthetic Efficiency for Onondaga Lake Compared with Results of Brylinsky and Mann<sup>3</sup>

M. BRYLINSKY AND K. H. MANN

đ.

(

C



Fig. 10. Relationship of (A) phytoplankton production to phytoplankton biomaps, (B) phytoplankton reduction to chlorophyll a concentration, (C) phytoplankton production to photosynthetic efficiency, and (D) chlorophyll a to photosynthetic efficiency.

a - Epilimnion average concentration in Onondaga Lake, May-Sept., 1974.

### 3.4.5 Mass Balances (continued)

()

crop (as chlorophyll), primary production, and photosynthetic efficiency are close to what one would expect based upon Brylinski and ... Mann's studies of a wide range of lakes. Besides providing perspective on Onondaga Lake, this lends some strength to the productivity estimate derived above.

Using this figure to represent internal sources of oxygen demand, an annual-average ultimate oxygen demand balance has been formulated (Table 3.4-4). The ratio of 5-day to ultimate carbon-aceous BOD has been assumed to be 0.65, corresponding to a first order decay rate of 0.21 day<sup>-1</sup>. Nitrogenous oxygen demands have been calculated from the TKN balances, assuming the following stoichiometry for the nitrification reaction:

$$NH_3 + 2O_3 \rightarrow NO_3 + H^{\dagger} + H_2O$$
 (3.4-14)

Table 3.4-4 shows that external sources of oxygen demand only slightly exceed internal (photosynthetic) sources on an annual average basis. During productive seasons, internal sources would be expected to dominate. The total net loading to the lake, after subtracting outflow, amounts to 14  $g-0_2/m^2$ -day, or, assuming an average depth of 12 meters, about 1.2  $g-0_2/m^2$ -day. Since gross

# Table 3.4-4

# Onondaga Lake

# Ultimate Oxygen Demand Balance

# Average Annual Conditions (1970-74)

|                      | Areal<br>Basis<br>(g-O <sub>2</sub> /m <sup>2</sup> -day) | Volumetric<br>Basis<br>(g-O <sub>2</sub> /m <sup>3</sup> -day) |  |
|----------------------|-----------------------------------------------------------|----------------------------------------------------------------|--|
| External Sources     |                                                           |                                                                |  |
| Carbonaceous         | 5.85                                                      | 0.49                                                           |  |
| Nitrogenous          | 3.28                                                      | 0.27                                                           |  |
| Internal Source      |                                                           |                                                                |  |
| Gross Photosynthesis | 7.91                                                      | 0.66                                                           |  |
| Total Input          | 17.04                                                     | 1.42                                                           |  |
|                      |                                                           |                                                                |  |
| Outflows             |                                                           |                                                                |  |
| Carbonaceous         | 0.92                                                      | 0.08                                                           |  |
| Nitrogenous          | 2.14                                                      | 0.18                                                           |  |
| Total Outflow        | 3.06                                                      | 0.26                                                           |  |
|                      |                                                           |                                                                |  |
| Net Demand           | 13.98                                                     | 1.16                                                           |  |
|                      | 1                                                         |                                                                |  |

.

\* 1972-74

ť

(

Ĺ

### 3.4.5 Mass Balances (continued)

C

photosynthesis would supply a maximum of 7.9  $g-O_2/m^2$ -day, assuming no atmospheric losses due to super-saturation, remaining oxygen sources or organic matter sinks would have to account for a minimum of 6.1  $g-O_2/m^2$ -day or 0.5  $g/m^3$ -day of oxygen demand. Oxygen sources include aeration, and nitrate and sufate reduction. Sedimentation of particulate organic matter may be an important oxygen demand sink, depending upon the extent to which this oxygen demand is exerted in a benthic form.

O'Brien and Gere<sup>24</sup> have attempted to quantify benthic oxygen demands by measuring oxygen uptake in disturbed sediment cores. Measured values ranged from 0.0024 to 0.0034  $g-O_2/m^2$ -day, well below the magnitude of external and other internal sources of oxygen demand. However, these data are suspect; Newbold and Ligget<sup>19</sup> reviewed the literature on benthic oxygen demands in lakes and streams and concluded that a value in the range of 0.1 to 1.0  $g-O_2/m^2$ -day was appropriate for Cayuga Lake. This range is about 50 times higher than that measured in Onondaga Lake. It seems unlikely that the Onondaga values would be so much lower, particularly considering its reportedly extensive organic sludge deposits<sup>24</sup>. On the other hand, the precipitation of calcium carbonate in Onondaga may serve to blanket the sludge deposits and prevent exertion of oxygen on demand in the overlying water. Calcite precipitation may also be an important mechanism for removal of yellow and humic organic compounds from the water column

3.4.5 Mass Balances (continued)

via an adsorption/sedimentation process<sup>25</sup>. If Newbold and Ligget's figures are used, the benthic component of the oxygen demand balance may be appreciable. However, since the source of the organic matter exerting the benthic demand is in the water column, the net balance in Table 3.4-4 is not affected.

Tertiary treatment of the Metro STP effluent to remove BOD and phosphorus is a measure designed to reduce both external and internal (photosynthetic) sources of oxygen demand. Based upon the calculations outlined above, the maximum impact of this plan on the oxygen demand budget of Onondaga Lake is presented in Table 3.4-5. In order to put an upper limit on the effects of phosphorus reduction, gross photosynthesis has been assumed to be reduced in proportion to the total phosphorus loading. This assumes that production is phosphorus-limited and ignores nutrient recycling effects. Accordingly, maximum reduction of 59% in the total oxygen demand loading is estimated.

### 3.5 Eutrophication Assessment

( :)

### 3.5.1 Phosphorus as a Controlling Factor

In order to develop some perspective on Onondaga Lake from a eutrophication viewpoint, its phosphorus balance has been compared

# Onondaga Lake

Present and Projected Ultimate Oxygen Demand Loadings

( )

(

(

|                      | Average Conditions<br>(1970-74) | Future Conditions<br>(1980) |
|----------------------|---------------------------------|-----------------------------|
|                      |                                 |                             |
| External Sources     | •                               |                             |
| Carbonaceous         | 5.85 .                          | 1.53                        |
| Nitrogenous          | 3,28                            | 3.28                        |
|                      |                                 |                             |
| Internal Source      |                                 |                             |
| Gross Photosynthesis | 7.91                            | 2.15*                       |
|                      |                                 |                             |
| Total                | 17.04                           | 6,96                        |
|                      | · · ·                           |                             |
|                      |                                 |                             |
|                      |                                 |                             |
|                      |                                 |                             |
|                      |                                 |                             |
|                      |                                 |                             |

\* Assuming primary production reduced in proportion to total inorganic phosphorus loading; represents a lower limit.

### 3.5.1 Phosphorus as a Controlling Factor (continued)

( )

 $( \ )$ 

Ć

with balances of a wide variety of other northern temperate lakes. This comparison has been done in the contexts of empirical models for predicting lake trophic state as a function of phosphorus loading and morphologic indices. The models employed have been discussed and evaluated in detail in Chapter 2. They include the two models of Vollenweider<sup>39,40</sup>, that of Dillon and Rigler<sup>6</sup>, and the model developed from the stepwise discriminant analysis in Chapter 2.

Table 3.5-1 presents the "permissible" (oligotrophic/mesotrophic) and the "dangerous" (mesotrophic/eutrophic) phosphorus loading levels for Onondaga Lake as estimated by these various models. These levels can be compared with estimated past, present, and future TIP loading also presented in Table 3.5-1. The projected TIP loading for 1980 is still over twice the highest "dangerous" level estimated by any of the models.

Two factors render these empirical models of limited value for application to Onondaga. First, the models are all based upon measurements of total phosphorus, whereas only total inorganic phosphorus readings were available in this case. Second, the uniquely-high calcium content of Onondaga promotes phosphorus removal via calcium phosphate precipitation and sedimentation<sup>24</sup>. This removal mechanism was generally not present to the same degree in the lakes upon which the empirical models were based. For this reason,

# Table 3.5-1

Allowable Phosphorus Loadings for Onondaga Lake According to

| Model             | "Permissible" | "Dangerous" |
|-------------------|---------------|-------------|
| Vollenweider I    | 0.11          | 0.22        |
| Vollenweider II   | 0.74          | 1.48        |
| Dillon and Rigler | 1.48          | 2.96        |
| ESP               | 0.66          | <b>1.90</b> |

**'**∔

×.

(

Various Models (grams / m<sup>2</sup> - year)

Observed and Projected Loadings

| Water Year | TIP Loading | % METRO STP |
|------------|-------------|-------------|
| 1970       | 156.7       | 43 %        |
| 1971       | 68.9        | 54 %        |
| 1972       | 42.2        | 76 %        |
| 1973       | 38.7        | 67 🐐        |
| 1974       | 28.7        | 81 % .      |
| Projected  | 7.8         | 30 %        |

• assuming 90 % reduction in 1974 METRO STP loading

### 3.5.1 Phosphorus as a Controlling Factor (continued)

1

the permissible and dangerous loadings specified by the various models may be too conservative. To some extent, Dillon's model may account for the effects of the additional phosphorus removal mechanism by employing the observed retention coefficient (0.63). Accordingly, the loadings specified by that model are higher than those specified by the others. The fact that the observed retention coefficients before and after detergent legislation were similar (see Table 3.4-3) suggests the validity of the first-order assumption inherent in the Dillon model. However, it is still uncertain whether the relationship between lake trophic state and phosphorus concentration holds in the unusual chemical environment of Onondaga Lake, i.e., the target levels of 0.02 and 0.01 g-P/m<sup>3</sup> may not be realistic.

An indication of the importance of calcium phosphate precipitation as a phosphorus removal mechanism can be derived by comparing the observed retention and phosphorus sedimentation coefficients with those estimated using the retention model developed in Chapter 2:

> $1 - RE = 1/(1 + 0.824 T^{.454}) = 0.70$ KE = 0.824 T<sup>-.546</sup> = 1.86 year<sup>-1</sup>

The corresponding observed values are given by:

```
1 - RO = 0.37
```

()

( )

 $KO = RO/[(1 - RO)T] = 7.53 year^{-1}$ 

The standard error of estimate for the above retention coefficient model is 0.137. The following t-test suggests that the observed and estimated retention coefficients are significantly different:

$$t = (RO - RE)/SEE_{D} = (0.63 - 0.30)/0.137 = 2.41$$

These calculations indicate that the retention coefficient in Onondaga is about twice what one would expect, based upon the model developed on data from other northern temperature lakes. The effective firstorder sedimentation coefficient/is about 4 times the expected value, suggesting the importance of additional phosphorus removal mechanisms in Onondaga. Some of these differences may be attributed to the fact that only inorganic phosphorus measurements were available. Since the organic faction in the lake's outflow has been ignored, the actual total phosphorus retention coefficient may be somewhat lower.

### 3.5.1 Phosphorus as a Controlling Factor (continued)

**(** )

<u>(</u> ,

As noted above, the Dillon model roughly accounts for any unusual phosphorus removal mechanisms in Onondaga. The projected loading for 1980 is still over twice the dangerous level predicted by that model. Since the Metro STP will account for only 30 percent of the projected loading, some rather extreme non-point source control measures may have to be implemented in order to reach acceptable loading levels. Estimation of confidence limits on the tributary and point-source loadings and consideration of combined sewer overflows (which may not be adequately reflected in the tributary loading estimates) would be required in order to provide an adequate basis for projecting the impacts of non-point source control measures. Generally, the results of this analysis supports the EPA's conclusion that "the best reason for instituting phosphorus removal at the Metropolitan Syracuse STP is the protection of Lake Ontario"<sup>38</sup>.

### 3.5.2 Light as a Controlling Factor

In Chapter 2, the rationale behind the Lorenzen-Mitchell model<sup>15</sup> for predicting the potential effects of mixing upon algal production in lakes and impoundments was discussed. A scheme for estimating the parameters of the model from transparency observations was also presented. The model is applied below to estimate the potential effects of mixing under present and future nutrient loading regimes.

# 3.5.2 Light as a Controlling Factor (continued)

According the development in Chapter 2, peak, light-limited biomass, expressed in terms of grams of chlorophyll-a per square mater, is given by:

$$C_{L}^{\max} Z = \frac{\mu^{m} F}{r\beta} - \frac{\alpha}{\beta} Z \qquad (3.5-1)$$

# The corresponding expression for nutrient-limited biomass is:

$$c_{N}^{\max} z = \frac{N_{o}}{y} z \qquad (3.5-2)$$

where,

X

 $\mathbf{C}$ 

$$C_{L}^{max} = \text{light-limited biomass } (g \text{ Chl-a/m}^{3})$$

$$C_{N}^{max} = \text{nutrient-limited biomass } (g \text{ Chl-a/m}^{3})$$

$$Z = \text{mixed depth } (m)$$

$$\mu^{max} = \text{maximum growth rate of algae } (day^{-1})$$

$$r = \text{respiration rate of algae } (day)^{-1}$$

$$\beta = \text{incremental light extinction coefficient}$$

$$due \text{ to algae } (m^{2}/g \text{ Chl-a})$$

$$\alpha = \text{background extinction coefficient } (m^{-1})$$

$$F = \text{light/depth integral}$$

 $N_0 =$ limiting nutrient concentration at spring overturn (g/m<sup>3</sup>)

y = nutrient content of algae
 (g nutrient/g Ch1-a)

The model suggests that peak biomass will be controlled by light or by nutrients, according to whether  $C_L^{\max} z$  is less than or greater than  $C_N^{\max} z$ , respectively. In Chapter 2, it was shown that the parameters of the model could be estimated from the following relationships:

$$F = 2.718\lambda/24$$
 (3.5-3)

$$\alpha = 1.44/2_{g}^{max}$$
 (3.5-4)

$$\frac{\mu^{m}}{r} = 1.44 \, z/F \, z_{s}^{min} \qquad (3.5-5)$$

where,

 $\lambda = \text{day length at peak biomass (hours)}$   $z_s^{\text{max}} = \max \text{maximum Secchi depth observed in the course of a year (m)}$  $z_s^{\text{min}} = \min \text{minimum Secchi depth observed in the course of a year (m)}$ 

### 3.5.2 Light as a Controlling Factor (continued)

Transparency observations in Onondaga Lake over the 1968-74 period are plotted in Figure 3.5-1. These generally range from about 0.5 meter to 2.5 meters, although some year-to-year variations in these ranges are apparent. During this period, nutrient concentrations were in excess, so the assumption is made that a 0.5 meter Secchi depth corresponds to peak, light-limited biomass. For a thermocline depth of 9 meters, the average epilimnion depth is estimated as the ratio of volume to surface area, or 7.15 (see Figure 3.3-1). Accordingly, the following parameters are assumed:

$$z_{s}^{\min} = 0.5 \text{ meter}$$
  
 $z_{s}^{\max} = 2.5 \text{ meters}$   
 $Z = 7.15 \text{ meters}$   
 $\lambda = 15 \text{ hours}$ 

Substituting into equations (3) to (5):

(

$$F = 1.70$$
  
 $a = 0.58 m^{-1}$   
 $\frac{\mu^{m}}{r} = 12.1$ 



Figure 3.5-1

YEAR

# 3.5.2 Light as a Controlling Factor (continued)

The estimated value of  $\frac{\mu^m}{r}$  is within the range of values reported in the literature<sup>7</sup>. Assuming that phosphorus is the potential limiting nutrient, the remaining parameters have been derived from the literature:

$$\beta = 20 \text{ m}^2/\text{g Chl-a}$$
 (ref.: 1, 26)  
y = 1 g-P/g Chl-a (ref.: 7, 26)

Thus, the expressions for light- and nutrient-limited biomass are:

( )

$$c_{L}^{\max} z = 1.03 - 0.029 z$$
 (3.5-6)

$$c_{N}^{\max} z = N_{O} z \qquad (3.5-7)$$

These lines are plotted in Figure 3.5-2 for values of  $N_0$ , the total inorganic phosphorus concentration at spring overturn, of 1.58, 0.32, and 0.086 g/m<sup>3</sup>. These concentrations correspond to 1970, 1974, and 1960 conditions, respectively. The figure for 1980 was estimated as the observed value for 1974 (Pigure 3.4-7) times the ratio of 1980 to 1974 TIP loadings (Table 3.5-1).



C,

C

MIXED DEPTH (meters)

3-90

### 3.5.2 Light as a Controlling Factor (continued)

(

Figure 3.5-2 indicates that under current mixing regimes (Z = 7.12 meters) and past and present phosphorus loading, peak biomass should be light-limited. The intersection of the Z = 7.15meters line with the light-limited biomass line occurs at a CZ value of 0.82 g Chl-a/m<sup>2</sup>. This suggests that peak biomass concentrations should be in the vicinity of 0.82/7.15 = 0.115 g Chl-a/m<sup>3</sup>, or 115 micrograms Chl-a/liter. Observed peak biomass levels for the 1972-75 period ranged from about 90 to 150 and averaged 110 micrograms per liter (Figure 3.4-6). These observations are not inconsistent with model predictions.

The model indicates that the 1980 loading reduction should result in a nutrient-limited peak algal biomass level of 0.62 g Chl-a/ $m^2$ , corresponding to a volumetric concentration of 87 micrograms/liter, or a 24% reduction of current levels. In order to bring peak biomass concentrations down to less than 30 micrograms Chl-a/liter, another 66% reduction in phosphorus loadings would be required. These conclusions are not greatly different from those derived from use of the Dillon model in the previous section, which suggested that a 62% reduction in the 1980 loadings would be required to reach "dangerous" loading levels (Table 3.5-1).

The fact that peak bicmass will be nutrient-limited under 1980 conditions suggests that increases in mixing rates may enhance
( )

productivity somewhat. If it is assumed that enhanced mixing effectively increases mixed depth, Onondaga would lie somewhere between current conditions (Z = 7.15 meters) and the completely-mixed conditions (Z = 11.7 meters). A maximim production of 0.77 g Chl-a/ $m^2$  would be realized at an effective depth of 9 meters, while the completely-mixed condition would result in a light-limited peak biomass of 0.70 g Chl-a/ $m^2$ . Thus, the model indicates that, from the point of view of total biomass production, no benefits would be derived from mixing the lake under the estimated 1980 loading conditions.

Under light-limited conditions, the expression for Secchi depth is:

Under nutrient-limited conditions,

$$(z_{s}^{\min})_{N} = \frac{1.44}{\alpha + \frac{\beta N_{o}}{\gamma}}$$
(3.5-9)

For the parameter values estimated above:

 $\bigcirc$ 

$$(z_{B}^{min})_{L} = 0.070 z , z \le 35.5$$
 (3.5-10)  
= 2.50 , z > 35.5

$$(z_{g}^{\min})_{N} = \frac{1.44}{0.58 + 20 N_{O}}$$
 (3.5-11)

Under 1980 conditions and the current mixing regime, it was estimated above the peak biomass would be nutrient-limited. Accordingly, the minimum Secchi depth would be given by:

$$z_{s}^{\min} = (z_{s}^{\min})_{N} = \frac{1.44}{0.58 + 20(0.086)} = 0.63 \text{ meter}$$

This would represent a slight improvement over current conditions ( $z_s^{min} = 0.5$  meter). It was also estimated above that if the lake were completely mixed (Z = 11.7 meters), peak algal biomass would be light-limited. Accordingly, under these conditions:

$$z_s^{\min} = (z_s^{\min})_L = 0.070 (11.7) = 0.82 \text{ meters}$$

Thus, mixing the lake under 1980 phosphorus loading conditions would increase minimum transparency by about 30%. This change would probably be difficult to measure with a Secchi disc in this range, although it could be detected with light transmission measurements. The two conditions examined above correspond to volumetric biomass concentrations of 87 and 59 micrograms Chl-a per liter, respectively, or a potential 31% decrease as a result of mixing.

Generally, in order for mixing alone to be successful as a means of restricting peak algal biomass to less than 30 micrograms  $Chl-a/m^3$ , equation (1) indicates that the following criterion must be satisfied:

$$C_{L}^{\max} = \frac{\mu F}{r \beta \overline{z}} - \frac{\alpha}{\beta} \leq 0.030 \qquad (3.5-12)$$

where,

a.

(

 $\overline{Z}$  = mean epilimnion depth (m)

mean lake depth, under completely mixed conditions

For the parameter estimates employed above, this expression reduces to:

$$\overline{Z}$$
 (1 + 1.67a)  $\geq$  34.3 (3.5-13)

or,

$$\overline{z} (1 + \frac{2.40}{z_{a}^{\max}}) \ge 34.3$$
 (3.5-14)

Lakes with greater mean depths or with greater background (non-algal) extinction coefficients would benefit more from artificial destratification. In the case of Onondaga, the value of the left side of equation (14) is 22.9, suggesting that additional controls on nutrient inputs would be necessary in order to restrict peak algal biomass to less than 30 micrograms of chlorophyll-a per liter. 3.6 Conclusions

(

64

- Significant differences in concentrations of some water quality components between the two lake monitoring stations indicate that the epilimnion at station 1, closer to the Metro STP discharge, is a relatively reduced environment. These differences are small compared with vertical and seasonal variations.
- Seasonal vertical stratification patterns reflect the effects of density and of chemical and biological reactions occurring in the epilimnion and hypolimnion.
- 3. A trend toward reduced density (temperature and chloride) stratification in the lake is possibly attributed to climatologic variations. This trend would indicate enhanced vertical mixing rates, in turn consistent with the following observations:
  - a. enhanced hypolimnic oxygen levels;
  - b. virtual disappearance of blue-green algae.
- 4. The apparent reduction in the blue-green algal population over the course of the monitoring period might be attributed to a number of factors possibly influencing interspecific competition, including:

a. reduction in phosphorus levels;

- b. reduction in chromium levels;
- c. enhanced vertical mixing rates.

If the last factor is ohiefly responsible, there is no reason why the blue-green population would not return in

### 3.6 Conclusions (continued)

:

 $( \cdot )$ 

6

the future, since the enhanced mixing rates have been attributed to climatologic variations.

- While nutrients may have controlled species dominance with season, the algal population appears to have been light-limited under the nutrient loading regimes of 1968-74.
- 6. Estimates of chloride inputs developed from water quality and quantity data balance outputs to within 2.4% over water years 1970-74. 68.3% of the total chloride flux can be attributed to Allied Chemical's waste bed overflow into Ninemile Creek.
- 7. Over this same period, fluxes of reduced nitrogen forms dominated over those of oxidized forms by about an order of magnitude. Apparent trends toward higher net accumulation rates of Kjeldahl N and lower accumulation rates of Nitrate and Nitrite N may reflect: (1) enhanced nitrification, conceivably a result of higher oxygen levels; (2) lower nitrogen fixation rates, resulting from the disappearance of blue-green algae.

8. Phosphorus balances indicate the following:

- a. The average loading in water years 1972-74 was 23% of that observed in water year 1970. This 77% reduction can be attributed to the effects of:
  - (1) phosphorus detergent legielation in 1971;
  - (2) Onondaga County's combined sewer maintenance program;

### 3.6 Conclusions (continued)

( )

(

( •

- (3) diversion of raw sewage formerly discharged into Ley Creek to the Metro STP in late 1969.
- b. The lake apparently retained 63% of the influent TIP during each of the above periods, suggesting the importance of a first-order reaction for phosphorus removal in the lake. Under steady-state conditions, the TIP retention coefficient and effective firstorder sedimentation coefficient were two and four times, respectively, what one could expect, based upon analysis of phosphorus balance data from other northern temperate lakes. These results, along with chemical equilibrium considerationa, suggest the importance of calcium phosphate precipitation as a dominant removal mechanism.
- c. Apparently negative accumulation rates and retention coefficients for TIP and ortho-P during water year 1971 may reflect the net release of inorganic P from the sediment or from organic P fractions in the water column during this transition period, in which loadings and ambient concentrations were markedly reduced. Sediment release is considered relatively unlikely because of the stability of apatite.
- 9. Under average 1972-74 conditions, external and internal (photosynthetic) sources of ultimate oxygen demand were about equal and totalled 17  $g-O_2/m^2$ -day, or 1.4  $g-O_2/m^3$ -day.
- 10. Installation of secondary and tertiary treatment facilities at the Metro STP is estimated to:

### 3.6 Conclusions (continued)

 $\left( \right)$ 

- a. reduce ultimate oxygen demand loadings by a maximum of 59%;
- b. reduce TIP loadings by a maximum of 73%, resulting in nutrient-limited peak algal biomass levels of about 76% of those observed under past and present lightlimited conditions.
- 11. Since future algal populations are projected to be nutrientlimited, no reduction in total biomass production would be realized by attempting to control available light through lake mixing.
- 12. Additional controls on nutrient inputs may have to be implemented in order to bring peak biomass levels down to mesotrophic levels. In this effort, further data and analysis are required in order to assess the importance of phosphorus loadings originating in combined sewer overflows, which may not be adequately reflected in the calculations outlined above.

#### **REFERENCES** - CHAPTER 3

 $\left( \cdot \right)$ 

- Bannister, T.T., "Production Equations in Terms of Chlorophyll Concentration, Quantum Yield, and Upper Limit of Production", Limnology and Oceanography, Vol. 20, No. 3, pp. 342-364, May 1975.
- Bella, D.A., "Simulating the Effect of Sinking and Vertical Mixing on Algal Population Dynamics", <u>Journal of the Water Pollution</u> <u>Control Federation</u>, Vol. 42, No. 5, Part 2, pp. R140-R152, May 1970.
- Blanton, J.O., "Vertical Entrainment into the Epilimnia of Stratified Lakes", <u>Limnology and Oceanography</u>, Vol. XVIII, No. 5, pp. 697-704, September 1973.
- Brylinsky, M. and K.H. Mann, "An Analysis of Factors Governing Productivity in Lakes and Reservoirs," <u>Limnology and Oceanography</u>, Vol. 18 , No. 1, January 1973.
- 5. Council on Environmental Quality, <u>Environmental Quality</u>, Sixth Annual Report, U.S. Government Printing Office, p. 396, 1975.
- Dillon, P.J., "The Phosphorus Budget of Cameron Lake, Ontario: The Importance of Flushing Rate to the Degree of Eutrophy of Lakes", Limnology and Oceanography, Vol. 20, No. 1, pp. 28-45, Jan. 1975.
- DiToro, D.M., D.J. O'Connor, R.V. Thomann, and R.P. Winfield, "Mathematical Modelling of Phytoplankton in Lake Ontario 1. Model Development and Varification", EPA-660/3-75-005, March 1975.
- Fair, G.M., O.C. Geyer, and D.A. Okun, <u>Water and Wastewater Engi-</u> neering, Vol. 2, John Wiley & Sons, New York, 1968.
- Harleman, D.R.F. and A.T. Ippen, "Salinity Instrusion Effects in Estuary Shoaling", <u>Journal of the Hydraulics Division</u>, <u>American</u> <u>Society of Civil Engineers</u>, Vol. 95, No. HY1, pp. 9-27, January 1969.
- Hervey, R.J., "Effect of Chromium on the Growth of Unicellular Chlorophyceae and Diatoms", <u>Botanical Gazette</u>, Vol. III, No. 1, September 1949.
- Hutchinson, G.E., <u>A Treatise on Limnology</u>, Vol. 1, John Wiley and Sons, 1957.
- International Business Machines, Inc., <u>IBM System/360 and System/</u> <u>370, Subroutine Library - Mathematics</u>, Program Product 5736-XM7, 1974.

5

- International Critical Tables, National Research Council, Edward
   W. Washburn, Ed., First Edition, McGraw-Hill, 1930.
- Lawler, D., Onondaga County Department of Drainage and Sanitation, personal communication, 1975.
- Lorenzen, M. and R. Mitchell, "Theoretical Effects of Artificial Destratification in Impoundments", <u>Environmental Science and Tech-</u> <u>nology</u>, Vol. 7, No. 10, pp. 939-944, October 1973.
- 16. Lorenzen, M.W., D.J. Smith, and L.V. Kimnel, "A Long Term Phosphorus Model for Lakes: Application to Lake Washington", presented before the Division of Environmental Chemistry, American Chemical Society, Philadelphia, April 1975.
- Malueg, K.W., P. Larsen, D.W. Schults, and H.T. Mercier, "A Six-year Water, Phosphorus, and Nitrogen Budget for Shagawa Lake, Minnesota", Journal of Environmental Quality, Vol. 4, No. 2, pp. 236-242, 1975.
- 18. Meyer, A.F., <u>Evaporation from Lakes and Reservoirs</u>, Minnesota Research Committee, St. Paul, Minnesota, June 1947.
- Newbold, J.D. and J.A. Liggett, "Oxygen Depletion Model for Cayuga Lake", Journal of the Environmental Engineering Division, American Society of Civil Engineers, Vol. EE1, pp. 41-59, Pebruary 1974.
- O'Brien and Gere, Engineers, "Onondaga Lake Monitoring Program", Annual Reports to Onondaga County, New York, Department of Public Works, 1970-74.
- O'Brien and Gere, Engineers, "Supplement to Wastewater Facilities Report - Metropolitan Sewage Treatment Palnt Expansion", prepared for Onondaga County, April 1971.
- 22. Odum, E.P., <u>Fundamentals of Ecology</u>, Third Edition, W.B. Saunders Co., 1971.
- Onondaga County, New York, Department of Drainage and Sanitation, "Performance Reports for the Metropolitan Sewage Treatment Plant", 1968-74.
- 24. Onondaga County, New York, "Onondaga Lake Study", US EPA, Water Quality Office Publication 11000 FAE 4/71, April 1971.
- 25. Otsuki, A. and R.G. Wetzel, "Calcium and Total Alkalinity Budgets and Calcium Carbonate Precipitation in a Small Hard-Water Lake", <u>Arch. Hydrobiol.</u>, Vol. 73, No. 1, pp. 14-30, February 1974.

### REFERENCES - CHAPTER 3 (continued)

())

[:

( )

- 26. Process Research, Inc., "Final Report on the Storrow Lagoon Demonstration Project", prepared for the Metropolitan District Commission, Commonwealth of Massachusetts, 1975.
- Reinsch, C.H., "Smoothing by Spline Functions", <u>Numerical Mathematics</u>, Vol. 10, No. 117, 1967.
- Rooney, J., "Evaluation of Interim Basin Plan, IBP-NY-07-07, for Lake Onondaga, N.Y. and Project WPC-NY-659 (Syracuse Metro STP)", EPA Region II Office, New York, 1973.
- Shannon, E.E., "Effects of Detergent Pormulation on Wastewater Characteristics and Treatment", <u>Journal of the Water Pollution Control</u> <u>Federation</u>, Vol. 47, No. 10, pp. 2371-2383, October 1975.
- 30. Sinton, D. and C. Steinitz, "GRID, Version 3", Laboratory for Computer Graphics and Spatial Analysis, Harvard University, October 1971.
- Spirtas, R. and H.J. Levin, "Patterns and Trends in Levels of Suspended Particulate Matter", Journal of the Air Pollution Control Association, Vol. 21, No. 6, pp. 329-333, June 1971.
- 32. Stefan, H. and D.E. Ford, "Temperature Dynamics in Dimictic Lakes", Journal of the Hydraulics Division, American Society of Civil Engineers, Vol. HY1, pp. 97-114, January 1975.
- 33. Stumm, W. and J.J. Morgan, <u>Aquatic Chemistry</u>, Wiley-Interscience, New York, 1970.
- 34. Sze, P., "The Phytoplankton of Onondaga Lake, N.Y.", Ph.D. Thesis, Cornell Univeristy, Limnology, 1972.
- 35. Thomann, R.V., D.M. DiToro, and D.J. O'Connor, "Preliminary Model of Potomac Estuary Phytoplankton", <u>Journal of the Environmental</u> <u>Engineering Division</u>, <u>American Society of Civil Engineers</u>, Vol. EE3, pp. 699-715, June 1974.
- US Department of Commerce, Weather Bureau, "Local Climatological Data", Syracuse, New York, Hancock Airport, 1968-74.
- 37. US EPA, "Effluent Limitation Guidelines and Standards of Performance and Pretreatment for Inorganic Chemicals Manufacturing, Point Source Category 40 CFR, Part 415", Federal Register 38/1967: 28174-28194, 1973.
- 38. US EPA, Region II, "Environmental Impact Statement on the Wastewater Treatment Facilities Construction Grants for the Onondaga Lake Drainage Basin", May 1974.

# REFERENCES - CHAPTER 3 (continued)

( )

(

()

- 39. Vollenweider, R.A., "Scientific Fundamentals of the Eutrophication of Lakes and Flowing Waters, with Particular Reference to Nitrogen and Phosphorus as Factors in Eutrophication", Office of Economic Cooperation and Development, Paris, September 1970.
- 40. Vollenweider, R.A., "Input-Output Models", unpublished manuscript, 1973.
- Wold, S., "Spline Functions in Data Analysis", <u>Technometrics</u>, Vol. 16, No. 1, pp. 1-11, February 1974.

# 4.0 A MODEL FOR VERTICAL STRATIFICATION IN ONONDAGA LAKE

ſ

The previous chapter has relied primarily upon monitoring data as a basis for analysis of Onondaga Lake water quality problems. This chapter employs some of these data in the context of a modelling study which illustrates some of the general parameter estimation, sensitivity analysis, and error analysis techniques discussed in Chapter 1. The focus of the study has been selected: (1) to be based upon some of the key relationships observed among the water quality variables in Chapter 3, and (2) to provide input to specific design decisions which will have to be made by Onondaga County in the near future, as discussed below.

### 4.1 Objectives - The Outfall Design Issue

The dissolved oxygen standard for Class B (contact recreational) waters in New York State is 4 mg/liter. This standard is often violated in the epilimnion, as well as the hypolimnion of Onondaga Lake, and compliance with this standard is a primary water quality management objective. In Chapter 3, an apparently increasing trend in dissolved oxygen levels over the course of the monitoring period was noted. This trend was shown to correlate with an apparently decreasing trend in density stratification. The suggestion that density stratification was partially controlling hypolimnic oxygen levels is consistent with results of a modelling study by Bella <sup>5</sup>, who has shown that hypolimnic DO levels in stratified levels are generally most sensitive to vertical

ſ

€

diffusion rates. These rates, in turn, depend upon density structure.

In Chapter 3, it was also estimated that future installation of tertiary treatment facilities would result in a maximum reduction of about 60% in the total oxygen demand loadings to the lake, including both external and internal (photosynthetic) sources. The response of ambient DO levels to this reduction will depend upon the availability of oxygen resources to satisfy this demand, i.e., upon reaeration rates, which, in turn, will be sensitive to density stratification through its controlling influence on vertical mixing rates. This raises concern over the potential impacts of the plan to combine the Metro STP effluent with the saline waste from Allied Chemical in order to effect phosphorus removal. Currently, the latter is discharged to Ninemile Creek, which affords about twice as much dilution on the average as will the Metro STP effluent. The plan will double the average density difference between the saline stream and ambient waters at the point of entry into the lake. By increasing the tendency of the effluent to sink into the hypolimnion, the plan may influence density stratification and vertical mixing rates. This tendency to sink could be reduced by discharging through a diffusing system which would force immediate dilution of the effluent in the ambient waters and thereby reduce the gravitational driving force for a density current.

The proposed method of discharging the tertiary effluent from the Metro STP is through a surface, shoreline outfall<sup>22</sup>. Because of

(

( :

the salinity of the Allied Chemical waste, the combined municipal/ industrial effluent is expected to have a chloride concentration in excess of 7000 mg/l, compared with an average lake concentration of 1700 mg/l. Under all realistic temperature regimes, the effluent will be much more dense than the ambient lake waters at the point of discharge. The tendency of the effluent to sink into the hypolimnion will depend upon the degree of initial dilution of the effluent in the ambient waters in the vicinity of the outfall site, as well as upon lake and discharge temperatures. The shoreline, surface outfall design has grown partially out of concern for the potential water quality impacts of discharging a relatively dense effluent<sup>22</sup>. Economic considerations have also been of importance, the shoreline discharge alternative being undoubtedly the least expensive one<sup>22</sup>.

James Rooney of EPA Region II<sup>26</sup> has reviewed and criticized this aspect of the Syracuse plan. In Rooney's discussion, reproduced below, the environmental reasons for the proposed shoreline discharge are cited and an opposing case for an offshore outfall design is made:

"The IBP and reference material do not adequately substantiate the desirability of the proposed surface-shoreline discharge. The items raised in support of this discharge method were as follows:

- a) '...to ensure distribution of the treatment plant effluent in the epilimnetic waters of the lake,' and thereby provide maximum mixing of the plume with ambient water and some degree of bio-degradation before sinking into the hypolimnion.
- b) '...to prevent the accumulation of any organic material in the hypolimnetic waters of the lake' and thereby preclude the possibility that the productive epilimnetic volume of the lake would decrease.

( :

C

Ċ

c) '...to prevent the tendency...to create a permanent stratification within the lake.'

The studies completed to date, however, are inconclusive with respect to the water quality impact of the proposed discharge and do not demonstrate the potential for any of the aforementioned responses. On the contrary, it appears that the discharge alternative consisting of a submerged outfall would probably be less detrimental to the lake quality and aesthetics due to the following considerations:

a). A submerged discharge would lessen the severity of the existing hypolimnetic conditions by the introduction of relatively high DO waters (5 mg/l) into a region which is essentially devoid of oxygen for approximately 7 months out of the year; the discharge, if located in the general vicinity of the thermocline (9-12 meters depth), would through proper design, also enhance the entrainment of some hypolimnetic waters into the epilimnion; this could possibly provide for a limited decomposition and subsequent reduction of the residual organic load in the hypolimnion where mean BOD5 values are on the order of 9.0 mg/l.

b). A submerged outfall will preclude the potential for future aesthetic degradation which may present a considerable problem due to the proposed introduction of the Allied Chemical wastewater into the municipal system; the plume visibility aspect and the related possibility of nearshore discoloration were not discussed in any of the reference material when evaluating the merits of surfaceshoreline discharge.

c). A submerged outfall will, by its very nature, provide greater initial dilutions than a surface discharge of comparable design; thus, the area affected by the relatively high concentrations of the effluent parameters (pH, ammonia, chlorides, total dissolved solids [TDS]) will be minimized under the submerged discharge alternative; in addition, the area most severely affected by the potentially adverse concentrations of these constituents will be located in the relatively unproductive waters of the hypolimnion.

d). The submerged discharge alternative will preclude the possibility of any accumulation of wastewater constituents along the shallow region (3 feet depth) adjacent to the proposed plant (22) site; the effluent dispersion model adapted for the plume analysis does not indicate that this may be a problem, however, the utilization of this model is questionable in that (1) the model itself and its application to non-thermal discharges is unverified, (2) the model does not account for the limiting effects of impingement and reduced entrainment of ambient waters due to the shallow nature of the lake ( $\leq$  3 feet) near the discharge point, and, (3) the model evaluates the proposed discharge under quiescient conditions and does not address potentially more critical conditions which may

( :

(3

occur due to specific meteorologic and/or hydrologic conditions.

In summary, there is insufficient data available at the present time concerning the lake hydrodynamics, in general, and circulation patterns in the vioinity of the proposed shoreline discharge, in particular, to permit adequate projections on the water quality response from the surface discharge alternative. It is likely, however, that due to the presence of the shoal area adjacent to the plant site, the natural assimilative capacity and the potential for significant entrainment of ambient waters into the proposed effluent plume will be minimal while the possibility for aesthetic damage will be maximized. Based on these conclusions, it appears that a submerged (or surface) discharge located further out in the lake would be the most practical (and conservative) alternative for the enhancement of water quality in the lake."

In an Environmental Impact Statement on the project, the EPA<sup>37</sup> discussed the advantages and disadvantages of discharging to the epilimnion and to the hypolimnion of the lake. The greater availability of oxygen in the eurface waters was considered an important justification for the surface discharge; however, it was considered likely that the dense effluent would sink into the hypolimnion before bio-oxidation of the relatively inert organic matter in the discharge. Because of uncertainty in the initial dilution rates and changes in lake and discharge temperatures, the report concluded that is was "impossible to predict how long the effluent would remain in the epilimnion".

The major disadvantage of the surface discharge was considered to be the plume visibility which would result from calcium carbonate precipitation. This reaction would occur when the excess calcium in the effluent mixes with alkalinity in the ambient lake waters. Control of effluent pH would limit this problem to some degree. The possibility was raised that a subsurface discharge could disturb the lake's bottom

( )

();

sediments, releasing nutrients, organic material, and mercury, although this was largely unsubstantiated.

The EPA<sup>37</sup> concluded that the major advantages of the surface outfall were low construction and maintenance costs. It was estimated that capital costs for a subsurface outfall would amount to an additional one million dollars. The flexibility of the shoreline, surface discharge alternative was also cited as an advantage. If problems (probably aesthetic) were to arise with the design, the outfall could be extended off-shore.

The outfall design problem has apparently been an issue of some controversy. In this chapter, a model is developed and applied for the purpose of addressing aspects of this issue. The study does not address the effects of specific details of outfall design, but rather it is concerned with general design specifications. The two major "decision variables" studied are: (1) discharge location (epilimnion or hypolimnion), and (2) degree of initial dilution. The model is concerned with vertical mixing rates in the lake and formulates heat and mass balances on the epilimnion and hypolimnion to simulate variations in lake temperatures and chloride levels. It could serve as a basis for future development of a more general model for predicting non-conservative water quality components. As noted above, this work also provides a context in which to illustrate some of the parameter estimation, sensitivity analysis, and error analysis techniques discussed in Chapter 1.

## 4.2 Review of Lake Vertical Stratification Models

()

ſ

(

One-dimensional models for vertical density stratification in lakes and reservoirs can be grouped into two general categories: dispersion models and mechanical energy balance models. In the former, the vertical transport of heat is modelled as a combination of diffusive, convective, and advective processes, based upon the mass-transport equation. In the latter, the vertical mixing process is governed by the relationship between wind-induced kinetic energy in the surface layers and the buoyant potential energy deficit resulting from density stratification. Of the two groups, the former have generally seen more widespread application in lake and reservoir modelling. The various discrepancies in the assumed dominant mechanisms both within and between the two groups of models reflect the fact that there is still some general disagreement as to the relative importance of various vertical transport mechanisms.

## 4.2.1 Mass Transport Models

Three major models of the first type have been developed by people from MIT<sup>8,11,13,14</sup>, Cornell University<sup>30,31,32,33</sup>, and Water Resources Engineers (WRE)<sup>42,24</sup>. Table 4.2-1 is a summary of the dominant features of the models, as derived from an evaluation by Parker et al.<sup>25</sup>. The only mixing mechanism shared by all three of these models is the convective (instantaneous) mixing

# Essential Aspects of Deep Reservoir Temperature Models Evaluated by Parker et al

| Aspect                                  | 30,31,32<br>Cornell Model                                                                                                                                                                                                                                                                            | 11,13<br>H.I.T. Model                                                                                                                     | 24,42<br>W.B.F. Model                                                                                                                                                                                                                                                                             |
|-----------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Surface Bondary<br>Conditions Specified | One of the following:<br>(1) water surface temperature<br>(2) heat flur at surface<br>(3) equilibries temperature<br>Priories velocity<br>(All B.C. input in sizesoidal form)                                                                                                                        | Networologic data:<br>[1] short-wave solar radiation<br>(2) wind speed<br>(3) cloud cover<br>(4) relative humidity<br>(5) air temperature | Neteorologic data <sup>b</sup><br>(1) short-wave solar radiation<br>(2) wind speed<br>(3) cloud cover<br>(4) relative humidity<br>(5) strospheric pressure<br>(6) wet bulb temperature<br>(7) dew point<br>(8) air temperature                                                                    |
| Diffusivity                             | Eddy diffusivity<br>$K_{H} = K_{H_{0}} (1 + oR)^{-1}$<br>$R = -(gr^2 a_{j}/a_{e}^{2}) \frac{32}{32}$<br>$K_{H_{0}}^{-} = (c_{\pm} + c_{\pm}x) u_{e}$<br>$\omega_{\pm} = friction valueity$<br>z = depth<br>$a_{T} = coefficient of values expansion \sigma, c_{\pm}, c_{\pm} = expirical parameters$ | Nolecular diffusivity                                                                                                                     | Empirical diffusion coefficient<br>$D_c = \lambda_1$ , $E \leq E_c$<br>$D_c = \lambda_2 E^{\lambda_2}$ , $E \geq E_c$<br>$E = -\frac{1}{\rho} \frac{3\rho}{2\pi}$ = stability criterion<br>$\lambda_1$ , $\lambda_3$ , $\lambda_3$ = empirical parameters<br>$E_c$ = critical stability parameter |
| Rydrologie Boundary<br>Conditions       | Crastant volume<br>Bo net inflows or outflows                                                                                                                                                                                                                                                        | Variable volume<br>Influent streams enter at point<br>of density equivalence, after<br>inicial dilution                                   | Variable volume<br>Influent streams enter at point.<br>of density equivalence                                                                                                                                                                                                                     |
| time Stap                               | Variable                                                                                                                                                                                                                                                                                             | Adjusted during simulation,<br>hased upon numerical stability<br>criterion<br>Typically 1 day                                             | Constant for a given simulation<br>Value based upon numerical stability<br>criterion<br>Typically 1 day                                                                                                                                                                                           |
| Depth Increment                         | Constant<br>Nations 100 intervals<br>Typically 5 feet                                                                                                                                                                                                                                                | Constant<br>20< Intervals < 50<br>Typically 1 - 2 meters                                                                                  | Constant<br>Maximum 200 intervals<br>Typically 1 - 2 meters                                                                                                                                                                                                                                       |
| Running Time <sup>4</sup>               | 11.86 min/ymar<br>dy = 5 feat                                                                                                                                                                                                                                                                        | 1.7 min/year                                                                                                                              | 3.3 min/year                                                                                                                                                                                                                                                                                      |

a - CPU time exclusive of program loading ; computer model not specified, but the same for all. b - Meteorologic data input on a daily basis; some variables computed internally as functions of others.

25

### 4.2.1 Mass Transport Models (continued)

( )

(

( )

which occurs in regions of instability, as determined by the sign of the density gradient. Solutions to the governing partial differential equations are achieved by finite difference techniques. All are characterized by relatively high degrees of spatial and temporal resolution (on the order of 1 meter and 1 day, respectively), and are accordingly relatively expensive to implement. The three models differ primarily in the mechanisms responsible for the vertical transport of heat during stably stratified periods.

In the Cornell and WRE models, diffusion (eddy or "effective", respectively) is the dominant transport mechanism. Application of either of these models requires specification of a number of parameters required to compute the time- and depth-variable diffusion coefficient as a function of some measure of stability (the Richardson number, in the case of the Cornell model, or normalized density gradient in the case of the WRE model). Neither of these models has been applied extensively enough in order to establish the validity of the functional forms or parameter values over a range of reservoirs. Hence, the parameters must be viewed as empirical and may need to be re-estimated for each new model application.

The MIT model, on the other hand, employs molecular diffusion and advection as dominant vertical transport mechanisms during stably stratified periods. Wind-induced mixing is ignored and mixing

## 4.2.1 Mass Transport Models (continued)

(

1

due to an unstable density profile accounts for convection in the epilimnion. One advantage of the MIT model is that it does not require the specification of empirical parameters for calculation of diffusion coefficients, since the molecular diffusion coefficient is constant.

In comparing the three models, Parker et al. 25 concluded that the MIT model was preferable, based upon ease of application and predictive capability. However, this conclusion may be biased, since the predictive capabilities of the three models were compared using data from Fontana Reservoir, the same reservoir used as a data source in the development of the MIT model. After correcting some apparent coding errors in the WRE diffusion coefficient calculation scheme, Parker et al. found that the WRE predictions were relatively insensitive to an order of magnitude increase in the effective diffusivity coefficient. Thus, the uncertainty in the eddy diffusivity parameters may not be a serious drawback to application of this model. The primary drawback of the WRE model was its relative sensitivity to assumed depth increment over the range tested (0.6 m - 2 m). The major disadvantages of the Cornell model were found to be: (a) its failure to account for variable depth versus surface area profile in the lake; (b) its lack of ability to consider advective flows through the reservoir (and the vertical mixing attributed to them); and (c) an apparent over-sensitivity of

· 4-10

(

(

()

the calculated eddy diffusivity to wind velocities over the lake.

### 4.2.2 Mechanical Energy Balance Models

The second major group of models characterizes vertical mixing in a lake by considering the relationship between kinetic energy input at the surface due to wind shear stress and the change in buoyant potential energy which occurs with mixing. The development of density stratification is attributed to the differential absorption of thermal energy in the surface layer. The works of Turner et al.<sup>34,35</sup>, Kato and Phillips<sup>18</sup>, and Stefan and Ford<sup>29</sup> are representative of this group of models. The first two are supported primarily by data from laboratory experiments and achieve qualitative agreement with the observed behavior of stratification patterns in the ocean and lakes. Stefan and Ford's model has been developed specifically to simulate conditions in a lake.

The laboratory experiments conducted by Turner et al. and by Kato and Phillips involved measurements of the rate of entrainment across the interface of two superimposed layers of fluid of slightly different densities. Entrainment rates were studied as a function of density gradient and rate of mechanical energy input to one or both of the layers. The major difference between the series of

# 4.2.2 Mechanical Energy Balance Models (continued)

experiments by Turner et al. and by Kato and Phillips was in the mode of mixing employed. Turner used grid stirring in one or both layers, while Kato and Phillips introduced kinetic energy by applying a known horizontal stress to the surface of the upper layer. The latter method, according to the authors, is more representative of field conditions (surface wind stress) and reduces the problem of having to define the length and velocity scales of the turbulence induced by grid stirring in order to apply experimental results to field conditions. Because of this problem, in some respects, it is difficult to relate the results of the two series of experiments quantitively.

Both Turner et al. and Kato and Phillips utilized a model of the following form to represent their data:

$$E = \frac{U_E}{U_*} = k R^n \qquad (4.2~1)$$

$$R = \frac{gZ\Delta\rho}{U_{\bullet}^{2}\rho} \approx \frac{g(\partial\rho/\partial Z)Z^{2}}{2\rho U_{\bullet}^{2}} \qquad (4.2-2)$$

where,

( )

1

E = entrainment ratio (dimensionless)  $U_p =$  entrainment rate (measured) - ( $\ell/t$ )

## 4.2.2 Mechanical Energy Balance Models (continued)

ſ

(

 $U_{\pm}$  = surface friction velocity - (l/t)  $\rho$  = density of surface layer - ( $m/l^3$ )  $\Delta \rho$  = density jump across interface - ( $m/l^3$ ) Z = depth of surface layer - (l) g = acceleration of gravity - ( $l/t^2$ ) R = Richardson number (dimensionless)  $n_{,k}$  = empirical parameters (dimensionless)

As discussed above, Kato and Phillips employed the surface friction velocity and depth of the mixed surface layer as surrogates for the velocity and length scales of the induced turbulence, respectively. Turner, on the other hand, did not explicitly define these scales and utilized stirring rates (cycles/time) as a surrogate for the velocity scale. The unknown length scale and the proportionality constant between grid stirring rate and the time velocity scale are thus incorporated into the empirical parameter k, to be estimated from the experimental data. The slope, n, is, however, independent of the unknown scale factors and thus can be used as a partial basis for comparing the two sets of experimental results.

Turner et al.<sup>35</sup> found that the slope parameter, n, was approximately -1 in the lower range of Richardson numbers investigated and

## 4.2.2 Mechanical Energy Balance Hodels (continued)

( :

T

approached -1.5 at higher Richardson numbers, where molecular diffusion became important. The molecular diffusion effects were deduced from observations that, at high Richardson numbers, entrainment rates were significantly lower when differences in salt concentration were used to impose the density gradient, as compared to when temperature differences were used. This could be explained by the fact that the molecular diffusivity of heat is on the order of one hundred times that of salt. It is not possible to determine whether the divergence between salt and heat transport is of consequence under field conditions because of the scaling problem, i.e., the range of effective Richardson numbers over which this phenomenon was observed may be higher than that typically observed in natural systems.

The experimental results of Kato and Phillips are shown in Figure 4.2-1. In fitting their data, the authors constrained the slope, n, to -1 and derived a k value of 2.5, stating that the latter is "uncertain to within about 30%". The authors noted that a least-squares estimate of the slope would have been somewhat greater than -1. The slope was constrained to illustrate the similarity of the results to those of Turner et al. at low Richardson numbers and to support the theoretical energy balance arguments discussed below. In these experiments, no evidence was found of the -1.5 dependence of E on R observed by Turner at high Richardson numbers. Because of the scaling problem discussed above, it is uncertain whether this

# Figure 4.2-1

Ć

C

( )

Entrainment Ratio Versus Richardson Number - Experimental Data of Kato and Phillips<sup>18</sup>



FIGURE 6. The entrainment coefficient B as a function of the overall Richardson number. The parameters represented by the symbols are indicated in the table below:

| •                 | $d\rho/dz = 1.92 \times 10^{-4}$ | $3.94 \times 10^{-3}$ | 7·09 × 10 <sup>-4</sup> c.g.s. |
|-------------------|----------------------------------|-----------------------|--------------------------------|
| 7₀ = 0.995 c.g.e. |                                  |                       | ×                              |
| 1-485             | Δ .                              | ▲                     |                                |
| 2.12              | 0                                | •                     | +                              |
| 8-75              | V                                | . <b>.</b>            | T                              |

## 4.2.2 Mechanical Energy Balance Models (continued)

constitutes a discrepancy between the two sets of experimental results.

As pointed out by Kato and Phillips, a slope of -1 in the E versus R relationship is evidence that the rate of change of potential energy in the system due to changes in the density stratification pattern is proportional to the rate of kinetic energy dissipation in the well-mixed surface layer. This is illustrated by calculations outlined in Figure 4.2-2. For a slope of -1, the rate of change of potential energy in the system is k/2 times the rate of kinetic energy input to the surface layer. Thus, the k value of 2.5 derived from the results of Kato and Phillips indicates that the rate of potential energy increase in their system was 125% of the rate of kinetic energy dissipation. (Their experimental system was adiabatic.) This apparent problem was not discussed in their article, although the authors did state that the k value was uncertain to within 30%, indicating that the observed energy conversion efficiency was not significantly different from 100%. However, this figure still seems higher than would be acceptable if other kinetic energy sinks (primarily heat generated by viscous damping) were taken into account. This suggests that the velocity and/or length scales selected by Kato and Phillips may not have been accurate and thus that the k value may not be appropriate for direct application in other systems, although the slope would be independent of such errors.

( )



Initial State

Pinal State

Buoyant Potential Energy Per Unit Area :  $PE_{i} = g[\rho_{E} z (\Delta z + z/2) + \rho_{H} \overline{\Delta z^{2}/2}]$  $PE_{g} = g[\rho (z + \Delta z) / 2]$ 

 $\frac{\text{Mass Balance :}}{\overline{\rho}(z + \Delta z) = \rho_E z + \rho_H \Delta z$ 

 $\frac{\text{Potential Energy Change With Mixing :}}{\Delta \text{PE} \Rightarrow \text{PE}_{f} - \text{PE}_{i} = .5 \text{ g z } (\rho_{H} - \rho_{E}) \Delta z$ 

Entrainment Rate :

 $U_e = \Delta z / \Delta t$ 

 $\Delta PE/\Delta t = .5 g z (\rho_H - \rho_E) \Delta z / \Delta t = .5 g z (\rho_H - \rho_E) U_e$ . .

Model :

$$E = U_{e}/U_{*} = k / R = k U_{*}^{2} \rho_{E} / [(\rho_{H} - \rho_{E}) g z]$$
  
$$\Delta PE/\Delta t = .5 k U_{*}^{3} \rho_{E}$$

 $\frac{\text{Rate of Kinetic Energy Dissipation :}}{\Delta KE/\Delta t} = \rho_E U_{\pm}^3$ 

Mechanical Energy Conversion :

 $\Delta PE/\Delta t = .5 k \Delta KE/\Delta t$ 

## 4.2.2 Mechanical Energy Balance Models (continued)

(

Stefan and Ford have extended the above two-layered models to a multi-layered one and have coupled the mechanical and thermal energy balance equations to permit simulation of temperature dynamics in dimictic lakes. Figure 4.2-3 depicts the energy transformations simulated by this model. All of the mechanical energy input from the wind is assumed to be converted to turbulence. The downward movement of the lower edge of the mixed surface layer is simulated using a mechanical energy balance criterion. At each time step, the model evaluates the change in potential energy which would occur if the next depth increment (of approximate thickness 0.25 m) below the mixed surface layer were to be entrained. This is compared to the rate of kinetic energy dissipation in the surface layer. If the ratio of kinetic energy input to potential energy change is greater than one, entrainment occurs; if the ratio is less than one, the kinetic energy is assumed to be dissipated as heat and no change in the depth of the mixed layer occurs. The thermal energy fluxes are modelled primarily using formulations suggested by Dake and Harleman<sup>6</sup>.

The model achieved general qualitative agreement with data from Lake Calhoun and Halstead Bay, Lake Minnetonka, Minnesota. In the simulation, heat energy and mechanical energy were input successively at time steps of either one day or three hours (in accordance with the availability of meteorologic data). The authors noted that the computed temperature profiles were rather sensitive to the time step





### 4.2.2 Mechanical Energy Balance Models (continued)

employed, as well as the assumed radiation attenuation coefficient. The latter conclusion was also reached by Parker et al. in their evaluation of the MIT deep reservoir model, discussed in Section 4.2.1. Since the attenuation coefficient is a partial function of algal density, Stefan and Ford suggested that this may be an important means by which the blota may influence the temperature distributions and vertical mixing rates in lakes and reservoirs. Light-limited algal populations would benefit from such effects.

## 4.2.3 Statistical Studies

(

A statistical study on vertical mixing data from several morphologically-different lakes and oceanic areas in the northern temperate zone was performed by Blanton<sup>6</sup>. The measure of mixing rate employed was the mean rate of entrainment into the epilimnion, as determined by the average rate of migration of the thermocline during the stratified period. The study excluded periods of rapid heating in the spring and convective mixing in the autumn. Entrainment was correlated with the mean stability across the thermocline, as computed from the temperature profiles at the beginning and end of the study period for each lake. The measure of stability employed was the mean density gradient. Because the mean stability was calculated using only data from the beginning and end of the stratified

# 4.2.3 Statistical Studies (continued)

period, the estimates may be somewhat biased. Stability tends to go through a maximum during the stratified period, so that estimates of mean stability calculated from data at the ends could tend to be too low.

Blanton presented his data as a plot of stability versus entrainment rate and summarized it with the following relationship:

> $X = a Y^b$   $a = 8.95 \times 10^{-7}$  b = -0.521r = -0.84

(4.2 - 3)

where,

(

**(**)

 $Y = \frac{1}{A_0} \frac{\Delta V_E}{\Delta t} = \text{entrainment rate (m/sec)}$  $X = \frac{q}{\rho} \frac{\Delta \rho}{\Delta z} = \text{mean stability (sec}^{-2})$ 

The statistical basis for the above relationship was not stated in Blanton's article. In the above form, the dependent variable is stability and the independent variable is entrainment rate. From a cause-effect standpoint, it would seem more logical to reverse

# 4.2.3 Statistical Studies (continued)

these roles. Accordingly, Blanton's data have been replotted in Figure 4.2-4 as entrainment rate versus stability on logarithmic scale. The least squares relationship derived from linear regression on log transformed data is:

> $Y = a x^{b}$  (4.2-4)  $a = 1.71 \times 10^{-11}$  b = -1.63r = -0.91

The correlation coefficient, -0.91, is somewhat higher in absolute value than that computed by Blanton, -0.84, suggesting that Blanton may have computed his coefficient on a linear scale.

In relating entrainment rate to stability, equation (4) is analogous to the relationship of E versus R [equation (1)], used to represent the experimental data of Kato and Phillips and of Turner et al.  $^{34,35}$ . The 95% confidence interval for the slope of the Y versus X relationship is -1.63 ± 0.34, somewhat higher in absolute value, but not significantly different from the range of slopes in the E versus R relationship observed in the laboratory experiments, -1.0 to -1.5.

C





(

 $\mathbf{C}$ 

1

## 4.2.3 Statistical Studies (continued)

There are two factors which might have contributed to the relative steepness of the slope in Blanton's data. First, one would expect a negative correlation between wind velocities and mean stability. Had the Richardson number been used as a measure of stability, the slope would have been flattened. Blanton noted that it would have been preferable to have expressed stability in the form of a Richardson number. However, the necessary wind data were not available. The second factor which might have contributed is the possible bias in Blanton's estimate of mean stability. If one assumes that the peaking in stability during the stratified period would tend to be more pronounced in lakes with higher stability, elimination of this bias in the mean stability estimate would also tend to flatten the slope. Even without these considerations, Blanton's data from a wide range of lakes and oceanic regions are not inconsistent with the experimantal results discussed previously.

Blanton also noted a strong correlation between entrainment rate and mean depth (Figure 4.2-5). Snodgrass<sup>28</sup> summarized Blanton's data with the following regression equation:

$$Y = 8.65 \times 10^{-9} \overline{z}^{1.08}$$
 (4.2-5)  
r = 0.887




C

(

## 4.2.3 Statistical Studies (continued)

where,

 $Y = \frac{1}{A_0} \frac{\Delta V_E}{\Delta t} = \text{entrainment rate (m/sec)}$ 

 $\overline{Z}$  = lake mean depth (m)

Snodgrass, using a different set of lakes, also found that the relationship between the vertical transport coefficient across the thermocline, expressed as a diffusivity, and mean depth (Figure 4.2-6) could be summarized by :

$$K_{\rm TH} = 0.00682 \,\overline{z}^{1.12}$$
 (4.2-6)  
r = 0.924

where,

T

()

 $K_{TH}$  = vertical transport coefficient (m<sup>2</sup>/day)

The apparent influence of mean depth on vertical transport rate, expressed either as entrainment or as diffusivity, was explained by Blanton with reference to the mechanical energy theory of Kato and Phillips. Blanton's hypothesis was that deeper lakes have

#### 4.2.3 Statistical Studies (continued)

( )

larger basins, greater available wind fetches, and, hence, greater kinetic energy input rates per unit area. This explanation is examined in more detail below.

In each set of data, the mixing rate variable ( Y or  $K_{m_{\rm H}}$  ) had a range of three orders of magnitude. A similar range of kinetic energy input rate per unit area would be necessary in order for the mechanical energy theory to account for this variation. The rate of kinetic energy input varies as the cube of the surface friction velocity (Figure 4.2-2). Friction velocity is proportional to wind velocity to the 1.25th power, over a moderate range of wind velocities, 1-15 m/sec<sup>44</sup>. Hence, kinetic energy input rate varies roughly as the 3.75th power of wind velocity, and about a 6.3-fold range in wind velocity would be required to explain the thousand-fold range in mixing rate. This range may be reasonable, although specific data on wind speed as a function of lake mean depth would be necessary in order to substantiate Blanton's argument. One factor ignored by Blanton is circulation induced by Coriolis forces, which would also tend to be more important in larger lakes<sup>15</sup>. This would serve as an additional source of kinetic energy and would thus reduce the 6.3-fold variation in wind velocities requires in order for the observed variation in mixing rates to be explained by variation in kinetic energy input rates.

#### 4.2.3 Statistical Studies (continued)

Another factor which might contribute to the observed correlation between mixing rate and depth is the apparent sensitivity of temperature profiles to radiation extinction coefficients, noted in modelling studies by Parker et al. and by Stefan and Pord. In many lakes, the rate of attenuation of radiation with depth is largely a function of algal density<sup>20</sup>. Vollenweider<sup>40</sup> has demonstrated the negative correlation between lake trophic state and mean depth. These facts suggest that lower algal densities and lower extinction coefficients would be typical of lakes with greater mean depths. A lower extinction coefficient would permit a more even distribution of radiation absorption with depth, giving rise to a more uniform temperature distribution and greater mixing rate. Thus, the apparent correlation between mixing rate and mean depth may be explained, in part, by biologic, as well as energetic effects.

#### 4.3 Model Development

Essential aspects of the model which has been developed for simulating vertical mixing in Onondaga Lake are summarized in Figures 4.3-1 and 4.3-2 and Tables 4.3-1 to 4.3-4. Figure 4.3-1 defines the system. The water, heat, and chloride fluxes corresponding to the various streams in Figure 4.3-1 are given in Table 4.3-1. "Forcing functions", which drive the model, are given in

(

{

Figure 4.3-1



H<sub>1</sub> = heat flux in stream 1 (KKcal/day)

M<sub>1</sub> = chloride flux in stream 1 (g/day) .

 $T_{\rm E}$ ,  $T_{\rm H}$  = epilimnion and hypolimnion temperatures (°C)

 $C_{g}$ ,  $C_{H}$  = epilimnion and hypolimnion chloride concentrations  $(g/m^3)$ 



Figure 4.3-2 .

Control Pathways in the Onondaga Lake Vertical Stratification Model

4-30

0

Table 4.3-1 Watar, Heat, and Chloride Fluxes

(

**(**)

| 74     | Deteription                            | Q <sub>1</sub> = 510w (a <sup>3</sup> /d <sub>m</sub> ) | E, - Hoat flux (Nical/day)                                          | N Citorida Nim Jakan                                                |
|--------|----------------------------------------|---------------------------------------------------------|---------------------------------------------------------------------|---------------------------------------------------------------------|
| -      | Tributury Inflow                       | <b>D</b>                                                |                                                                     |                                                                     |
| ~      | - pilomt                               | 1<br>1<br>1                                             |                                                                     | "1"23<br>11 h f                                                     |
| =      | " Discharge to Mail.                   | v, û+e,) û-r,)                                          | 1 1 2<br>4 (0,.42)(1-7.)                                            |                                                                     |
| •      | · Discharge to Bypo.                   | U, (144, 12,                                            | U, (U, A, T, )7.                                                    |                                                                     |
| ŝ      | Natto STP Inflow                       |                                                         |                                                                     | 1.21.62.1                                                           |
|        | * pilomt                               |                                                         | 2 21<br>1 a.T.                                                      | 12,24                                                               |
| •      | - Discharge to Epil.                   | 0_(1++_)(1-F_)                                          | 2 2 2 2 1 1 2 2 2 1 1 - 2 1 2 2 1 1 - 2 1 2 2 1 2 2 1 2 2 2 2       | 2722<br>0.01 + c 10-0                                               |
| 8      | Discharge to ligo.                     | U_(2+m_)P,                                              | 1 0.(0+4.7.]P.                                                      |                                                                     |
| •      | Winchile Inflow                        | <u>u</u> ,                                              | 10 <sup>-10</sup> -2<br>10-10-2                                     | 21-24 - 242-2                                                       |
| 3      | - pilone                               | 0,4<br>1                                                |                                                                     | 73725<br>1 6 F                                                      |
| #      | * Discharge to thil.                   | [0,(2++,)+0,,+0,,10-7.)                                 | 3 3 5<br>  [0, (V+a, T. ) +H+H) (3+P. )                             |                                                                     |
| 3      | - Discharge to hypo.                   | [U,(1+*,)+Q,,+Q,,]F,                                    | [ [0.(0*a.7.)+f+f. ]F.                                              | ا روعیت ایورست رواست محد ۲۵ میل این ا<br>مدر ۲۰۰ میل ۲۰۰ معد ۲۰۰ ۲۱ |
| 2      | Allied West Flume Discharge            | 2, {0, c+0, c+0, -)                                     | ₹.{8+H+H)<br>  ₹.{8+H+H)                                            | C. 17                                                               |
| ä      | <ul> <li>Waste Bed Overflow</li> </ul> |                                                         |                                                                     | ,4t.,9L.,5L.,P-                                                     |
| 3      | " Thereal Load.                        |                                                         |                                                                     | ,<br>,                                                              |
| 2      | - Bype. Intaka                         | (1-0.19                                                 |                                                                     | 5                                                                   |
| 5      | - Epil. Intake                         |                                                         |                                                                     | T-n4) ACE                                                           |
| 8      | Past Num Discharge                     |                                                         |                                                                     | 14°5°                                                               |
| 2      | Pracinitation                          |                                                         | (+4) (H15 <sup>+1</sup> 26 <sup>+1</sup> 17)                        | (1-24) (M15+M26+M17)                                                |
| 1      |                                        | 2,0,7                                                   | ushe'ut                                                             | 0.                                                                  |
| 2 5    | Structure Bad and                      |                                                         | ر م <sup>2</sup> /د                                                 | •                                                                   |
|        |                                        | ŝ                                                       | e <sup>6</sup> (0,11+377-11)~(1+4-5 <sup>1</sup> 2,0)A <sub>2</sub> | •                                                                   |
| N      | Conduction in                          | •                                                       | **************************************                              |                                                                     |
| ព      | Conduction out                         | ó                                                       | A. 7. U. A. D.                                                      | ė                                                                   |
| ž      | Surface Enlandon                       | <b>.</b>                                                |                                                                     |                                                                     |
| ង      | Everotetion                            | *s <sup>U</sup> , 1 <sup>6</sup> 6 <sup>b</sup>         | A. U., (A., 46., T. 48., T. )A. P. D.                               | i d                                                                 |
| 36     | Outitor                                | 2,0-2,c+2,42,40,                                        | 1 0 - 1                                                             |                                                                     |
| 2      | Turbulent Exchange (E-8)               |                                                         | 2 07 L                                                              | *26'E                                                               |
| 8      | (3-4)                                  |                                                         |                                                                     | ्र र                                                                |
| 5      | Molecular Diffusion (B-8)              | -<br>                                                   | 1_11_R_E                                                            |                                                                     |
| 8      | (3-4)                                  |                                                         |                                                                     |                                                                     |
| ភ      | Advective Flow (B-H)                   | P.                                                      |                                                                     | <u>ר</u>                                                            |
| 2      |                                        |                                                         | 10,<br>10,                                                          | 3-63                                                                |
|        | 1                                      | a.'                                                     | F10TH                                                               | P105R                                                               |
| а.     | for stream definitions                 | , see Figure 4.3-1                                      |                                                                     |                                                                     |
| н<br>А | · assuming a volumetric <u>r</u>       | weat capacity of 1 REcal/                               |                                                                     |                                                                     |

.

4-31

•

•

Table 4.3-2. "System functions", which depend upon values of the state variables, are given in Table 4.3-3. Parameter identifications and values are given in Table 4.3-4. Figure 4.3-2 summarizes the important control pathways. Essential features are discussed in detail below.

#### 4.3.1 System Definition

The model represents the lake as two, completely-mixed compartments of constant volume, corresponding roughly to the epilimnion and the hypolimnion of the lake. Based upon the analysis of density profile data in Chapter 3, the boundary between the two compartments has been set at the average thermocline depth of 9 meters. Because of thermocline migration with season, the compartments do not exactly correspond to the epilimnion and hypolimnion. It was hoped that this highly-aggregated representation of the lake would be adequate for analysis of the outfall design issue, the primary emphasis of which is on potential impacts of the design upon general mixing in the lake. The test of the adequacy of this representation is in its ability to simulate observed temperature and chloride variations with time.

A total of four state variables are integrated in the model:

4.3.1 System Definition (continued)

$$T_E = epilimnion temperature (OC)$$
  
 $C_E = epilimnion chloride (g/m3)$   
 $T_H = hypolimnion temperature (OC)$   
 $C_H = hypolimnion chloride (g/m3)$ 

Density, an important factor governing the mixing process, is computed as a function of temperature and chloride levels according to the following equations:

$$D = 1 + b_1 x C + b_2 x + b_3 C$$
 (4.3-1)

$$x = 16 - 2 T_{max} T + T^2 \qquad (4.3-2)$$

$$T_{max} = 4 - 0.211 S$$
 (4.3-3)

$$S = 0.0021 C$$
 (4.3-4)

where,

(

**(**)

D = water density 
$$(g/cm^3)$$
  
X = function of T  $(^{\circ}C)^2$   
T = temperature  $(^{\circ}C)$   
C = chloride concentration  $(g/m^3)$   
T<sub>max</sub> = temperature of maximum density  $(^{\circ}C)$ 

S = salinity = total dissolved solids (kg/m<sup>3</sup>) $b_1 \approx regression parameter estimate = 7.87 \times 10^{-11}$  $b_2 = regression parameter estimate = -6.78 \times 10^{-6}$  $b_{2} \simeq regression parameter estimate = 1.70 \times 10^{-6}$ 

The parameter estimates  $b_1$  ,  $b_2$  , and  $b_3$  have been derived from a linear regression analysis of data on density as a function of temperature and salinity given in Williams<sup>43</sup> ( $R^2 = 0.999$ )\*. The linear model [equation (1)] is designed so that D is maximum at  $T = T_{max}$ . The relationship between  $T_{max}$  and S [equation (3)] is also derived from data in Williams. The relationship between salinity and chloride concentration is typical of Onondaga Lake waters, according to the Onondaga Lake Study<sup>23</sup>. Density differences in the fourth or fifth decimal place can be important in the model calculations. In order to minimize possible effects of round-off errors, actual computations are done using the following linear transformation commonly employed in oceanographic work 43:

$$D' = 1000 (D - 1)$$
 (4.3-5)

\* SEE =  $5.4 \times 10^{-6}$ , 0 < T < 25, 0 < S < 10

#### 4.3.1 System Definition (continued)

(

 $\mathbf{O}$ 

As shown in Figure 4.3-1, inflows to the lake have been aggregated into four sources: Allied Chemical (streams 14 and 15); Ninemile Creek (stream 9); Metro STP (stream 5); and "Tributary" flows (stream 1). The latter represents the sum of Onondaga Creek, Ley Creek, Harbor Brook, Bloody Brook, and local drainage. Stream 14 from Allied Chemical represents the waste bed overflow to Ninemile Creek, which contains most of the salinity entering the lake. When the new sewage treatment plant is put into operation, stream 14 will be diverted and mixed with stream 5. Streams 16 and 17 represent cooling water withdrawals from the hypolimnion and epilimnion, respectively. Stream 15 is primarily waste heat from Allied Chemical. This is discharged in streams 13 ("West Flume") and 18 ("East Flume") to Ninemile Creek and to the lake epilimnion, respectively.

The nodes associated with each inlet stream in Figure 4.3-1 represent entrance mixing zones, in which the entering streams (1, 5, 9) are mixed with diluent epilimnion waters (2, 6, 10). The mixtures subsequently enter the epilimnion (3, 7, 11) or the hypolimnion (4, 8, 12), according to whether their densities are less or greater than the estimated density at the thermocline, according to functions described in Section 4.3.3.

The lake surface streams (19-25) consist of precipitation (19)and five thermal energy fluxes, which depend upon meteorologic

#### 4.3.1 System Definition (continued)

conditions and upon lake surface temperature. Most of the functional forms and parameter values for the latter have been taken from Harleman and Markofsky<sup>11</sup>.

Streams 27 - 32 represent three basic types of internal exchanges between the epilimnion and hypolimnion. Bulk, density-dependent, turbulent transport is represented by streams 27 and 28. Mechanisms involved include eddy diffusivity, seiches, and thermocline erosion. Streams 29 and 30 represent molecular diffusion across the thermocline. These are generally on a much small scale than 27 and 28, but have been distinguished because of their density-independence. The last pair of streams, 31 and 32, represent advective flows, which are required in order to satisfy constant volume constraints placed upon the hypolimnion. Finally stream 26 represents lake outflow, determined by the epilimnic water balance.

#### 4.3.2 Forcing Functions

( )

Forcing functions are defined as factors which influence, but are not influenced by the lake system. The functions in Table 4.3-2 constitute the boundary conditions which drive the system and are comprised chiefly of hydrologic, meteorologic, and industrial variables. All of these variables have been estimated on monthly-average time

#### **Table 4.3-2**

#### Forcing Function Definitions

| Fur             | action                                         | Value   | Units                   | Ref  |
|-----------------|------------------------------------------------|---------|-------------------------|------|
| U               | Tributary Inflow                               | а       | m <sup>3</sup> /day     | c    |
| U2              | Metro STP Inflow                               | a       | m <sup>3</sup> /day     | с    |
| U3              | Ninemile Creek Inflow                          | a       | m <sup>3</sup> /day     | c    |
| U4              | Praction of Allied Chem Intake from Epilimnion | ь       | <b>-</b> .              | .7   |
| U <sub>5</sub>  | Temperature Rise Through AC Power Plant        | 20      | degrees C               | 7    |
| U <sub>6</sub>  | Allied Waste Bed Overflow Chloride Load        | 1.70x10 | 9<br>g/day              | 7,21 |
| 07              | AC East/West Plume Chloride Load               | 1.23x10 | 8<br>g/day              | 21   |
| v <sub>8</sub>  | Precipitation                                  | a       | m/day                   | 38   |
| U <sub>9</sub>  | Solar Radiation                                | ь       | KKCal/m <sup>2</sup> da | уЬ   |
| U10             | Cloud Cover                                    | a       | -                       | 38   |
| U <sub>11</sub> | Air Temperature                                | a       | degrees C               | 38   |
| U<br>12         | Fraction of Possible Sunshine                  | a       | -                       | 38 🤺 |
| U<br>13         | Wind Velocity                                  | a       | ∎/sec                   | 38   |
| U<br>14         | Saturation Vapor Pressure at Air Temperature   | b       | mm Hg                   | ь    |
| U <sub>15</sub> | Relative Humidity                              | b       | -                       | b    |
| U<br>16         | Dew Point                                      | a       | degrees C               | 38   |
| U<br>17         | Atmospheric Pressure                           | a       | mm Hg                   | 38   |
| <sup>U</sup> 18 | Equilibrium Water Temperature                  | b       | degrees C               | ь    |
| U<br>19         | Air Density                                    | Ь       | g/cm <sup>3</sup>       | ь    |
| บ <sub>20</sub> | Tributary Temperature                          | b       | degrees C               | ь    |
| U21             | Metro STP Temperature                          | b       | degrees C               | ь    |
| U22             | Ninemile Creek Temperature                     | b       | degrees C               | ь    |
| U<br>23         | Tributary Chlorides                            | Ь       | g/m <sup>3</sup>        | ь    |
| U_24            | Metro STP Chlorides                            | b       | g/m <sup>3</sup>        | ь    |
| <sup>U</sup> 25 | Ninemile Creek Chlorides                       | ь       | g/m³                    | ь    |

a - input on a monthly-average basis

b ~ defined below

\$

(

(

c - calculated from hydrologic budget developed in Chapter 3

n States in the states in the

# Table 4.3-2 (continued)

.

- -

**(**)

(

4-38

| Function                                                                                                       | Evaluation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                              |                                                     |                                             | Re              |
|----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------|-----------------------------------------------------|---------------------------------------------|-----------------|
| U <sub>4</sub>                                                                                                 | Dates         U         Dates         U         Date         Date         U         Date         Date <thdate< th=""> <thdate< th=""> <thdate< th=""></thdate<></thdate<></thdate<> | es<br>05/73<br>21/74<br>18/74<br>01/75       | U <sub>4</sub><br>0.<br>.78<br>0.                   |                                             | 7               |
| U <sub>9</sub>                                                                                                 | $U_{9} = 1.70 + 1.21 \cos \theta + 2.82 U_{12} + .985 U_{12} \cos \theta$<br>$\theta = 2\pi(t-172)/365$<br>t = day of year<br>[linear regression analysis of monthly-average<br>and sunshine data from East Lansing, Mich.;<br>Ithaca, N.Y.; Boston, Ma.; Portland, Me (N<br>SEE =260)]                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | ge soj<br>Cleve<br>= 28;                     | lar rad<br>eland,<br>3, R =                         | liation<br>Ohio;<br>.985;                   | <sup>•</sup> 38 |
| <sup>U</sup> 14                                                                                                | $U_{14} = \exp[20.59 - 5199./(U_{11} + 273.1)]$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                              |                                                     |                                             | 10              |
| <sup>ย</sup> 15                                                                                                | $v_{15} = [(1121 v_{11} + v_{16})/(112 + .9 v_{11})]^8$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | <u></u>                                      |                                                     |                                             | 19              |
| ช <sub>18</sub>                                                                                                | solution of following transcendental energy by<br>$(.211+.042U_{13})(U_{18}-U_{11}) - U_9$<br>$+ 2.59 \times 10^{-4}(107597U_{18})(1+.224U_{13})[p(U_{18})-U_{18})$<br>$p(U_1) = \exp[20.59 - 5199/(U_1 + 273.1)]$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | alance                                       | equat<br>_)] ≈ 0                                    | ion :                                       | 39<br>10        |
| <sup>U</sup> 19                                                                                                | $U_{19} = .00129[273.1/(U_{11}+273.1)][(U_{17}378U_{15})]$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 5 <sup>U</sup> 14 <sup>}/</sup>              | /760]                                               | -                                           | 10              |
| <sup>U</sup> 20<br><sup>U</sup> 21<br><sup>U</sup> 22<br><sup>U</sup> 23<br><sup>U</sup> 24<br><sup>U</sup> 25 | $U_{20} = 6.14 \pm 4.19 \cos\theta \pm .33 U_{18,0}$<br>$U_{21} = 11.8 \pm 2.77 \cos\theta \pm .23 U_{18,1}$<br>$U_{22} = U_{20}$<br>$\ln(U_{23}) = 6.601621 \ln(U_1/86400)$<br>$\ln(U_{24}) = 5.989 \pm .140 \ln(U_2/86400)$<br>$U_{25} = U_{23}$<br>[regression analysis of survey data]<br>$U_{18,1} = equilibrium temperature, lag i months$<br>$\theta^{-} = 2\pi (t-220)/365 ; t = day of year$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | <u>N</u><br>91<br>91<br>-<br>115<br>127<br>- | R <sup>2</sup><br>.914<br>.815<br>-<br>.434<br>.003 | SEE<br>1.91<br>2.09<br>-<br>.48<br>.41<br>- | 21              |

f.

## 4.3.2 Forcing Functions (continued)

đ,

Ć

(

scales for the period of interest (1968-74). They are plotted against time in Appendix E.

Tributary flows  $U_1$ ,  $U_2$ , and  $U_3$  are derived from the hydrologic balance developed in Chapter 3. Tributary temperatures,  $U_{20}$ ,  $U_{21}$ , and  $U_{22}$ , are estimated based upon regression models which compute temperature from day-of-year and equilibrium temperature. The latter,  $U_{18}$ , is defined as the water temperature which is in equilibrium with ambient meteorologic conditions. It is computed according to a formulation suggested by Velz<sup>39</sup>. The temperature regression models for stream 1 and the Metro STP have been estimated based upon data from the O'Brien and Gers surveys<sup>21</sup>. Chloride levels for these streams,  $U_{23}$  and  $U_{24}$ , are computed as functions of flow, using regression models also developed from O'Brien and Gere's data. Ninemine Creek temperatures and chloride levels upstream of the Allied Chemical discharges are assumed to be equal to those of stream 1.

A 20<sup>o</sup>C temperature rise through the Allied Chemical power plant has been assumed, based upon analysis of survey data<sup>21</sup> and personal communications with the Allied Chemical engineering staff<sup>7</sup>. The chloride fluxes from the solvay plant,  $U_6$  and  $U_7$ , have also been estimated from survey data<sup>21</sup> on Ninemile Creek and East Flume, respectively. The waste bed overflow component,  $U_6$ , has been calculated

#### 4.3.2 Forcing Functions (continued)

T

from the Ninemile Creek flow and chloride concentration data, with adjustment for upstream chloride levels. Both the magnitude and the temporal stability of this flux have been verified by soda ash production data provided by Allied Chemical<sup>7</sup>.

 $U_4$ , the fraction of the Allied Chemical cooling water taken from the epilimnion, has been estimated as a function of time from information provided by Clough<sup>7</sup>. The cooling water intake is roughly regulated to minimize pumping and chlorine demand costs. Hypolimnic waters are cooler, have more heat capacity, and therefore require somewhat less flow to dispose of a given quantity of heat. Allied chlorinates the cooling water to oxidize hydrogen sulfide and minimize corrosion damage to its equipment. Normally, the plant uses hypolimnic waters exclusively. When hydrogen sulfide levels become excessive, an epilimnion intake is opened, to reduce chlorine demand costs.

The remaining forcing functions in Table 4.3-2 are meteorologic variables, some of which are computed as functions of others. Solar radiation measurements  $(U_9)$  were not available for Syracuse, so U.S. Weather Service data<sup>30</sup> from stations at roughly the same latitude were employed to estimate a regression model for radiation as a function of time-of-year and percent of possible sunshine. The saturation vapor pressure of air  $(U_{14})$ , relative humidity  $(U_{15})$ , and air density  $(U_{19})$  are all computed from other reported meteorologic variables,

#### 4.3.2 Forcing Functions (continued)

according to the functions given in Table 4.3-2.

#### 4.3.3 System Functions

×.

**(** )

 $\bigcirc$ 

System functions depend upon the values of the state variables, and comprise the primary relationships within the model. They are defined and described in Table 4.3-3.

The first three functions represent the fractions of the inflowing waters which sink into the hypolimnion due to density differences. As described above, each stream enters the lake and is initially diluted with epilimnic waters. The density of the diluted stream is compared with the density of the lake at the thermocline, estimated as the average of the computed epilimnion and hypolimnion densities. If the influent stream density is less than the interface density, it is assumed to stay in the epilimnion and the corresponding function value is set to zero. If the influent density is greater, a certain fraction of the flow is assumed to sink into the hypolimnion. The exponential functions employed to compute this fraction are merely used to smooth the transition from zero to the respective maximum values,  $a_{20}$ ,  $a_{21}$ , and  $a_{22}$ . This is done to prevent instability in the numerical integration scheme. The value of  $a_{10}$  is sufficiently small that the transition occurs

## Table 4.3-3

## System Function Definitions

.

C

 $\mathbf{C}$ 

()

| Fun             | ction                                                | Units               |
|-----------------|------------------------------------------------------|---------------------|
| F               | Fraction of Tributary Flow Entering Sypolimnion      | - ,                 |
| $\mathbf{F}_2$  | Fraction of Metro STP Flow Entering Hypolimnion      |                     |
| F3              | Fraction of Ninemile Creek Flow Entering Hypolimnion | -                   |
| F <sub>4</sub>  | Allied Chemical Total Cooling Wator Flow             | m <sup>3</sup> /day |
| F <sub>5</sub>  | Average Inlet Temperature in AC Cooling Water        | degrees C           |
| F <sub>6</sub>  | Evaporation Driving Force                            | mm Hg               |
| F7              | Partial Pressure of Water at Surface Temperature     | um Hg               |
| F8              | Hypolimnic Flow Surplus                              | m <sup>3</sup> /day |
| F9              | Advective Flow [Epilimnion + Hypolimnion]            | m <sup>3</sup> /day |
| F10             | Advective Flow [Hypolimnion + Epilimnion]            | m <sup>3</sup> /day |
| F11             | Vertical Exchange Rate                               | m/day               |
| F <sub>12</sub> | Richardson Number                                    | -                   |
| F <sub>13</sub> | Friction Velocity                                    | m/sec               |
| F14             | Ice Cover Function                                   | -                   |

•

Table 4.3-3 (continued)

| Function                                                          | ·                                        | Ref  |
|-------------------------------------------------------------------|------------------------------------------|------|
| P <sub>1</sub> = 0, D                                             | $S_{\rm I} = (D_{\rm E} + D_{\rm H})/2.$ | •    |
| $= a_{20} \exp[a_{19}/(D_1 - D_3)] = a_{20}, D_3$                 | > D <sub>I</sub>                         |      |
| $\overline{x}_2 = 0.$ , $D_7$                                     | , ≤ D <sub>I</sub>                       |      |
| $= a_{21} \exp[a_{19}/(D_1 - D_7)] \approx a_{21} , D_7$          | , > D <sub>I</sub>                       |      |
| $F_3 = 0.$ , $D_1$                                                | .1 <sup>≤ D</sup> 1                      |      |
| $= a_{22} \exp\{a_{19}/(D_{1} - D_{11})\} \approx a_{22} , D_{1}$ | 1 <sup>° D</sup> I                       |      |
| $F_{A} = a_{2A} + a_{25}F_{5}$                                    |                                          | 7    |
| $F_5 = U_4 T_E + (1 - U_4) T_H$                                   |                                          |      |
| $F_6 = F_7 - U_{15} U_{14}$                                       |                                          | 11   |
| $F_7 = \exp[20.59 - 5199/(T_{\rm E} + 273.1)]$                    |                                          | 10   |
| $F_8 = Q_4 + Q_8 + Q_{12} - Q_{16}$ [Hy]                          | polimnic Nater Balance]                  |      |
| $F_9 = 0.$ , $F_8 \ge 0.$<br>$= -F_9$ , $F_9 < 0.$                |                                          |      |
| $F_{10} = F_{6}$ , $F_{8} > 0.$<br>= 0, , $F_{6} \le 0.$          |                                          |      |
| $F_{11} \approx 86400 a_{14} F_{13} / [F_{12} + (a_{14}/a_{15})]$ | , F <sub>12</sub> > 0.                   | 18   |
| = 86400 a <sub>16</sub> F <sub>13</sub>                           | $, F_{12} \leq 0.$                       |      |
| $F_{12} = 9.8 a_{17} (D_H - D_E) / (F_{13}^2 D_E)$                |                                          | 18   |
| $F_{13} = .0224 U_{13}^{1.25} F_{14} (U_{19}/D_E)^{.5}$           |                                          |      |
| F, = 1.                                                           | ,T_ ≥0.                                  | 4,44 |
| $\approx a_{23} + (1 - a_{23}) \exp(3 T_E) \approx a_{23}$        | , T <sub>B</sub> < 0.                    |      |

(

Ċ

#### 4.3.3 System Functions (continued)

rapidly. Examination of the lake and tributary density data revealed that the major tributary stream (1) and the Metro STP stream (5) were always less dense than the epilimnic waters, due to the salinity of the latter. Thus, under past and present conditions, these streams would not be expected to sink, and the corresponding parameter values  $a_{20}$  and  $a_{21}$  have been set to zero. Likewise, in these cases, initial dilution would be of no consequence, so the corresponding dilution ratios  $a_1$  and  $a_2$  have also been set to zero. Survey data indicated that Ninemile Creek was considerably more dense than the epilimnion and the hypolimnion of the lake during most seasons. Accordingly, the density current phenomenon would be important in this case. Estimation of the parameters for this stream will be discussed in a subsequent section.

The rate of cooling water withdrawal by Allied Chemical,  $F_4$ , is assumed to be a weak function of the average inlet temperature,  $F_5$ . The linear relationship has been estimated from monthly-average flow data provided by Allied Chemical<sup>7</sup> and corresponding lake temperature data<sup>21</sup>. Flows decrease with decreasing inlet temperature because of increased heat disposal capacity per unit of flow.

Evaporation from the water surface is governed in part by  $F_6$ , the evaporation driving force, which depends upon  $F_7$ , the partial pressure of water vapor at the lake surface temperature and upon the

#### 4.3.3 System Functions (continued)

8

(

partial pressure of water vapor in the air, computed from air temperature and relative humidity.

The hypolimnic flow surplus,  $F_8$ , is required to compute the advective flows to and from the hypolimnion,  $F_9$  and  $F_{10}$ , respectively. These are necessary to satisfy the constant volume constraint placed upon the hypolimnion.

The primary vertical exchange function in the model is  $F_{11}$  , which is computed from the Richardson number ( $F_{12}$ ) and the friction velocity ( $F_{1,3}$ ). This functional form was suggested in part by the experimental work of Kato and Phillips, as discussed in Section 4.2.2. Their model essentially couples the vertical mixing process with the mechanical energy balance of the lake. The Richardson number,  $F_{12}$  , is defined as the ratio of the buoyant potential energy deficit in the system due to density stratification to the rate of kinetic energy dissipation due to wind action. The friction velocity,  $F_{13}$ , is computed from wind speed ( $U_{13}$ ), air density ( $U_{19}$ ), surface water density ( $D_{\rm E}$ ), and the ice cover function ( $F_{14}$ ), using functions given by Banks<sup>4</sup> and Wu<sup>44</sup>. The ice cover function,  $F_{14}$  , describes the effect of ice cover upon wind-induced mixing. When the computed surface temperature is greater than zero,  $F_{14}$  is set to one. For surface temperatures less than zero, P<sub>14</sub> rapidly approaches a minimum value, a<sub>23</sub>. Again, the exponential function has been employed only

#### 4.3.3 System Functions (continued)

as a means of smoothing transitions and preventing instabilities in the numerical integration. The southern portion of Onondaga Lake generally remains free of ice cover, due to the discharge of heated effluents in this area<sup>23</sup>. This is the reason for the non-zero value of  $a_{23}$ .

With the exception of the second term in the denominator,  $a_{14}/a_{16}$ , the functional form for the exchange rate,  $F_{11}$ , corresponds exactly to the model employed by Kato and Phillips (see Figure 4.2-1). The added term effectively places an upper limit on the vertical mixing rate as the Richardson number approaches zero. This is necessary to prevent numerical instability in the calculations. The upper limit, determined by  $a_{16}$ , is sufficiently high to achieve essentially complete vertical mixing.

#### 4.3.4 Parameter Values

(

**(**)

The values and bases for the various parameter estimates are given in Table 4.3-4. Some of the parameters are associated with processes or relationships which are general and have defined in other modelling or experimental studies. Values for these have been taken directly from the literature  $(a_6 - a_{11}, a_{18})$ . Other parameters relate specifically to this particular lake or to this model. Some of these have been defined by measurements or data which are independent of the

#### Parameter Definitions and Values

| Par                   | aneter               |                |                      | ·                  |         | Value                       | Units                                                  | Ref  |
|-----------------------|----------------------|----------------|----------------------|--------------------|---------|-----------------------------|--------------------------------------------------------|------|
| a,                    | Initial              | Diluti         | on Ratio             | - Tributari        | es      | 0.                          | -                                                      | a    |
| a,                    | 11                   | 10             | 47                   | - Metro STP        | 1       | 0.                          | -                                                      | a    |
| a_3                   | **                   | **             | 61                   | - Ninemile         | Creek   | ọ.                          | -                                                      | a    |
| a_4                   | Fraction<br>Discharg | of Al<br>ed to | lied Coo<br>West Flu | ling Water ৰ<br>me |         | 0.273                       | -                                                      | 23,7 |
| a <sub>5</sub>        | Surface .            | Mass 1         | ransfer 1            | Parameter          |         | 0.204                       | sec/(day-mmHg)                                         | ъ    |
| <b>a</b> <sub>6</sub> | Conducti             | on Par         | ameter               |                    |         | 0.279                       | munHg                                                  | 11   |
| a <sub>7</sub>        | Back-Rad             | iation         | Paramete             | er                 | 1.      | 136x10 <sup>-5</sup>        | KKCal/m <sup>2</sup> <sup>o</sup> K <sup>4</sup> -day  | 11   |
| a <sub>8</sub>        | Atmosphe:            | rio Ra         | diation 1            | Parameter          | 1.      | 064x10 <sup>-1</sup>        | <sup>4</sup> KKCal/m <sup>20</sup> K <sup>6</sup> -day | 11   |
| a <sub>9</sub>        | 11                   |                | •                    | 81                 |         | 0.17                        | -                                                      | 11   |
| a_10                  | Heat of '            | Vapori         | zation at            | ⊧ 0°c              |         | 596.                        | Kcal/Kg                                                | 11   |
| a <sub>11</sub>       | Effect o             | f Temp         | erature o            | on Heat of Va      | ар.     | -0.54                       | Kcal/Kg- <sup>O</sup> C                                | 11   |
| a <sub>12</sub>       | Vertical             | Heat :         | Exchange             | via Molec. A       | Dif: 2. | 74x10 <sup>-3</sup>         | m/day                                                  | с    |
| a <sub>13</sub>       | Vertical             | Chl.           | Exchange             | via Molec. 1       | Dif. 2. | 74x10 <sup>-5</sup>         | m/day                                                  | ۰    |
| a <sub>14</sub>       | Vertical             | Excha          | nge Paran            | neter              |         | 0.770                       | -                                                      | b    |
| a <sub>15</sub>       | et                   | 11             | 4                    | •                  |         | 1.001                       | -                                                      | ь    |
| a <sub>16</sub>       | "                    | 96             | •                    | •                  |         | 0.020                       | -                                                      | a    |
| a17                   | Thermocl:            | ine Dej        | pth                  |                    |         | 9.0                         | m                                                      | 21   |
| a <sub>18</sub>       | Heat Capa            | acity (        | of Water             |                    |         | 1.0                         | Kcal/Kg- <sup>O</sup> C                                | 10   |
| a <sub>19</sub>       | Density (            | Curren         | t Paramet            | er                 | 0       | .00001                      | g/cm <sup>3</sup>                                      | а    |
| a_20                  | Max. Dens            | sity C         | ur. Fract            | ion - Tribut       | taries  | 0.0                         | -                                                      | a    |
| a <sub>21</sub>       | ff (                 | n a            | it ei                | - Metro            | STP     | 0.0                         | <b>-</b> .                                             | а    |
| a22                   | н і                  | N 1            |                      | - Ninem:           | ile Ck. | 0.267                       | -                                                      | ь    |
| a_3                   | Ice Cove             | er Para        | ameter               |                    |         | 0.46                        | -                                                      | a    |
| a24                   | Allied (             | Coolin         | g Water B            | low at 0°C         | 2.      | 47x10 <sup>5</sup>          | m <sup>3</sup> /day                                    | е    |
| a <sub>25</sub>       | Effect d             | of Inl         | et Temp.             | on AC CW Flo       | w 8.3   | 21 <b>x1</b> 0 <sup>3</sup> | m <sup>3</sup> /day- <sup>o</sup> C                    | e    |

a - assumed value.

:

(

 $(\cdot)$ 

b - estimated empirically.

- c estimated from molecular diffusivities of heat and sodium chloride, assuming a thermocline thickness of 4 meters  $^{23}$ .
- d estimated from observed ice cover<sup>23</sup>; corresponde to a 90% reduction in wind-induced mixing under ice cover.

e - based upon analysis of flow data supplied by Clough  $^7\,$  and lake temperature data supplied by Onondaga County  $^{23}$ 

#### 4.3.4 Parameter Values (continued)

model (e.g.  $a_4$ ,  $a_{12}$ ,  $a_{13}$ ,  $a_{17}$ ,  $a_{23}$ ,  $a_{24}$ ,  $a_{25}$ ). In some cases, model calculations are insensitive to particular parameter values, and reasonable values have been assumed ( $a_1$ ,  $a_2$ ,  $a_3$ ,  $a_{16}$ ,  $a_{19}$ ,  $a_{20}$ ,  $a_{21}$ ). The remaining four parameters are lake- or model-specific, have important effects on the model calculations, and cannot be estimated accurately based upon independent evidence. These parameters ( $a_5$ ,  $a_{14}$ ,  $a_{15}$ ,  $a_{22}$ ) have been estimated by fitting model simulations to observed temperature and chloride data, using Bard's algorithms for parameter estimation in nonlinear dynamic systems. Demonstration and evaluation of the utility of these algorithms in ecosystem modelling is a primary emphasis of the work described in this chapter. The specifics of the parameter estimation exercises will be described in Section 4.5.

#### 4.4 Implementation

A Fortran IV program has been written to perform the calculations specified by the model. The structure of the program is summarized in Figure 4.4-1. The roles and sizes of each of the 17 subroutines are given in Table 4.4-1. Programming and computations have been done on an IBM 370-168.

The system of equations is integrated numerically using a fourthorder, Runge-Kutta procedure with variable step size, contained in the IBM Scientific Subroutine Package<sup>.16</sup>. This subroutine (RKGS) automatically adjusts integration step size for control of integration errors. The criterion used for step size adjustment is:

## Table 4.4-1

# Model Subroutines

.

(

 $C^{+}$ 

| Name  | Function                                                                                      | Number of<br>Statements | Reference |
|-------|-----------------------------------------------------------------------------------------------|-------------------------|-----------|
| MAIN  | initializes and controls program execution                                                    | 129                     |           |
| BOXES | computes compartment surface areas and volumes<br>from morphometric data                      | 27                      |           |
| Setup | reads observed temperature and chloride data fro<br>disc file and computes spatial averages   | m 32                    |           |
| CAV   | computes weighting factors for spatial averaging of profile data                              | 28                      |           |
| RKGS  | controls fourth-order Runge-Kutta numerical<br>integration                                    | 115                     | 16        |
| DERIV | controls time-derivative computations                                                         | 34                      |           |
| OUTP  | called after each Runge-Kutta step; traces calcu<br>lations if specified                      | - 12                    |           |
| PFUNC | updates forcing function vector                                                               | 180                     |           |
| ₽UP   | updates parameter vector for sensitivity testing                                              | 30                      |           |
| DERY  | computes derivatives of state variables for give<br>parameter vector                          | n 91                    |           |
| INLET | distributes inflows and computes water balances                                               | 91                      |           |
| RESLT | prints, plots, stores, and analyzes output at specified times and at end of program execution | 287                     |           |
| resan | analyzes and plots residuals                                                                  | 156                     |           |
| HIST  | prints histograms                                                                             | 54                      |           |
| PLT   | prints line plots                                                                             | 75                      |           |
| TALLY | summarizes residual statistics                                                                | 37                      | 16        |
| DENSY | function to compute density from temperature and chloride concentration                       | 6                       |           |

.



Figure 4.4 - 1 Schematic of Model Subroutines

\* Subroutines are called from left to right and top to bottom

**(**)

4.4 Implementation (continued)

$$\delta = \frac{1}{15} \sum_{i=1}^{4} w_i |y_i^{(1)} - y_i^{(2)}| \leq \delta_{\max} = 0.004 \qquad (4.4-1)$$

4-51

where,

( )

()

 $w_i = weighting factor for state variable i,$ <math>i = 1,4 ( $\Sigma w_i = 1.0$ )  $y_i^{(1)} = state variable value one time step ahead$  $<math>y_i^{(2)} = state variable value two time steps ahead$ 

If the parameter  $\delta$  is greater than 0.004, step size is halved and the computations are repeated. A reasonable value for  $\delta_{max}$  was selected by trial-and-error. The weighting factors,  $w_i$ , are approximately inversely proportional to the annual ranges of the respective state variables, so that each variable contributes equally to the total integration error. A maximum step size of 0.025 years (9.13 days) has been specified.

The Runge-Kutta method requires evaluation of the derivative vector four times per each integration step. It can be relatively expensive compared to second-order, predictor-corrector methods<sup>16</sup>. The advantages of the former are that it is self-starting and relatively stable. The accuracy of this method is particularly important to successful implementation of the parameter estimation algorithms, as described in the next section. Adjustment of step size according to the above procedure can save considerable computation time. With the above error bound

#### 4.4 Implementation (continued)

parameters, step size varies between about 4.5 days, during relatively stable periods, to less than 0.5 days, during unstable periods (e.g., overturn). The yearly-average step size is about 3 days. Exclusive of loading, the program consumes approximately 1.7 CPU seconds per year of simulation on the IBM 370-168 for integration of one set of state variable equations.

The program permits sensitivity testing by integrating more than one set of state variable equations simultaneously. One additional set is added for each parameter or forcing function sensitivity being evaluated. For each set of state equations added, the value of the corresponding parameter is increased by a given percent (typically 3%). This permits calculations of sensitivity coefficients via finitedifferences as functions of time. The sensitivity coefficients are essentially normalized first derivatives of the state variables with respect to the parameter values.

A linear interpolation scheme has been employed to input the values of time-variable forcing functions. As noted in Section 4.3.2, these functions have been defined on monthly-average time scales. Step changes in these variables would introduce instabilities in the numerical integration routine. Accordingly, transitions from one month to the next have been smoothed by adding a linear segment extending from the average value for a given month to that of the next in a time interval of one-half a month. Thus, the first quarter of each month is the last part of a linear segment extending from the

#### 4.4 Implementation (continued)

previous month's average to the current month's average, the middle half is constant at the average value, and the last quarter is the first part of a linear segment extending to the next month's average.

#### 4.5 Parameter Estimation

A total of four parameters have been estimated by implementing Bard's algorithms for parameter estimation in nonlinear dynamic systems<sup>2</sup>. The general parameter estimation problem has been outlined in Chapter 1. Sections 4.5.1 and 4.5.2 discuss some of the specific options and alterations which have been implemented in this application. Section 4.5.3 describes the data base employed for parameter estimation purposes. Results are presented and interpreted in Section 4.5.4 and 4.5.5, respectively. Finally, the residuals are analyzed in 4.5.6.

#### 4.5.1 Methods

**(**)

<u>(</u>);

As discussed in Chapter 1, the basic problem is to find a set of parameter values which maximizes an objective function computed from the residuals, subject to constraints imposed by the modal equations. As also noted in that chapter, the general problem can also include the affects of prior distributions of parameter estimates on the objective function, as well as constraints on parameter values. In this particular application, these additional options have not been exercised. The parameter solutions obtained are sufficiently far from any implied constraints (e.g., positivity) that such constraints

#### 4.5.1 Methods (continued)

are of no consequence to the solution. It was not found necessary to include constraints in order to guide convergence. Accordingly, the only constraints actually imposed are those dictated by the model equations.

The particular objective function employed depends upon the assumed form of the covariance matrix of residuals. As distributed, a total of six options are available in the Bard program:

diagonal x known to within a multiplicative constant unknown

The most general form of the covariance matrix (non-diagonal x unknown) has been assumed. The corresponding form for the constant portion of the maximum likelihood objective function is :

$$\phi = -\frac{n}{2} \ln[\det(\frac{A}{n})] \qquad (4.5-1)$$

where,

(

(

- 🗢 🗢 objective function
- n = number of observations
- moment matrix of residuals
- $a_{ij} =$  element of matrix <u>A</u>; i, j = 1, k
- e = residual for variable i and observation m; i = 1,k , m = 1,n

k = number of observed variables = 4

Ľ

(

The Bard program provides two optional algorithms for guiding the convergence of an initially-assumed parameter vector to the optimal value: the Gauss-Newton method<sup>1</sup> and the Davidon-Pletcher-Powell method<sup>9</sup>. The former has been implemented here, based upon preliminary comparisons of the effectiveness of these algorithms on test systems<sup>41</sup> and upon recommendations in the Bard program manual <sup>3</sup>.

In order to guide the convergence of the parameter vector to the solution, an estimate of the derivative of the objective function with respect to each of the parameter values is required. This is computed within the Bard program from the derivatives of the state variables with respect to the parameters. As distributed, the Bard program is designed for use with analytic derivatives, i.e., sensitivity equations. The discontinuities associated with the excessive number of conditional statements in the model render analytic differentiation impractical in this application. Accordingly, substantial modifications have been made in parts of the Bard program to permit derivative computation via

finite-difference methods. For maximum efficiency, a finite-difference scheme should select parameter perturbation sizes which balance truncation error (increasing with increasing step size) against round-off error (decreasing with increasing step size). An algorithm suggested by Bard <sup>3</sup> has been employed to achieve this. This finite difference scheme enhances the flexibility of the model and the parameter estimation algorithms by facilitating coding changes made in model development and implementation.

Parts of the Bard program have also been modified in order to facilitate estimation of the 95% confidence regions for the parameter estimates. The confidence region is defined as that portion of parameter space which satisfies the following criterion:

$$\Phi \left( \underline{\theta}^{*} \right) - \Phi \left( \underline{\theta} \right) \leq \varepsilon \qquad (4.5-3)$$

where,

( e.

- $\phi$  ( $\theta$ ) = objective function value for parameter vector  $\theta$
- $\Phi(\theta^*) = \text{objective function value for parameter}$ vector  $\theta^* = \text{solution}$

 $\varepsilon$  = indifference interval

The region is estimated based upon a quadratic approximation of the response surface in the vicinity of the solution. A Taylor series expansion around the solution gives:

$$\Phi \left(\underline{\theta}^{*} + \underline{\delta\theta}\right) \simeq \Phi \left(\underline{\theta}^{*}\right) + \underline{q}^{*T} \underline{\delta\theta} + \frac{1}{2} \underline{\delta\theta}^{T} \underline{H}^{*} \underline{\delta\theta} \qquad (4.5-4)$$

where,

$$\delta\theta = \theta - \theta^*$$

- a\*<sub>T</sub> ≈ transpose of vector of first partial derivatives of the objective function with respect to the parameters at the solution
- <u>H</u>\* = matrix of second partial derivatives of the objective function with respect to the parameters at the solution

At the solution, the first derivatives are near zero:

$$\mathbf{q^{*T}} \simeq \mathbf{O} \tag{4.5-5}$$

Thus,

C

 $\mathbf{C}$ 

$$\Phi (\underline{\theta}^* + \underline{\delta \theta}) - \Phi (\underline{\theta}^*) = \Delta \Phi \simeq \frac{1}{2} \underline{\delta \theta}^T \underline{H}^* \underline{\delta \theta} \qquad (4.5-6)$$

The covariance matrix of the parameter values,  $\underline{v}_{\theta}$ , is given by  $\overset{3}{:}$ 

$$v_{\theta} \simeq \underline{H}^{\star -1}$$
 (4.5-7)

( ;

( )

Thus, from equations (6) and (7), the indifference region is given by:

$$\Delta \Phi \simeq \frac{1}{2} \underbrace{\delta \theta}^{\mathrm{T}} \underbrace{v_{\theta}}^{-1} \underbrace{\delta \theta} \qquad (4.5-8)$$

Bard<sup>3</sup> has also shown that, for normally-distributed errors and if the estimate of  $V_{\theta}$  is assumed correct, the quantity  $| \underline{\delta \theta}^{T} \underline{v}_{\theta}^{-1} \underline{\delta \theta} |$  is distributed as  $\chi^{2}$  with  $\ell$  (= number of parameters) degrees of freedom. Thus, according to equations (3) and (8), the size of the indifference interval is given by:

$$|\Delta \phi| = \frac{1}{2} |\underline{\delta \theta}^{\mathrm{T}} \underline{v}_{\underline{\theta}}^{-1} \underline{\delta \theta}| \leq \epsilon = \frac{\chi^2}{2} \qquad (4.5-9)$$

This definition of the  $\varepsilon$ -difference region is essentially equivalent to a likelihood ratio test applied to the log-likelihood function,  $\phi^{12}$ . In order to translate this into a statement about actual parameter values, a principal component analysis of the parameter covariance matrix is first employed. The analysis, contained in the original form of Bard's program, computes linear functions of the parameter values which are independent of each other: 4.5.1 Methods (continued)

ei.

(:

$$\rho_{i} = \sum_{j=1}^{k} b_{ij} \delta \theta_{j}, \quad i = 1, k \quad (4.5-10)$$

When defined in terms of  $\delta\theta$ , the expected value of each principal component is 0. and the standard deviation is  $\sigma_j$ . Expressed in terms of the principal components, the  $\varepsilon$ -difference region is given by:

$$\underline{\rho}^{\mathrm{T}} \underline{v}_{\rho}^{-1} \underline{\rho} \leq x^{2} \qquad (4.5-11)$$

The advantage of this representation is that the off-diagonal elements of  $v_{\rho}^{-1}$  are zero, because the principal components are independent of each other. Accordingly, equation (11) can be evaluated as:

$$\sum_{i=1}^{k} \frac{\rho_i^2}{\sigma_i^2} \leq \chi^2 \qquad (4.5-12)$$

The end points of each principal axis  $\rho_j$  are given by setting  $\rho_i = 0$  for  $i \neq j$  and solving  $\rho_j$  in equation (10):

4.5.1 Methods (continued)

(

 $C_{i}$ 

For each principal component, the end points are determined in terms of the original  $\theta$  coordinates by inverting and solving equation (10) for  $\delta \theta_1$  in terms of  $\rho_1$ .

#### 4.5.2 Implementation

Parameter estimation subroutines are listed and described in Table 4.5-1. Figure 4.5-1 depicts the calling sequences for the various subroutines in the program version designed for parameter estimation purposes. Substantial modifications to the Bard subroutine DER have been made to permit: (1) use of Runge-Kutta integration; (2) derivative (sensitivity coefficient) calculation by finitedifference methods; (3) output of the residual sums of squares for each observed variable after each evaluation of the objective function. Likewise, modifications to the Bard subroutine OUT, called after the solution has been reached, have been made to permit:
# Table 4.5 - 1

# Parameter Estimation Subroutines

| Name  | Function                                                                                                                                      | Number of<br>Statements | Reference        |
|-------|-----------------------------------------------------------------------------------------------------------------------------------------------|-------------------------|------------------|
| MAIN  | initializes and controls program execution; creates<br>interface between model and Bard subroutines                                           | : 123                   |                  |
| NIMAX | controls parameter convergence via Gauss-Newton<br>method                                                                                     | 222                     | 2                |
| MAP   | maps values of objective function for specified parameter values                                                                              | 20                      |                  |
| ACCUM | computes value of objective function for given parameter vector                                                                               | 175                     | 2                |
| EIG   | computes scaled eigenvalues and eigenvectors of a given matrix                                                                                | 82                      | 2                |
| BOUND | computes constraint penalty functions and their .<br>derivatives                                                                              | 6                       | 2 <sup>8</sup>   |
| PRIOR | computes effect of prior distribution of parameter<br>on objective function                                                                   | s 3                     | 2 <sup>a</sup>   |
| DER   | controls integration of model equations between<br>specified time limits; computes derivatives of sta<br>variables with respect to parameters | 140<br>te               | 2 <sup>b</sup> . |
| RUN   | sets initial conditions, if unknown                                                                                                           | 3                       | 2 <sup>a</sup> . |
| XTOY  | relates state variables and their parameteric<br>derivatives to observed variables                                                            | 11                      | 2 <sup>a</sup>   |
| out   | provides detailed output after solution has been reached; analyzes response surface in vicinity of solution                                   | 184                     | 2 <sup>b</sup>   |
| MINV  | inverts a matrix                                                                                                                              | 89                      | 16               |
| SIMQ  | solves a set of simultaneous equations for end point<br>of principal component axes of parameter estimates                                    | nts 52                  | 16               |

()

(

(

a ~ not of consequence in this application
b ~ original subroutine substantially modified; see text.



\* Subroutines are called from left to right and top to bottom

#### 4.5.2 <u>Implementation</u> (continued)

( -

(1) computation of residual and parameter correlation matrices; (2) computation of residual serial correlation coefficients; (3) implementation of a multivariate test for significant bias in the residuals; (4) estimation of the 95% confidence region for the parameter vector. In addition, the subroutine MAP has been added to permit direct sampling of the response surface for a given set of parameter values. Any modifications of the Bard subroutines have been designed for general application, i.e., they are not specific for use with this particular model.

#### 4.5.3 Lake Data

Temperature and chloride profile data from the O'Brien and Gere surveys<sup>23,21</sup> have served as a basis for parameter estimates. From April, 1968 through December, 1974, a total of 136 complete profiles at 3-meter depth increments were available. These data have been spatially aggregated to provide epilimnion- and hypolimnion-average values for each sampling date. Within each level, averaging has been done by integrating concentration times surface area with respect to depth (using the trapezoidal rule), and dividing the result by total compartment (epilimnion or hypolimnion) volume. Accordingly, the weighting factors applied to each seven-membered profile vector (0-18 meters) to compute the epilimnic average (0-9 meters) are:

.

•

· .

٠

•...

瘀

(:

**(**)

# Temperature and Chloride Data Used for Parameter Estimation

• ,

. . .

| Cbs,         | Tim              |             | Ty      | c,     | с <sub>и</sub> | Obe.       | 7386            |              | Ŧ.,  | °,     | <u>с</u> , |
|--------------|------------------|-------------|---------|--------|----------------|------------|-----------------|--------------|------|--------|------------|
| . 1          | 1.243            | 10.6        | 7.6     | 1464-4 | 1929+1         | 64         | 3.073           | 20.8         | 12.0 | 1569-1 | 1428.1     |
| - <b>2</b> . | 1.332            | 11-3        | 8.7     | 1866.3 | 1393.7         | 10         | 3.707           | 20+0         | 13.3 | 1927-6 | 2630-0     |
| 4            | 1.927            | 18.5        | 12.2    | 1557.7 | 2120.6         | 12         | 3.749           | 14.7         | 13.4 | 1681-1 | 2054 -0    |
| في           | 4+444            | 20.4        | 12.9    | 1524+1 | 2140.0         | 13         | 485 .6          | 10.5         | 10.6 | 1619-2 | 1906.9     |
| ģ            | 1.400            | 20.1        | 14.3    | 1582.0 | 2146.6         | . 74       | 3.923           | 7.6          | 1.1  | 1183+1 | 1968-5     |
| •            | 1.523            | 21.1        | 13-1    | 1292.9 | 2007.2         | 16         | 4.055           | i.5          | 2.0  | 1462.0 | 1753.1     |
| 9            | 1.542            | 22.1        | 13.4    | 1337.3 | 2050.4         | 77         | 4.099           | 2.0          | 2.9  | 1405.6 | 1737-8     |
| 10           | 1.547            | 23-1        | 12.0    | 1470.2 | 1937-8         | 14         | 4.132           | 1.4          | 2.2  | 1362.2 | 1921-3     |
| -45          | 1+244            | . 22.1      | 12-5    | 1654.7 | 210202         | 14         | 4.247           | 3-3          | 2.4  | 901-1  | 1621-3     |
| ii -         | 1.058            | 22.2        | 13.0    | 1505.9 | 2189.4         | ăĨ         | 4.238           | 6.2          | 4.5  | 861.6  | 1329.4     |
| 14           | 1.0 40           | 20.8        | 14.2    | 1805.9 | 2303.1         | 62         | 4.355           | 11+1         | 7.0  | 970.4  | 1341-2     |
| 15           | 1.751            | 19.1        | 14-4    | 1332.7 | 1165.7         | 63         | 4. 242          | 10+9         | 7.9  | 1461.0 | 1244.4     |
| 17           | 1.408            | 10+3        | 14.3    | 1843.2 | 2462.1         | 0 <b>4</b> | 5.058           | 1.0          | 1.2  | 1863.4 | 1442.4     |
| ii ii        | 1.610            | 12.2        | 12.8    | 1367.0 | 1924.6         | 65         | 5.107           | 0.7          | 1.0  | 1050.4 | 2326-5     |
| 19           | 1.047            | 12-5        | 12+4    | 1626.9 | 2136.4         | 87         | 5.247           | 2.4          | 1.3. | 1196.4 | 2237-5     |
| 20           | 1+923            | 6.3         | 7.5     | 1374.9 | 1750.0         | 54         | 94200           | 2.4          | [+]  | 1001.0 | 1804.0     |
| 21           | 2.018            | 2+1         | 2.0     | 1456.0 | 1619.6         | . 97       | 5.397           | 10.7         |      | 1179.1 | 1741.5     |
| 23           | 2.063            | 1.0         | 1.8     | 1168-1 | 1731.0         | 91         | 5+436           | 16.9         | 8.2  | 1051.7 | 1767.7     |
| 24           | 2.1.4            | 0.5         | 2.7     | 1454.8 | 1616.5         | 92         | 5.512           | 18.9         | 8.B  | 697.1  | 1584-1     |
| 25           | 2.123            | 1.9         | 2.0     | 1405-2 | 1403-3         | 93         | 54551           | 21.6         | 7.9  | 966.8  | 1772.5     |
| 27           | 2.175            | 1.6         | 1 2.6   | 1450.3 | 1034.5         | 95         | 5.627           | 19.4         | 12.1 | 1261.0 | 1017.6     |
| zà           | 2.238            | 3.3         | 3.0     | 1362.6 | 1447.4         | 96         | 5.605           | 21.2         | 14.5 | 1461.8 | 1949 . 4   |
| 24           | 2.258            | 3.0         | 3.0     | 1323.0 | 1434-0         | 97.        | 5.704           | 55+0         | 10.6 | 1490.2 | 1979.2     |
| 30           | 2.271            | 5.4         | 3.0     | 1228.0 | 1672.0         | 98         | 5.742           | 18+3         | 11+0 | 1474.0 | 2010+ 1    |
| 12           | 2.310            | 5-3         | 5.8     | 1093.4 | 1741.5         | 100        | 5.819           | 10.5         | 8.9  | 1706.6 | 2117.2     |
| <b>3</b> 3   | 2.329            | 10.2        | 6.2     | 1278.3 | 1685.8         | 101        | 5.800           | 9.1          | 9.4  | 1405.5 | 1941.6     |
| 34           | 2+348            | £1.9        | 0.5     | 1150.0 | 1668.1         | 102        | 5.953           | 3.4          | 5.3  | 1302-5 | 1497-3     |
| 32           | 2.367            | 12.2        | 7.9     | 1277.4 | 1651.5         | 103        | 0+071           | 3.0          | 3.2  | 1224-6 | 1300-2     |
| 37           | 2.403            | 15.1        | 7.0     | 963.9  | 1654.1         | 105        | 4.222           | 3.3          | 3.2  | 1144.4 | 1435-0     |
| 38           | 2.425            | 17.0        | 9. L    | 1081.3 | 1534.1         | 100        | 6.275           | 5.0          | 5+0  | 1017.9 | 1146-5     |
| 39           | 2.444            | 17-6.       | 8.2     | 1479+1 | 1935-3         | 107        | . 0. Jiš        | 9.8          | 7.4  | 1008+1 | 1410-7     |
| 20           | 2.4903           | 10+3        | 9.2     | 1211.7 | 1771.3         | 140        | 6.390           | 11.9         | 9.9  | 1114.0 | 1474.6     |
| 42           | 2.501            | 20.4        | 9.5     | 1537.4 | 1674.5         | 110        | 6.452           | 20.1         | 11.2 | 1044.4 | 1483.9     |
| 43           | 2.521            | 21.3        | 5.4     | 1487.0 | 1 142. 7       |            | 6.525           | 21.4         | 12.5 | 1276.5 | 1534.2     |
| 44           | 2-540            | 23.0        | 10.3    | 1395.8 | 1449.4         | 112        | 0+507<br>A. 445 | 22.0         | 15.1 | 1481.7 | 15//.0     |
| 46           | 2.674            | 24.1        | 13.0    | 1774.5 | 1231.7         | 114        | 6.644           | 21.6         | 15.1 | 1527.6 | 1714.6     |
| 47           | 2.715            | 20.9        | 13.7    | 1674.1 | 1722.3         | 115        | 6.032           | 23.5         | 10.1 | 1543+1 | 1782.9     |
| 40           | 2,132            | 19.9        | 12.4    | 1790.3 | 1896.2         | 114        | 6.721           | 10-4         | 14+7 | 1554.5 | 1641.6     |
| 49           | 2.770            | 17-1        | 13.0    | 1847.3 | 1042.2         | 117        | 4.747           | 10.8         | 12.7 | 1707.8 | 1725.9     |
| 51           | 2.649            | 10-6        | 16.9    | 1908.8 | 2045.7         | - 119      | 6.932           | 7.4          | 6.4  | 1691.3 | 1731.5     |
| 52           | 2.690            | 8+4         | 9.3     | 2025-1 | 20:14.3        | 120        | 7.101           | 3.0          | 3.0  | 1633.0 | 1593+1     |
| 53           | 2.920            | <b>5</b> •2 | 5.9     | 1891.8 | 2310-1         | 121        | 7.230           | 0.3          | 0.5  | 1100 4 | 1425 -8    |
| 54           | 4.044            | 3.1         | 3.7     | 1512.7 | 2217.4         | 123        | 7.315           | 7.0          | 5.7  | 1069-0 | 1254-5     |
| 56           | 3.679            | 1.7         | 1.8     | 1503.7 | 2171.3         | 124        | 7.353           | 9.3          | 10.0 | 1117.2 | 1141+3     |
| 51           | 3.116            | 2+6         | 4+4     | 1371-1 | 25672          | , 125      | 1.342           | 14.7         | 12.7 | 997.4  | 1106.5     |
| 58           | 3,153            | 2.7         | 4.3     | 1635.8 | 2245+0         | 126        | 1.430           | 17.7         | 12.4 | 1022+2 | 1144+4     |
| 22.          | 34320<br>1.1.1.1 | ****        | 12.4    | 1445.9 | 2118.5         | 128        | 7.376           | 21.3         | 14.9 | 1117.5 | 1413.7     |
| 61           | 3.425            | 19.4        | 10.2    | 1483.2 | - 1821.6       | - 129      | 7. 564          | 20.0         | 14.2 | 1384.7 | 1576.4     |
| 62           | 3.463            | 21+5        | 11.0    | 1616.0 | 1401.8         | 130        | 7.003           | 21.7         | 14.6 | 1632.7 | 1739-6     |
| 63           | 3,432            | 19,8        | 9.6     | 1642.2 | 1971.0         | 131        | 7+641           | 8+15         | 15.8 | 1779.3 | 1774-5     |
| 07<br>45     | 3,541            | 20.7        | 10.6    | 1148.4 | 1567.2         | 133        | 7.718           | 17.4         | 15.4 | 1864.3 | 2033.8     |
| 66           | 3,575            | 22.4        | 11.7    | 1104.4 | 1507.2         | 134        | 1.756           | 13.2         | 12.5 | 1945.3 | 1951.8     |
| 67           | 3.614            | 23.3        | 10.1    | 1043.9 | 2039-2         | 115        | 7-195           | 12+5         | 12.0 | 1010.1 | 1814.1     |
| 68           | 3.074            | 20.4        | 12.4    | 1043.4 | 170 1.7        | 130        | 1.833           | 10.7         | 10.9 | 1107.5 | 1943*3     |
|              |                  | •           |         | •      |                |            |                 |              |      |        |            |
|              |                  |             | •       | •      |                |            |                 | т., <b>т</b> | •    |        |            |
| i + 70       | me in Yes        | rs from Jan | магү 1, | 1967.  | •              | •          |                 | • •          |      |        |            |
|              |                  |             |         |        |                | •          |                 |              |      |        |            |

•

4-64

÷

#### 4.5.3 Lake Data (continued)

( )

( )

()

0.210, 0.351, 0.305, 0.134, 0., 0., and 0. The corresponding set of weights for the computation of the hypolimnic average (9 - 18 meters) are: 0., 0., 0., 0.210, 0.364, 0.286, and 0.140. The data are listed in Table 4.5-2. In estimating the parameters, the first observation has been employed as an initial condition, leaving a total of 135 observations as a basis for parameter estimation.

#### 4.5.4 Results

Initial estimates of the parameters are required in order to start the formal parameter estimation routine. Independent evidence suggested values for each of the four parameters.  $a_{14}$  and  $a_{15}$  represent the intercept and slope of the exchange rate function,  $P_{11}$ , and correspond to the intercept and slope on the plot of Kato and Phillips's experimental data on exchange rate versus Richardson Number (Figure 4.2-1). The data of Kato and Phillips suggested that values of 2.5 and 1.0 would be appropriate for  $a_{14}$  and  $a_{15}$ , respectively. Because of the logarithmic nature of this relationship, the logarithm of  $a_{14}$  was actually estimated. This effectively increased the linearity of the response surface in the region of the solution, a desirable aspect from the point of view of estimating the parameter covariance matrix. The surface heat and mass transfer parameter,  $a_5$ , is specified as 0.18 by Harleman and Markofsky<sup>11</sup>.

This parameter is multiplied by wind speed to compute the transfer coefficient for evaporation and conduction. It has been included as an unknown here partially because of uncertainty as to the adequacy of wind speed data measured at Hancock Airport as representative of conditions over the lake. Thus, the final estimate for this parameter may include a correction factor for average wind speed, as well as any deviations of the actual parameter from the suggested value.

Initial simulations indicated that a value of 1.0 for parameter  $a_{22}$ , the maximum fraction of Ninemile Creek flow entering the hypolimnion, resulted in consistent over-prediction of mid-summer hypolimnic temperatures, even if the turbulent exchange parameter,  $a_{14}$ , was set to zero. The over-prediction became more severe with increasing values of the initial dilution ratio,  $a_3$ . This suggested that, despite its relative density, all of Ninemile Creek does not flow into the hypolimnion. Apparently, much of it is dispersed in the epilimnion before it has a chance to sink. If the fraction entering the hypolimnion is assumed to be zero, the chloride gradients are consistently under-predicted. Thus, it has been assumed that some portion of the flow enters the hypolimnion without initial dilution. The value for parameter  $a_{22}$  has accordingly been assumed to lie in the range of zero to one.

( )

C

()

The results of the final parameter estimation run are given in Table 4.5-3. Principal characteristics of temperature and chloride residuals obtained from this run are summarized in Table 4.5-4. The initial parameter guesses employed in this case correspond to a solution previously obtained with parameter a<sub>16</sub> equal to 0.01. This parameter determines the maximum rate of mixing under unstable (negative density gradient) conditions. Examination of residuals from this fit suggested that a higher maximum rate would be desirable. With some testing it was found that a doubling of a<sub>16</sub> would enhance the rate without appreciably affecting the stability of the calculations. Accordingly,  $a'_{16}$  was set to 0.02 and the parameter estimation routine was re-started at the solution obtained for  $a_{16} = 0.01$ . As shown in Table 4.5-2, the intercept and slope parameters,  $a_{1A}$ and  $a_{1K}$ , were reduced somewhat, but the best estimates of the other two parameters changed only slightly. Convergence to the solution was achieved with 7 iterations (derivative evaluations) and 16 objective function evaluations, or a total of  $16 + 4 \times 7$  or 44 equivalent function evaluations. This consumed a total of 8.15 minutes of CPU time on the IBM 370-168. The performance of the algorithm will be discussed in greater detail in Section 4.7.

Strictly speaking, it cannot be guaranteed that the solution obtained above is a global one, because of the inherent nature of the nonlinear estimation problem. Generally, the robustness of the solution Parameter Estimates and Confidence Regions Based upon All Data

|                                      |                 | Parameter       |                   |                |                                  |  |  |  |  |
|--------------------------------------|-----------------|-----------------|-------------------|----------------|----------------------------------|--|--|--|--|
|                                      | a <sub>14</sub> | a <sub>15</sub> | <sup>6</sup> 22 . | <sup>a</sup> 5 | Function                         |  |  |  |  |
| Estimate                             | -0.2617         | 1.0012          | 0.2669            | 0.2062         | -1542.65 = \$                    |  |  |  |  |
| Standard Deviation                   | 0.2872          | 0.0367          | 0,0122            | 0.0095         | •                                |  |  |  |  |
| Correlation Matrix                   |                 |                 |                   |                |                                  |  |  |  |  |
| a <sub>14</sub>                      | 1.0             | ÷               | ·                 |                |                                  |  |  |  |  |
| a15                                  | 0.989           | • 1.0           |                   |                |                                  |  |  |  |  |
| a22                                  | 0.249           | 0.144           | 1,0               |                |                                  |  |  |  |  |
| a <sub>5</sub>                       | 0,224           | 0,205           | -0,108            | 1.0            |                                  |  |  |  |  |
| Principal Component Co               | pefficients     |                 |                   |                | øj<br>Standard<br>Deviation      |  |  |  |  |
| P1                                   | 2.36            | 18,10.          | 17.42             | 24.60          | 1.458 -                          |  |  |  |  |
| ρ <sub>2</sub>                       | -0.12           | 0.62            | -59.61            | 72.43          | 1.053                            |  |  |  |  |
| P3                                   | ~0.59           | -7,58           | 53.43             | 72.36          | 0,872                            |  |  |  |  |
| P4 .                                 | -2.49           | 18.89           | 6.71              | 2.92           | 0.066                            |  |  |  |  |
| End Points of Princip                | Al Axes .       |                 |                   |                | <sup>و</sup> ب ه <sup>• بو</sup> |  |  |  |  |
| $\overline{\rho_1} = -\chi \sigma_1$ | -1.135          | 0.892           | 0,255             | 0.196          | -3,23                            |  |  |  |  |
| ρ <sub>1</sub> = +χ σ <sub>1</sub>   | 0.612           | 1.111           | 0.279             | 0.216          | -5.04                            |  |  |  |  |
| $\rho_2 = -\chi \sigma_2$            | -0.230          | 0.998           | 0,295             | 0.185          | -4.75                            |  |  |  |  |
| $\rho_2 = +\chi \sigma_2$            | -0.293          | 1.004           | 0.238             | 0.227          | -2.76                            |  |  |  |  |
| ρ <sub>3</sub> = -χ σ <sub>3</sub>   | -0.131          | 1.029           | 0.246             | 0.189          | -5.34                            |  |  |  |  |
| ρ <sub>3</sub> = +χ σ <sub>3</sub>   | -0, 392         | 0.974           | 0.288             | 0.224          | -3,27                            |  |  |  |  |
| P4 = -X 04                           | -0,220          | 0.996           | 0.267             | 0.206          | -4.45                            |  |  |  |  |
| $\rho_A = + \chi \sigma_A$           | -0.303          | 1.006           | 0.267             | 0.206          | -4.25                            |  |  |  |  |

b natural logarithm transform

( ;

()

e correspond to approximate bounds of 95% confidence region (see eq. 4.5-13 to 15)  $\dot{\chi} = \sqrt{\chi^2_{,05}}$  with 4 degrees of freedom = 3.08

d sampled objective function value at end point of principal axis value at solution ; expected value, based upon quadratic approximation of response surface (equation 4.5-6 ),=-4.75 =  $-\chi^2$  /2.0 G Total of 135 profiles between April, 1968 and December, 1974

- finitial guesses : .5096, 1.091, .2632, .2110,  $\phi_0 = -1552.17$

| Variable                          | т <sub>.</sub>          | т <sub>н</sub>          | с <sub>в</sub> .        | с <sub>н</sub>          |
|-----------------------------------|-------------------------|-------------------------|-------------------------|-------------------------|
| Units                             | ိင                      | °c                      | g/m <sup>3</sup>        | g/m <sup>3</sup>        |
| Mean                              | 0.280                   | 0.289                   | -33.5                   | -15.8                   |
| Standard Error <sup>a</sup>       | 1.676                   | 1,873                   | 198.9                   | 185.3                   |
| Serial Correlation<br>Coefficient | 0.453                   | 0.603                   | 0,511                   | 0.517                   |
| Correlation Matrix                |                         |                         |                         |                         |
| T <sub>e</sub>                    | 1.0                     |                         |                         |                         |
| т <sub>н</sub>                    | 0.370                   | 1.0                     |                         |                         |
| с <sub>в</sub>                    | -0.103                  | 0.070                   | 1.0                     | •                       |
| с <sub>ң</sub>                    | 0.076                   | 0.132                   | 0.461                   | 1.0                     |
| овв <sup>р</sup>                  | 0.788 x 10 <sup>4</sup> | 0.260 x 10 <sup>4</sup> | 0.994 x 20 <sup>7</sup> | 0,102 x 10 <sup>8</sup> |
| RSS <sup>C</sup>                  | 0.377 x 10 <sup>3</sup> | 0.470 x 10 <sup>3</sup> | 0.530 x 10 <sup>7</sup> | 0.460 x 10 <sup>7</sup> |
| R <sup>2</sup> đ                  | 0,952                   | 0,819                   | 0.467                   | 0.549                   |

Table 4.5-4 Characteristics of Temperature and Chloride Residuals

()

Ċ

( •

a - Standard Error = (Residual Sum of Squares/134)<sup>0.5</sup> b - O S S = Observed sum of squares of deviations from mean B. c - R S S = Residual sum of squares of deviations from zero. d -  $R^2$  = 1 - (R S S / O S S)<sup>-</sup>

(\_\_)

C

would be tested by starting the estimation routine at a variety of different initial parameter values and testing whether they all converge to the same point. However, the dimension of the parameter vector and the cost of execution renders this approach impractical in this application. The adequacy of the solution is indicated first by the apparent feasibility of the parameter estimates, compared to the expected values and ranges discussed above. Secondly, the parameters are all relatively well-defined, with coefficients of variation ranging from 0.036 to 0.287, or corresponding t values ranging from 27.3 to 3.5\*. This indicates that this maximum of the objective function is a relatively distinct region in parameterspace. Using the mapping option added to the Bard program, the objective function has been evaluated at the end points of the principal component axes. These evaluations (Table 4.5-3) indicate that the response surface is fairly well-behaved in the vicinity of the solution, since the computed values of the objective function at the end points are not greatly different from those estimated based upon a quadratic approximation of the response surface (equation 4.5-6), which predicts that the solution should be 4.75, as compared with computed values ranging from 3.23 to 5.34. This indicates that response of the function to changes in the parameter

a14 was estimated on a log scale; the standard deviation of the logarithm of a14 approximately equals the coefficient of variation of the parameter on a linear scale.

С.

()

values near the solution is not highly non-linear. The solution is not a "peak" in the objective function, but rather a high point on a ridge, because of the high correlation between parameters  $a_{14}$ and  $a_{15}$  (r = 0.989).

As discussed in Chapter 1, parameter stability is a desirable property of a valid model. In order to develop further evidence for the verification of the model, the stability of the parameter estimates has been examined by dividing the data set into six groups and estimating optimal parameters for each group separately. The six groups correspond to years, with the exception that data for 1970 and 1971 have been combined, because only nine observations were available for the latter. To ease parameter estimation difficulties and reduce costs, only two of the four parameters have been estimated for each group:  $a_{14}$  and  $a_{22}$ . The remaining two,  $a_{15}$ and a<sub>5</sub> , have been constrained to the values obtained above. As will be discussed in the next section, the value of and has a theoretical basis and conforms to the value derived by Kato and Phillips. It is also highly-correlated with a14 . a5 corresonds roughly to the value suggested by Harleman and Markofsky<sup>11</sup>. As will be ehown, this parameter is not as critical to analysis of the outfall design issue as are the other parameters, because it does not strongly influence vertical exchange rates. The other two parameters are more empirical and do not have external bases for

their values. Accordingly, these have served as foci for parameter stability studies.

Sampling error in the covariance matrix of the residuals can become important as sample size is reduced. In order to minimize such effects and permit a focus on parameter variations alone, the covariance matrix of the residuals for the individual data groups has been assumed to be equal to that obtained from all the data combined (Table 4.5-4). In this case, the constant portion of the log-likelihood objective function employed in the Bard program is:

$$\Phi = -\frac{1}{2} \frac{4}{12} \frac{4}{12} \frac{4}{12} \frac{4}{12} \left[v\right]_{1j}^{-1}$$
(4.5-16)

where,

This amounta to weighted least squares, with known weights determined from the inverse of the specified covariance matrix.

C

**(**)

Ľ

( :

**(**)

The assumed initial conditions for each period have been determined by the last observation in the previous period, with the exception of the 1968 and 1972 groups, in which cases the first observations in the respective years have served as initial conditions. The solution obtained for all years together has also been used as a starting point for the parameter vector in the estimation routine. Results of the analysis are given in Table 4.5-5. Approximate 95% confidence regions for the parameter estimates are illustrated in Figure 4.5-2.

1968 was the only year in which the simulations were begun in April, after the onset of stratification. It is possible that the differences in the optimal parameter estimates for that year could

#### Table 4.5-5

#### Best Parameter Estimates for Individual Years

| Years            |           | 1968     |                | 1969           |                | 1970-71 | L     | 1972     |       | 1973   |                | 1974   |       |
|------------------|-----------|----------|----------------|----------------|----------------|---------|-------|----------|-------|--------|----------------|--------|-------|
| N                |           | 20       |                | 33             |                | 30      |       | 17       |       | 17     |                | 17     |       |
| ф <sub>о</sub>   | a         | -41.75   |                | -47.83         |                | -69,42  |       | -41.19   |       | -29.06 |                | -33.93 |       |
| Φ                | ь         | -35.25   |                | -46.93         |                | ~68.42  |       | -39.17   |       | -28.47 |                | -32.05 |       |
| φ φ <sub>0</sub> | C         | +6.50    | ÷              | +0.90          |                | +1.00   |       | +2.02    |       | +0.59  |                | +1.88  |       |
| Parameters       | đ         | θι       | Θ <sub>2</sub> | 0 <sub>1</sub> | θ <sub>2</sub> | θ1      | θ2    | θ1       | θ2    | Θ1     | θ <sub>2</sub> | Θ1     | θ2    |
| Mean             |           | -0.690   | 0.299          | -0.341         | 0.235          | -0.269  | 0,242 | -0.414 · | 0.298 | -0.206 | 0.266          | -0.140 | 0.242 |
| Standard Dev     | 1.        | 0.237    | 0.030          | 0.093          | 0.023          | 0.064   | 0.020 | 0.110    | 0.032 | 0.077  | 0.038          | 0.103  | 0.039 |
| End Points       | <b>of</b> | Principa | al Axe         | e<br>S         |                |         |       |          |       |        |                |        |       |
|                  |           | -1.181   | 0.231          | -0.559         | 0.181          | -0.339  | 0.264 | -0.677   | 0.220 | -0.270 | 0.297          | -0,257 | 0.287 |
|                  |           | -0.099   | 0.368          | -0.123         | 0.290          | -0,199  | 0.221 | -0.150   | 0.376 | -0.140 | 0.235          | -0.021 | 0.197 |
|                  |           | -0.433   | 0.273          | -0.278         | 0.220          | -0.408  | 0.199 | -0.360   | 0.283 | -0.303 | 0.180          | -0.363 | 0.156 |
|                  |           | -0.847   | 0.325          | -0.404         | 0.251          | -0.130  | 0.286 | -0.465   | 0.313 | -0.028 | 0.351          | 0.083  | 0.327 |
| L                |           | 1.       |                |                |                |         |       | 1        |       |        |                |        |       |

a -  $\Phi_0$  = objective function value at starting point = best estimate for all years [ $\theta_1$  = -0.262,  $\theta_2$  = 0.267]

- b  $\Phi_{\pm}$  = objective function value at solution for given year
- c \* indicates that  $\phi_* \phi_*$  is significantly different from 0 at the 95% confidence level if normally-distributed errors are assumed
- $d \theta_1 = \ln (a_{14}), \theta_2 = a_{22}, a_{15} = 1.0012, a_5 = 0.2062$
- e corresponds to approximate bounds of 95 % confidence region for parameter estimates, assuming normally-distributed errors



(;

# Figure 4.5-2

(

Ć

be attributed to the initial conditions. Errors in the initial conditions at this point could have profound effects on model simulations for the entire stratification period. Some of the seasonal biases in model performance, discussed in Section 4.5.6, could also be involved. In addition, the sensitivity patterns to parameter and forcing function values appear to be different in 1968 than in other years, as will be discussed in Section 4.6.1.

# 4.5.5 Interpretation of Parameter Estimates

1

Estimates of the parameters dealing with the vertical exchange function,  $F_{11}$ , can be interpreted from a mechanical energy balance point of view. The basis of the mixing model is the relationship between kinetic energy input at the surface due to wind shear stress and the increase in buoyant potential energy due to mixing. From the definition of the surface friction velocity<sup>4</sup>, energy input to the lake is given by<sup>29</sup>:

$$\frac{d \ \overline{KE}}{dt} = D_{E} F_{13}^{3} A_{E} \qquad (4.5-17)$$

#### 4.5.5 Interpretation of Parameter Estimates (continued)

where,

F<sub>13</sub> = surface friction velocity (m/sec) (see Table 4.3-3)

For vertical exchange at a rate  $F_{11}$  between the hypolimnion and the epilimnion, the rate of change of buoyant potential energy is given by:

$$\frac{d \overline{PE}}{dt} = g F_{11} A_H \overline{\Delta Z}_G (D_H - D_E) \qquad (4.5-18)$$

where,

( ·

()

 $g = acceleration of gravity = 9.8 m/sec^2$ 

 $\Delta Z_{G}$  = difference in center of gravity between the epilimnion and hypolimnion = 8.5 meters\*

In units of meters per second, the exchange rate function at moderate to high Richardson Numbers is given by\*\*:

\* Determined from lake surface area versus depth plot (Chapter 3).

\*\* The second term in the denominator of the exchange rate function in Table 4.3-3 is negligible during stratified periods.

$$P_{11} = \frac{a_{14} P_{13}}{a_{15}}$$
(4.5-19)  
$$F_{12}$$

$$F_{12} = Richardson Number = \frac{g_{a_{17}} (D_H - D_E)}{D_E F_{13}^2}$$
 (4.5-20)

Substituting in the appropriate parameter values, and assuming

 $a_{15} = 1.0012 = 1:$ 

()

 $(\cdot)$ 

()

$$F_{11} = \frac{0.086 \ D_E \ F_{13}^3}{g \ (D_H - D_E)}$$
(4.5-21)

Combining with equation (18):

$$\frac{d \overline{PE}}{dt} = 0.73 D_{E} F_{13}^{3} A_{H} \qquad (4.5-22)$$

This result can be compared with the expression for the rate of kinetic energy input [equation (17)]:

4.5.5 Interpretation of Parameter Estimates (continued)

( :

( ;

$$\frac{d \overline{PE}}{dt} = 0.73 \frac{A_{H}}{A_{E}} \frac{d \overline{KE}}{dt} = 0.47 \frac{d \overline{KE}}{dt} \qquad (4.5-23)$$

Thus, the parameter estimates indicate that 47% of the kinetic energy supplied to the system by the wind is used to increase the potential energy of the system by mixing the hypolimnic and epilimnic waters. The remainder is presumably dissipated as heat of friction.

Implicit in the estimate of  $a_{14}$  and, therefore, in the estimate of 47% conversion, is a correction factor for average wind speed over the lake relative to average wind speed at Hancock Airport. Thus, while the model calculations for this lake are not sensitive to any differences in average wind speeds, use of the same parameter values in modelling other lakes may not be appropriate, unless adequate adjustements are made. The estimate of the slope,  $a_{15}$ , is independent of such errors and would be expected to be more generally applicable.

The formulation employed by Kato and Phillips to represent their experimental data has been shown to be useful in modelling the mixing process in a real lake. The best estimate of the slope,  $a_{15}$ , is essentially the same as that employed by Kato and Phillips. The value of 1 gives rise to the mechanical energy balance arguments presented

#### 4.5.5 Interpretation of Parameter Estimates (continued)

()

(° ;

above. The intercepts,  $a_{14}$ , are appreciably different (0.77 versus 2.5). As discussed in Section 4.2, Kato and Phillips's value of 2.5 indicates that the rate of change of potential energy in their system was 125% of the rate of kinetic energy input. It was suggested that this could be attributed to possible errors in their kinetic energy input rates. The same general type of problem may exist in this case, with the possible errors in average wind speed discussed above. One important difference between the two applications is that Kato and Phillips applied the model to represent entrainment (i.e., one-way transfer from the bottom to the top layer, resulting in a moving interface), whereas the function has been applied here to represent two-way exchange between the bottom and top layers with a fixed interface. Apparent differences in the  $a_{14}$  parameter estimates derived from the two studies might also be attributed to this factor.

( )

( )

 $\bigcirc$ 

Essential properties of the residuals are summarized in Table 4.5-6. Comparisons of model predictions have been made with observations of temperature, chloride, and density for the epilimnion, hypolimnion, and vertical gradient (hypolimnion-epilimnion). Time series plots of observations and prediction are displayed in Figures 4.5-3 to 4.5-5. Residuals are plotted against time in Figures 4.5-6 to 4.5-8. Observations are plotted against predictions in Figures 4.5-9. Histograms and normal probability plots are depicted in Figures 4.5-10 and 4.5-11, respectively.

t-tests (line b, Table 4.5-6) indicate statistically-significant bias in the model predictions for six out of the nine components. However, the bias is generally small compared to the size of the residuals, since less than six percent of the total residual variance can be attributed to deviations of the mean from zero (line d). Effective  $R^2s$  range from 0.26, for chloride gradient, to 0.95 for surface temperature (line e). Both the Durbin-Watson statistics (line f) and the non-parametric runs tests (line g) indicate significant auto-correlation in the time series of residuals in all cases. This serial correlation is also evident in residual time series plots (Figures 4.5-6 to  $4.5-8_{J}$ .

The Kolmogorov-Smirnov test has been applied to test for normality. If the calculated residual means and standard deviations are employed in the test, non-normality is indicated at the 95% confidence level

Table 4.5-6 Results of Residuals Analysis

Variable

|                    |                                |     | T <sub>e</sub> | T <sub>H</sub> | T <sub>H</sub> - T <sub>E</sub> | с <sub>в</sub> | c,     | ° <sub>E</sub> ∸ ⊂ <sub>E</sub> | D <sub>E</sub> | D <sub>H</sub> | D <sub>H</sub> -D <sub>E</sub> |
|--------------------|--------------------------------|-----|----------------|----------------|---------------------------------|----------------|--------|---------------------------------|----------------|----------------|--------------------------------|
| Observed Variable  | Mean                           | Т   | 13.25          | 9.03           | -4.21                           | 1448.          | 1811.  | 364.                            | 1.39           | 2.68           | 1.30                           |
|                    | Standard Dev.                  |     | 7.64           | 4.39           | 4.38                            | 271.           | • 275. | 232.                            | 1.00           | 0.52           | 0.85                           |
| Estimated Variable | Nean                           |     | 12.97          | 8.75           | -4.22                           | 1491.          | 1796.  | 315.                            | 1.42           | 2.62           | 1.20                           |
|                    | Standard Dev.                  | - 1 | 8.21           | 5.30           | 4.67                            | 294.           | 260.   | 202.                            | 1.09           | 0.49           | 0.95                           |
| Residual           | Nean                           |     | 0.28           | 0,29           | 0.01                            | -33.           | 16.    | 49.                             | -0,03          | 0.06           | 0.10                           |
|                    | Standard Dev.                  | a   | 1.65           | 1.84           | 1.99                            | 195.           | 284.   | 193.                            | 0.43           | 0.35           | 0.47                           |
| T-test for Bias    | :                              | ъ   | 1.97*          | 1.83*          | 0.06                            | -1.98*         | 0.99   | 2.95*                           | 0.87           | 2,08*          | 2.37*                          |
| Standard Error     |                                | •   | 1.67           | 1.87           | 1.99                            | 198.           | 185.   | 199.                            | 0.43           | 0.36           | 0.48                           |
| (Mean/ Std. Err    | <del>or</del> ) <sup>2</sup> . | ٩   | .028           | .024           | <.001                           | .028           | .007   | .060                            | .006           | .031           | .040 ·                         |
| R <sup>2</sup>     | •                              | •   | .95            | .82            | .79                             | .47            | .55    | .26                             | .82            | .53            | .69                            |
| Durhin-Watson S    | tatistic                       | £   | 1.07*          | 0.79+          | 0.83*                           | 0.96+          | 0.97+  | 1.09*                           | 1.10           | 0.84+          | 0.99*                          |
| Runs Test Paraz    | eter ·                         | 9   | -5.43*         | -6.47*         | -6.28*                          | -4.74*         | -4.16* | -4.46*                          | -4.79*         | -4.27*         | -4.25*                         |
| Prob(normality)    |                                | h   | .855           | . 349          | .045*                           | .463           | .657   | .387                            | .446           | .111           | .982                           |
| Prob(normality)    |                                | 1   | .104           | .013*          | .041*                           | .016*          | .160   | .019*                           | .107           | .001*          | .057                           |
| Yearly P Statis    | tic                            | 5   | . 3.65*        | 1.66           | 1.72                            | 5.29*          | 4.78*  | 4.48*                           | 4.98*          | 5.55* ·        | 5.24*                          |
| Variance           | Praction                       | ×   | .146           | .072           | .074                            | .199           | .183   | .174                            | .189           | .206           | .197                           |
| Seasonal P. Stat   | datic                          | 1   | 8.87*          | 13.48*         | 13.32*                          | 2.94-          | 3.24*  | 5.11*                           | 2.72*          | 1.25           | 5.08*                          |
| Variance           | Fraction                       | ₽   | .390           | .492           | .490                            | .174           | .189   | .269                            | .163           | .083           | .268                           |

a - Standard Deviation around mean.

- b t = mean/standard error of mean ; degrees of freedom = 135.
- c Standard Error around zero = square root of mean squared residual.
- d fraction of residual variance attributed to bias.
- $e R^2 = 1 (Residual Sum of Squares)/(Observed Sum of Squares of Deviations around Mean).$
- f autocorrelation present in time series at the 95% level if d <d = 1.59 (raf. 17)
- g Runs test parameter = (observed number of runs expected number)/standard deviation of expected number; distributed as N(0,1.) (ref. 2).
- h Kolmogorov-Smirmov Test, assuming mean and standard deviation given in lines 5 and 6 (ref. 12)
- i Kolmogorov-Smirnov Test, assuming mean = 0., standard deviation = standard error { line 8} (ref. 1)
- j Analysis of Variance Statistic for Yearly Stratification ; degrees of freedom = 6,128.
- k Fraction of total residual variance explained by yearly groupings = between-group sum of squares/ total sum of squares.
- 1 Analysis of Variance Statistic for Seasonal Stratification, degrees of freedom = 9,125.
- m Fraction of total residual variance explained by seasonal groupings = between-group sum of squares/ total sum of squares.
- \* Null Hypothesis rejected at the 95% confidence level.



()

(



(

Ć

E. A



()

(:

 $C^{\circ}$ 



**(** )

**(** ,

(





C

**(**)



**(**)

C

**(**)



1

(:

C





Figure 4.5-10

#### Histograms of Residuals

 $\langle \rangle$ 

()







(T

C

 $C_{\pm}$ 

#### 4.5.6 Analysis of Residuals (continued)

only in the case of temperature gradient (line h). If the residuals are assumed to have means of zero and standard deviations corresponding to the computed standard errors in line c, the hypothesis of normality is rejected in cases of five out of the nine variables (line i). It should be noted that, since the mean and standard deviation have been calculated directly from the eample, the power of the Kilmogorov-Smirnov test to detect non-normality is somewhat hindered. As a result, the computed probabilities of normality are biased upwards. Histograms (Figure 4.5-10) and normal probability plots (Figure 4.5-11) support the conclusion that the temperature gradient residuals are the least normal of any of the variables examined. Their distribution appears to be skewed toward positive values. This could be a consequence of the approximate lower limit of zero in this variable, since observed hypolimnion temperatures rarely exceed those of the epilimnion. Some of the probability plots tend to curve downwards at the low tails and upwards at the high tails, indicating a tendency for the tails to be lighter than normal, or for the distributions to be on the uniform side of normal.

The residuals have also been examined for seasonal and yearly patterns. A total of ten seasonal groups have been formed by aggregating at increments of 0.1 years. For each type of grouping, an analysis of variance has been done to determine the statistical significance of variations of the means of the residuals between groups. The confidence levels for the statistics computed from these analyses

#### 4.5.6 Analysis of Residuals (continued)

()

( :

()

are only approximate because of the serial correlation within the groups. These analyses have been done chiefly to characterize general features of the residual patterns. It should also be noted that the yearly and seasonal groupings are not completely independent of each other, since sampling strategy shifted in the later years of the lake survey. During that period, sampling crews were apparently less eager to venture out onto the lake ice, so that cold-season samples are relatively scarce in the later years (Table 4.5-2). Hence, apparent yearly effects may be partially seasonal effects and vice-versa.

Seasonal- and yearly-mean residuals for each variable are plotted in Figures 4.5-12 and 4.5-13, respectively. F statistics in these figures and in Table 4.5-6 indicate significant variations between yearly groups in seven out of the nine variables and between seasonal groups in eight out of the nine variables. The variance fractions indicate that a maximum of 20% of the residual variance (in the case of  $D_{\rm H}$ ) can be attributed to year-to-year variations, while up to 50% can be attributed to seasonal variations ( $T_{\rm H}$ ,  $T_{\rm H} - T_{\rm E}$ ).

Generally, seasonal effects on temperature residuals appear to be the strongest of the relationships examined and also appear to be quite periodic. Time series plots of temperature residuals support this (Figure 4.5-6). Epilimnion temperature is generally over-predicted by about  $2^{\circ}$ C during the cold seasons and under-predicted by about  $1^{\circ}$ C during the warm seasons. Periodicity in the hypolimnion and gradient



Figure 4.5-12

 $\frown$ 



Figure 4.5-13

Mean Residuals Versus Year

 P statistic + between-years dans symme deviation/ a significant at the 15 % local
a hotmonopass sets of spatnes/total out of spatnes
= function of variance exclained by yearly gravying • man algorithmatly different from sero at the 15 % confidence lowel a starte minimum exercise 420

 $\mathbf{)}$ 

4-95

#### 4.5.6 Analysis of Residuals (continued)

( )

( :

()

residuals may reflect the observed periodicity in the epilimnion residuals, with an apparent lag time of about 0.3 years. As will be shown, the epilimnion temperatures are sensitive only to a<sub>5</sub>, the surface mass transfer parameter, and not sensitive to any of the other estimated parameters dealing with exchange between the hypolimnion and epilimnion. Average epilimnic temperatures are fixed by the overall heat balance on the lake.

The periodicity in the epilimnion temperature residuals can be attributed to two factors. First, during cold seasons, a minimum temperature of 0°C has been arbitrarily specified for the epilimnion. In actual fact, observed temperatures rarely go below 2°C, even under ice cover. It is apparent that if this minimum temperature were specified at 2°C instead of 0°C, much of the periodicity in the residuals would be removed. Secondly, the function employed for calculation of solar radiation (U<sub>q</sub> in Table 4.3-2) may have been inadequate. This function includes only a single cosine curve. A recent article by Thompson<sup>36</sup> outlines a general method for estimating solar radiation from sky cover for locations in the United States. In this scheme, the function used to compute clear-sky radiation involves five periodic components. Comparisons of the predictions of the two methods using Syracuse meteorologic data indicate that the method employed in the model tends to under-predict radiation in cold seasons and to over-predict it in warm seasons. An examination of the residuals obtained in estimating U<sub>o</sub> (see Table 4.3-2) has revealed similar
## 4.5.6 Analysis of Residuals (continued)

÷.

( ) i

C ;

patterns, despite a high  $R^2$  of 0.970. Thus, some of the seasonal periodicity in the temperature residuals may have been due to improper specification of the boundary conditions. Despite these problems, the model explains 95% of the variance in the epilimnion temperature data. The apparent decreasing trend in the yearly-average epilimnion temperature residuals (Figure 4.5-15) may be due to decreasing sampling frequencies during cold seasons, as discussed above. Because of these problems, the estimate of the surface mass transfer coefficient,  $a_5$ , may be somewhat biased.

Yearly effects on epilimnic chloride residuals appear to be substantial (Figure 4.5-13). Again, the surface conditions are not very sensitive to the mixing mechanisms or parameter values. In fact, average surface chloride concentrations are essentially fixed by the hydrologic and salt-influx boundary conditions, due to mass balance constraints. Thus, variations in the mean yearly surface chloride residuals can be attributed to errors in specifications of the hydrology or salt loadings.

Patterns in the density residuals reflect those in the temperature and chloride residuals. An under-prediction of the density gradient is evident during cold seasons (Figure 4.5-12), particularly during the seasons 0.7 and 0.8. While some of this may be due to the temperature problems discussed above, model errors could also be involved. One aspect of the model which could be important here is the

()

 $\cap$ 

assumption of constant epilimnion and hypolimnion volumes. Erosion of the thermocline effectively increases epilimnion depth and volume in the late seasons prior to fall overturn. This effectively increases the volume over which the kinetic energy due to wind shear stress is dissipated. With an increase in depth  $(a_{17})$ , the Richardson Number  $(F_{12}, Table 4.3-3)$  would increase, the vertical exchange rate  $(F_{11})$ would decrease, and the computed density gradient would, in turn, increase. Thus, if effecte of a migrating thermocline were incorporated into the model, the tendency to under-predict late summer and early fall density gradients might be reduced.

Serial correlation in the residuals can be attributed to periodic or auto-correlated errors in the specification of boundary conditions, to the effects of factors not considered in the model, and to aspects of the measurement process. Bard<sup>3</sup> notes that serial correlation in the residuals is more generally the rule than the exception in cases of dynamic models. These results are not grounds for rejection of the model. The primary effect of the serial dependence of the errors is upon the confidence regions of the estimated parameter values.

If the model were linear, and the residuals were uncorrelated, unbiased, and normally-distributed, the confidence regions calculated for the parameter estimates would apply exactly. In this case, however, the model is nonlinear, and the residuals are auto-correlated, somewhat biased, and, in some cases, not normally distributed. A standard

#### 4.5.6 Analysis of Residuals (continued)

approach to determining confidence regions for parameters under such conditions would involve Monte-Carlo simulation. Such an approach is precluded in this case by the expense of implementing the parameter estimation routine. Thus, in applying the model and in assessing errors in model projections, the approximate nature of the estimate of the parameter covariance matrix must be considered.

#### 4.6 Model Applications

( )

()

#### 4.6.1 Sensitivity Analysis

Before applying the model to assess the potential effects of the sewage outfall design upon general aspects of vertical mixing in Onondaga Lake, an analysis has been performed to estimate the sensitivity of the model simulations to some of the parameters and forcing functions. The parameters studied have included a 14 (the numerator of the vertical exchange rate function,  $P_{11}$  ) ,  $a_5$  (the surface mass transfer parameter for evaporation and conduction), and a<sub>22</sub> (the maximum fraction of Ninemile Creek flow entering the hypolimnion). Sensitivities to U6 (the chloride flux from Allied Chemical's waste beds) and  $F_{\Delta}$  (the Allied cooling water flow) have also been evaluated. The sensitivity to a<sub>14</sub> can also be used as a measure of the sensitivity to the average rate of wind-induced kinetic energy input to the lake. In the numerator of  $P_{11}$ , the exchange rate function,  $a_{14}$  and  $F_{13}^3$  occur as a product.  $P_{13}^3$ , the cube of the friction velocity, is proportional to the rate of kinetic energy input, according to equation 4.5-17. The purpose of the analysis is to assess the relative importance of various mechanisms in controlling lake mixing under past and present conditions.

The period of record (1968-74) has been simulated using the best estimates of the parameters developed in Section 4.5 and recording results at 0.05-year intervals. Normalized sensitivity coefficients have been computed as a function of time for each factor studied, using the finite-difference methods discussed in Section 4.4. Sensitivity coefficients are essentially normalized first derivatives of the state variables with respect to the factor values:

$$S_{i,j,t} = P_i \frac{\partial Y_{j,t}}{\partial P_i}$$
(4.6-1)

where,

()

 $( \ )$ 

The coefficients can be interpreted as the marginal changes in the state variables which would result from fractional changes in the factors. The normalization adjusts for differences in factor scales. Sensitivity coefficients for temperature, chloride and density are plotted in Figures 4.6-1 to 4.6-3 respectively. The numbers labelling the curves in each plot correspond to the eubscripts of the  $p_i$  variables defined above.





( :







**(**,

T



C

(

C





One general aspect of the temperature and chloride sensitivies is the tendency for "spikes" to occur during overturn periods, particularly in late 1972 and 1974. This is essentially because changes in the factors influence the times at which density equivalence is reached and rapid mixing occurs. Just prior to overturn, both temperature and chloride gradients may exist, but with opposing effects on the density gradient, which is approaching zero. When mixing occurs, temperature and chloride variables change rapidly, giving rise to the spikes in the sensitivity plots.

Epilimnion temperatures appear to be insensitive to all factors examined, except for the surface transfer parameter,  $p_2$ . As vertical stratification develops during each season, sensitivities of hypolimnion temperature with respect to the dominant factors,  $p_1$  and  $p_4$ , increase in absolute values and are opposite in sign. Temperature gradients (hypolimnion - epilimnion) are also chiefly controlled by these factors. Chloride sensitivities (Figure 4.6-2) indicate the dominance of  $p_4$ , the Allied Chemical chloride flux, in both the epilimnion and hypolimnion. Chloride gradients are controlled chiefly by  $p_1$  and  $p_4$ . Epilimnion density is controlled by  $p_4$  and  $p_2$ , while hypolimnion density and density gradient both respond most dramatically to changes in  $p_4$  and  $p_1$  during most years.

The density gradient sensitivities are particularly important and warrant more detailed analysis. Generally,  $p_1$  and  $p_4$  appear to be controlling, as in the cases of temperature and chloride gradients.

During peak stratification seasons,  $P_1$  sensitivity coefficients are on the order of -1.5, while  $P_4$  coefficients are approximately 2.0. Thismeans that a 1% increase in the chloride flux is estimated to have about the same effect on peak density gradients in the lake as a 2/1.5 = 1.33% decrease in  $a_{14}$ , or, equivalently, a 1.33% decrease in the average rate of kinetic energy input. Since the latter depends upon the 3.75 power of wind speed, this would correspond to a 0.35% decrease in average wind speed over the lake. This helps to quantify the stratifying effect of the industrial discharge relative to the natural forces tending to mix the lake. Because of the nonlinear nature of the model, these figures are valid only for small changes in the factors.

Cooling water flow,  $p_5$ , has a negative impact on a density gradient during late summer seasons, except during 1970. In that year, cooling water was consistently withdrawn from the epilimnion (see  $U_4$ , Table 4.3-2). Thus, this factor was not as important a mixing mechanism in the lake during 1970 as it was during other years, in which the cooling water was withdrawn from the hypolimnion.

In 1968, the sensitivity equations are markedly different from those observed during other years, in that factors 3 and 5 appear to be controlling peak density gradients. The reasons for this are unclear, but may be related to the fact that the simulation was started after the onset of stratification in the spring. This fact, coupled with some of the seasonal biases in the model discussed in Section 4.5-6, may account for the differences. During this period, the sensitivity

coefficients are not in equilibrium with the factors driving the model, but are still heavily influenced by initial conditions. The qualitative differences in the sensitivity coefficients during 1968 may in part account for the differences in optimal parameters estimated for that year in Section 4.5.

The degrees and types of mixing in the lake over the study period are characterized in Figure 4.6-4. In units of year<sup>-1</sup>, the total hypolimnic dilution rate (HDR) is defined as the total outflow rate from the hypolimnion divided by the hypolimnion volume:

$$HDR_{T} = \frac{Q_{16} + Q_{28} + Q_{30} + Q_{32}}{V_{H}} \times 365. \qquad (4.6-2)$$

The non-advective portion of this is attributed to the density-dependent exchange between the epilimnion and hypolimnion,  $F_{11}$ , and is given by:

HDR<sub>na</sub> = 
$$\frac{Q_{28}}{V_{\rm H}}$$
 x 365. (4.6-3)

These variables are particuarly important as measures of the oxygen resources of the hypolimnion. The logarithms of these two variables are plotted against time in Figure 4.6-4. During fall overturn periods, maxima are reached in the vicinity of 2.6. These upper limits are

()

Simulated Total and Non-Advective Hypolimnic Dilution Rates



# Figure 4.6-5





(

determined by the assumed value of  $a_{16}$ , which is somewhat arbitrary. Accordingly these periods should be interpreted as periods in which the lake is essentially completely-mixed.

During periods of peak stratification, the advective mixing mechanisms become important. If cooling water is withdrawn exclusively from the hypolimnion, the advective component of the HDR amounts to 2.24 year<sup>-1</sup>, or 0.35 on a logarithmic scale. The non-advective component generally reaches a minimum in the vicinity of -0.2 on a log scale, or about one quarter of the advective cooling-water component. During low flow seasons, Ninemile Creek flow amounts to about 4  $m^3$ /sec. With a value of 0.267 for  $a_{22}$ , this corresponds to an HDR of 0.63 year<sup>-1</sup>, or -0.2 on a log scale, about equivalent to the non-advective mixing component. Nowever, this does not contribute to the total HDR, because it merely satisfies some of the demand imposed by cooling water withdrawal. Thus, during peak stratification, cooling water use accounts for about 80% of the total hypolimnic displacement, if it is withdrawn exclusively from the hypolimnion. When the epilimnion intake is opened, it is estimated that 78% of the water entering the plant comes from the surface waters. Under thess conditions, at peak stratification, the total displacement rate is reduced by about 60%.

The simulations indicate that mixing during spring overturn periods is not as complete or as rapid as in the fall. This is chiefly due to the chloride gradients which are built up under ice cover during the

winter. An increasing trend in the intensity of the spring mixing period is evident. Computed total dilution rates are higher by about an order of magnitude during the spring of 1974, as compared with the springs of 1969 and 1970. This correlates well with corresponding increases in hypolimnic dissolved oxygen levels noted in Chapter 3 and shown again in Figure 4.6-5. Comparisons of patterns in mixing rate with periods of hypolimnic anaerobiosis indicates that the latter correspond roughly to periods in which the logarithm of the total dilution rate is less than about 0.8. The lengths of these periods from spring overturn to fall overturn, estimated from Figure 4.6-4, range from about 0.3 years in 1974 to about 0.5 years in 1970. These factors suggest that the observed enhancement in hypolimnic dissolved oxygen levels in later years of the survey may have been due to climatologic influences on lake mixing, as simulated by the model developed here. More specific evidence for this could be developed by increasing the complexity of the model to permit direct simulation of dissolved oxygen. This will be discussed further in Section 4.7.

## 4.6.2 Case Simulations

( )

Ĺ

A total of thirteen cases have been formulated in order to permit further analysis of the impacts of coding water use, salt discharge, and Metro STP outfall design on general aspects of lake mixing. Case conditions are summarized in Table 4.6-1. The first three examine the effect of cooling water withdrawal under present conditions of salt discharge to Ninemile Creek. Cooling water is withdrawn exclusively from the hypolimnion in Case 1, primarily from the epilimnion in Case 2, and is totally absent in Case 3. The next three cases examine the same cooling water conditions in the absence of the salt discharge. Case 6 can be considered as a baseline condition or undisturbed state. The last seven cases examine the effect of increasing initial dilution ratio for discharge of the combined municipal/industrial effluent into the epilimnion. Case 7, with a dilution ratio of 0., is essentially equivalent to discharge of the effluent into the hypolimnion, due to density current effects. Initial dilution ratios increase by factors of two from a value of 1 for Case 8 to a value of 16 for Case 12. In Case 13, the discharge has been prevented from sinking into the hypolimnion by setting a 21 equal to zero. This is essentially equivalent to an infinite initial dilution ratio. All of the last seven cases have assumed exclusive cooling water withdrawal from the hypolimnion, which has been Allied Chemical's operating policy since mid-1974<sup>.7</sup>

One of the factors which introduces uncertainty into the analysis of the impact of the initial dilution ratio is uncertainty in parameter

|                | C              | OOLING WA       | TER               |                |                | SALT                           |                                           | COMMENTS                   |         |               |  |  |
|----------------|----------------|-----------------|-------------------|----------------|----------------|--------------------------------|-------------------------------------------|----------------------------|---------|---------------|--|--|
| Case           | Flow           | Withdra<br>Epil | Wl Level<br>Hypo. | Load           | Stream         | Initial<br>Dilution<br>n Ratio | Max. Fraction<br>Sinking<br>Into Hypolim. | Cooling Water<br>Withdrawl | Salt    |               |  |  |
| 1              | F <sub>4</sub> | 0.00            | 1.00              | U <sub>6</sub> | Ω,             | $a_3 = 0.$                     | $a_{22} = 0.266$                          | Hypolimnion                | Present | Conditions    |  |  |
| 2              | F              | 0.78            | 0.22              | "              | n <sup>-</sup> | - 11                           | "                                         | Epilimnion                 | .*      | **            |  |  |
| 3              | о.             | -               | -                 |                |                | <b>H</b>                       | • ;                                       | Absent                     |         | 91            |  |  |
| 1              |                |                 |                   |                |                |                                |                                           |                            |         |               |  |  |
| 4              | ₽ <sub>4</sub> | 0.00            | 1.00              | 0.             | -              | n                              | $a_{22} = 0.0$                            | Hypolimnion                | A       | bsent         |  |  |
| 5              | Fa             | 0.78            | 0.22              | •              | -              | n                              | n t                                       | Epilimnion                 |         | m             |  |  |
| 6              | o.             | -               | -                 |                | -              | er.                            | •                                         | Absent                     |         |               |  |  |
|                |                |                 |                   |                |                |                                | 1                                         |                            | Notres  | Toitial Dil   |  |  |
| 7 <sup>a</sup> | FA             | 0.00            | 1.00              | U              | ۵ <sub>5</sub> | a <sub>2</sub> = 0.            | $a_{21} = 1.0$                            | Rypolimnion                | STP ;   | d = 0.        |  |  |
| 8              | "              | -               | **                | "              | "              | <b>1</b> .                     | 21                                        | n (                        | N       | đ = 1         |  |  |
| 9              |                | ••              | **                | -              |                | 2.                             | n                                         | "                          | n       | d = 2         |  |  |
| 10             |                | u               | n                 | n              | **             | 4.                             | n                                         | •                          |         | d = 4         |  |  |
| 111            |                | ņ               | n                 |                | **             | 8.                             | 17                                        |                            |         | d = 8         |  |  |
| 12             | - 11           | 14              |                   | n              |                | 16.                            | n                                         | ør                         | 11      | <b>d</b> = 16 |  |  |
| 13             | 1 "            | п               | 10                |                | **             | -                              | $a_{21} = 0.0^{b}$                        | -                          |         | d = ∞         |  |  |

#### Table 4.6-1

Definitions of Cases Studied

a - see Table 4.3-1,-2,-3, and -4 for definitions of Q, U, F, and a variables, respectively.

b - setting a<sub>21</sub> equal to zero forces all of the influent stream to remain in the epilimnion and is equivalent to an infinite initial dilution ratio, a<sub>2</sub>.

c - Fraction of cooling water withdrawn from epilimnion  $= U_4$ d - Essentially equivalent to discharging into the hypolimnion, due to density current effects.

 $a_{21}$ , the maximum fraction of the diluted effluent which is allowed to sink into the hypolimnion, subject to density constraints (see  $P_2$ , Table 4.5-3). A value of 1.0 has been assumed for this parameter in Cases 7-12. The analogous coefficient for Ninemile Creek,  $a_{22}$ , has been estimated empirically at 0.267. It would be difficult to predict these parameters a priori without more detailed modelling of the hydrodynamics of the outfall site, including specifics of wind velocities (speeds and directions), horizontal currents, and bottom topography. However, Cases 7 and 13 represent extreme conditions, which are insensitive to the value of  $a_{21}$ . Case 7 essentially represents disposal in the hypolimnion, while Case 13 represents infinite initial dilution in the epilimnion. Thus, while uncertainty in  $a_{21}$  may introduce uncertainty as to the specific path from Case 7 to Case 13 as dilution ratios increase, the end points are not influenced.

Simulation of each of the cases has been done under the hydrologic and meteorologic conditions from October, 1967 through December, 1974. Metro STP flow  $(U_2)$  has been assumed constant at the design value of 3.78 m<sup>3</sup>/sec. Results have been recorded at 0.05-year increments. The first 15 months of each simulation have not been analyzed because of sensitivity to assumed initial conditions. Thus, the cases have been compared based upon model simulations over the six-year period from 1969 through 1974. In Cases 1-3, sensitivity coefficients to each of the four empirically-estimated parameters  $(a_{14}, a_{15}, a_5, and a_{22})$ have also been calculated. In Cases 4-13,  $a_{22}$  is of no consequence 4--112

(

and has not been included in the sensitivity calculations. The primary incentive for computing sensitivity coefficients is to permit estimation of the effects of uncertainty in the parameter estimates on errors in model projections.

Four primary "objective functions" have been used to compare lake responses to the conditions specified by the various cases: (1) mean density gradient; (2) maximum annual density gradient; (3) mean hypolimnic dilution rate (HDR); (4) minimum annual HDR. A final supplementary criterion is average turnover frequency, defined as the average number of periods per year of simulation in which the Richardson Number,  $F_{12}$ , drops below 100. The Richardson Number is a measure of the resistance to vertical mixing. The vertical exchange rate,  $F_{11}$ , is roughly inverse to its value, which ranges from zero during unstable periods to about 5 x 10<sup>4</sup> during peak density stratification. A value of 100 has been arbitrarily employed as a turnover definition. This corresponds to an HDR value of 123 year<sup>-1</sup>, or a complete exchange of hypolimnic waters in about 3 days.

For the first and third objective functions specified above, a total of three factors has been assumed to contribute to variations within each case: season, year, and parameter values. Seasonal variations in the functions are attributed to annual meteorologic and hydrologic cycles. Yearly variations represent the effects of variations in average annual meteorologic and hydrologic conditions. To provide a basis for assessing the severity of the effects of parameter uncertainty on model projections, the seasonal and yearly variations have been compared with those possibly attributed to parameter variations. To determine the latter for each case, estimates of objective function values at the end points of the parametric principle component axes (Table 4.5-3) have been derived from the value at the center of the parameter confidence region, the sensitivity coefficient matrix, and the vector of parameter values at the end points of each principal component axis. For a given state variable  $y_i$  at time t :

$$y_{j,t,k} = y_{j,t,0} + \sum_{i=1}^{4} \left(\frac{\partial y_{j,t}}{\partial \theta_{i}}\right) \left(\theta_{1,k} - \theta_{1,0}\right) \qquad (4.6-4)$$

where,

6

( )

 $y_{j,t,k} = value of state variable j at time t and$ parameter vector k $<math display="block">y_{j,t,o} = value of state variable j at time t and$  $optimal parameter vector <math>\frac{\theta_{1,o}}{\theta_{1,o}}$  $\frac{\partial y_{j,t}}{\partial \theta_{i}} = sensitivity coefficient of state variable y_{j}$  $with respect to parameter <math>\theta_{i}$  at time t  $y_{j}$  $\theta_{i,k} = value of parameter \theta_{i} at principal component$ axis end point k $\theta_{i,o} = optimal value of \theta$ 

(

This scheme has been employed to develop a three-dimensional array of values for the density gradient and  $\log_{10}$  (HDR) for each case. Each array consists of 20 seasons, 6 years, and 9 parameter sets. The ranges of the means within each of the three factor groups have been used as bases to assess the relative effects of the factors on mean density gradient and on  $\log_{10}$  (HDR). In analysis of yearly extreme values (maximum density gradient and minimum HDR), the extremes for each year and case have been extracted from the general, three-dimensional array to generate a two-dimensional matrix (6 years x 9 parameter sets) for each case.

Results are presented in Table 4.6-2 and in Figures 4.6-6 to 4.6-9. Simulations of Cases 6, 7, and 13 are presented in Figures 4.6-11to 4.6-13, respectively. These can be compared with the simulation of actual 1968-74 conditions in Figure 4.6-10.

Comparing the results for Cases 1, 2, and 3 with Cases 4, 5, and 6 provides a basis for evaluating the sffects of the salt discharge to Ninemile Creek. Average turnover frequencies for Cases 1-3 range from 0.50 to 1.00, while 2 turnovers per year are consistently observed in the absence of the salt discharge. The primary effect of the discharge is to inhibit the spring turnover period. The difference in average density gradient between the two groups of cases is 0.68 mg/cm<sup>3</sup>, corresponding to 62% of the Case 1-3 msan. Thus, if the salt discharge were eliminated, it is indicated that average density gradients in the

## Table 4.6-2

## Results of Case Simulations

| COLE® |  |
|-------|--|
|-------|--|

| CORPORAT               |                      | 1       | 2       | 3     | 4    | 5_         | 6     | 7    | 8            | 9     | 10      | 11      | 12          | 13    |
|------------------------|----------------------|---------|---------|-------|------|------------|-------|------|--------------|-------|---------|---------|-------------|-------|
| Turnover Frequ         | ency <sup>b</sup>    | 1.00    | 0,63    | 0.50  | 2.00 | 2,00       | 2.00  | 0.00 | 9.00         | 0.50  | 0,83    | 1.00    | 1.83        | 1.83  |
| Density Gradie         | at <sup>C</sup> Bean | 0.96    | 1.17    | 1.21  | 0.35 | 0.49       | 0.44  | 5.15 | 2.71         | 1.64  | 1.13    | 0.64    | 0.41        | 0.40  |
| Seaaco                 | Hinima               | 0.00    | 0.07    | 0.15  | 0.00 | 0.00       | 0.00  | 3.25 | 0,90         | 0.23  | 0.01    | 0.00    | 0.00        | 0.00  |
|                        | Hereiman.            | 2.55    | 2.68    | 2.81  | 1.49 | 11.69      | 1.72  | 7.18 | 4.31         | 3.10  | 2.02    | 1.24    | 0.91        | 1.42  |
|                        | Iance                | 2.55    | 2.01    | 2.67  | 1.52 | 1.69       | 1.72  | 3.93 | 3.41         | 2.87  | 2.01    | 1.24    | 0.91        | 1.42  |
| Tenr                   | Hinimu               | 0.71    | 0.86    | 0.66  | 0.25 | 0.34       | 0.30  | 4.89 | 2.52         | 1.70  | 1.03    | 0.57    | 0.30        | 0.27  |
|                        | Jac in m             | 1.36    | 1.61    | 1.58  | 0.44 | 0.62       | 0.56  | 5.47 | 3.03         | 2.10  | 1.31    | 0.75    | 0.72        | 0.90  |
|                        | Bance                | 0.65    | 0.77    | 0.73  | 0.19 | 0.28       | 0.26  | 0.57 | 0.50         | 0.40  | 0.29    | 0.18    | 0.42        | 0.68  |
| Derassian <sup>®</sup> | Minimum              | 0.90    | 1.08    | 1.05  | 0.32 | 0.45       | 0.41  | 4.94 | 2.64         | 1.80  | 1.10    | 0.62    | 0.36        | 0.37  |
|                        | Tariana.             | 1.01    | 1.27    | 1.37  | 0.37 | 0.53       | 6.47  | 5.32 | 2.78         | 1.88  | 1.14    | 0.64    | 0.45        | 0.44  |
| 1                      | -                    | 0.13    | 0.19    | 0.31  | 0.05 | 0.08       | 0.07  | 0.34 | 0.14         | 0.09  | 0.04    | 0.02    | 0.09        | 0.07  |
| Annual Series          |                      |         |         |       |      |            |       |      |              |       |         |         |             |       |
| Density Gradi          | mt Rean              | 2.57    | 2.92    | 2.87  | 1.51 | 1.92       | 1.74  | 7.22 | 4.37         | 3.16  | 2.04    | 1.24    | 1.02        | 2.47  |
| Tear                   | Hinima               | 2.00    | 2.37    | 1.32  | 1.10 | 1.44       | 1.30  | 6.71 | 4.07         | 2.94  | 2.85    | 1.08    | 0.61        | 1.04  |
|                        | Having               | 13.52   | 3 81    | 3.67  | 1 89 | 2 34       | 2.16  | 8 17 | 5.11         | 3 70  | 2 32    | 1 17    | 2.16        | 2 82  |
|                        | Panna                | 1.6     | 1 44    | 1 30  | 0 77 | 0.90       | 0.86  | 1.46 | 1 04         | 0 76  | 0 48    | 0.39    | 1 56        | 1 78  |
|                        | Misimu               | - 46    | 2 00    | 2 74  | 1 42 | 1 20       | 1 45  | 7 07 | 4 35         | 3 /10 | 1 00    | 1 22    | A 00        | 1 76  |
| and managements        |                      | 1.1     | 2.00    | 2 66  | 1.46 | 3.04       | 1 00  | 7 47 | 4.43         | 3,00  | 1.77    | 3 36    | 3.66        | 1 60  |
|                        |                      | 1.00    | A 76    | 0.76  | 2.00 | 0.77       | 1.07  | 0.43 | A 73         | 0.17  | 2.00    | 0.05    | 0.00        | A 33  |
|                        |                      | 10.24   | 0.40    | 0.20  | 0.10 | <u></u>    | V.20  | 0.44 | <u>v.</u> 43 | 0.17  | 0.10    | 0.03    | 0.00        | 4.44  |
| In the second          | d The second         | 1       | 0.79    | 0.70  | 1 70 | 7 44       | 3 45  | 0.40 | A 80         | 1 01  | 7 10    | 1 50    | 1 73        | 3 40  |
|                        |                      | 10.44   | A 04    | 0.04  | A 66 | -0.11      | -0 07 | A 41 | 0.71         | 0.89  | 1 10    | 1.34    | 1.24        | 0 56  |
|                        |                      | 10.22   | 3 21    | 1 01  | 3 67 | - 44 A     | 3 67  | 0 61 | 1 01         | 1 04  | 2 43    | 2 41    | 2 55        | 2 61  |
|                        | Paner                | 12.16   | 2 16    | 1 96  | 2.07 | 2.78       | 2.74  | 0.20 | 0 33         | 0.96  | 1 22    | 1 07    | 1.32        | 2.07  |
| Tear                   | Minimum              | 0.86    | 0.53    | 0.49  | 1.54 | 1.24       | 1.24  | 0.46 | 0.77         | 0.95  | 1.23    | 1.51    | 1.41        | 7.74  |
|                        | Marri ana            | 1.19    | 1.01    | 0.97  | 1.89 | 1.67       | 1.67  | 0.51 | 0.85         | 1.12  | 1.41    | 7 66    | 1 44        | 1 73  |
|                        | Danne                | 6.55    | 0 49    | 0.49  | 0 33 | 0 41       | 0 44  | 0.05 | 0.00         | 0.16  | 0 70    | 0.15    | 0.44        | 0.59  |
|                        | wishing a            | 10.33   |         | 0.40  | 3 64 | 1 79       | 1 17  | 0.03 | 0.00         | 1.03  | 3 20    | 1.0     | 1,00        | 1,39  |
| - ALLERCEL             |                      | 1.3     | 0.70    | 0.01  | 1.00 | 1.30       | 1.34  | 0.47 | 0.15         | 1.02  | 1.40    | 1.3/    | 1.00        | 1.40  |
|                        |                      | 1.10    | 0.00    | 0.19  | 1.11 | 1.47       | 1.37  | 0.30 | 0.82         | 1.04  | 1,34    | 1.33    | 1.42        | 1.30  |
|                        |                      | 0.14    | 0,18    | 0.13  | 6.13 | <u>0-Ц</u> | 9,25  | 0.03 | 0.03         | 0,03  | 0.03    | 0.02    | 0.19        | _0.18 |
|                        | -                    | 1       | 0.04    | ~ ~~  |      | -0.17      |       | ~ ~  |              |       |         |         |             |       |
| 10 <sup>-1</sup>       |                      | 10.33   | -0.04   | 0.00  | 0,22 | -0.70      | -0.00 | 0,40 | 0.70         | 0.87  |         |         | 1.31        | 0.52  |
| TOOL                   |                      | 0.20    | -0.02   | -0.03 | 0.40 | -0,32      | -0.28 | 0,39 | 0.03         | 0.00  | 1.00    | 1.33    | 0.48        | 0.41  |
|                        | Dance                | 0.14    | 7 15    | 0.11  | 0.00 | 0.09       | 0.13  | 0.41 | 0,11         | 0.85  | 1.00    | 1.34    | 7.33        | 0.30  |
|                        |                      | <u></u> | <u></u> | 2.0   | 0.12 | 0.40       | -0.37 | 0.02 | 0.02         | -0.02 | <u></u> | <u></u> | <del></del> |       |
|                        |                      | 6.31    | 0.00    | -0.04 | 0.30 | -0.24      | -0.17 | 0.40 | 0.70         | 0,87  | 1.09    | 2223    | 1.11        | 0,49  |
| 1                      |                      | 10.41   | 0.08    | 0,00  | 0.36 | -0.04      | 0,00  | 0,41 | 0,70         | V,87  | 1.09    | 1.33    | 1.31        | 0.54  |
|                        | <b>Mange</b>         | 10.04   | 0.07    | 0.09  | Q.05 | 0.17       | 0.17  | 0.01 | 0.01         | 0.01  | 0.00    | 0.00    | _0.71       | 0.65  |

- a for two definitions, see Table 4.6-1. b Turnover Proposity defined as everys number of times per year Richardson Master drops halow 100. c Density gradient = hypolismion density epilismion density ( $a_2/cm^3$ ). d Hypolismic Dilution Rate (NUR) = total flow through hypolismion / hypolismics volume ( $year^{-1}$ ). e Emges of within-group maste for anaposal, yearly, and parameter factors.

Simulated Mean Density Gradients for Various Cases

C

**C**<sup>1</sup>

Ĉ



Figure 4.6-7

Simulated Maximum Annual Density Gradients for Various Cases





Simulated Hean Hypolimnic Dilution Rates for Various Cases

**(**)

 $\mathbf{C}$ 

(



# Figure 4.6-9

Simulated Minimum Annual Hypolianic Dilution Rates for Various Cases





4-119

Figure 4.6-10 Simulations of Actual 1968-74 Conditions



# Figure 4.6-11 Simulations of Case 6 Conditions

1 J.I

4-120

۰.....





4-122

Figure 4.6-13 Simulations of Case 13 Conditions

## 4.6.2 Case Simulations (continued)

C

Ć

lake would decrease by 62%. The corresponding reduction for maximum annual density gradient is estimated to be 38%. The difference in the logarithm of the mean HDR between the two groups of cases is about 0.7, indicating a ratio of five in average displacement rates. The minimum HDR values are governed chiefly by advective flows due to cooling water withdrawal and are not as severely impacted by the salt discharge.

Comparing Cases 1, 2, and 3 provides as basis for evaluating the effects of cooling water use under current salt disposal conditions. Results indicate that the lake mixing afforded by withdrawal of cooling water from the hypolimnion and subsequent discharge to the epilimnion does have a calculable impact upon density gradiente and total HDR values. Results for Cases 1 and 3 indicate that if the cooling water use were stopped, turnover frequencies would decrease by 50%, mean and maximum annual density gradients would decrease by 26% and 12%, respectively, and mean and annual minimum HDR's would decrease by factors of 2 and 2.5, respectively. The differences between Cases 1 and 2 suggest that observed increases in lake mixing rates over the 1968 to 1974 period may have been due in part to more frequent withdrawal of cooling water from the hypolimnion in later years (see  $U_A$ , Table 4.3-2). Corresponding effects of cooling water use without the salt discharge (Cases 4-6) are somewhat less marked, because of the increase in density-dependent mixing, which reduces the relative importance of the advective mixing afforded by cooling water use. Density gradients for Case 5, in which cooling water is withdrawn primarily from the

## 4.6.2 <u>Case Simulations</u> (continued)

epilimnion, are somewhat higher than for Case 6, without cooling water. This is an effect of heat disposal in the epilimnion increasing the density stratification. The average heat disposal rate amounts to about 19% of the average annual solar radiation incident on the lake.

Simulations of Case 7 indicate that discharge of the combined municipal/industrial effluent into the epilimnion without initial dilution, or, equivalently, dischargs into the hypolimnion, would result in permanent stratification under the hydrologic and meteorologic conditions of 1968-74 (Figure 4.6-10). At the other extreme, results for Case 13 (Figure 4.6-13 suggest that infinite initial dilution in the epilimnion would result in an average of 1.63 turnovers per year, and in density gradient and HDR values not much different from the "undisturbed state", Case 6. As noted above, the path between these two cases as a function of dilution ratio is uncertain because of uncertainty in parameter  $a_{21}$ . Results obtained for  $a_{21}$  equal to 1.0 are conservative in that they would tend to over-predict density currents and resultant effects on chloride and density gradients in the lake. The consequences of these results on outfall design will be discussed in Section 4.7.

For all cases and objective functions, estimated variations due to parameter values are small compared with those attributed to season and year and compared with differences in means among the various cases. The parameter variations have been derived from the 95%

()

( Th

()

C

(

confidence regions of the parameter estimates. As noted in Section 4.5, the estimate of the size of this region is inexact because of nonlinearity in the model and the serial dependence and non-normality of the residuals. Observed yearly ranges are on the order of four times those attributed to the parameters. The former have been derived from a sample of six years, and thus do not represent 95% confidence ranges. If the distribution of yearly means is assumed to be normal, the standard deviation of the yearly variations would be given by 0.395 times the range for a sample size of  $6^{27}$ , and the 95% confidence range would be about 15.8 times the observed range for six years. Thus, in order for the parameter variations to equal yearly ones, the estimate of the size of the parameter confidence region (expressed as distance along the principal component axee) would have to be low by a factor of 1.58 x 4 = 6.32.

Bard<sup>3.</sup> has derived an expression for the  $\varepsilon$ -indifference region of the maximum-likelihood objective function for the situation in which the sampling distribution (multivariate distribution of the parameters) cannot be assumed to be normal:

 $\Delta \Phi = \varepsilon = \frac{1}{2} \frac{2}{1-\gamma}$ 

(4.6-5)

where,

C

 $\ell$  = number of estimated parameters = 4  $\gamma$  = confidence level = 0.95

Paralleling the development in Section 4.5 for the normal case, the corresponding distance along the principal component axes, expressed in multiples of the respective principal component standard deviations is given by:

$$z = \sqrt{2 \Delta \Phi} = \sqrt{\frac{2}{1 - \gamma}} = 8.94$$
 (4.6-6)

For the normal case,  $\Delta\Phi$  is given by  $\chi^2/2$ , and z is given by  $\sqrt{\chi}$ , or 3.08. Thus, if an attempt is made to account for non-normality, the dimensions of the confidence region increase by a factor of 8.94/3.08 = 2.90. The same proportionate increase would be expected in the ranges of the density gradient and HDR's attributed to parameter variations, according to equation (4.6-4) Under these conditions, the 95% confidence range attributed to parameter ranges. Generally, the higher this ratio, the more difficult it would be to detect projection errors due to errors in the parameter estimates. The non-linearity of the model and serial dependence of the residuals have not been accounted for, however. This is about as far as the analysis can be taken without resorting to Monte-Carlo type methods to determine actual parameter confidence regions.

An analysis has been done to estimate some of the various error components of model projections in each case studied. Assuming computation error to be negligible, total projection error can be broken down into three basic components<sup>3</sup>:

$$s_{T}^{2}(t) = s_{I}^{2}(t) + s_{p}^{2}(t) + s_{R}^{2}$$
 (4.6-7)

where,

 $\mathbf{C}$ 

(

$$S_T^2$$
 (t) = variance of a projection of a given  
variable at time t  
 $S_I^2$  (t) = variance attributed to independent  
variable error  
 $S_P^2$  (t) = variance attributed to parameter error  
 $S_R^2$  = residual variance

In the case of a dynamic model, all of the above are time-variable, with the exception of  $S_R^2$ . The first term is given approximately by:

$$s_{I}^{2}(t) \approx \sum_{\substack{i=1 \ i=1}}^{n_{X}} \frac{n_{X}}{\partial x_{i}} \left( \frac{\partial y(t)}{\partial x_{i}} \right) \left( \frac{\partial y(t)}{\partial x_{i}} \right) Cov(x_{i}, x_{j})$$
(4.6-8)

4.6.3 Error Analysis (continued)

where,

O

0

()

y (t) = value of predicted variable y at time t

- n, = number of independent variables
- $\mathbf{x}_i$  = value of independent or forcing variable i

Similarly, the parameter error term is given by:

$$s_{p}^{2}(t) \simeq \sum_{i=1}^{n_{p}} (\frac{\partial y(t)}{\partial P_{i}}) (\frac{\partial y(t)}{\partial P_{i}}) Cov (p_{i}, p_{j}) \qquad (4.6-9)$$

where,

$$p_i = value of parameter i$$
  
 $n_p = total number of parameters$ 

The last term is attributed to residual error and is determined by the standard error of estimate of the predicted variable. This component can be further dissected into two components :

$$s_R^2 = s_E^2 + s_M^2 = \overline{SEE}_y^2$$
 (4.6-10)

## 4.6.3 Error Analysis (continued)

where,

( )

= model error = measurement error square of standard error of estimate of predicted variable

In order to estimate the two components of the residual variance, some independent evidence is generally required. For example, if replicate samples, chloride analyses, and temperature measurements were available, the measurement error components could be estimated and subtracted from the observed residual variance to obtain estimates of the model error components. In the absence of such data, the aggregated form of the residual variance has been employed below. As noted in Section 4.5, residual standard errors are not strictly independent of time or season, so the assumption that they are constant is not completely valid, but it sufficient for the analysis below.

As discussed in Chapter 1, a method of assessing the adequacy of data for parameter estimation purposes is to compare the error components attributed to parameter uncertainty,  $s_p^2$  (t) with the residual variance,  $s_R^2$ . The former can usually be reduced by increasing the size of the data sample used for parameter estimation, while the latter would not necessarily change under such conditions, provided that the additional data were of the same quality as that

## originally employed.

(

C

The magnitudes of the parameter and residual error components have been compared for each case simulated and for predictions of the epilimnic, hypolimnic, and gradient values of temperature, chloride, and The parameter error term is time-variable and its distridensity. bution for the entire 6-year simulation has been summarized by the extreme values and the mean. The variance components for each case and variable are summarized in Table 4.6-3. Generally, mean parameter variance amounts to a small fraction of the sum of the two error components. The maximum fraction observed is 0.18, for prediction of temperature gradient in Case 3. Case 3 appears to be the most sensitive to parameter values of the cases studied. For prediction of density gradient the fraction of total variance attributed to parameters ranges from 0.0004 in Case 11 to 0.034 in Case 7. Despite the approximate nature of the estimate of the covariance matrix of parameter values, it appears that the estimate would have to be considerably in error in order for the parameter variance to become a substantial component of the total projection error.

A more complete error analysis would also incorporate the effects of independent variable error. This would involve considerable effort, due to the number of independent variables and the difficulties of estimating their covariance matrix. To some extent, this type of error has already been incorporated into the parameter and residual error terms, since the parameter and residual covariance matrices have

# Table 4.6-3

. . . . .

C

C

**(**)

Parametric and Residual Variance Components for Each Case

## and Predicted Variable

|            | -             |               | - Papameter    |           |                                       |                                       |                  |               |                         | Parene        | ter         |                    |                      |                   |
|------------|---------------|---------------|----------------|-----------|---------------------------------------|---------------------------------------|------------------|---------------|-------------------------|---------------|-------------|--------------------|----------------------|-------------------|
|            |               | _ حلاظ        | Hex.           | ALC: N    | Pesidual                              | Totel                                 | PRAN PATH        | <u>e</u>      | He,                     |               | ' Pirga     | Besiduel           | Total                | Neen Param        |
| Case       |               | VM-           | Var.           | Var.      | Variance                              | Vet Janco                             | Total            | <b>_</b> CL94 | Var,                    | , Vat,        | VAC.        | _ Yaslance         | Variance             | Total             |
|            |               | _             |                |           | £                                     |                                       |                  | -             |                         |               | v           |                    |                      |                   |
|            |               |               |                |           | 1                                     |                                       | A                |               |                         |               |             |                    |                      |                   |
| - F        | 644           |               | 6.0.26         | 0-012+    | 2.1400                                | 2+6139                                | 0-0030           |               | 1 0.0301                | 0.0314        | 943434      | 34.9013            | 34,9260              | 3.0362            |
| - 2        | - Yan         |               | 2.33(3         | 4-9146    | 2.1144                                | 2.0340                                | 0.0101           |               | 1                       |               |             | - 14-2011          | 40.0374              |                   |
|            | 0.0           |               | <u>ettin</u> t | 2+9127    | 2+1444                                | 2.0003                                | 0.0030           |               | 3 0+3143                | 23-0613       | 4.3634      | 34.2010            | 41,9290              | 0.0556            |
| - 4        | a+a           | 1             | 0.0432         | 3.9131    | 5*1455                                | 5-0121                                | B-332P           |               | 4 0.0343                | 9.0231        | 0.2333      | 34.0314            | 34.9315              | 2.3799            |
| - 5        | - Q.O         | ·             | _ J.J911_      | 0.364     | 2.1400                                | L. odan                               | _d,d326          |               | 2 0.000                 | 4.9113        | 1.3031      | 24.0913            | 7414371              | 0.0000            |
| - 6        |               |               | . 9.9379.      |           | 2.7900                                | _5*A31A                               | 2020.0           | · •           | e e-1999                | c.0010        | بدر ورم و   | 34*0913            | 14.6414              | 1.0001            |
| • •        | 3.3           | •             | Ç.J33+         | 2.31+9    | 2.7933                                | 2.0146                                | 0.3353           | 1             | 7 1.5530                | 4+7234        | 3.6274      | 34.5010            | 43.2243              | 4.3.14            |
| - i        | Č.U           |               | 0.0.11         | 0.0150    | 2.7933                                | 2.6350                                | 4-4422           |               | <b>0</b> .1679          | 1.4739        | 4.0297      | 19.5013            | 41.2217              | 9.3154            |
| - i        | C. 0          |               | 3.11+2+        | 4.0133    | 2. 1533                               | 2. 4353                               | 0.0055           |               | دېږ. د و                | C. ( B. C     | 3.2151      | 39                 | 14.3141              | * a. 635 a        |
| 10         | 3.3           | •             | 3.35.34        | 4.31.41   | 2.7930                                | 2.attal                               | 0.3054           | te            | e d.J.ev                | PL04.1        | 4.4274      | 19.6363            | 44.07.4              | 0.0111            |
| - 11       |               |               | 4.45.1         | 3, 31 51  | 2.7935                                | 4.6161                                | 4.4347           |               | 0.01.14                 | 0.14.54       | 0.03.0      | 34                 | 10 4130              | 0.0023            |
|            |               |               |                |           | 78.33                                 | · · · · · · · · · · · · · · · · · · · | 10.3161          |               | 3.30.04                 | 1.0473        | 1 0161      | 36 6010            | 34.04.14             | 6+010r            |
|            |               |               | 1 44 16        | 0.0110    | 2.1703                                | 1 4046                                | 0.000            |               | 6.6.00                  | 0,0012        |             | 93.6010            | 34-0161              | 0.0004            |
| 4.9        | 4.0           |               | MAAA 13 '      | . 013133  | . 5.1300                              | E. 0037                               | 010033           |               | 1                       | u+44 f /      | 4.4110      | \$3100fû           | 3446439              | 0*0311            |
|            |               |               |                |           | e                                     |                                       |                  | -             | +                       |               |             | °                  |                      |                   |
|            |               |               |                |           |                                       |                                       |                  |               |                         |               |             | <b>-</b>           |                      |                   |
| - 1        | - <b>4</b> .4 | <b>324</b>    | 8.1136         | J.0371    | 3,5003                                | 3.5371                                | 0.0103           | 1             | 0.0149                  | 2.5493        | 0.6151      | 19273003           | 185.6141             | 0.0131            |
| - 2        | 0.0           | J44           | . 640550_      |           | - 3.5030                              |                                       | 0,0355           | - 2           | 1 0.0444                | 3.2148        | 0.8453      | 182.3077           | 185.8753             | 3-33+4            |
| ġ          | 0.0           | 075-          | 11.5141        | B         | 1.5015                                |                                       | 0.1920           |               | 1 0+25/9                | 2,3541        | 0.5273      | 183-2000           | 105.0390             | 1.3317            |
| - A        | 0.0           | ouo —         | 9.9344         | J.3100    | 3.5030                                | 3.5160                                | 0.0043           | - 4           | 0.0333                  | 2.3011        | 0.4475      | 192-2003           | 145.4475             | 3. 3325           |
| ÷.         | 4.3           | 040           | 0.0017         | 3.5226    | 3.5000                                | 3.5/28                                | 0.0365           |               | 0.0301                  | 2.52JB        | 4.5240      | 185.0000           | 181.4416             | 3.1124            |
| - 2        | 3.3           |               | 1.3021         | 3.32.34   | 3-2303                                | 3.5634                                | 3.3338           | Ē             | 6.3331                  | 1             | 3.5743      | 144.3033           | Idea Deal            |                   |
|            | 3.0           | 333           | 0.04/5         | 0.3137    | 3.50.22                               | 4.5107                                | 0.0011           |               | 0.3145                  | 2. 1441       | 11. 44.14   | 145.0045           | 145.44.45            | 0.0011            |
| - 1        | 4.6           |               | 0.01 33        | 0.0224    | 1.5000                                | 1.5120                                | 0.0014           | <b>-</b>      | 0.0211                  | 2.10.1        | 3.4374      | 145.1010           | 100 6143             | 0.0110            |
| - <u>-</u> |               |               |                |           |                                       |                                       | 0.0114           |               | 0.3143                  | 4.474         | 4 2018      |                    | 16247410             | 0+0335            |
|            | لوجون ا       |               | 044993         | 6.415     | 2.2033                                | 3+3133                                | 0+0430<br>A 0016 |               | 0.0113                  | 2.1131        | A*2013      | 100-1110           | 442+2413             | 0.0321            |
| 10         | 4.0           | 933, <u>-</u> | 64 35 14       |           | 3.2033                                | 3+2163 -                              | 0,0032           | 10            | 0.0014                  | 2+1/90        | 0+2102      | 1 22. 3003         | 145.5105             | 3,3329            |
| 11         | 0.0           | 330           | 0.3459         | 0.2121    | 3. 9.000                              | 1.2121                                | 9.0014           | 11            | 0.0379                  |               | 045208      | _11>++++           | 185.5288             |                   |
| 13         | 9.0           | 300           | 3.3442         | 4.3192    | 3.50JU                                | 3. 21 65                              | U. CU 4 F        | - 12          | 0.0022                  | 3.0344        | 9.6144      | 112, 2022          | 152.6744             | 0.0330            |
| 13         | . 6. 6        | JJ2 -         | 0,1745         | J.6353    | 3*2000                                | 3.5363                                | 0.0193           | - 13          | C+0335                  | 1.9620        | J.4912      | 145.0000           | L45.4911             | 0.3326            |
|            |               |               |                | -         |                                       | •                                     |                  |               |                         |               | _           |                    |                      |                   |
| _          |               |               |                | T         | · · · ·                               |                                       |                  |               | *                       |               | P           |                    |                      |                   |
| - 4        | 0.0           | 000           | 0.1409         | 0+0309    | 3,9600                                | 3,9969                                | 0.0092           | 1             | ¢+1016                  | 3,1291        | ີ່ໄ+2546    | 130.0000           | 131.2548             | 3.3395            |
| ÷          | 0.0.          | 512           | 3.2256         | 3.1537    | 3. 4000                               | 4.1137                                | 0.0124           | - 2           | 0.2057                  | 29.7515       | 3.7555      | 133. 3383          | 133. 2555            | 0.0268            |
|            | 0.0           | 435           | 11.59.96       | فيطقل     | 3.9630                                | 4.8265                                | 0.1795           | - ī           | 0- 564 8                | \$6.36.84     | 7.0415      | 150.3503           | 117.0414             |                   |
| - 1        | 0.0           | ini '         | 0.0491         | 3.0124    | 1.9600                                | 3.9724                                | 0.0332           | - 7           | 0.0303                  | 0.4544        | 4.26.72     | 1                  | 1 4 3 26 73          | 240217            |
| - 7        | 1 1           | 11.0          | 4.12.51        | .1. 13.46 | 3 84 73                               | 1.0448                                | 3.3374           | - 2           | 0.4301                  | 1.0414        | 0.211       | 130,2004           | 133-2012             | 0-0021-           |
| 2          | -             |               | 3.1052         |           |                                       | ·····                                 | 0.0366           |               | C. 0301                 | 7.6 d.b.      |             | 13013000           | 130.2/10             | 010051            |
|            |               |               | 941076         | W. W. OZ  | 3. 7000                               | 307392                                | 0.0344           | - 1           | 4. 14.11                |               | V-4611      | 130-3303           | 112-251              | 0.3317            |
| 7          | 0.01          | 107           | 6+6242         | 4+4422    | 3. 4000                               | 347922                                | 01000            |               | 4.1003                  | 11.2271       | 6+1134      | 120+2022           | 130.1139             | 0.3544            |
|            | 9.9           | 21            | 9.9495         | 0.3131    | 3.9903                                | 3+4131                                | 0.0014           |               | 0.0474                  |               |             | 110-2022           | 13110125             | 0.0125            |
|            | 0.0           | 21.2          | C.0433         | 9*0241    | 3.9033                                | 3* 40 4E                              | 0-0023           | •             | 9.2244                  | 1,3072        | 0.7344      | 129'3933           | 130.7344 -           | しょうりょう            |
| 30         | 6-64          | 332           | 0.0170         | 9.0331    | 3.9600                                | 3.9037                                | 9.9994           | 10            | 0.0776                  | [.1927        | 0.4637      | 130.3993           | 133.4037             | 0.0336            |
| 21         | 0,00          | )10           | _0.4[J5_       | 0.0016_   |                                       |                                       | 0.0034           | - 11          | 0.3210                  | 118914        | 0.4209      | 133-3933           | 133.4239             | 0.0032            |
| 13         | 0.00          | 170           | C.J246         | 0.0025    | 3.9500                                | 3.4625                                | 0.0306           | 17            | 0.3397                  | 3+0527        | 0.6326      | 130, 3333          | 134.6326             | 3.3346            |
| 17         | 0.33          | 333           | 0.2194         |           | 3.9600                                | 3.4952                                | 0.0088           | - 13          | C.C395                  | 2.9270 -      | 0.7337      | 110. 1000          | 130.7347             | 4.4356            |
|            |               | • • •         |                |           | e                                     |                                       |                  |               | <u> </u>                |               |             |                    |                      |                   |
|            |               |               |                |           | <b></b>                               |                                       |                  |               | 1                       |               |             | '''ı ————          |                      |                   |
|            | 6.00          | <b>37</b> .   | 0.0560         | ້ວ. ນເຈ   | 39.2040                               | 49, 22 59                             | 4.0376           | 1             | 0.0004                  | 4.4040 -      | 1.7695      | 2 44. 2020         | 211.7661             | 4.4574            |
|            | 0.01          | 165           | 6.5107         | 0.3693    | 19.2040                               | 39.2733                               | 0.0316           |               | 0.0256                  | 24.6313       | 4.2316      | 234.0003           | 214.2726             | 2.31.43           |
|            | 5.03          |               | 0.000          | - 0.6513  | 19. 2040                              | 19.2459                               | 0.0321           |               | 0.1212                  | 14. TG UN     | 7. 331 3    | 210.0000           |                      |                   |
|            | 0.5           | 10            | · A. J. J. M.  | 4.5318    | 34.2363                               | 39.2041                               | 0.0.00           |               | 0.0303                  | 1.8457        | A. 174.     | 23.1. 330.3        | 210 . 31.            | 010323            |
| -          |               |               | 0.3317         | 4.3332    | 30. /04.)                             | 44. 2.14.0                            | 0.033.0          | - 7           | 1.4444                  | 1.6100        |             | 2300 3000          | 23043120             | 0.0010            |
|            | 6 63          |               | 0.0017         | 0.0011    | 30.2040                               | 10.2041                               | 0.0303           |               | 0.0000                  | 2.46.40       |             |                    | 233-1341             | 010030            |
|            | 4.44          | 130           | A 4601         | 4 3337    | 30 305 4                              | 54.33.4                               | 0.03/16          |               | A. 1740                 | 240404        | 0.3335      | 230-3043           | 2H1 122H             | · J.JJZ4          |
| 7          | 4.00          |               | 0.0071         | 0.0467    | 19.1043                               | 3444403                               | 4.0116           |               | 74 17 47<br>A. A. B. A. | 1742216       | 441124      | 6666 July 2        | 238,1755             | <b>د ب دن , ن</b> |
|            | V+00          |               | 0.0342         |           | - 'AAA 5040'-                         |                                       | 0,0004           |               | 0,0(8)                  | J.4867        | 1.11.31     | \$30*3019          | 231 47137            | 0.0014            |
|            | 0.00          | 116           | 0.0034         | 0.0111    | 37.2043                               | 3742127                               |                  |               |                         | 1+6649        | 0.111       | 520.0070           | 233.7336             | 0.0332            |
| 10         | 0403          | 194           | 0.3245         | 0.537#    | 39.2040                               | 34, 2948                              | 0+0001           | 10            | 0.0103                  | 4.6928        | J+4924      | ₹ <b>?</b> ?* 3833 | 233.2429             | للددرد            |
| 11         | 0.00          | 135           | 0.0101         | 0.001     | 39.2040                               | \$9.2363                              | 010301           | - 11 .        | 0-0303                  |               | 0.1314      | 579,3339           | 672.1019             | 0.0334            |
| 75 -       | 0.01          | isi 👘         | 8.9755         | 0.0437    | 39,2343                               | 39,2477                               | 0.0311           | - 14 -        | 0+0375 -                | 1.3895        | ~ Q.1214    | 250,9000           | 213.1234             | 4.0345            |
| 13         | 0.04          | 120           | 0.0173         | 0.0350    | 39.2360                               | 39.2090                               | 0.0331           | - 12 .4       | 0.0032                  | 3.7066        | J.7626      | 230. 3000          | 23.1. 2626           | 0.0033            |
|            |               |               |                |           |                                       |                                       |                  | {             |                         |               |             |                    |                      | 414477            |
|            | -             |               |                |           | · · · · · · · · · · · · · · · · · · · |                                       | i                |               |                         |               |             |                    |                      |                   |
| 1          | 0.01          | 76            | 0.6243         | 0.2480    | 36.2250                               | 34.473.2                              | 0.0072           |               |                         |               |             |                    |                      |                   |
| 2          | 0.03          | 31            | 6.3643         | 1.1444    | 34.2255                               | 33.4040                               | 0.0335           |               | •                       |               |             |                    | · • •                |                   |
| τ.         | 8.14          | 74 .          | 22.43.27       | 2.45.44   | 44. 2 460                             | 86. 8440                              | 0.0194           |               |                         |               |             | · .                |                      |                   |
| 1          | a             | ก่อ (         | 0.0.316        | 0.0.11    | 34.2240                               | 34.2253                               | 0.33.13          |               |                         |               |             | •                  | •                    |                   |
| 2          | 0.01          | 55            | 0.014#         | A. 33 14  | 14.334.                               | 14.7255                               | 0.0103           | 1             | To get was              | lentes en ori | almat ecolo | a. mittiela        | usture               |                   |
|            | ****          |               | C 30.470       |           | 746670                                | 3444434                               | 0.0000           |               | in Sable P              | W 1.0. 1000.  | and .001 /  | or T. C            | TURE AND AND A STATE |                   |
| 2          |               | 40            |                | 4.44333   | 42+6634                               | 3944673                               | weddda<br>a abad |               | respect Ive             | 1.            |             |                    | n Ascreptes          | •                 |
| 1          | 10.1          |               | 6.10 VJ        | 2.1612    | 20+2220                               | 3 F 4 Y 4 6 7                         | 010701           |               |                         | ••            |             |                    |                      | •                 |
|            | 9.74          | 11            | 1.41.22        | 0.5152    | 14+2220                               | 39+0414                               | 040177           |               |                         |               |             |                    |                      |                   |
|            | 0.00          | -26           | 0.5114         | 9.2164    | \$4.2250                              | 34.4417                               | 8.3063           |               |                         |               |             |                    |                      |                   |
| 10         | 0,01          | 21            | 0.1435         | 0.0618    | 34.2253                               | 34.2040                               | 0+0419           |               |                         |               |             |                    |                      |                   |
| 11         | h*d3          | 24            | £+V044         |           | 34.2250                               | 39+2462                               | 040036           |               |                         |               | •           |                    |                      | •                 |
| 13         | 6.00          | JB .          | 1.2646         | 0,0160    | 34,2250                               | 34,3007                               | 0.0322           |               |                         |               |             |                    | •                    | •                 |
| 13         | 0.03          | 22            | C.1>50 _       | 0.6324    | 34.2200                               | 34.2284                               | 0.0014           |               |                         | •. •          |             |                    |                      |                   |
| -          |               |               |                |           |                                       |                                       |                  |               |                         |               |             |                    |                      |                   |

.

٠

()

(:

(

been estimated by supplying the model with a set of independent variables which were subject to error and subsequently comparing dependent variable observations with model predictions. Generally, the same set of independent variables has been used in simulating each case, so that the same covariance matrix of independent variable errors [Cov  $(x_i, x_j)$  in equation (8)] would apply. However, the sensitivities of the state variables with respect to the independent variables [ $\partial y(t) / \partial x_i$  in equation (<sup>8</sup>)] may change as a function of case, suggesting that total independent variable error may also change with case, just as total parameter error has been shown to change.

Probably the most important concern with regard to the effects of this type of error is the possibility that consistent errors or biases in one or more of the independent variables may have introduced biases in the parameter estimates. One variable which would be of potential concern in this regard is  $U_6$ , the chloride flux from the Allied Chemical waste beds. As demonstrated in Section 4.6, the hypolimnion and gradient variables are highly sensitive to the values of this variable. This indicates that the optimal parameter estimates, particularly  $a_{14}$ , would also be sensitive to bias in  $U_6$ . Fortunately, as discussed in Section 4.3, the value assumed for this variable has been verified by two independent methods: (1) Ninemile Creek flow and chloride concentration data, and (2) a mass balance on the Allied Chemical solvay plant developed from soda
# 4.6.3 Error Analysis (continued)

ash production, reaction stoichiometry, and reaction efficiency data. Because of the possibility for introduction of bias, the quality of the independent variable data can be critical to proper implementation of the empirical parameter estimation methods employed above.

#### 4.7 Conclusions and Recommendations

### 4.7.1 Model Adequacy

Œ

C

( )

One distinctive feature of the model which has been developed, estimated, and applied above is its high level of spatial and temporal aggregation. Monthly-average boundary conditions have been employed to drive a system consisting of two mixed layers. This is to be compared with the M.I.T.<sup>1,2</sup> and W.R.E.<sup>9</sup> reservoir models discussed in Section 4.2. These models employ a minimum of 20 mixed layers of 1- to 2-meter thicknesses and require daily-average boundary conditions. Aggregation in space and time go together, because the former tends to enhance the stability of the system and to render it less sensitive to high-frequency variations in boundary conditions.

This aggregation has resulted in a number of distinctive properties. First, the model obviously has less potential resolution in time and space. The resolution was sufficient for generally characterizing vertical exchange in the lake, as evidenced by the ability of the model to simulate observed temperature and chloride variations in the hypolimnion and epilimnion with time. Second, the model requires less data to implement. Obviously, the specification of daily-average meteorologic and hydrologic boundary conditions would

#### 4.7.1 Model Adequacy (continued)

 $( \cdot )$ 

()

be a much more demanding requirement. Third, the model is less expensive to implement, because it involves fewer state variables and longer time steps. As noted in Section 4.2, the M.I.T. model, with spatial and temporal resolutions of one meter and one day, respectively, consumes 2.4 minutes of CPU time per year of simulation, while the model developed here uses 0.028 CPU minutes per year\*. Use of the parameter estimation routine would probably have been economically infeasible if a model with the spatial and temporal resolution of the M.I.T. model were employed.

The standard errors of estimate of the model developed here are compared with those typical of the M.I.T. model, as derived from the study of Parker et al.<sup>11</sup> in Table 4.7 ~l. For predictions of surface temperature, the standard errors for the latter averaged  $2.15^{\circ}$ C and ranged from 1.38 to  $2.98^{\circ}$ C in a sample of seven different reservoirs and years. These can be compared with the standard error of  $1.67^{\circ}$ C obtained for predictions of epilimnion temperature in Onondaga Lake. Corresponding values for predictions of reservoir outlet temperatures averaged  $1.90^{\circ}$ C and ranged from 1.01 to  $2.88^{\circ}$ C using the N.I.T. model, compared with a value of  $1.87^{\circ}$ C obtained above for prediction of hypolimnion temperature. Thus, it appears that the two models have about the same residual standard errors, despite the differences in their levels of spatial and temporal resolution.

\* Some of the differences in CPU timemay, be attributed to differences in computers. The computer model employed in the 1975 study of the M.I.T. model was not specified by the authors.<sup>25</sup>

Table 4.7-1

Comparisons of Standard Errors of Estimate for Temperature Predictions in Onondaga Lake with Those Typical of the M.I.T Deep Reservoir Model

|                  |                      |         | STANDARD ERRORS OF E   | STIMATE ( <sup>o</sup> c) |
|------------------|----------------------|---------|------------------------|---------------------------|
| Model            | Lake or<br>Reservoir | Years   | Surface<br>Temperature | Bottom<br>Temperature     |
| ESP              | Onondaga             | 1968-74 | 1.67 <sup>a</sup>      | 1.87 <sup>b</sup>         |
| MIT <sup>d</sup> | Fontana              | 1966    | 1.74                   | 1.19                      |
| MIT              | Douglas              | 1969    | 2.03                   | 2.11                      |
| MIT              | Cherokee             | 1967    | 2.07                   | 2.70                      |
| MIT              | Norris               | 1972    | 2.31                   | -                         |
| MIT              | So. Holston          | 1953    | 2.44                   | 2.88                      |
| MIT              | Hiwassee             | 1947    | 1.38                   | 1.01                      |
| MET              | Fort Loudon          | 1971    | 2.98                   | 1.53                      |

a - epilimnion-average.

(

ſ

b - hypolimnion-average. c - Model of Harleman et al. , as evaluated by Parker et al. ; bottom temperature refers to outlet level.

( )

(-

One feature which renders the model developed above somewhat unrealistic is the assumption of constant epilimnic and hypolimnic volumes. Because of thermocline migration, these volumes actually vary with season and could possibly be influenced by some of the conditions specified by the various cases in Section 4.7. Realism could be enhanced by incorporating another state variable which would permit simulation of thermocline migration. This would involve a modification of the exchange function  $(F_{11})$ , but mechanical energy balance considerations would still apply. Because the relative changes in volume could have substantial effects on the dynamics of non-conservative components (in particular, algal growth rates), this element of realism should probably be added to the model before expanding it to permit direct simulation of water quality components.

#### 4.7.2 Data Adequacy

Generally, two types of data have been employed above to serve two purposes: definition of system conditions and definition of boundary conditions. The adequacy of the former as a basis for parameter estimation is indicated by the relatively small contributions of parameter variance to total model prediction variance, as noted in Section 4.6.3. The adequacy of the boundary condition data is more difficult to assess. Much of the auto-correlation and periodicity in the residuals has been attributed to concomittant errors in the specification of meteorologic or hydrologic boundary conditions (e.g.,

### 4.7.2 Data Adequacy (continned)

solar radiation). These residual properties have been shown to introduce uncertainty into the estimates of the parameter covariance matrix and confidence region. Residual standard errors can actually be attributed to three components: measurement error, model error, and independent variable error. An assessment of these components is not possible without independent evidence. The primary concern over the potential effects of independent variable error is that consistent errors or biases in these data may have given rise to biases in the parameter estimates. Such biases, in turn, may have influenced the policy recommendations derived from model applications. It would be difficult if not impossible to make the analysis robust to such errors.

# 4.7.3 Methods Adequacy

This work has provided a partial basis for assessment of the values and limitations of Bard's parameter estimation algorithms in modelling studies of this type. A number of modifications and additions to these algorithms have been made, specifically to permit : (1) mapping of objective function values and residual sums of squares for specified parameter values; (2) use of Runge-Kutta integration; (3) computation of derivatives of the objective function with respect to the parameter values using finite-difference methods; (4) estimation of the limits of the parameter confidence region. These additions have enhanced the flexibility of the parameter estimation program, which has demonstrated reasonable potential.

The methods have been employed to estimate optimal sets of parameters

## 4.7.3 Methods Adequacy (continued)

( )

()

for each year individually and for all years combined. In the yearly exercises, in which two parameters were estimated, convergence was achieved using the Gauss-Newton maximization algorithm in from 3 to 7 interations (derivative evaluations), 8 to 19 objective function evaluations, or from 15 to 33 equivalent function evaluations. When all data were combined and four parameters were estimated, convergence was reached in 7 iterations, 16 objective function evaluations, or 44 equivalent function evaluations. The numbers of iterations are typical of those required in nonlinear estimation problems<sup>33</sup>. The use of finite-difference methods, as opposed to sensitivity equations, for computation of derivatives does not appear to have impaired convergence rates. In these problems, the objective functions and parameter values changed fairly rapidly over the first one or two iterations and rapidly approached assymptotic values as the solutions were approached.

Most of the cost of estimating the parameters is attributed to numerical integration of one set of state variable equations throughout the time period for each equivalent function evaluation required. As noted above, such costs could become prohibitive as the level of temporal and spatial resolution in the model increases. The algorithms appear to be most practical for use with aggregated models. If high resolution is not required, the use of aggregated models offers an additional advantage in that it would entail the aggregation of observations, which would also serve to remove some of the measurement error. This, in turn, could improve convergence rates by rendering the solution a more well-defined region in parameter space.

#### 4.7.3 Methods Adequacy (continued)

t

 $(\cdot)$ 

One of the most important benefits derived from use of the algorithms is that both the first and second moments of the parameters can be estimated. These can provide a basis for estimation of the confidence region and for use of parameter stability criteria for model verification. The parameter and residual covariance matrices can also be used to estimate the covariance matrix of model predictions. Comparisons of parametric and residual variance components can lead to an assessment of the adequacy of the amount of data employed for parameter estimation purposes. Generally, because of nonlinearity in the model and serial dependence and non-normality in the residuals, the estimates of the residual and parametric covariance matrices are only approximate. For simple models, better estimates could be obtained using Monte-Carlo techniques.

Applications of the parameter estimation algorithms should be restricted to cases in which the independent or forcing variables are well-known. Biases in these variables could lead to biases in parameter estimates, a distorted picture of system dynamics, and biases in model predictions and policy recommendations. In such situation, prior parameter estimates, based upon independent experimental evidence might be employed, if available. Alternately, the unknown boundary conditions could be treated as unknown parameters and estimated along with the unknown system parameters, provided that the dimensions of the problem do not become excessive." Although not exercised above,

\*"Excessive" would probably be defined by economic constraints. As distributed, the Bard program can handle up to 20 unknown parameters. By increasing the sizes of the appropriate dimension statements, this upper limit could be increased. The feasibility of estimating more than four parameters simultaneously has not been demonstrated here. Generally, convergence problems might be encountered as the number of estimated parameters increases if some or all of the parameters are highly correlated with each other. <sup>2</sup>

# 4.7.3 <u>Methods Adequacy</u> (continued)

**(**)

()

the option to include the effects of prior distributions of parameter estimates on the objective function would seem potentially useful. In such a case, the optimal parameter estimates would be based upon information obtained both from the system under study and from independent evidence obtained from experiments or from other natural systems. The information content of the prior parameter distributions could improve the convergence properties of the algorithms, which could be particularly effective when the size of the unknown parameter vector is large.

In approaching parameter estimation problems of this type, preliminary studies should involve mapping of objective function and residual sums of squares for observed variables over feasible regions of parameter space. This can provide a feeling for the general location of the solution and for the shape of the response surface, including the locations of any local optima. Comparing the residual sums of squares for various dependent variables can also be useful as a means of detecting independent variable errors or model inadequacies. If such errors or inadequacies are not significant, and if there are no biases in the measurements, minimal sums of squares for all predicted variable residuals should all be located in approximately the same region of parameter space. This type of preliminary analysis can also provide reasonable initial guesses for implementation of the formal parameter estimation routine. "()

 $( \cdot )$ 

Model aimulations indicate that the mode of discharge of the combined minicipal/industrial effluent could have relatively dramatic effects on the degrees of vertical mixing in Quondaga Lake. Discharge into the hypolimnion would apparently result in permanent stratification under the hydrologic and meteorologic conditions in the 1968 to 1974 pariod. At the other extreme, simulations indicate that discharge into the epilimnion with an infinite initial dilution ratio would minimize the impact of the industrial salt loading and induce a mixing regime substantially more vigorous than currently exists and not much different from what would exist if the unnatural salt discharge were eliminated. As discussed in Section 4.7, the path between these two limiting cases as a function of initial dilution ratio is uncertain because it depends upon the value of a parameter  $(a_{21})$ which cannot be estimated a priori without a more detailed study of the hydrodynamics of the outfall site. Simulations with a dilution ratio of 16 and a conservative value of 1.0 for  $a_{21}$  (Case 12), are not greatly different from simulations with an infinite initial dilution ratio (Case 13). If allowance for this conservatism is made, it would appear that discharge into the epilimnion at a dilution ratio of 16 or greater would likely be equivalent to infinite intial dilution in that it would essentially destroy the driving force for sinking of the effluent into the hypolimnion. It is unclear whether the current plan to discharge through a shoreline, surface outfall would achieve this level of initial dilution. A ratio of 16 would not be particularly difficult to achieve if an off-shore diffuser were employed.

4.7.4 Implications for Outfall Design (continued)

()

(

Decreases in density gradients over the course of the 1968-1974 period have been attributed to : (1) climatologic variations (possibly including (a) warmer winters in 1973-74, (b) higher wind speeds in spring of 1974, and (c) residual effects of the high flows experienced during Hurricane Agnes in June of 1972 ) and (2) the increased frequency of cooling water withdrawal from the hypolimnion by Allied Chemical in later years. Some of the water quality improvements observed over this period may have been partial responses to enhanced mixing rates. These include increased hypolimnic dissolved oxygen levels and reduced dominance of blue-green algae.

These observations suggest that future water quality could be sensitive to the design of the Metro STP outfall. If the outfall were designed to achieve a high rate of initial dilution of the effluent in ambient epilimnic waters, resultant enhanced mixing could improve hypolimnio oxygen levels. Based upon the analysis of Chapter 3, it could also result in somewhat higher transparencies and a less favorable environment for the growth of bluegreen algae. Specfic water quality impacts could be more thoroughly investigated with an expanded model. Increasing the intitial dilution ratio would serve other purposes by reducing the impact of the effluent in the immediate discharge zone and possibly abatingaesthetic problems by reducing the driving force for rapid precipitation of calcium carbonate, as discussed in Section 4.1. If economics are considered, it would appear that the flexibility and relative inexpense of the current plan for a surface, shoreline outfall are attractive. Should this alternative prove unacceptable from a lake mixing or aesthetic viewpoint, the analysis here indicates that

## 4.7.4 Implications for Outfall Design (continued)

\$

 $C_{i}$ 

an extended outfall should provide for dilution of the effluent in the epilimnion, if permanent density stratification is to be avoided.

Advective mixing provided by cooling water withdrawal from the hypolimnion and discharge to the epilimnion is significant compared with wind-induced mixing rates during midsummer seasons. Continued use of hypolimnic waters by Allied Chemical would result in a variety of environmental and economic impacts. Allied chlorinates its cooling water to minimize the impact of reducing compounds on its equipment<sup>7</sup>. Enhanced dissolved oxygen levels in the influent water would lower chlorine demand costs, while simultaneously permitting constant use of hypolimnic waters, the lower temperatures of which would reduce volume requirements and associated pumping costs. If advective flows into the hypolimnion due to density currents are eliminated (e.g., Case 13), withdrawal from the hypolimnion may result in a lower average thermocline level, a result not derivable directly from the model because of its constant volume constraints. Allied's withdrawal rate is about sufficient to lower the thermocline from 6 meters (approximate springtime level) to about 10 meters in a 120-day period. Coupled with normal wind-induced thermocline migration, this may cause the hypolimnic volume to shrink more rapidly during the summer season than under current conditions, in which the hypolimnion is partially replenished by flow from Ninemile Creek. The hypolimnion would likely not be destroyed completely due to cooling water withdrawal, because Allied's deep intake ports would be exposed to the epilimnion, should the thermocline reach the vicinity of the 15-meter level.

#### **REFERENCES - CHAPTER 4**

()

- Bard, Y., "A Function Maximization Method with Application to Parameter Estimation", New York Scientific Center Report 322.0902, I.B.M., May 1967.
- Bard, Y., "Nonlinear Parameter Estimation and Programming", Program No. 360D-13,6,003, Share Program Library Agency, Triangle Universities Computation Center, Research Triangle Park, N.C., December 1967.
- 3. Bard, Y., <u>Nonlinear Parameter Estimation</u>, Academic Press, New York and London, 1974.
- Banks, R.B., "Some Features of Wind Action on Shallow Lakes", Journal of the Environmental Engineering Division, American Society of Civil Engineers, Vol. 101, No. EE5, pp. 813-827, October 1975.
- Bella, D., "Dissolved Oxygen Variations in Stratified Lakes", Journal of the Sanitary Engineering Division, American Society of Civil Engineers, Vol. 96, No. SA5, Proc. Paper 7628, pp. 1129-1146, October 1970.
- Blanton, J.O., "Vertical Entrainment into the Epilimnia of Stratified Lakes", <u>Limmology and Oceanography</u>, Vol. XVIII, No. 5, pp. 697-704, September 1973.
- Clough, R., Allied Chemical Corporation, Syracuse, New York, personal communication, 1976.
- 8. Dake, J.M.K. and D.R.F Harlaman, "Thermal Stratification in Lakes: Analytical and Laboratory Studies", <u>Water Resources</u> Research, Vol. 5, No. 2, pp. 404-495, April 1969.
- Fletcher, R. and M.J.D. Fowell, "A Rapidly Convergent Descent Method for Minimization", <u>The Computer Journal</u>, Vol. 6, No. 2, pp. 163-168, 1963.
- 10. Handbook of Physics and Chemistry, 36th Edition, 1955.
- 11. Harleman, D.R.P. and M. Markofsky, "A Predictive Model for Thermal Stratification and Water Quality in Reservoirs", MIT Department of Civil Engineering, Laboratory for Water Resources and Hydrodynamice, Report No. 134, January 1971.
- 12. Hoel, P.G., <u>Introduction to Mathematical Statistics</u>, Fourth Edition, John Wiley and Sons, 1971.

#### REFERENCES - CHAPTER 4 (continued)

(

 $(\cdot)$ 

- Huber, W.C. and D.R.F. Harleman, "Laboratory and Analytical Studies of Thermal Stratification in Reservoirs", MIT Hydrodynamics Laboratory Technical Report No. 112, October 1968.
- Huber W.C., D.R.F. Harleman, and P.J. Ryan, "Temperature Prediction in Stratified Reservoirs", <u>Journal of the Hydraulics</u> <u>Division, American Society of Civil Engineers</u>, Vol. 98, No. HY4, Proc. Paper 8839, pp. 645-646, April 1972.
- Hutchinson, G.E., <u>A Treatise on Limnology: Vol. 1 Geography</u>, Physics, and Chemistry, Wiley, New York, 1957.
- International Business Machine Corp., <u>System/360 Scientific</u> <u>Subroutine Package</u>, Program No. 360A-CM-03X, Fifth Edition, 1970.
- Johnston, J., <u>Econometric Methods</u>, Second Edition, McGraw-Hill Book Co., New York, 1972.
- Kato, H. and O.M. Phillips, "On the Penetration of a Turbulent Layer into Stratified Fluid", <u>Journal of Fluid Mechanics</u>, Vol. 37, Part 4, pp. 643-655, 1969.
- 19. Linsley, R.K., M.A. Kohler, and J.L.H. Paulhus, <u>Hydrology for</u> Engineers, McGraw-Hill Book Co., Second Edition, 1975.
- 20. Lorenzen, M. and R. Mitchell, "Theoretical Effects of Artificial Destratification on Algal Production in Impoundments ",<u>Environ-</u> <u>mental Science and Technology</u>, Vol. 7, No. 10, pp. 939-944, October 1973.
- 21. O'Brien and Gere, Engineers, "Onondaga Lake Monitoring Program", Annual Reports to Onondaga County, Department of Public Works, 1970-74.
- 22. O'Brien and Gere, Engineers, "Supplement to Wastewater Facilities Report - Metropolitan Sewage Treatment Plant Expansion", prepared for Onondaga County, April 1971.
- 23. Onondaga County, New York, "Onondaga Lake Study", US EPA, Water Quality Office, Publication No. 11000 FAE 4/71, April 1971.
- 24. Orlob, G.T. and L.G. Selna, "Temperature Variations in Deep Reservoirs", Journal of the Hydraulics Division, American Society of Civil Engineers, Vol. 96, No. HY2, Proc. Paper 7063, pp. 391-410, February 1970.

#### REFERENCES - CHAPTER 4 (continued)

( )

()

- Parker, F.L., B.A. Benedict, and C. Tsai, "Evaluation of Mathematical Models for Temperature Prediction in Deep Reservoirs", National Environmental Research Center, Office of Research and Development, US EPA, Document No. EPA-660/3-75-038, June 1975.
- Rooney, J., "Evaluation of Interim Basin Plan IBP-NY-07-07, for Lake Onondaga, New York and Project WPC-NY-659 (Syracuse Metro STP)", EPA Region II Ofice, New York 1973.
- 27. Snedecor, G.W. and W.G. Cochran, <u>Statistical Methods</u>, Sixth ... Edition, Iowa State University Press, 1967.
- Snodgrass, W.J., "A Predictive Phosphorus Model for Lakes -Development and Testing", Ph.D. Thesis, University of North Carolina at Chapel Hill, Environmental Sciences and Engineering, 1974.
- Stefan, H. and D.E. Ford, "Temperature Dynamics in Dimictic Lakes", Journal of the Hydraulics Division, American Society of Civil Engineers, Vol. 101, No. HY1, pp. 97-114, January 1975.
- Sundaram, T.R., C.C. Easterbrook, K.R. Prech, and G. Rudinger, "An Investigation of the Physical Effects of Thermal Discharges into Cayuga Lake", Cornell Aeronautical Laboratory, Buffalo, New York, Report VT-2616-0-2, November 1969.
- 31. Sundaram, T.R., R.G. Rehm, and G.E. Merritt, "A Study of Some Problems on the Physical Aspects of Thermal Pollution", Cornell Aeronautical Laboratory, Buffalo, New York, Report VT-2790-A1, June 1970.
- 32. Sundaram, T.R. and R.G. Rehm, "Formation and Maintenance of Thermocline in Temperate Lakes", <u>American Institute of Aeronautics</u> <u>and Astronautics</u>, Vol. 6, No. 2, pp. 1322-1329, 1971.
- 33. Sundaram, T.R. and R.G. Rehm, "Effects of Thermal Discharges on the Stratification Cycle of Lakes", <u>American Institute of</u> <u>Aeronautics and Astronautics</u>, Vol. 10, No. 2, pp. 204-210, 1972.
- Turner, J.S. and E.B. Kraus, "A One Dimensional Model of the Seasonal Thermocline I.A Laboratory Experiment and its Interpretation", <u>Tellus</u>, Vol. 19, pp. 88-97, 1967.

#### REFERENCES - CHAPTER 4 (continued)

()

C

Cit

- 35. Turner, J.S., "The Influence of Molecular Diffusivity on Turbulent Entrainment Across a Density Interface", <u>Journal</u> of Fluid Mechanics, Vol. 33, pp. 639-656, 1968.
- Thompson, E.S., "Computation of Solar Radiation from Sky Cover", <u>Water Resources Research</u>, Vol. 12, No. 5, pp. 859-865, October 1976.
- 37. US EPA, Region II, "Environmental Impact Statement on the Wastewater Treatment Facilities Construction Grants for the Onondaga Lake Drainage Basin", May 1974.
- U.S. Weather Service, "Local Climatological Data", U.S. Department of Commerce, 1965-74.
- Velz, C.J., <u>Applied Stream Sanitation</u>, Wiley-Interscience, New York, 1970.
- 40. Vollenweider, R.A., "Scientific Fundamentals of the Eutrophication of Lakes and Flowing Waters, with Particular Reference to Nitrogen and Phosphorus as Factors in Eutrophication", Organization for Economic Cooperation and Development, Paris, 1970.
- 37. Walker, W.W., "Techniques for Parameter Estimation in Nonlinear Dynamic Systems Applied to Two Ecosystem Models,", Technical Paper No. 750512, Onondaga Lake Modelling Project, Environmental Systems Program, Harvard University, 1975.
- 42. Water Resources Engineers, Inc., "Mathematical Models for the Prediction of Thermal Energy Changes in Impoundments", Federal Water Quality Administration, Water Pollution Control Research Series 16130 EXT 12169, December 1969.
- Williams, J., <u>Introduction to Marine Science</u>, Little and Brown Co., Boston and Toronto, 1964.
- 44. Wu, J., "Wind Stress and Surface Roughness at Air-Sea Interface", <u>Journal of Geophysical Research</u>, Vol. 74, pp. 444-455, January 1969.

()

( )

The previous chapters have demonstrated the use of a variety of quantitative techniques for assessments of lake water quality problems. These techniques have been generally characterized in Chapter 1. Rather than reiterate here the specific conclusions already expressed at the end of each chapter, it would be of value to summarize what has been learned about the potential roles and limitations of these methods and approaches.

Preliminary data analyses have been useful in summarizing important relationships, both in the cross-sectional studies of Chapter 2 and in the spatial and temporal studies of Chapter 3. In Chapter 2, marked stratification of all independent and dependent variables with lake trophic state was observed. In Chapter 3, a variety of temporal associations were evident. For example, the disappearance of blue-green algae occurred simultaneously with the reductions in ambient phosphorus levels, chromium levels, and density gradients. Lake phosphorus concentrations decreased simultaneously or immediately following the implementation of detergent restrictions, combined sewer maintenance programs, and diversion of a raw sewage discharge to primary treatment facilities. Because of the general problem of multicollinearity, either in variables characterizing different lakes or in events occurring in or around Onondaga Lake, data analyses alone have not provided sufficient bases for inferences of causation. At a practical level, we have had to rely upon independent evidence as a basis for functional understanding

**2\*\***1

(i

[ ]

of system behavior. The empirical modelling approach of Section 2.5 would not permit clear separation of the effects of depth and hydraulic residence time upon lake phosphorus dynamics. Some more mechanistic modelling efforts might be helpful in this regard. It is unclear whether current understanding of multi-species algal population dynamics would permit us to attribute the apparent disappearance of blue-green algae from Onondaga Lake to a specific factor. Thus, preliminary data analyses have provided important descriptive information, but not the functional understanding required to predict lake behavior.

Regression techniques for estimating parameters have been applied to linear, nonlinear, and nonlinear/dynamic models. In the last case, the use of nonlinear programming algorithms has demonstrated reasonable potential, subject to economic constraints on model complexity. The variety of assumptions inherent in the use of maximum likelihood criteria for parameter estimation have been discussed in Chapter 1. A difficulty which has been encountered in Chapter 2 and Chapter 4 modelling efforts concerns the possible effects of errors (as bias or variance) in the independent variables on the estimates of the parameter vector and of the parameter and residual covariance matrices. In Chapter 4, observed serial correlation in the residuals was attributed partially to such errors. A Monte Carlo approach to this problem would be feasible only in relatively simple systems. Alternative methods for obtaining prior estimates of such errors are not well-developed. This tends to emphasize

( :

the importance of independent variable data as bases for modelling efforts. Specification of models, particularly with regard to complexity, should not be done without considering the quality and quantity of available independent variable data. The same comments apply to the importance of parameter estimates which are derived exclusively from prior information.

As a corollary to this, assessments of lake water quality problems can be particularly dependent upon estimates of the quantities of materials entering the lake from various sources. Thus, mass flux boundary conditions represent an important type of independent variable in these analyses. If a key component is identified (e.g., phosphorus), in many instances, management policy recommendations can be based upon accurate estimation and comparison of the various problem sources, without use of a particular lake model, other than the assumption that the particular component is controlling lake water quality. In other cases, lake models may be required to estimate the degree of source control necessary to achieve specific water quality objectives. The estimation and verification of such models depend upon the accuracy of source estimates. Thus, the state-of-the-art of source models, as discussed in Section 2.3, suggests the importance of source monitoring, as well as lake monitoring, in providing an analytical basis.

Error analyses have been useful in assessing model and data

( ·

adequacies and in providing bases for model predictions on probabilistic terms. In Section 1.5, prediction errors were described as consisting of three basic components: parameter error, independent variable error, and residual error. The last was further dissected into model and measurement error components. In applications, it has been difficult to obtain complete separation of these components. For example, if it is estimated by comparing model predictions with system observations, residual error actually contains an independent variable error component. Prior estimates of errors in independent variables and in specified parameters are usually difficult to obtain with much accuracy. Posterior estimates of the parameter covariance matrix are likewise approximate in the case of a nonlinear model, particularly if the residuals are autocorrelated. It must also be considered that the error equation (1.5-2) is based upon a first-order approximation. Thus, three factors suggest the approximate nature of error analyses, particularly in cases of complex models: (1) smearing of error components; (2) the approximate natures of the parameter, independent variable, and residual covariance matrix estimates; (3) first-order truncation of the fundamental equation. Despite these factors, comparisons of error sources can still yield useful information for assessing model and data adequacies, because, as demonstrated in Sections 2.5.3 and 4.6.3, the terms of the total error equation often differ by orders of magnitude. The approximate nature of the total error estimate for a given model prediction suggests that this second moment should not be relied upon too heavily as a basis

>~4

for probabilistic projections and rational designs, particularly if a relatively complex model is employed.

Generality is a key model attribute. In order to be applied properly in addressing specific management problems in specific natural systems, models should be general enough to be valid both under present and projected states of the systems. Parameter stability criteria have been employed in Chapters 2 and 4 as partial bases for assessments of model generality. The parameters of the phosphorus retention model developed in Chapter 2 were shown to be stable across lake trophic states. While this model was apparently general enough to give unbiased predictions in different types of lakes, its validity for simulating behavior of single lake in time was not substantiated. In Chapter 4, two principal parameters of the vertical stratification model were found to be reasonably stable when estimated independently, based upon data from separate years. However, the model was applied to predict lake behavior under conditions of salt disposal which were substantially different from those under which estimation took place. The theoretical basis of the model was relied upon to permit its use in extrapolating or projecting system behavior beyond previously observed states. Demonstration that the same model and parameter estimates could successfully simulate behavior of a different lake would have provided stronger evidence for model generality and further justification for its use in a projective mode.

C

The qualified successes of empirical approaches have shown that average water quality conditions in a cross-section of lakes in the same geographical region can be associated with such factors as phosphorus loading, mean depth, and hydraulic residerce time. However, the accuracy and validity of these models for use in predioting response of a given lake to changes in the associated factors have yet to be adequately determined. As discussed in Sections 2.4.2 and 2.6, theoretical ecosystem models have not as yet demonstrated wide generality for use in eutrophication assessments, as evidenced by apparent variations in parameter values appropriate for simulating different lake ecosystems. This suggests inadequacies in the data and/or models and indicates that our ability to project the detailed behavior of a given lake ecosystem much beyond previously observed states is still rather limited.

Based upon the weaknesses inherent in these analyses, it would seem that modelling efforts should strive to demonstrate generality both among lakes and along temporal dimensions within lakes. This suggests that time series data from more than one lake should serve as a basis for model estimation and verification. The success of such an approach would depend upon the availability of adequate data and upon the feasibility of expressing essential functional relationships in concise terms, based upon current understanding of lake ecosystems. Applied to key parameters, the estimation techniques demonstrated above would partially eliminate the necessity of having to use parameter

()

values derived from laboratory experiments, the results of which are often of limited validity under field conditions. The stability of optimal parameter estimates along temporal dimensions and among lakes could be used as a partial basis for assessment of model generality. Demonstration of parameter stability would help to insure that the estimation step has been more than a curve-fitting exercise and that important functional aspects of system behavior have been captured. Comparisons of optimal parameter estimates with ranges of values found in the literature or measured experimentally could provide further evidence. However, it is re-remphasized that demonstration of parameter stability or satisfaction of any criterion for model "verification" in these non-experimental situations would not establish model validity, but only test whether that validity could be rejected. Thus, application of a rigorous battery of tests would reduce the dependence of model evaluation upon subjective judgment, but could not eliminate it entirely. - PALALE T-YEST FOR LIGATFICART COMPRICACES BUILDER STATIONS AT BACH DEPTH CONTINUES OF CALEGORIES OF THE STATE OF THE -----· .... . **A1** 14 100 a (1 LEVIL . 471 5 111-111 4.4344 5-4752. 4.1424 2.1112 5.1045 8-4128 1.2341 43 58444-0861 ERDE-448 51.6864 -13-7489 38-6679 -0.1625 8.1161 6.3734 4.4641 0.3689 1.4500 1.1811 4.1117 10 1411-0159 31.0447 . 1407.0000 11.2643. -35.9000. 24-1971. 1.3477 4-1588. 2.1477 8.4738 +B-4984 0.7147 10.2416 1448-0217 29-8174 30.0141 31.1993 71 1.4344 82 4.2761. 0.7603 A.8385 6.2013 -8-1477 4-5255 -0.0176 71 1749.7485 28.8678. -1.1941 22.2243. -9.8420 4.7914 .6371 11.0014 LATTE. -1.2541 5.0002 diam'r. here-ates 12.4118 1440.4597 31. 6787 50.2793 21.5377 2.4717 12.0041 1.1901 13.1004 1.5225 -1.8471 2.0012 71 1948-0179 20.1608 1945-8877 37.1274 -1-5712 27.0791 6.34.34 -0.0044 41.2321 30.5519 22 13-4413 1.011 11.1101 2.3955 -1.4952. باللك ل 1.1019 1963-4161 42-1475 1937-0433 11.044 2.6713 9.5191. 4.1443 0.1321 1.1220. 1.2132 1.1704 71 1474-4157 12.1151 -2.6262 21.6212 R. Shirt 3 . 42 الملاهماء التراقية 0.6407 7.1441 17.7828 2.2234 CONFORTAT. L. COTHE-B. PRIALD 1-1637 FEB \$1841910417 CIFFLEENCES ETTUEDE STUTIOUS AT BACK OFFM \_ CONTRACT 1\_\_\_ELECTORY CATERA\_\_\_\_\_\_ MALAD 1-151 FCB STATESCAT CITERALS OFTALON STATIONS AT INCO DETTO LEVIL A 10-UTEL 4 AT1-472 493 414 201 100 T 111 9 412 104 5 S.ATST. 5-4411 7.0001 4.5317 -1.381+ 3.3474 -1.9911 6.9392 ..... 0.0621 0.3744 8.0741 0.0942 1.3534 2.2852.\* 0.0421 5.9999 4.1321 C.317a 9.TTAC 0.3124 -0.9211 -1.+145 6.4910 0.0411 -0.0004 4.0111 -0.0142 .... 0.6110 2.6791 6.4407 6.1503 -0.0761 1.04.11 1.1167 1.4745 2.7961 A. 188 C. SATEST. -8.4326 9-2451 -7.6083 2.1125 ¢.228 2.3614 4.2947 -0.9540 1.1483 +2.3222 5.1161 6.0477 1.0144 4.4404 -0.0345 -0442 -0.3347 74 1.7410 a. 0759 3.3822 A.ATAT -4-0372 -0.1411 ..... 1.2920 9.2125 1.2120 0.2197 4.3244 8.3987 0.2478 .41 1.4732 A.8134 1.4810 A.OTOF -10.019 0.0673 -A. 714A 6.3itt 0.7813 A. HOM. 0.1817 0.1379 8.3163 1.5188 1.1617 1.4692 0.1019 1.19411 8.0011 .0020 0.0418 4.4714 ... 6.2010 0.4185 0.71841 8.1951 8.3371 1.4825 6.9128 0.0344 4.9112 0.0111 -0.0009 -CD 94 -0.0217 .75 4.1102 C-3714 4.7785 4.1184 1.4188 1.7449 ···· 2995. 1.6643 0.0444 1,015. 0.0004 0.0314 4.0432 6.4113 1.JH1 E.1114 6.4154 ...... 8.1418 a. 34 a4 \*\*\*\* - COMPONENT A TETAL THEREARD P - PLILLE T-TEST FOR STARFECTAT BLFFREDUCAS ATTALES STATIONS AT ENDED -LEVIL 4 **1** ÷. 691 492 411-472 500 LIVEL . mt 500 . 192 **B**\* 441-472 400 t 2.1601 8.2756 2.1668 0.7939 0.2003 ..... 1-3454 9-9363 1.1101 1.1242 -1. iata B-1953 -1.1988 \*.\*\*\* 0.7190 9.7893 0.4034 -0.1711 0.3709 #1.07b1 .99 1.8148 °0.1431 4.1044 8.1133 -4.2118 0.4165 -2.4991 ...... 8.2520 2.4495 0.1711 -0.4997 -1.4454 1000 28. 1.4161 6.1912 4.7133 5.1454 6.1011 6.1529 8.4914 0.2411 2.2527 6.32Th 2.2415 \$119.0. 4.1977 a. 211\_\_\_ 2.5544. 0.2498 2.4595 0.1511. -0-0454 9.1431 4.1927 1.1644 0.1118 8.7549. 0.6373 2.2752 1.3745 3.9003 8.3010 2.8286 8.2462. BALLINT. 1.2475 e.7413 2,1221 0.1761. 0.2112 4.5725 2-3048 0.4410 \_1.200 7.3012 2.1914 0.3181 -0-1905 4.7194 -9-1671 20 2.4612 7.1139 9-6162 0.1201. -0.1544 0.0411 -1.9033 3.7718 ê. 2109. 0.2814 4.2452 \_\_\_\_\_ 1.0112 \_\_\_\_\_ B.11% 9.1922 0.2169 8-6328 0.2763 P. 47M

2

#### APPENDIX A - Results of t-Tests for Horizontal Mixing in Onondaga Lake

 $\frown$ 

APPENDIX A (continued)

|                 |     |       |         |           | 1.111  | THE PARTY |             | 1112-1- |         |        |              | -       | Article - | -0.6414 | -1.812   | 1111.4 |        |       | 1111  |       |       |                  | ŀ        |         |       | 1014-6- | 16141  | ette f-                                  |        | -1.481 |      |          |
|-----------------|-----|-------|---------|-----------|--------|-----------|-------------|---------|---------|--------|--------------|---------|-----------|---------|----------|--------|--------|-------|-------|-------|-------|------------------|----------|---------|-------|---------|--------|------------------------------------------|--------|--------|------|----------|
|                 | 5   | 1440  |         |           | - 4114 |           |             |         | 1.44    |        |              | ā       | 1118-8    | 0.467   |          |        |        |       | 1127  | 1121  |       |                  | ž        | 11.00.6 |       | THE P   | 0152.£ | PAD-D                                    | 10,01  | Har    |      | A 11-14  |
|                 |     | 1111  | THE     |           | A 2444 |           |             |         | A late  |        |              | 217-114 | 0.011     | -0-0010 | - not-   |        | 1100.4 | 1     | 1000  | -     |       | Co MATR          |          | 1       |       |         |        | WILL F                                   | E.MMI. |        | 104  |          |
| 1041 EL 14      |     | 1982  |         |           |        |           | -           | 2       | -       |        |              | 8       | 1.11      | 2.0015  | - 4600-0 |        | 1001   | A MAL |       | 1224  |       | il in traili     |          | 1.0413  | 1111  | 1.411   | - NAM  |                                          | - TANK | - HUDA | 1100 | A114.4   |
| AT CLERK        | 215 | 1111  | - INLAI | int       | 1.414  |           | 1.111       |         | 1012-01 | THE P  |              | Ē       | 1004      |         |          | Ĩ      | 12271  |       |       |       | N N N | i activectis str | ĩ        | 1.001   | 6.4U7 | - the   |        |                                          | 100    | 1000   |      | 0.1141   |
| LIMB PALS       | 1   |       |         |           | 1.144  |           | <b>1.17</b> |         |         |        | ETH BOAT     | Æ       | 2,004     | A00.4   | Nea a    | Ĩ      | 1.00M  | 1.004 | . 112 |       |       |                  | B        |         |       | ING     | 100    | LE L |        | - uni  |      | 0.444    |
| LINE CONTRACTOR | H   | 17274 |         | THE COLOR | 1,750  |           | 1 mil       |         |         | I IIII | The Articler | щ       | 1111      | 1.000   | 1100     |        | 1.674  |       |       | THE R | Eller | Statister        | 611<br>1 | 4.4671  |       | A.MIL   | · HIT  | 1111                                     |        |        | HILA | P. P4 74 |
|                 |     | 1     | -       |           | 1      |           |             | 1       |         |        |              |         | 3         | \$      | 1        | Ę      |        | Į     | 1     | ļ     | 1     | First no         | •        | 8       | f     | •       | 4      | Ţ                                        | =      | -      | -    | Ģ        |
|                 | Ĩ   |       |         | 1         | 1      | 1         |             | 1       | 1       | 1      | ļ            | Ĭ       |           | Î       | 1        | 1      |        | 1     | 1     | 1     | 1     |                  | LIMB     |         | 1     | 1       | 1      | 1                                        | 1      | 1      | 1    | 1        |

| •                | -       | INT    |       |           | - Ward   | 1111    |         | 1420-1 | 1101-1 | - HART |            | -       |        | 2-047   |         | 1-4178 | Buttle | A.T.M. | 144-6- |               |         |               | -       | 1.111  | 1.000         | - 7414 | <b>D.401</b> | 142.1  | Interes | Chord. |       | 1014     |
|------------------|---------|--------|-------|-----------|----------|---------|---------|--------|--------|--------|------------|---------|--------|---------|---------|--------|--------|--------|--------|---------------|---------|---------------|---------|--------|---------------|--------|--------------|--------|---------|--------|-------|----------|
|                  | 4       | E. IN  | 111.6 | ALL L     |          | 1 Hills | A THA   |        | 1114   |        |            | 1       |        |         | 0 U U   | 1.1100 | 4.224  | 1.00   |        |               | 121-3-  |               | a       | 8.2219 | <b>6.1310</b> | TAL .  | MILA         | 1111   | 1111    |        |       |          |
|                  | 749-144 | -      | ARL C | THE PARTY |          | 4111-0  |         | 1111   |        |        |            | 441-442 | 1.1017 | ATM.A   | A.111.A | 6.2247 | 200    | 1173   |        | a mu          |         |               | Lin-ita | 9.3677 | MALS          |        | A100         | 202    | -       | E THE  | HIL   | Ş        |
|                  |         | ALTEN  | AILA  | PHI I     | - THIN   |         |         |        |        |        | Taritad av | a       | 6.141  | 6.177   | 0.147   |        | T11.A  | - TEL- | 1 HILE | <b>e</b> 11.6 |         | TATION AT     | a       | 9.1511 | . 2013.       | 2112   |              | 5102.E | - alfe  | A1114  | 1001  | and a    |
| L MINUT          | 111     | 1422   |       | 1111      | MILL     |         | 100     | THE P  | 1414-  |        |            | Ĩ       | HMH    | . 17240 | 1.7145  |        | 1494   | and i  | -      | THE PARTY     |         | th official i | 244     |        | 1944          | PINT   | 1111         |        |         |        |       | -<br>In- |
| <b>HO</b> 140 In | 8       | IIII   | HI I  | HILE      | - MILLO  | E111    |         | - KULA | 101.0  |        | us tifree  | 5       | 9-1145 | A11.9   | E I     | 29149  | 2.1622 | HILLS  | THIS . | HILE          |         | AT DIMPLEX    | \$      | - Ne   | -182.6        | I.     | - HILL       |        |         | NUT    | and a | 112.4    |
| 1101475 au       | Ĩ       | 2.35.6 | 1111  | N00-1     | - 1111-1 |         | - 1111- | 1121   | -1984  | 1111   | Distros en | 2       | 1.4441 | 2.6764  | 1.100   | 19171  |        | THE    | 10111  | TIN.          | - 11212 | Ditati        | Ē.      | 1111   | 44114         |        |              | 1111   |         | - ADM  | 1111  |          |
|                  | 1       |        | 3     |           | -        |         |         |        |        |        |            | ונשו    | 4      | 3       |         | 8      | 1      |        |        |               |         | State and a   | Ĭ       |        |               | 1      | 1            |        |         |        |       | 4<br>1   |

()

(

()

#### APPENDIX A (continued)

|            | -11157           | 12100<br>14 11411114 | 17.01FTE DE    | ts erturne s | 7471993 AT. C  | K+ 0000     |         |         |
|------------|------------------|----------------------|----------------|--------------|----------------|-------------|---------|---------|
| FØYEL      |                  |                      |                |              |                |             | 584     |         |
|            | - 11             | - <del>31.141</del>  | 34.174         |              | M. TIL         | -4.3415     | 18.9932 | -0.5410 |
|            |                  |                      | 27.444         | 467.7WM      | <b>N.</b> 11 M | -4.1941     | 14.9862 | -4.3713 |
|            | - 64             | 245.5325             | 44.2573        | 345.1004     | 19.4393        | 6.703       | 24.4678 | 8.0442  |
| •          |                  | 823,6448             | 21.000         | 613-1010     | 30,004         | -1-0000     | 14-3445 | -0.1102 |
|            | -                | 440.3463             | 14.16          | Test State   | 34, 244        | -4.441      | 21-1444 |         |
| 4          | **               | \$43.8543            | 21.543         | 414-2744     | 12,4010        | -0.1000     | 21.4644 | -9.1112 |
| 1          | Į.               | 638-6587             | 14.614         | 611.3443     | 36.3989        | 21-2244     | F8-696F | 1.4126  |
| •••••      | -44              | 449-3184             | 24, (1)4       | 675.1420     | 14.4465        | -4.4331     | 11.3174 | -4.3341 |
|            | -                | 94-931               | \$3*9640       | . 483(1343   | 31.40b0        | 2.7623      |         | - e.011 |
| ALTER I    | ا ـ د ک<br>آمادا |                      | 143 C197621    | cas emilia   | STATIONS AT    | ICA DEFTA   |         |         |
|            |                  | AL                   | 924            | M2 .         | 100            | MJ-N2       | jim .   |         |
|            | - 12             |                      | 7.4937         |              | 1.2190         | 0.9772      | 2.9947  | 1.0001  |
|            | . 73             | 149-0321             | <u>7-3514</u>  | 149-4434     | 2.7418         | 3.3699      | L.7718  |         |
| · · · - 3  |                  | 141.0134             |                |              | 2.3714         |             | 1.4984  | tata    |
| 4-         |                  |                      | 2.3147         | 178.9861     |                |             | 1.7619  | 0.4T14  |
| s.         |                  |                      | 2.4417         | 111.7000     |                | -0.4841     | 1.917   |         |
|            |                  |                      |                |              | 9. 1922        | 0.7797      | 7.\$173 |         |
| <b>i</b> . | 12               | 201.1210             | 3.410          | 269.6798     |                |             | 1.4540  |         |
| البجين     |                  | 149.2284             | 2.4474         | 144.0175     | . 3.9421       | 4.1001      | 1-3334  |         |
| a 🖡        | <u></u>          |                      | 2.15%          | 192.6194     |                | 32351       | 1.104   | 2.1154  |
| _C30/304   | 1-101            |                      | unt clyfele    |              | STATEONS AT    | LACE DI Pho |         |         |
| UNI.       |                  | 401                  | 104            | - 442        |                | 981+973     |         |         |
|            | 42.              |                      | afe <u>a.a</u> | 1.6853       | 0,0444         | -0.1001     |         | -1.9704 |
|            | 41               | 7.184                | 0.0444         | 1.1210       | 0.6501         | -0.1378     |         | +7.7547 |
| 1          | 48               | 7.4411               | 0.0145         | 7.8972       | 0,0331         |             | 4.4755  |         |
|            |                  | 7.4157               |                |              | 0.0101         | -1.0152     | 9.0222  |         |
| <u> </u>   |                  | 1.4112               | 4.5797         | 2.4413       |                | -6.6191     |         | -2.1991 |
|            |                  |                      | 0.0302         | 7.4193       | 0.4214         | -0-0794     |         |         |
|            | - 14             | 7-8763               | 0.0130         | 7.4358       | <u></u>        |             |         | -8.2108 |
|            |                  | 7.4615               | 0.0440         | T.TAN        |                | -0.071      | 8.0743  | 1.1101  |
| <b>_</b>   |                  |                      | 9,0284         |              |                |             | 9.0148  |         |

.

| enterne a later and electricity circlements estimate stations or and dryn |                  |                     |              |                     |                   |          |                  |         |  |  |  |  |  |
|---------------------------------------------------------------------------|------------------|---------------------|--------------|---------------------|-------------------|----------|------------------|---------|--|--|--|--|--|
| 14966                                                                     |                  | 11                  | \$9 <b>0</b> | 445                 | 100               | 447-445  | 12 <b>0</b>      | T       |  |  |  |  |  |
|                                                                           | <u>19</u>        | 5.2267              | <u></u>      | 0.2019              | 0.0171            | -0.4119  | 0.0755           | 0.4680  |  |  |  |  |  |
| ,                                                                         | 57               | 0.1584              | Q.6119       | 4.1917              |                   | -9.4013  | 8.0197           | -0.4814 |  |  |  |  |  |
|                                                                           | 41               | 4.1591              | 6.6143       | 4.1937              | 0.0127            | 8.0044   | 0.0141           | 9.303.0 |  |  |  |  |  |
|                                                                           | <u> </u>         | E.1117              | 8.8107       | d.1004              | 0.8205            | -8.0166  | 8.3122           | -1.1010 |  |  |  |  |  |
|                                                                           |                  |                     | .0.0214      | 8.2614              |                   |          |                  | -2.0746 |  |  |  |  |  |
|                                                                           | <u>.</u>         |                     |              | 8-2418              | 0.0170            | -0.4230  |                  | -1-1971 |  |  |  |  |  |
|                                                                           |                  |                     |              | 2.1268              | <u>- 6,6109 -</u> | 0.6101   |                  | -2.0417 |  |  |  |  |  |
|                                                                           | 57               |                     | C. (121      | C.2101              | 0.0137            | 8.0011   |                  | 0.4004  |  |  |  |  |  |
|                                                                           | _Я               | e.7471              | 6.0196       | 0.2444              |                   | -0.8143  | <u>. 6010a .</u> |         |  |  |  |  |  |
| <u>Contesta</u><br>Pillado 1                                              |                  | 403148<br>114818 PC | AT BUTULOG   | <u>41 9170/14 1</u> | Intions at        |          |                  |         |  |  |  |  |  |
| LEVEL                                                                     |                  |                     | 10=          | 544                 | 100               | AF1-472  | 199              | Ĩ.      |  |  |  |  |  |
|                                                                           | -41              | 0.0202              |              | 0.0322              | . <b>4.0039</b>   |          | 3.9927           | -0.453  |  |  |  |  |  |
| . 2                                                                       |                  | 6,6123              | 0.2017       | 0.0322              | - 0.0054          | a.0000   | 8.0071           | 0.4174  |  |  |  |  |  |
|                                                                           | _ 44 _           | 0.4370              | 0.5075       | 4.0370              | . 0. 0011         | _ 0.0010 |                  | 0.1760  |  |  |  |  |  |
| - +                                                                       |                  |                     | 0.0159.      |                     |                   |          |                  |         |  |  |  |  |  |
|                                                                           | · 47             | 0.0304              | 0.000        | A-0123              |                   | _+0.0019 |                  |         |  |  |  |  |  |
| -                                                                         | - 42             | 4.0224              | 4.5044       | 0.0111              | 4.9647            | -0.0075  |                  | -1.4744 |  |  |  |  |  |
|                                                                           | _1t              | <u> </u>            | 0.0011       |                     | 0.0011            |          | 0.0029           | -2.3093 |  |  |  |  |  |
|                                                                           |                  | 0.0331              |              | 0.0323              | 0.0014            |          |                  | rt.1374 |  |  |  |  |  |
|                                                                           | - 42             | <u>0.9774</u>       |              | 8-0344              |                   |          |                  |         |  |  |  |  |  |
| CONTINUES 1                                                               | 17 + U<br>1-1111 | INA A               | NY ESFFEREN  | Es Altretes         | TATIONS OF        |          |                  |         |  |  |  |  |  |
| LETTE                                                                     |                  | <u>(11</u>          | <b>Sin</b>   | EV2                 |                   | _#1-#¥2  | \$00             |         |  |  |  |  |  |
|                                                                           | st               | 1,4143              |              | 8-9301              | 9.6(13            | -6.6018  | 8.0048           | -9.3763 |  |  |  |  |  |
|                                                                           | - 54             | - 1116-5            | 0.6447       | 6.6951              | 6.644             | 0.0412   | 8.4041           | 0.7799  |  |  |  |  |  |
|                                                                           | -                | 4.0252              | 9.6617       | 4.0414              | 0.9834            | 8.0054   | 9.0027           | 2.0009  |  |  |  |  |  |
| •                                                                         | 30               | 0.0112              | 8.0049       | 8.843               | 0.5CM             | 8.4145   | 6.0044           | 3.3H3   |  |  |  |  |  |
| -                                                                         | - 14             | 0,0450              | 0-0033       | 4.9443              | - 4.8ELS          | 0.0427   | 6.8439           | 0.7427° |  |  |  |  |  |
|                                                                           | - 14             | 6.633               | 4.007        | 6.6413              | 0.0019            | 0,0022   | 6.0013           | 0.7850  |  |  |  |  |  |
|                                                                           | -23              | 8.0411              | 6.8643       | 0.0301              | 0.0631            | 0-0110   | 0.0014           | 1.1744  |  |  |  |  |  |
|                                                                           | 34               | 0.0344              | 1.4634       | 4-0315              | - 0.2634          | 0.0025   | 1.0025           | 0.1634  |  |  |  |  |  |
|                                                                           | н                | à.0415              | 8.5834       | 8.041               | - 0.4EE           | 0.0034   | +                | 1.4174  |  |  |  |  |  |

·····

|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | VIIIA PLU - GAVIRGANGARA BRATON - MARADA - MARADA VIIIA - VIIIA ATAS                                                                                                                                                                                                                            |  |
|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| initiation and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | ALE I TIME IN MENTHE FACE JAMANA LEAD                                                                                                                                                                                                                                                           |  |
| VERTICAL BEAL                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | ) ( GENIM IN MAXAAS JAGAM LANASA MAARLING POINT (10 - CONVERSION)                                                                                                                                                                                                                               |  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | ESHPOHENT ; TEMPENATUMA (DEGAELS 2)<br>N A chima dathaan duunan na alaa ann talaa mala a lalah du bala di. Baya a datt                                                                                                                                                                          |  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | a names applying to deputation and an an                                                                                                                                                                                                                                                        |  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                                                                                                                                                                 |  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | 1011 10138:38 101 10 182.08 10 182.08 10 182.08 10                                                                                                                                                                                                                                              |  |
| 1478L5                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                                                                                                                                                                 |  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | TEMPÉRATURE GRADIEN7<br>1848 1849 1875 1873 1873 1874 1876                                                                                                                                                                                                                                      |  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | f 141 fevere a fry all a cligation and a state and a state a st                                                                                                                                                                                 |  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                                                                                                                                                                 |  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | Jara flag - Enversmental Systems process - Marvand universiter                                                                                                                                                                                                                                  |  |
| UN 12011746. GCA                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | ui e temp en avandar almuart esca                                                                                                                                                                                                                                                               |  |
| CATSCAL <sup>®</sup> BCALO                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | A DEPTH IN WATERS ADDRE COURT SAMPLING POINT (18 + SUPPACE)                                                                                                                                                                                                                                     |  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | COMPONENT & TERRETATIONE GRADIENT CONS-C/HS                                                                                                                                                                                                                                                     |  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                                                                                                                                                                 |  |
| 474 #4 <b>4420 (</b> #                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | 4 LEVELE BÊTVEEN SKITEFE VALUFÊ QÊ Ê.G AND F.DG MEAK 4 6.33 BT, DEY, O G.QO<br>Bande Abel verd di darm lêvel                                                                                                                                                                                    |  |
| 174 24 <b>0000</b> (2<br>1171163 <sup>value</sup>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 4 LOVELE BETWEEN EXTERNE VALUES OF 0.0 AND 1.06 MEAN - 0.22 BT. DEV 0.49<br>RANGE APPLYENG TO BAEN LOVEL<br>BUTS LIBE EISE FIE<br>BUTS LIBE EISE                                                                                                                                                |  |
| ATA AAA <b>REA</b> (M<br>1991-1991 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 1994 - 19 | ialar vəşafðill <sup>a</sup> larna <sup>n</sup> gifte vaaflikt 10 event<br>uvar vəşafliklalar<br>uvar vəşaflikla ia avin falar<br>Fils bils<br>Fraare nylaeen uxuelee avrale on 4°40 trop metn a 8°33 alf bahr = 4°40<br>Fraare nylaeen uxuelee avrale on 4°40 trop metn a 8°33 alf bahr = 4°40 |  |

APPENDIX B - GRID Displays of Onondaga Lake Water Quality Data

1970

()

 $C^{i}$ 

 $\mathbf{C}^{\mathbf{k}}$ 

TEMPERATURE

1971

1874

1975

1971

| DECOMPOSAS LAST PAR<br>HONISCONTAL SCALE S<br>VERTICAL SCALE S<br>DATA HANNEL IN A<br>ANTENIA VALUE HA<br>PERCENTAGE DF TOF<br>CEVELS CONTACT PARTY<br>CEVELS CONTACT PARTY<br>FUER WEAK CONTACT FOR | TA GLET - ENVIRONMENTAL OVOTENE PROGRAM - MADVARD UNIVERSITY<br>> TING IN ANNUARY 1100<br>COMPONENT : CHARGE (MAL)<br>COMPONENT | - <b>--</b> - <b>-</b> |
|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------|
|                                                                                                                                                                                                      | DISSOLVED OXYGEN                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                        |
| Churdaga Lan<br>Hurtssunta, S.<br>Vantssa, S.<br>Vantssa, S.<br>Uata Aappeo<br>Ministan Van<br>Marchataur C.<br>Status V.<br>Status V.<br>Status V.                                                  | <sup>1</sup> <sup>1</sup>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                        |

-

CHLORIDE

1971

1972

(973

1974

1975

# APPENDIX B (continued)

**(** )

**.** 

1116

1969

1970

A Destart POLICIPALICE OF DATA FORM VALUES IN GARA LINE MISH 

executed as core resorned and the selfing to strater

# ALIENTER VALUE AAAAA ATTALEER IS AAEAA AISA AISA AISA AISA

DATA HAPPED IN + LEVELS ARTWEEN BEIDENS VALVES GP - Quad And - 4400 MEAN - 1400 STA DEVA A

COMPRIMENT & TOTAL INCREMES AND AND CALL

VENTICES SCALE I DEFIN IN NETERS ACOVE LANSEST CANALING PAIRS (18 - SUMPACE)

HURSZONTAL BEALT I TINE IN HONTH'S PARM JANUARY 1948

GNGNDAGA LANG GATA PLOT - GAVERANDENTAL BYETANS PAGERAN - HARVARD LAITHERSTY









()

C

KA STOLDTOTICH ON OULD WELHA ANTING IN BUCH FEAST LVIS LANCE 671 Pecoulac

ALECTATION OF JOLN' THOUPS'ANT A' GIRDE THAT AN TO THE THAT

# ANDIAN ANTRA ANTRA ANTRA AN ANTRA SIST

CONFIGNANT & NETRETA NETROGEN - CHAPLE DATA MARRED IN A LEVELS RETAILS BUTADA WALLES OF

8:1\$

VENTION SCALE & DEPTH IN PETERS ABOVE LONGST BANPLING PUBLY ISS - BURPACE)

MUNISCRETCH SCALE & TERS IN MONTHE PROP ANNUARY (1914

ONCHORGE LAKE DATA FLOT - ENVIRONMENTAL SYSTEMS FREGRAM - HERVERD UNLUCHASTY





-coccestor as total secondification for the states as efficient

ANTONIA VALUE AANGE PERLYINE TO EACH LEVEL. 8:18 8:11

DATA BAPPED IN A LEVELS BETWEEN ESTATION VA

(:

HENT & HETMATE METROGEN COMP 480/L3

VERTICAL SCALE & DEPTH (IN HETERE ABOVE LOWERT BANKLING MOINT (IS A SUMPACE)

NTAL ECALS I TEME IN AGRING PRON JANUARY INCO W74220



| anano aga                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | LACH GATA <sup>1</sup> PLET - ENVERCHMENTAL SYSTEME PROBADY - MARVARO GUEVERSETY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| NOAL SONT                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | , SCALE & TIDE IN MARTING PROV JANUARY 1968                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| WEATECAL                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | icald a dealer in maters found lovest sampling point the or support                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | CONFERENT I ARRENTA MITHODEM (MC/L)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| 0ATA 94 <b>P</b>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | ID EM & LEVELO BEFORD ARTHEME VALUES OF                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
| "NUNKA                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | ALLAN MANDE APPLITATIO DECHI LEVEL. 5:58 2:28                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| PERCENTAC                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | De JOINT VEROFALS ANTHE RESER NEWS IN STER PART                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
| P Regulations of                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | DISTRIBUTION OF GATA POINT VALUES IN BACH LEVEL                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | "ITTELETT ITTELETT BURKLAND SAMAAN PAINTIN ANDELAN                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | ,,                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | ORGANIC NITROGEN                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
| 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | 1964 1969 1870 1071 1872 1873 1974 1975                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
| 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | HARRANCE LINETTER STREETERS IS STREETERS IS STREETERS IN STREETERS IS STREETERS IN THE STREETER STREETERS IN THE STREETERS IS STREETERS IN THE STREETERS IN THE STREETERS IN THE STREETERS IS STREETERS IN THE STREETERS IN THE STREETERS IN THE STREETERS IS STREETERS IN THE STREETERS IN THE STREETERS IN THE STREETERS IS STREETERS IN THE STREETERS IN THE STREETERS IN THE STREETERS IN THE STREETERS IS STREETERS IN THE STREETERS INTO STREETERS IN THE STREETERS IN THE STREETERS INTO STREETERS                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | siz forefabresserer prinke: treatistication presserer forefit |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
| 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | hannandian an a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
| ļ                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
| UNINGA                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | A LAKE DATA ALAT - ENVIRONMENTAL SVETENE FRASBAN - RATVARD UNIVERSITY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
| -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | Ing acade i fine in nonthe room January 1944                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
| -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | . SCALE & DEPTH IN WETERS ADONE LEWERT BANKLING POINT THE & SUMPACE?                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | COMPONENT & CREANSC INSTROOM SHEALS                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| 04TA 8/                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | MED 44 6 LEWELS BETVER SXIAGAS VILLES OF 0.86 AND 4.68 MEAN 4 6.48 17. OCV. 4 5.41                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
| 1868                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | S VALUE RANKE APPLYED TO BACKANE AND END                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
| weat of                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
| Autor At                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | A SPACE AND A CALL AND AN AND IN AND A CALL HIGH AVENUE                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
| PLICENT<br>PECOR<br>LEVELS                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
| EA MECH MI<br>Vincon Ex<br>Vincon Ex<br>Vinco |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
| PLNCENS<br>CPECUES<br>CPECUES<br>PRECUES                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
| PLNCU NA<br>PLNCU NA<br>LO VELA<br>SY NBQL<br>PREQUE                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |



( )

C

(

n v

. . .



RECENTION OF TOTAL ABORTALS AFTING 2:235 WATTING TO ETC. THAT

#### 

DATA MAPPED IN & LEVELS SETURER SATAERS VALUES OF 1.08 AND 16.00 MEAN . 6.26 ST. DEL

CORPORANT 1 0-DAY 000 (-0/L)

VERTICAL BCALE I OFFICE IN RECEASE ABOVE LIVEST SAMPLING POTHT (LD - TUMPACE)

MONTAGE ACALS & TINE IN MONTHS PARM ANNUARY LYCE.

CHOMONON LARE DATA PLUT - ENVIRONMENTAL STOTEME PROGRAM - MARVAGO UNIVERSITY





| PALQUÍNET<br>LEVELS | EV9-OTERSTon | OP 0474 | -     |        | • Level |                                        |
|---------------------|--------------|---------|-------|--------|---------|----------------------------------------|
| \$Y#801.5           |              |         |       |        | ×       | HEAH                                   |
| endautincy          |              | 17 17   | 78-27 | 314-14 | 144     | ************************************** |

MENCENTACE OF FOTAL ABIOLUME AND AND AND THE TO THE STREET

ALTINIA VALUE RANGE ANALYINE TO EASTA LEVEL 4:88 1:88

COMPONENT I SELIGATE (MOVE) DATA NAPPED IN 4 LÉVELS BEFAEIN EXTAINE VALUES OF 4-66 AND 4-60 MEAN # 4-73 67, DEV. # 1-66

L SEALS I OPPH IN MITCHE AGOVE LOAGET BACPLINE POINT (10 - SUMPACE)

HORESCHITAL BEALS & TENS BE HONTHIS PROF JANUARY LOSS





()

C

.



APPENDIX B (continued)

()

A-10



 $\mathbf{C}$ 

C



APPENDIX C - Line Plots of Volume-Averaged Onondaga Lake Water Quality Data

 $(\cdot)$ 

()

C;



**WPPENDIX C (continued)** 

()

: ];

¥~73



**₽1-**¥


A-15

()



 $\sum_{i=1}^{n}$ 



0



....

 $\int_{\Omega}$ 

(

APPENDIX D - Mass Balances on Onondaga Lake

()

[



ιj

\_1





. . .









Ŧ







.

÷. •









